

Are oral health conditions associated with children's school performance and school attendance in the Kingdom of Bahrain?

A Life course approach

Seham Mohamed

A thesis submitted in the fulfilment of the requirement for the Degree of Doctor of

Philosophy

Unit of Oral Health, Dentistry and Society School of Clinical Dentistry University of Sheffield United Kingdom

Dedication

To my mother, the pure and innocent soul, who taught me that believing in Allah,

good intention, and hard work is the key to success.

COVID-19 statement

"It was the best of times; it was the worst of times" (Dickens, 1949). A cliché? I do not think so. Not after most of us has experienced in the past months.

It was the summer holiday of 2019, and things could have never been better. I am at home with my family, just finished a brilliant year one of my PhD. My mother responded well to chemotherapy, and her lymphoma is gone. I started collecting data from schools and children. I was there for day one when the pupils came back to begin a new academic year. The children's smiles and anticipations were so refreshing. The job was done, and I headed back to the United Kingdom (UK) to start my year two, planning to revisit the children by the following autumn.

My husband was supposed to join my three children and me in the UK for the Christmas holidays. That was my last sense of normality. A health professional by himself, my husband withheld his plan as the news of mysterious disease started to grow up like a snowball. By February 2020, Covid-19 cases were confirmed in Bahrain. Soon, my mother was admitted to a hospital with difficulty breathing. I wanted to go back home and see her, but travel restrictions were in place at that time meant that I should quarantine for two weeks. By February the 23rd, she was in ICU isolation with an unknown infection, and by February the 26th, 2020, we lost her. I was 20 days late and could not say goodbye.

On March the 11th, 2020, WHO declared Covid-19 a pandemic. Bahrain entered a draconian lockdown, followed by the UK. The hospitals became hotspots for contracting the disease, and people tried to avoid them. My father had chest pain, and because of fears and restrictions in accessing the hospital, he developed a severe heart attack. The risk of losing him too was so eminent, but surgical interventions kept

ii

his heart going. He also started developing dry gangrene in his diabetic feet, and a lengthy journey of lifestyle, non-surgical and surgical management just managed to save his feet.

Education was severely hit by this pandemic. As in many other countries, a shift to online learning swiftly replaced conventional education in the UK. My children and I had to adapt to this e-learning. As a teacher by profession, I find this one of the hardest things. The value of face-to-face learning can never be overestimated. With much appreciation for my supervisors' and tutors' efforts, I have always missed the traditional interaction. However, my children enjoyed being at home and online studying. Nevertheless, this has put lots of pressure on me to manage a total of nine sessions per day for all my 4-, 9- and 10-years old children, instantly helping them to do their homework and posting them online.

As schools in Bahrain implemented online tutoring, and even if the kids went back to school in the 2020-2021 academic year (which they did not till now), It was clear that the impact of Covid-19 on children's school outcomes would have proven more influential than the measured variables in the study. Soon I realised that I would not be able to conduct the proposed longitudinal design for my research. This was morally devastating for me, as I was so eager to have it done. Disappointment aside, I promptly amended the study design to get the best of the data I collected. To do that, I had to apply for changes in the ministry of education in Bahrain and the University of Sheffield, get their approvals, and gather children's school performance through the parents via a phone application and their school attendance via school registers.

Back to the UK in 2020 and through 2021 with two more lockdowns and further travel restrictions meant my children and I could not see our beloved family as much as we

iii

would like to. The constraints challenged our mental health to the limits. I also applied to extend my study duration, resulting in more strain on my financial resources.

Ultimately, reading and writing about life course puts enormous pressure on me as a mother living away from my family with three children and doing a PhD degree at this pandemic time. I am aware such an event might leave its scars not only in everyday lives, but it can leave a long-lasting one. However, I am very proud of how I managed to go through it, even publishing a paper during this time (Appendix 1). Summing up my experience, for myself, I will say, "It is a far, far better thing that I do than I have ever done" and for my mother, I hope "it is a far, far better rest that you go to than you have ever known" (Dickens, 1949).

Abstract

Background: The link between oral health conditions (OHCs) and school performance and attendance remain unclear among Middle Eastern children. The association has been studied extensively in the Western region; however, several concerns have been raised regarding the reliability and validity of measures, low quality of studies, inadequate inclusion of potential confounders, and the lack of a conceptual framework. These limitations have meant that, to date, there has been no detailed understanding of the association or of the key social, clinical, behavioural and parental factors which may impact the association.

Aim: To examine the association between OHCs and children's school performance and school attendance at Grade 2 in Muharraq city in the Kingdom of Bahrain (KoB) using Heilmann et al.'s (2015) life course framework for oral health (OH).

Objectives: To (1) describe the prevalence of OHCs among 7-8 years old schoolchildren of the good, rated schools in the city of Muharraq; (2) analyse the social, biological, behavioural, and parental pathways that link early and current life exposures with children's current OHCs by testing the critical, and the accumulation life course models with consideration of social determinants of OH; (3) examine the association between OHCs and school performance and school attendance among schoolchildren including the direct and indirect (mediated) pathways; (4) explore the early and current life course social, biological, behavioural and parental factors associated with children's school outcomes in addition to OHCs.

Design: A time-ordered-cross-sectional study was conducted with 466 schoolchildren aged 7-8 years and their parents from Muharraq city in KoB. Data were collected through parents' self-administered questionnaires, children's face-face interviews, and

V

dental clinical examinations. Outcome variables, including school performance and school attendance data, were obtained from the parents and school records. The data were analysed using confirmatory factor analysis and structural equation modelling (SEM).

Results: Dental caries, the consequence of dental caries (PUFA/pufa), and enamel developmental defects (EDD) prevalence were 93.4%, 25.7%, and 17.2%, respectively. The findings from the SEM showed that children born in families with high SES were less likely to suffer from dentine dental caries (β = -0.248) and more likely to earn high school performance (β = 0.136) at 7-8 years of age in Muharraq. From the current life course of children, the dental plaque was associated significantly and directly with enamel caries (β = 0.094), dentine caries (β = 0.364), treated teeth (filled or extracted because of dental caries) (β = 0.121), and indirectly associated with dental pain (β = 0.057). Further, dentine dental caries was associated significantly and directly with low school performance (β = -0.155). At the same time, the dental plaque was indirectly associated with low school performance via dental caries ($\beta = -0.044$). Conversely, treated teeth were associated directly with high school performance (β = 0.100). Besides OHCs, parents' early and current SES were significantly and indirectly associated with children's school performance via parental characteristics (β = 0.457). Notably, none of the OHCs, biological, SES, behavioural, or parental conditions was related to school attendance in children.

Conclusion: The life course approach was adequate to examine the role of OHCs on children's school performance and attendance. Birth and current (7-8-year-olds) social factors were significant predictors of poor OH and poor school performance. Dental caries was associated with poor school performance but not with school attendance. Treated dental caries was associated with good school performance.

vi

Acknowledgements

First, I would like to recognize the generous financial support of the University of Bahrain, without which this work would not be possible.

My most profound appreciation goes to this thesis's supervisors, Professor Sarah Baker, for her guidance, insightful suggestions, encouragement and unwavering support throughout this project. Professor Mario Vettore for the research idea and his constructive criticism, and Professor Chris Deery for his helpful advice and practical suggestions.

I gratefully acknowledge the Ministry of Health and Ministry of Education in the Kingdom of Bahrain to facilitate the conduction of this project fieldwork.

I owe many thanks to schools, pupils and families, whom this work was conducted, for their effective and wholehearted involvement.

I also had the great pleasure of working with many volunteers whose professionalism and enthusiasm in conducting the fieldwork was unparalleled.

I extend my sincere gratitude to my father and my late mother, who, despite their illiteracy, have relentlessly endorsed my academic interests since my school days and their prayers and blessing were always my comfort zone.

My special thanks go to the people who witness this journey step by step, my husband and our three boys. Thank you for being my life's cornerstones.

Finally, Sheffield, the university, the place, and the people, I am deeply indebted to the kindness, the inclusiveness and the warmness that my family and I were shown.

vii

Table of Contents

Dedicationi
COVID-19 statementii
Abstractv
Acknowledgementsvii
List of Tablesxiv
List of Figures xvii
List of Abbreviations xix
Chapter 1. Introduction 1
1.2 Aim 4
1.3 Objectives 4
Chapter 2. Literature review 5
2.1 Introduction to life course approach5
2.1.1 Emergence of the life course approach5
2.1.2 Early life circumstances and chronic diseases in adulthood 6
2.1.3 The concepts and the principles of the life course approach11
2.1.4 Conceptual models in life course epidemiology14
2.1.5 Key stages of the life course affecting health
2.2 Life course and oral health28
2.2.1 Oral health over the life course29
2.2.2 Early life factors and oral health in adulthood
2.2.3 Life course framework for oral health40
2.3 Dental caries life course studies among children and adolescents42
2.3.1 Study design46
2.3.2 Origin of the studies46
2.3.3 Participants' age47
2.3.4 Oral examination and indices47

2.3.5 Non-clinical data collection	48
2.3.6 Dental caries life course factors	50
2.3.7 Life course models and analysis methods	51
2.3.8 Summary of the results of the studies	52
2.3.9 Conclusion	60
2.4 The impact of OHCs on school performance and attendance adolescents	in children and 62
2.4.1 Overview	62
2.4.2 Aim	62
2.4.3 Study question	62
2.4.4 Inclusion criteria	62
2.4.5 Search strategy	64
2.4.6 Review method	64
2.4.7 Summaries of the systematic reviews	67
2.4.8 Discussion	88
2.4.9 Conclusion	90
2.4.10 Updates of the systematic reviews	91
2.5 Predictors of school performance and attendance	102
2.5.1 Overview	102
2.5.2 Child's characteristics	104
2.5.3 Microsystem/mesosystem	108
2.5.4 Conclusion	117
2.6 Theoretical framework of the present study	118
2.6.1 Overview	118
2.6.2 Modifications to the Heilmann et al. (2015) framework	120
2.7 Rationale for the study	122
Chapter 3. Methodology	125

3.1 Study design	125
3.2 Geographic location	126
3.3 Selection of population	126
3.3.1 Target population	126
3.3.2 Population source and setting	126
3.3.3 Study group	127
3.3.4 Exclusion criteria	128
3.3.5 Sampling and recruitment	128
3.3.6 Sample size calculation	131
3.4 Data collection instruments	131
3.5 Variables	132
3.5.1 Parent's self-administered questionnaire variables	132
3.5.2 Child's face-face interviews variables	137
3.5.3 Clinical examination	138
3.5.4 Outcome variables	144
3.6 Design of study instruments	153
3.6.1 Questionnaire design and translation	153
3.7 Ethical considerations	155
3.7.1 Permissions and liaison	155
3.7.2 Ethical principles	155
3.8 Conduct	158
3.8.1 Training	158
3.8.2 Equipment	159
3.8.3 Procedures	160
3.8.4 Personnel	163
3.8.5 Pilot study	163
3.8.6 Data transfer	164

3.9 Data analysis strategies	164
3.9.1 Descriptive	164
3.9.2 Data screening	165
3.9.3 Structural equation modelling (SEM)	166
Chapter 4. Results	183
4.1 Introduction	183
4.2 Response rate	183
4.3 Descriptive statistics	184
4.3.1 Socio-demographics	184
4.3.2 Current socioeconomic circumstances	185
4.3.3 Current family structure	187
4.3.4 Current OH behaviours	187
4.3.5 Current health: perceptions, behaviours, and status	190
4.3.6 Early life factors	194
4.3.7 Child's early learning	200
4.3.8 Child's school outcomes (Grade 2)	202
4.3.9 Parent's characteristics and social support	204
4.3.10 Paediatric oral health-related quality of life (POQL)	205
4.3.11 Clinical data	207
4.4 Reliability of the study measures	215
4.4.1 Reliability of the questionnaires	215
4.4.2 Reliability of dental clinical measures	216
4.5 Data screening	217
4.5.1 Missing values	217
4.5.2 Outliers	218
4.5.3 Data normality	218
4.6 Structural equation modelling	219

4.6.1 Measurement model	219
4.6.2 Parsimonious model	228
4.6.3 Summary of the results	245
Chapter 5. Discussion	246
5.1 Introduction	246
5.2 Summary of the key findings	247
5.3 Discussion of Objective 1	247
5.3.1 Prevalence of dental caries	248
5.3.2 Prevalence of EDD	257
5.4 Discussion of Objective 2	259
5.4.1 Early and current life factors linked to OHCs	259
5.5 Discussion for Objective 3	267
5.5.1 The association between OHCs and school performance	267
5.6 Discussion for Objective 4	277
5.6.1 Other predictors of school performance	277
5.7 Limitations and strengths	281
Chapter 6. Conclusions and recommendations	288
6.1 Conclusions	288
6.2 Recommendations	290
6.2.1 Recommendation for social determinants in early life	290
6.2.2 Recommendation for OH promotion	292
6.2.3 Targeted OH promotion for pregnant and new mothers	296
6.2.4 Targeted OH promotion for preschool- and school-age child	lren.297
6.2.5 Recommendations for better school outcomes	299
6.2.6 Recommendations for future research	301
Chapter 7. References	304
Chapter 8. Appendices	336

Appendix 1 Published paper	.336
Appendix 2. Search strategies for life course dental caries studies amo and adolescents (2012-2019)	ong children .337
Appendix 3. Amstar (Shea 2007)	.343
Appendix 4. JBI Data Extraction Form for review for the four systematic	c reviews .344
Appendix 5. The summary of the included studies: exposures, outcome results	es and .348
Appendix 6. Summary of the confounders in the included studies	.354
Appendix 7. Search strategies to update evidence related to the impact on children's and adolescent's school performance and attendance	ts of OHCs .358
Appendix 8. Muharraq's schools quality rating (2018-2019)	.362
Appendix 9. Participant Information Sheet and consent form	.363
Appendix 10. Parent's questionnaire (English)	.369
Appendix 11. Children's face-face questionnaire	.379
Appendix 12. Dental examination form	.381
Appendix 13. Weight and Height form	.382
Appendix 14. University of Sheffield ethical approval	.383
Appendix 15. Bahrain's Ministry of Education ethical approval	.384
Appendix 16. Bahrain's Ministry of Health ethical approval	.385
Appendix 17. University of Sheffield ethical amendments	.386
Appendix 18. MOE ethical approval for schoolchildren attendance	.387
Appendix 19. Summary of data screening	.388
Appendix 20. The initial CFA model	.392
Appendix 21. Standardised Regression Weights, total, direct and indire BC 95% CI, and SE for full model	ct effects, .393

List of Tables

Table 1. Life course studies to understanding dental caries indicators and risk factors
among children and adolescents (2012-2021) 43
Table 2. DMFT's mean threshold used in life course studies 47
Table 3. DMFT's mean threshold used to assess severity of dental caries in life course
studies
Table 4. Use of questionnaires and interviews in dental caries life course studies 49
Table 5. The quality assessment results of four systematic reviews using AMSTAR
(Shea et al. 2007)
Table 6. Summary of systematic reviews exploring the impacts of OH on school
performance and attendance in children and adolescents
Table 7. Studies included in the four systematic reviews (Ribeiro et al., 2018, de Paula
and Mialhe, 2013, Ruff et al., 2019, Rebelo et al., 2019)
Table 8. Country of origin of the studies included in the systematic reviews
Table 9. Study design of the studies included in the systematic reviews
Table 10. Sample size and age group of the studies included in the systematic reviews
Table 11. The result of the relation between OH measures and school outcomes from
Rebelo et al. (2019) metanalysis
Table 12. The risks of bias results from the three reviews (Rebelo et al., 2019, Ruff et
al., 2019, Ribeiro et al., 2018)
Table 13. Summaries of recent studies on the impacts of OHCs on school performance
and attendance among children and adolescents (2018-2021)
Table 14. Dental Trauma Index (UK-CDHS, 2013)
Table 15. Modified FDI, EDD index (Clarkson and O'Mullane, 1989)

Table 16. ICDAS codes for restoration (Ismail et al., 2007)	. 142
Table 17. ICDAS codes for caries severity (Ismail et al., 2007)	. 143
Table 18. PUFA codes (Monse et al., 2010)	. 143
Table 19. Grading classifications in the KoB	. 144
Table 20. Summary of the measurement and analysis of the study variables	. 145
Table 21. Unweighted Cohen's Kappa for first ICDAS code	. 158
Table 22. Weighted Cohen's Kappa for second ICDAS code	. 159
Table 23. Cut-off points for classification of constructs variables	. 175
Table 24. Demographic profile of 466 participants at baseline	. 184
Table 25. Participant's current socioeconomic circumstances	. 185
Table 26. Participant's household income and housing conditions	. 186
Table 27. Participant's current family structure	. 187
Table 28. Oral health behaviours of children and parents' participants	. 189
Table 29. Health perceptions, behaviours, and status of participants	. 191
Table 30. Participant's socioeconomic circumstances at the time of child's birth	. 195
Table 31. Participant's family structure at the time of child's birth	. 196
Table 32. Mother's demographic and health during pregnancy	. 197
Table 33. Children's birth outcomes and feeding practices in infancy	. 199
Table 34. Children's OH status and OH behaviours at 3-6 years old	. 200
Table 35. Children's early learning outcomes	. 201
Table 36. Children's school performance and school attendance for Grade 2	. 202
Table 37. Parent's characteristics and social support of participants	. 205
Table 38. POQL scores of 466 children (100%) child	. 206
Table 39. POQL items scored by 466 children (100%)	. 207
Table 40. Number of children and teeth examined	. 207

Table 41. Dental caries measures according to ICDAS and dmfs/DMFS 209
Table 42. ICDAS mean and total counts in primary and permanent dentition of 466
children
Table 43. Prevalence and most prevalent consequences of untreated caries in primary
and permanent teeth among participants 212
Table 44. Dental trauma in permanent incisors of children
Table 45. Malocclusion conditions among children
Table 46. EDD prevalence and most common type among children
Table 47. Prevalence of dental plaque among children
Table 48. Internal consistency of parenting style questionnaires
Table 49. Test-retest results for parenting style questionnaires by ICC 216
Table 50. Inter- and intra-rater reliability of dental clinical measures
Table 51. Latent variables distributions according to children's school performance and
school absence days
Table 52. The recommended and the obtained fit indices for the measurement model
using CFA
Table 53. Standardised regression weights of the measurement model: ß, R-square,
BC 95% CI, and bootstrapping SE according to the CFA model
Table 54. Fit indices for the full structural model and the parsimonious model 228
Table 55. Observed variables, according to children's school performance and school
absence days
Table 56. Significant direct and indirect relations between the variables in the
parsimonious model
Table 57. The specific indirect paths and mediators in the parsimonious model 234

List of Figures

Figure 1. Life course causal model, accumulation of risks: (a) dependant risk; (b)
accumulation. (Ben-Shlomo and Kuh, 2002)17
Figure 2. Chain of risk models with (c) additive effect; (d) trigger effect. (Ben-Shlomo
and Kuh, 2002)
Figure 3. Life course framework for oral health (Heilmann et al. 2015)
Figure 4. Prisma flow chart for identifying systematic reviews on the impacts of OHCs
on school performance and attendance in children and adolescents (2019)
Figure 5. The Bronfenbrenner (2005) bioecological model for child's development 103
Figure 6. Life course framework for OH (Source: Heilmann et al, 2015) 119
Figure 7. Modified from Heilmann et al.'s (2015) life course framework for OH 121
Figure 8. Flowchart of the sampling process
Figure 9. The proposed full model for the Structural Equation Modelling analysis . 173
Figure 10. Life course model (Ben-Shlomo and Kuh, 2002), critical period model; a)
independent risks; b) clustered risks 178
Figure 11. The accumulation of risks models 180
Figure 12. The relation between BSES and CSES and children's current OH 181
Figure 13. OH rating of parents, parent's rating of the child's OH, and child's own rating
Figure 14. Health rating of parents; parent's rating of the child health, and child's own
rating 192
Figure 15. Prevalence and type of chronic health conditions among children 193
Figure 16. Prevalence of overweight and obesity by gender
Figure 17. School performance per gender for the academic year (2019-2020) 203
Figure 18. School absence days per gender for the first term (2019-2020)

Figure 20. Distribution of ICDAS among 466 children 210 Figure 22. Measurement model with six latent variables and 20 observed variables Figure 23. Structural model of the life course framework for OH as proposed by Heilmann et al. (2015), after mapping latent variables from CFA adding observed Figure 24. Significant direct and indirect effects for the parsimonious model 231 Figure 25. The critical period model: the path estimates between BSES (a), birth outcomes (b), EDD (c) and child's OHCs independently and dependently from BSES, Figure 27. The association between BSES and CSES...... 240 Figure 28. The mediation effects of CSES between BSES and child's OHPs (a) and Figure 30. The mediation effect of treated teeth and dentine caries on the link between Figure 31. The mediation effect of dental caries on the link between BSES and school Figure 32. The mediation effect of parent's characteristics in the link between CSES Figure 33. The association between social and parental factors with children school

List of Abbreviations

Aberdeen Maternity Neonatal Databank	AMND
Adjusted Goodness-of-Fit Index	AGFI
Assessment of Multiple Systematic Reviews	AMSTAR
Attentive Deficit Hyperactivity Disorder	ADHD
Bahrain Ministry of Education	BMOE
Bahraini Dinar	BD
Bias-corrected	BC
Birth SES	BSES
Bleeding on Probing	BOP
British Association for The Study Of Community Dentistry	BASCD
Caries Assessment Spectrum and Treatment	CAST
Caries Management by Risk Assessment	CAMBRA
Child Oral Impact on Daily Performance	C-OIDP
Child Perception Questionnaire	CPQ8-10
Children's Dental Health Survey in the UK	CDHS-UK
Community Periodontal Index	CPI
Comparative Fit Index	CFI
Confirmatory Factor Analysis	CFA
Cooperation Countries	CC
Current SES	CSES
Dental Aesthetic Index	DAI
Deoxyribonucleic Acid	DNA
Direct Acyclic Graphs	DAG
Directorate of Government Schools	DGS
Dunedin Multidisciplinary Health and Developmental Study	DMHDS
Early Childhood Caries	ECC
Early Childhood Protein-Energy Malnutrition	EC-PEM
Early Longitudinal Study in Kindergarten Cohort	ECLS-K
Economic Development Board	EDB
Economic Freedom Index	EFI
Education & Training Quality Authority	BQA
Education Quality and Qualification Assurance	QQA
Enamel Developmental Defects	EDD
Fédération Dentaire Internationale	FDI
Focusing Resources on Effective School Health	FRESH
Fragile Families and Child Wellbeing Study	FFCWS

Goodness of Fit	GOF
Grey Matter Density	GMD
Gross Domestic Product	GDP
Government employee	GE
Gulf Cooperation Council	GCC
High Development Status	HDS
Human Development Index	HDI
Index of Orthodontic Treatment Need	IOTN
Infant Feeding Survey	IFD
Informed Consent	IC
Intelligent Quotient	IQ
Interclass Correlation Coefficients	ICC
International Caries Detection and Assessment System	ICDAS
International Federation of Gynaecology and Obstetrics	IFGO
Ischemic Heart Disease	IHD
Joanna Briggs Institute	JBI
Kingdom of Bahrain	КоВ
Life Course Health Development	LCHD
Low Birth Weight	LBW
Maternal Chronic Health Conditions	MCHCs
Maximum Likelihood	ML
Ministry of Education	MOE
Ministry of Health	МОН
Missing Completely at Random	MCAR
Modification Indices	МІ
Molar Incisor Hypomineralization	MIH
Mouth Wash	MW
National Health Regulatory Authority	NHRA
National Immunisation Survey	NIS
National Institute for Health and Care Excellence	NICE
National Survey of Children's Health	NSCH
Non-Communicable Diseases	NCD
Oral Health	ОН
Oral Health Conditions	OHCS
Oral Health Education	OHE
Oral Health Impact Profile	OHIP
Oral Health Practises	OHPs
Oral Health-Related Quality of Life	OHRQoL
Oral Hygiene Index Simplified	OHI-S

Oral Impacts on Daily Performance	OIDP
Paediatric Dental Service	PDS
Paediatric Oral Health-Related Quality of Life	POQL
Parental–Caregivers Perceptions Questionnaire	P-CPQ
Parenting Style Dimension Questionnaire	PSDQ
Preterm Birth	PB
Programme for International Student Assessment	PISA
Prostaglandin E	PGE
Public Health England	PHE
Pulpal Involvement, Ulceration, Fistula and Abscess	PUFA
Quality and Qualification Assurance	QQA
Quality of Life	QoL
Randomised Controlled Trial	RCT
Root-Mean-Squared Error of Approximation	RMSEA
Scale of Oral Health Outcome for 5-Year-Old Children	SOHO-5
Socioeconomic Status	SES
Specific Learning Disability	SPLD
Standardised Root Mean Residual	SRMR
Statistical Package for The Social Sciences	SPSS
Streptococcus Mutans	SM (S. mutans)
Structural Equation Modelling	SEM
Structural Model	SM
The Centre for Disease Control	CDC
Traumatic Dental Injury	TDI
UK Children's Dental Health Survey	UK -CDHS
United Nations Development Programme	UNDP
United Nations Educational, Scientific and Cultural Organization	UNESCO
United Nations International Children's Emergency Fund	UNICEF
United Nations Programme On HIV/AIDS	UNAIDS
United States Dollar	USD
University of Bahrain	UOB
Value Added Tax	VAT
Very Low Birth Weight	VLBW
United Arab of Emirates	UAE
United Kingdom	UK
World Llooth Organization	WHO

Chapter 1. Introduction

Despite a greater understanding of the aetiology and causes of OHCs, the global burdens of OHCs have continued (Peres et al., 2019). Moreover, it is estimated that these conditions have been exacerbated, especially in low-income countries (Kassebaum et al., 2017). Over 3.5 billion individuals worldwide suffer from OHCs, with untreated dental caries being the most significant public health problem (Watt et al., 2019). The concerns are even more severe across low-income countries where the prevalence of OHCs is increasing (Peres et al., 2019).

It should be mentioned that OHCs cause pain infection and impair people's quality of life, especially among the disadvantaged population (Martins et al., 2016). The impacts of OHCs may influence children's school outcomes, such as school performance and school attendance (de Paula and Mialhe, 2013). Such distress, especially in a child's early life, could irreversibly impact their current and later life course by restricting their education and future career (Banerjee, 2016). Thus, although attempts have been made to close the global gap in social inequality by investing in education and improving the health and OH of the population (Dornan and Woodhead, 2015), OHCs remain a significant health problem that might have a detrimental impact when fighting inequality. It has been suggested that promoting OH contributes positively overall to human development, including an individual's cognitive development (Dornan and Woodhead, 2015), by relieving the disease, and together with the economic and social burdens that OHCs cause (Watt et al., 2019).

A comprehensive framework to investigate health-related outcomes, named "life course", was developed and used in oral epidemiology (Halfon et al., 2018). The life course approach provides a theoretical framework to evaluate how social inequalities

and socially patterned exposures during the life span may affect health later. The framework examines the early life social, biological, behavioural and psychosocial exposures that may influence health through independent, cumulative and subsequent effects (Ben-Shlomo and Kuh, 2002, Kuh et al., 2004). The life course approach emphasises that children's early life social circumstances are essential for later development throughout the stages of development (Barker, 2001).

Through the life course approach lens, common risk and protective factors related to OH and education development could be highlighted. Accordingly, it would facilitate our recognition of the nature of the association and its' pathways which might be pertinent to improving OH and educational attainment. Therefore, this study examined the association of OHCs with children's school performance and attendance at Grade 2 in the city of Muharraq at KoB using Heilmann et al.'s (2015) life course framework for OH.

This thesis is structured as follows:

Chapter 1: Introduction

Chapter 2: Literature review, which outlines all of the relevant literature for the PhD study with the following objectives:

- Describe the emergence of the life course approach
- Outline the influence of early life circumstances on the development of adult's chronic diseases
- Identify the common concepts and principles of the life course approach
- Explain Ben-Shlomo and Kuh's (2002) life course models
- Describe the key stages of the life course affecting human development

- Review the application of the life course approach in oral health research
- Explain the association between life course factors and the development of dental caries in children
- Assess the available evidence on the association of OHCs with school performance and school attendance
- Provide an overview of the factors related to children's school performance and school attendance
- Explain the study's life course theoretical framework
- Describe the rationale of the study

Chapter 3: Methods which includes a description of the study design, ethical approvals, population, sample size, data collection and analysis

Chapter 4: Results including the descriptive statistics for all the clinical, demographic, study variables, alongside the findings from the structural equation modelling, which tested the life course framework

Chapter 5: Discussion of the findings, including the strengths and limitations of the study

Chapter 6: Conclusions and recommendations for research and practice

Chapter 7: References

Chapter 8: Appendices

1.2 Aim

To examine the association of OHCs with children's school performance and attendance at Grade 2 in the city of Muharraq at KoB using Heilmann et al.'s (2015) life course framework for OH.

1.3 Objectives

- To describe the prevalence of OHCs among 7-8 years old schoolchildren of the good, rated schools in the city of Muharraq.
- 2. To analyse the social, biological, behavioural, and parental pathways that link early and current life exposures with children's current OHCs by testing the critical, and the accumulation life course models with consideration of social determinants of OH
- To examine the association between OHCs and school performance and school attendance among schoolchildren, including the direct and indirect (mediated) pathways.
- To explore the early and current life course social, biological, behavioural and parental factors associated with children's school outcomes in addition to OHCs.

Chapter 2. Literature review

This chapter is presented in seven sections. Section 1 is an introduction to the life course approach. Section 2 reviews the application of life course in relation to oral health. Section 3 explains the association between life course factors and the development of dental caries in children. Section 4 assesses the available evidence on the association of OHCs with school performance and school attendance. Section 5 provides an overview of the factors related to children's school performance and attendance other than OH. Section 6 explains the study's life course theoretical framework, and Section 7 describes the rationale of the PhD study.

2.1 Introduction to life course approach

This introductory section on the life course approach addresses five objectives. First, it describes the origin and emergence of the life course approach. Second, it outlines the influence of early life circumstances on the development of chronic conditions in adulthood. Third, it identifies the common concepts and principles of the life course approach. Fourth, it explains Ben-Shlomo and Kuh's (2002) life course models. Fifth, it describes the key stages of the life course affecting human development.

2.1.1 Emergence of the life course approach

The life course approach has been used across several disciplines over the last 40 years by exploring how people's lives, transitions, environment, and social change shapes the individual's life from birth to death (Elder et al., 2003).

Early applications of the life course can be traced from 1918-1920 in the social sciences. William Thomas, a pioneer researcher in sociology, advocated the importance of longitudinal studies of life history in 1920 using data recorded throughout

life. He proposed studying many types of individuals concerning their experiences in different times and circumstances, and following them up in the future, with continuous documentation of their experiences. Through his book titled: *The Polish Peasant in Europe and America* (1918-1920), co-authored with Florian Znaniecki, a Polish immigrant from Chicago, they emphasised the importance of an individual's social network, believing that it was a vital factor regarding social change. He argued that Polish society shaped their culture and social networks rather than the policies found in the United States (US). The book was unique and considered a 'society scandal' at that time (Elder et al., 1985).

In the 1950s, Wright Mills proposed the study of the life course in behavioural science with the intention to explore the societal pathway of people's lives, particularly in their historical place and time. However, there was scant information concerning how people spent their lives from childhood to adulthood; neither the pathways of impacts on their course of development and ageing nor the importance of historical or geographical settings were acknowledged. Furthermore, World War II and the social and financial consequences concealed life course research for a time (Kuh et al., 2004). Consequently, Mill's proposal did not come to light at that time. Likewise, the thoughts of William Thomas began to be implemented in sociology in the 1960s by Elder and Clausen (Halfon et al., 2018). Subsequently, his initial thoughts became the imperative driver of life course and longitudinal research in most current disciplines, including sociology and medicine (Elder et al., 1985).

2.1.2 Early life circumstances and chronic diseases in adulthood

The epidemic of coronary heart disease and lung cancer raised concerns regarding the aetiological factors of specific chronic conditions as possible reasons for the incremental rates of these diseases in the inter-war period. Increasing attention was placed on the influence of social and lifestyle factors on adults' health (Kuh and Ben-Shlomo, 2004). However, the failure of the lifestyle model to explain the underlying causes of chronic conditions or clarify the social and geographical health disparities stimulated some investigators to explore the contributions of early life exposures to the development of chronic health conditions (Kuh et al., 2004).

The assumption that an individual's early life circumstances may influence an adult's health status and mortality was not new. It was the main view of public health in Britain during the Boer War in the first forty years of the 20th century and substantially influenced early social reforms to enhance maternal and child health care (Gilbert, 1993).

Thus, the first attempt to link environmental factors during early life to diseases during adulthood was raised by ecological investigations using official mortality statistics, which showed geographical associations between past mortality rates and current adult mortality rates rather than cohort analysis (Kuh et al., 2004).

In 1977, Forsdahl provided the first evidence relating early life factors to susceptibility to disease during adulthood. He explored the association between living standards using the infant mortality index and mortality rates from ischemic heart disease in Norwegians. Geographic area was associated with infant mortality and current adult mortality rates among those born in the same year. It was observed that high infant mortality in the same cohort exhibited high mortality rates during middle age. In conclusion, living in a poor environment during childhood followed by wealth was associated with high mortality rates of ischemic heart disease later in life (Forsdahl, 1977). Furthermore, the observations hypothesised that the weakest infants died

during infancy, whilst the healthy ones survived. However, unfavourable living circumstances in early life carried lifelong susceptibility to diseases that increased the likelihood of acquiring other risk behaviours during their life, suggesting the accumulation of risks over the life course (Forsdahl, 1977).

Studies from the UK also replicated Fordahl's findings (Barker and Osmond, 1986, Barker and Osmond, 1987). The studies by Barker and Osmond (1986) revealed a significant geographical relation between ischaemic heart disease mortality rates in 1968-1978 and neonatal and postnatal mortality in 1921-1925 and between, stroke mortality rates in 1968-1978, and maternal mortality rates in 1911-1914 (Barker and Osmond, 1987). Other demography researchers adopted the same ecological approach to explore whether early life exposures were related to high death rates later in life in Sweden (Fridlizius, 1989), Italy (Bengtsson and Lindström, 2000), US (Caselli, 1990), and Japan (Okubo, 1981).

In addition, others discovered a reverse relation between adult height and different causes of mortality rates from cardiovascular and respiratory diseases. The causes were attributed to earlier life circumstances since height is a natural process determined in childhood and adolescence (Kuh et al., 2004). A cohort study in Finland argued that mortality rates from coronary heart disease, myocardial infarction, and IHD related to small height in adulthood were influenced by childhood malnutrition or hereditary factors (Notkola et al., 1985). Similar results were observed in the UK by the Whitehall study (Marmot et al., 1984) and in Norway (Waaler, 1984).

However, the above studies have been criticised as they failed to separate poverty in early life from the one through life (Elford et al., 1992). Similarly, the geographic association between past death rates and present adult death rates logically depicted

persistence in the geographic distribution of poverty (Ben-Shlomo and Kuh, 2002). To overcome these flaws, Barker and colleagues adopted a retrospective study design to examine the pathways between early life circumstances and adults' chronic conditions. They recruited men and women in middle age whose birth records were available in three areas in England (Hertfordshire, Preston and Sheffield). Data from the Hertfordshire study, involving 16,000 individuals born between 1911-1930, revealed that low weight at one year of age was associated with high mortality rates from IHD in adulthood (Barker and Osmond, 1986).

Moreover, Barker and Osmond (1986) discussed the pathways between prenatal and postnatal exposures and IHD later in life. Their research findings highlighted the role of several mediators, including plasma cholesterol concentration, high blood pressure, diabetes and glucose tolerance, body fat and fibrinogen concentration in plasma. Research in Sweden (Leon et al., 1998), Finland (Eriksson et al., 2001), the US (Rich-Edwards et al., 1997), and India (Stein et al., 1996) confirmed those findings.

Barker (1990) illustrated the cornerstone of 'biological programming' or the 'Foetal Origins' hypothesis, through which he argued that deprivation and undernutrition during the prenatal and postnatal period (critical period) were the principal reasons behind the development of coronary heart diseases later in life (Barker and Osmond, 1986). The hypothesis suggested that environmental factors can interact with the body's cells, genes and tissue growth, resulting in permanent health impacts later in life.

Roseboom et al. (2001) supported Barker's hypothesis from their study on the impacts of the Dutch Famine during the gestational period on the development of chronic conditions in later life. A cohort study conducted in the Netherlands involving 2,414

participants born between 1943 and 1947 demonstrated that people who experienced the famine during the first and mid-trimesters of gestation were at greater risk of having diabetes, hypertension, and obesity than those who experienced the famine in the third trimester of gestation. Hoek et al. (1998) observed a high rate of schizophrenia among people exposed to famine during gestation. Other studies also investigated the effect of famine during the gestation period on chronic conditions. However, they did not find such an association in Finland (Kannisto et al., 1997) and Leningrad (Stanner et al., 1997).

Nonetheless, Barker's hypothesis has been criticised. Paneth and Susser (1995) argued that Barker's hypothesis was too broadly defined and vague. They argued that proposing nutritional intake during gestation affected disease risk in adulthood permitted investigators to examine the influence of several nutritional measures on any diseases. Specifically, such a hypothesis can generate 'Type-I' errors (Paneth and Susser, 1995). Pedhazur (1991) criticised Barker's hypothesis for being difficult to replicate, which is considered crucial in scientific theory because the hypothesis was a general formulation. However, further modifications to the original hypothesis have addressed several of these criticisms, including explaining the hypothesis concerning specific diseases and replication in animal models (Barker, 2001).

Barker's hypothesis has also been criticised for evaluating the association between birth's anthropometric measures and health status in adulthood without adjustment for relevant confounders (Susser and Levin, 1999). Some later investigations have controlled for possible confounders and confirmed the initial findings (Barker, 2001, Huxley et al., 2007). Further, a meta-analysis suggested the occurrence of publication bias concerning the inverse association between birthweight and high blood pressure (Schluchter, 2003). However, the re-evaluation of the data with control for the

estimated bias determined that the association weakened but remained significant (Skogen and Overland, 2012).

Moreover, incomplete sample, attrition on follow-up, and missing records were significant criticisms of the initial studies of Barker's hypothesis since the health records of forty- to fifty-year-old people were used to analyse the association between early life circumstances and chronic health conditions (Susser and Levin, 1999). Despite initial debate in this area, further studies with high follow-up rates, using appropriate research designs and controlling for confounders confirmed the correlation between social deprivation in early life and chronic health conditions (Kuh et al., 2004). Eighty studies revealed a significant association between low birth weight and high blood pressure (Huxley et al., 2000), whereas thirty-eight investigations showed the link between low birth weight and insulin resistance (Hales et al., 1991).

2.1.3 The concepts and the principles of the life course approach

Many researchers have defined life course as a concept, while others consider it a theoretical framework. As a concept, a life course can be defined as "a sequence of socially defined events and roles that the individual enacts over time" (Giele and Elder, 1998, p. 22). Thus, the concept of life course suggests age-differentiated social events discrete from the life span and life-cycle concepts (Elder et al., 1985). Life span is the length of an individual's life rather than age-related social events (Baltes, 1987). Conversely, a life cycle is related to life's event stages or the transformation from one generation to the next in a population (Elder et al., 1985).

However, Elder's recent writings on this subject defined life course as a theoretical framework focused on studying human lives, development and ageing (Elder, 1998).

Thus, five principles were derived from the social and behavioural sciences research using the life course as a framework.

The first principle relates to historical time and location. As research started to study family and individual life pathways, it has been observed that people born in different years confronted different historical events with different life opportunities. Thus, historical time and geographic location may produce cohort effects when social transition impacts one cohort differently from the following cohorts (Elder, 1985). For example, Elder's (1974) longitudinal study Children of the Great Depression found that the influence of family hardship on the life course of the Berkeley cohort (1928-1929) who were born or were young during the time of the economic depression was more extreme than the Oakland cohort (1920-1921) who were older or at late teenage years.

The second principle of the life course is timing. Developmental antecedents, events, the effect of transitions, and behaviours patterns fluctuate depending on their timing in the individual's life (George, 1993). The same events may influence people differently depending on when they happen in the life course and developmental stage (Elder et al., 2003). Lee and Ryff (2016) concluded that women who entered parenthood early, particularly teenagers, were more likely to have heart problems, economic difficulties, depression, and a higher risk for smoking and obesity; indicating that life transitions should occur within a specific period so that human development is compatible with those changes. The transition is a different life event within a person's trajectory, whereas the trajectory is a lifelong pathway with temporal sequences of development within a social institution like family or education (Elder et al., 1985). The transition can be stable or change over time and varies between individuals according to economic, environmental and social factors (Halfon et al., 2018). Thus, the life course intends to explain the societal pathways of these transitions and trajectories.

The third principle is social networks. Life course acknowledges that lives are connected and lived interdependently via social networks. Social and individual experiences are connected via the family and its shared interrelationships (Elder, 1998). Elder's (1974) study of children issued during the Great Depression revealed that parents' financial pressures resulted in depression and marital conflicts, negatively affecting their children's ability to nurture. Consequently, their children experienced many emotional, academic, and behavioural problems. Therefore, this emphasizes that one person's life changes can affect the lives of other family members.

The fourth hallmark of the life course is human agency. In life course, people are active agents in their own development (Featherman, 1981). Individuals may influence their own life course through the decisions they encounter and their choices within the contexts of history and social circumstances (Elder, 1998). These choices, however, are highly determined by the available opportunities faced during the life course. Furstenberg et al. (2000) claimed that the environment does not passively draw people to individual choices; they can still have control and discover alternatives. They illustrated that by analysing urban families who lived in disadvantaged environments with limited resources, poverty, and violence. However, despite such adversities, they could raise their children away from this by considering other environments like churches and youth clubs.

The last tenet concerning life course is ageing. Human development defines ageing as an everlasting process that does not stop at eighteen (Baltes, 1987). In effect, transitions from childhood can shape an individual's life forty to fifty years later (George, 1993). Thus, studying lives over a long period is essential to understand the impact of the interplay between social changes and human development.

2.1.4 Conceptual models in life course epidemiology

In epidemiology, Kuh et al. (2004, p. 1) defined life course as "the study of long-term biological, behavioural and psychosocial processes that link adult health and disease risk to physical or social exposures acting during gestation, childhood, adolescence, earlier in adult life, or across generations". This definition stemmed from Barker's hypothesis and the lifestyle approach (Kuh et al., 2004). The life course approach acknowledges the interrelationship between biological and social risk processes as determinants of health rather than separating them. Its aim is to investigate the early and later life exposures together to recognise risk and protective factors across the life course. It suggests that several biological and social factors may act independently and cumulatively throughout life to influence adult health (Kuh et al., 2003).

Considering the different determinants of health, the significance of timing and duration of those factors might affect health in various ways. Thus, Ben-Shlomo and Kuh (2002) proposed several life course models to explore the links between exposures throughout the life course and health outcomes later in life. The following section explains the well-established life course models in health, including critical period, accumulation of risk and the chain of risk models.

2.1.4.1 Critical period model

The critical period model acknowledges that health discrepancies observed in a different social group are due to exposures during a critical development period. Such exposures may include social deprivation, malnutrition, smoking, or other risk factors. The timing of exposure to environmental variables plays an essential role in influencing health outcomes (Kuh et al., 2004). Moreover, this model considers the intrauterine
period and childhood as critical or sensitive periods for developing diseases throughout life.

Ben-Shlomo and Kuh (2002) differentiated between the critical and sensitive periods. They defined the critical period as "the limited time window in which an exposure can have adverse or protective effects on development and subsequent disease outcome". In contrast, the sensitive period is "the time period when an exposure has a stronger effect on development and subsequent disease risk than it would at other times" (Ben-Shlomo and Kuh, 2002, p. 288). The critical model proposed by David Barker is also known as the 'programming' or 'latency model'. The model suggested that inadequate diet during the gestational period was critical, potentially leading to physical alteration to some of the body's organs or structure, producing irreversible or permanent deviation in later life (Barker, 1990). The latency model correlates exposure at one stage in the life course with the possibility of health outcomes in later life regardless of transitional events in life (Hertzman and Power, 2003).

2.1.4.2 Critical period model with later effect modifier

The critical period model with a later effect modifier is a modified version of the critical period model that presumed that physiological or psychosocial risks in later life might increase the influence of exposure via a critical period of development (Kuh et al., 2004). For example, it was suggested that the associations of high blood pressure, coronary heart disease, and insulin resistance with low birth weight were higher among obese adolescence or adulthood (Lynch and Smith, 2005).

2.1.4.3 Accumulation of risk model/cumulative model

The accumulation model hypothesises that health conditions in later life result from the accumulation of risks throughout the life course regardless of their timing (Ben-Shlomo and Kuh, 2002). This model explores the role of risk accumulation during the life course on health outcomes, considering the number of risks, exposure duration, and interactions between them.

Accordingly, risk factors can be independent (Figure 1a) or clustered (Figure 1b) in socially predicted ways. For instance, disadvantaged pregnant women were more likely to deliver a child with low birth weight, grow up in a poor environment, be exposed to passive smoke, have inadequate nourishment, and experience poor educational attainment (Ben-Shlomo and Kuh, 2002). The risk may accumulate differently in a birth cohort, specific cohort or when a generation was exposed to disadvantaged situations in a specific period, resulting in undesirable health outcomes in the entire cohort. It could be an incident-based event such as the Dutch Famine or a sudden social and economic change like the 'baby boom' generation (Smith, 2007).

The cumulative risk model may follow a dose-response association since the extent of accumulated damage increases if the amount, length and severity of exposures increases. Additionally, the ageing process limits the ability to cope with the harmful effects of exposures (Kuh et al., 2003).



Figure 1. Life course causal model, accumulation of risks: (a) dependant risk; (b) accumulation. (Ben-Shlomo and Kuh, 2002)

2.1.4.4 Chain of risk model

The chain of risk model designates a sequence of linked factors that contribute to impaired function and increase disease risk because one factor introduces the next one and then another (Ben-Shlomo and Kuh, 2002). Each risk factor in the chain can increase the subsequent risk and have an independent 'additive effect' on the latter function or health outcomes (Figure 2c). Alternatively, 'trigger effect' defines a chain of risk where only the final exposure in the chain impacts health outcomes (Figure 2d). For instance, unemployment cause's financial problems, which may result in losing one's house. Besides, family conflicts can lead to abuse and neglect, resulting in psychosocial stress and subsequently poor health-related behaviours like alcohol addiction, inadequate diet, and ultimately chronic health conditions (Ben-Shlomo and Kuh, 2002).



Figure 2. Chain of risk models with (c) additive effect; (d) trigger effect. (Ben-Shlomo and Kuh, 2002)

2.1.5 Key stages of the life course affecting health

Understanding the key stages of peoples' lives relevant to their health is required within the life course approach. This section highlights the key stages of human development from pregnancy to adulthood.

2.1.5.1 Pregnancy period

Barker's hypothesis raises awareness regarding the factors associated with foetal growth and patterns, including maternal and perinatal health roles on later health outcomes (Kuh et al., 2004). Those factors include malnutrition, smoking and alcohol, chronic health conditions, and age during pregnancy.

Unfavourable conditions like malnutrition during gestation can produce epigenetic changes to the foetus's development, increasing the risk of diseases in later life

(Barker, 1998, Halfon et al., 2018). Epigenetics was defined as the "changes to an individual's genetic code that can alter gene expression without changing the DNA sequence passed from one cell generation to the next" (Waddington, 2012). Maternal nutritional status is considered a vital determinant of fetal growth before, during and after pregnancy. The International Federation of Gynaecology and Obstetrics (IFGO) suggested that nutrition during the prenatal period is a steppingstone for maternal and child health development. In turn, this adds better health outcomes to the next generation, assuming a healthy lifestyle before conception (Hanson et al., 2015).

Poor SES can result in malnutrition, undesirable pregnancy outcomes, and less optimum foetal and child development (Inskip et al., 2009). For instance, the lack of folate during pregnancy can change the epigenetic status of the foetus (Aisling et al., 2016), resulting in LBW (McGee et al., 2018). Conversely, a systematic review reported a two-fold rise in folate consumption linked with a 2% gain in newborn's weight (Fekete et al., 2012). Micronutrient deficiency during pregnancy, like vitamin B12, was linked to LBW and risk of diabetes in later life (Yajnik et al., 2008). Inadequate iron and calcium intake has been associated with maternal death, while maternal anaemia through pregnancy has been associated with LBW (Black et al., 2013).

Social disadvantages were also linked with unhealthy lifestyles during pregnancy, including smoking, high sugar consumption and alcohol misuse (McGovern, 2013). Maternal smoking can reduce thirty grams of the baby weight at birth, while non-smoking can increase the baby weight by 180 grams (Durmuş et al., 2011). Overall, smoking during pregnancy accounted for 40% of infant mortality in the UK (Department of Health - UK, 2007). Alcohol consumption has also been linked to risk factors associated with congenital anomalies like heart defects (Liu et al., 2015). Both smoking

and alcohol consumption have been significantly correlated with miscarriage, preterm delivery and LBW (Stephenson et al., 2011).

Other maternal related factors correlated with fetal growth impairment during pregnancy is maternal chronic health conditions (MCHCs), such as gestational diabetes, obesity, hypertension and kidney diseases. Gestational diabetes can cause epigenetic changes that increase the risk of premature delivery, LBW and type 2 diabetes in adulthood (Darnton-Hill et al., 2004). High maternal insulin levels during pregnancy might also increase the insulin level in offspring, which can continue during childhood (Dabelea et al., 2000). Evidence has illustrated that obesity during pregnancy could cause epigenetic changes in the mother and the foetus, increasing the risk of obesity during childhood (Yuanyuan, 2018). Likewise, maternal hypertension and renal diseases have also been linked to developmental programming and, therefore, intrauterine growth impairment, LBW and preterm birth, hypertension in later life and low nephron numbers (Luyckx et al., 2017).

Whether young or old, maternal age during pregnancy is another cause of concern (McCall et al., 2015). Population's analysis in Europe from 1950 to 2010 using data from the Aberdeen Maternity Neonatal Databank (AMND) confirmed that although teenage pregnancy rates had reduced, in the period mentioned above, deprived areas still sustained higher rates of teenage pregnancies (McCall et al., 2015). Teenage pregnancy is connected with many adverse birth and health outcomes due to insufficient maternal physiological and psychosocial development and young women not being prepared for pregnancy (McCall et al., 2015). Besides, young women's reproductive systems' immaturity and nutritional status are in flux (Scholl et al., 1990). Likewise, Chen et al. (2007) contended that reduced foetal growth in young pregnant women resulted from nutritional competition between the growing young body of the

mother and the foetus for growth. However, low SES status can be attributed to many unhealthy behaviours in young people like smoking, alcohol misuse and inadequate diet, which might influence adverse birth outcomes independently (Smith and Pell, 2001). Also, evidence has revealed that advanced maternal age was associated with various adverse pregnancy outcomes, including LBW, preterm delivery, foetal death, and high caesarean section rates (Blomberg et al., 2014).

A recent cohort study completed in Saudi Arabia compared pregnancy outcomes between young and older women in a sample of 14,514 women (Fayed et al., 2017). The study found that younger women (age < 20 years) were more likely to be educated than those aged 40 years or older. The former group were also more likely to have a vaginal delivery and preterm birth. Pregnant women at higher ages were less likely to be educated and more likely to have childbirth by caesarean section, have gestational diabetes and preterm delivery than those younger than 20 years (Fayed et al., 2017).

2.1.5.2 Infancy period

Children acquire essential developmental skills according to their physical, social, and emotional development during infancy and early childhood. This should involve healthy eating styles and activities, besides developing self-regulation skills, language, and cognitive and learning skills (Jacob et al., 2017).

Infancy is defined as the period from birth until eighteen months, while toddlers span from eighteen months to three years of age (Erikson, 1980). According to Erikson, infancy is when a child develops trust and hope. Children during this period require care and visual and sensible attention from parents. A lack of those attributes can result in low self-esteem, and a sense of insecurity, besides the risk of developing temper tantrums and a stubborn attitude or antisocial behaviour (Erikson, 1980). Children start

to gain body control during the toddler stage and develop independent skills. Successful completion of this stage can enhance self-esteem in children, while the opposite can cause an unstable personality (Erikson, 1980).

One of the main predictors of adequate physiological and psychosocial development during this period is breastfeeding. The World Health Organization (WHO) classified breastfeeding as exclusive breastfeeding when the infant is fed with breast milk only until six months of age or predominant breastfeeding when other liquids or solid foods are also used fed to the infant besides breast milk. The WHO and United Nations International Children's Emergency Fund (UNICEF) recommend that breastfeeding should commence during the first hour after birth, and exclusive breastfeeding should be the only source of food during the first six months and continue until two years of age or more (Walters et al., 2016). Breastfeeding can enhance a newborn's sense of security by establishing a solid bond between the infant and the mother (Bai et al., 2009). Breastfeeding can be a protective factor for infants and mothers against undesirable health outcomes. Evidence regarding breastfeeding and child's health has revealed that breastfeeding decreases the likelihood of ear infections, gastroenteritis, respiratory infection, dermatitis and sudden infant death. It can also reduce the risk of asthma, obesity, diabetes type 1 and diabetes type 2 in early childhood. Furthermore, breastfeeding may protect the mother against maternal breast cancer and ovarian cancer (Chowdhury et al., 2015). However, the same study concluded that formulafeeding practices could increase ear infections by 100%.

Although breastfeeding is considered one of the oldest and least expensive ways to feed a newborn baby, different factors influence breastfeeding practices and breastfeeding duration. These factors can include the mother's age, ethnicity,

education level, occupation, type of delivery, night feeding, social support and use of contraceptives (Meedya et al., 2010).

Data from the UK Infant Feeding Survey (IFD) in 2005 indicated that ethnicity and SES significantly predicted breastfeeding practices and duration. The survey concluded that Asian, African and Chinese women were more likely to engage in breastfeeding practices than Caucasian women. African women had the highest prevalence of exclusive breastfeeding and a longer duration of breastfeeding (until nine months of age). Moreover, mothers over thirty years exhibited the highest prevalence of exclusive breastfeeding than younger ones. Similarly, the higher the years of education, the higher the breastfeeding practices and duration (Meedya et al., 2010).

Social support can positively influence the initiation and continuation of breastfeeding. A prospective cohort study involving 1,069 Australian women has shown that social support and encouragement from their partners positively influenced women's breastfeeding initiation while at the hospital and its continuation afterwards (Scott et al., 2001). It is worth noting that professional support for working women also reinforced breastfeeding practices and duration (Trivedi, 2018).

Breastfeeding is a maternal behaviour that also can be influenced by religious practices and cultural norms. For example, in Islam, women are encouraged to breastfeed until the child is two-years-old by a biological mother or a lactating woman if the former cannot provide it. This possibly explains the high prevalence of breastfeeding until six months (98%) among Muslim countries (Anggraeni et al., 2020). The decision to discontinue breastfeeding should be made by couples considering practical reasons (Anggraeni et al., 2020).

Cultural norms can also enforce or inhibit breastfeeding practices. For instance, in India and some areas of Turkey, women believe that colostrum is a harmful and dirty material. Therefore, breast milk is thrown away rather than given to the infant. In Lebanon, ordinary people believe that abdominal cramps women experience is transmitted to the infant when the child is breastfed. In Egypt, there is a common belief that breastfeeding in public can expose the mother and infant to an evil eye, whereas in Kenya, breastfeeding is encouraged but only after ritual cleaning (Osman et al., 2009).

2.1.5.3 Early childhood

Early childhood is from 3 to 6 years old (Erikson, 1980). Children during this period are also susceptible to environmental or economic disadvantages in developing diseases. At this age, children begin to follow their parents' behaviours, who are their role models. Adequate care can increase a child's self-initiative. Nonetheless, children may experience self-guilt and weak personalities (Erikson, 1980).

It has been suggested that childhood is a critical period since exposure might have an enduring influence in later life. A cohort study in Ireland assessed the impact of improvements in public health services, such as water and sanitation, on children's health after World War II. They found that upward social mobility during childhood dramatically decreased mortality and disability in adulthood (Delaney et al., 2011). Another longitudinal study conducted in poor villages in Guatemala ascertained a significant association between health conditions during early childhood, including diarrhoea, pulmonary diseases, anorexia and other infectious diseases, besides adulthood heart disease and type 2 diabetes, and low SES circumstances (Margolis, 2008).

Interventions targeting this age group may prevent many chronic health conditions, including sedentary lifestyles and obesity in adulthood (Mark et al., 2017). The Canadian Society for Exercise for Physiology suggested that children aged 1-4 years should have at least three hours of physical activity spread out over the week to achieve positive health outcomes like musculoskeletal, motor, adiposity, psychosocial and cognitive skills (Mark et al., 2017).

2.1.5.4 Middle childhood

Middle childhood referring to 6-12 years-old children, is considered one of the foremost critical periods in childhood development, as manifested by the eruption of the first permanent molars and androgen released by the adrenal gland (Bogin, 1997). The period is characterised by slow body growth, rapid brain development, increased muscle mass gain and adiposity of fat, with sex features in the body structure becoming distinct (Halfon et al., 2018).

The stage is manifested by increased perceptual and motor skills and problem-solving and reasoning skills. However, the most distinct features happen in the domain of selfregulation: they become much more capable of resisting undesirable behaviour, preserving constant attention, creating and following plans (Weisner, 1996). School plays an integral role in skills and knowledge acquisition and psychological development during this period due to social interactions with peers and school faculties. During middle childhood, children develop friendships outside the family environment, build their self-competence, and reduce feelings of inferiority (Erikson, 1980).

The adrenarche at this stage plays a role as a developmental switch (regulator) that enable adaptive plasticity (Halfon et al., 2018). West-Eberhard (2003) defined a switch

as "a regulatory mechanism that activates at a specific point in development, collects input from the external environment or the state of the organism, and shifts the individual along alternative pathways, ultimately resulting in the development of alternative phenotypes". In contrast, plasticity is an organism's capacity to adjust its phenotype as a reaction to environmental stimuli (West-Eberhard, 2003). For instant, a switch may determine the acquisition of aggressive behaviour; thus, living in a broken family environment might trigger aggression, while a stable family environment entrains the development of a low level of aggression (Del Giudice et al., 2009). Concerning health, the middle childhood period appears as a switching period for further growth and metabolic functions that may leave enduring influences on health, like the risk of obesity and type 2 diabetes (Hochberg, 2010). Thus, it has been argued that metabolic changes in middle childhood reflect both the effect of the early environment and the stimulation of new genetic factors (Halfon et al., 2018). Consistent with this, Day et al. (2015) found a significant association between genetic and puberty timing, diabetes type 2 and heart disease.

2.1.5.5 Adolescence

Adolescence is the continuous period marked by rapid brain growth and hormonal changes, accompanied by physical, emotional and psychological changes (Patton and Viner, 2007). The WHO (2014) defined adolescence as the age group between ten and nineteen years old and the transition period from childhood to adulthood. Erikson defined this age group as 12-18 years, corresponding to the time when a person is looking for his/her identity and experiences a shift from being attached to parents to peer influence by seeking to develop independence and build their personality (Erikson, 1980). Adolescence can be a critical period affected by previous and current emotional and physiological changes that influence health and wellbeing (Patton and

Viner, 2007). Adolescence is also an exceedingly challenging period, as the individual has the physical features of an adult, but their psychological development is not yet concluded.

Blakemore et al. (2010) suggested that hormones can affect brain function. The limbic system, which controls one's appetite and desires, is established prior to the prefrontal cortex that regulates decision-making functions (Blakemore et al., 2010). Consequently, adolescence is a crucial period for adopting many risky behaviours, such as smoking and alcohol misuse, that, if not prevented, can be maintained through life (WHO, 2014). Most adult smokers start smoking before nineteen years of age (UK Parliament Department Of Health, 2010) and one in four teenagers reported alcohol consumption (Viner et al., 2015).

Psychological problems like depression and anxiety are frequently initiated by fourteen years (UK Parliament Department Of Health, 2010). If ignored, they can influence health and social life in adulthood (WHO, 2014). In addition, puberty associated with sedentary activities, such as watching TV for a long time, unhealthy snacks and responses to advertisements, especially in females, may contribute to obesity. The initiation of a sedentary lifestyle during adolescence may have future health consequences since eight out of ten obese teenagers remain obese in adulthood (UK Parliament Department Of Health, 2010).

2.1.5.6 Adulthood

Adulthood is the period from 18-64 years old. On the one hand, it is the period of love and developing commitments: engaging in healthy associations produces a happy ageing life. Conversely, failing to engage in healthy associations might result in isolation, loneliness and depression (Erikson, 1980).

During this period, most health-related risk factors from early life, childhood, and adolescence accumulate and manifest, resulting in undesirable health conditions (Kuh et al., 2004). Poor SES have a significant impact on adults' health behaviours. People with poor SES are more likely to have an inadequate diet, smoke, and a sedentary lifestyle, and consequently, are at a greater risk of chronic diseases (UK Parliament Department Of Health, 2010). Approximately half of all deaths related to chronic diseases affected adults aged seventy years or younger, and 30% affected people younger than 60 years (Jacob et al., 2017). Other risks that cause concern at age 50-69 years old are smoking, second-hand smoking, high blood pressure and insufficient intake of vitamins (Lim et al., 2012).

Further, the influence of poor SES on an adult's health can cause suicide attempts, psychological disorders, and heart disease (UK Parliament Department Of Health, 2010). Marmot et al. (1991) concluded that the lower the occupational status, the higher the probability of being exposed to risk factors for heart diseases, high blood pressure, stress and mortality.

2.2 Life course and oral health

This section aims to review the application of a life course approach in OH. It describes the main reasons for applying life course in OH research and provides an overview of Heilmann et al.'s (2015) life course framework. Subsequently, it discusses the possible factors associated with children's OH according to life course considering social and family aspects. Finally, it explains the association between life course factors and dental caries in children.

2.2.1 Oral health over the life course

The life course approach is considered applicable to OH research due to five main features (Nicolau et al., 2007). First, the main OHCs are prevalent chronic conditions; thereby, retrieving a representative cohort is practicable. Dental caries is often the most prevalent chronic condition, affecting 3.5 million people globally (Kassebaum et al., 2017). Second, OHCs are cumulative and can affect people from early childhood to adulthood (Heilmann et al., 2015). Thus, a comparison between individuals could be drawn based on disease severity and extent rather than counting all who develop the condition together (Halfon et al., 2018). Third, OHCs can be easily detected through clinical examination even when the individual fails to report them (Halfon et al., 2018). Fourth, reliable and valid measures to detect OHCs are available and affordable. Fifth, oral conditions have a significant adverse impact on guality of life and wellbeing (Martins et al., 2016). The recent review of the global burden of disease emphasised that those OHCs accounted for 16.9 million disability-adjusted life years globally (Kassebaum et al., 2017). Untreated dental caries in the primary dentition of preschool children can be the cause of pain, discomfort and infection (Wong et al., 2012). Thus, it can disturb their eating and sleeping habits and disturb children's attention in school, eventually affecting their academic performance (Sheiham, 2006). Also, severe dental caries may adversely affect a child's growth and development (Duijster et al., 2015). Besides the adverse impact of oral conditions on one's everyday life, OHCs can, directly and indirectly, impact the economy. Petersen (2003) estimated that OHCs are the fourth most expensive condition to manage (direct cost) in industrialised countries. The WHO (2010) estimated the economic impact of the global burden of OHCs as US\$442 billion in 2010, with direct estimated treatment costs worldwide equal to US\$298 billion yearly. The indirect costs associated with loss of productivity were estimated to be US\$144 billion yearly (Listl et al., 2015).

Thus, OHCs are considered a public health problem, and therefore, an investigation of oral conditions is ethically sound and can have economic consequences.

Studying OH using a life course approach provides a comprehensive explanation of risks and protective factors of OH diseases because it considers the interaction of socioeconomic, behavioural and psychosocial exposures over time (Kawachi et al., 2002). Using a Life courses approach provides a framework that conceptualises the OH and OH development process in a unified definition. OH within a life course is "the capacity to adapt to ubiquitous environmental and microbial challenges, maintaining the robustness of teeth and other structures in the mouth, and OH development refers to the pathways and trajectories of change of these assets over the life course" (Halfon et al., 2018, p. 301).

2.2.2 Early life factors and oral health in adulthood

Most of the empirical OH research that adopted the life course approach has explored the association between childhood SES and an OH in adulthood. Considering the scarcity of prospective cohort studies, most of the available evidence stemmed from three birth cohort studies: the Newcastle Thousand Families 1947 birth cohort in England, the Dunedin Multidisciplinary Health and Development Study (DMHDS) in New Zealand (1972–1973), and the Pelotas birth cohorts in Brazil (1982 and 1993) (Heilmann et al., 2015). This section briefly explains the association and the pathways between childhood SES and OH in adulthood and the impact of social mobility on OH.

2.2.2.1 Childhood SES and oral health in adulthood

Poulton et al. (2002) examined the association between childhood SES and health status, including the OHCs at 26 years, in a cohort of 1000 participants from the Dunedin study. SES of the cohort had been assessed according to the parent's occupation status recorded at birth and seven life stages (3, 5, 7, 9, 11, 13, and 15 years). At twenty-six, the adult's physical, mental and dental health was examined. The results revealed that all OHCs, such as dental caries, dental plaque, periodontal disease and bleeding status, were linked with childhood SES after adjusting for adult SES. As SES increased, the level of OHCs decreased. Similar results were also documented by Thomson et al. (2004).

These results contrast with studies that examined the relations between childhood SES and tooth retention and OHRQoL at age 50 in the Newcastle Thousand Families cohort (Pearce et al., 2004, Mason et al., 2006). To illustrate, Pearce et al. (2004), with a sample of 337, evaluated the significant risk factors during childhood and adulthood that could influence tooth retention at age 50. The difference in the number of retained teeth was mostly related to adult risk factors like SES, smoking, and alcohol consumption. After adjusting for adult SES, childhood SES predicted tooth retention among middle-aged women only, and the association was no longer statistically significant. Within the same sample, childhood SES was associated with OHRQoL among men. However, the association was mediated by adult SES (Mason et al., 2006). Despite that, the number of retained teeth in middle age was associated with OHRQoL among men and women.

2.2.2.2 Pathways between childhood SES and OH in adulthood

The previous sections indicated that childhood SES predicted OH in adulthood but without describing the pathways linking early life socioeconomic position and later OH status. The following sections highlight the findings of the potential pathways.

2.2.2.1 Early malnutrition

The pathways between poverty in early life and OH within the life course have been linked to malnutrition, a condition described by protein, energy, and micronutrient deficiencies (Schroeder, 2001). Most of the evidence discussed the association between early malnutrition and dental caries.

In line with the critical period, disturbing dental enamel during development may result in irreversible, visible anomalies, which provides relevant evidence on the influence of early life disadvantages on OH (Suckling, 1989). Thus, it has been suggested that early childhood malnutrition influenced dental caries via two different mechanisms: via enamel developmental defects and impairment of salivary function (Psoter et al., 2005, Caufield et al., 2012). Teeth with an enamel defect and irregular surface will favour *Streptococcus mutans* (S. mutans) accumulation increasing the risk of having dental caries (Li et al., 1994). Longitudinal evidence following Australian children from birth until 24 months showed that children with enamel hypoplasia were 4.4 times more likely to have SM than those with non-enamel hypoplasia in a disadvantaged population (Wan et al., 2003).

In parallel with that, enamel anomalies have been frequently linked with preterm and low birth weight. Such association has been studied in a systematic review that involved twenty-three studies: 19 cohort studies, two case-control studies, and two cross-sectional studies (Jacobsen et al., 2014). The result revealed a significant link

between preterm birth and enamel hypoplasia based on thirteen studies. Additionally, limited studies conveyed an association between primary teeth's enamel opacities and children with low birth weight (<1500 g). Seow (2014) suggested that the primary pathway linking enamel defects and low birth weight was via mineral deficiencies caused by systematic conditions related to low birth weight, like infection, malnutrition, and renal diseases.

Another indirect marker of childhood malnutrition linked to OH status is stunted growth (Psoter et al., 2008b, Psoter et al., 2008a). Psoter et al. (2008a), in a retrospective cohort study, evaluated the effects of Early Childhood Protein-Energy Malnutrition (EC-PEM) and stunted condition on the dental exfoliation and eruption status in 498 11- to 13-year-old Haitian's adolescents. The results revealed that EC-PEM and stunting conditions in early childhood were associated with the delayed exfoliation of primary teeth and eruption of permanent teeth. Their further study (Psoter et al., 2008b) with a sample of 1017 11-19-years old found that severe EC-PEM and continuing nutritional stress, as evidenced by stunting, were linked with reduced salivary flow for a prolonged period.

In addition, deficiencies in vitamin D at the time of teeth development have also been linked with enamel hypoplasia and early childhood caries (ECC). A Canadian prospective cohort study recruited 207 expectant mothers from low SES urban areas to examine the association between prenatal vitamin D status and ECC (Schroth et al., 2014). The study concluded that children over 14 months of age with prenatal vitamin D deficiency were more likely to have enamel hypoplasia and ECC in the primary dentition. Vitamin D is essential to facilitate calcium absorption and calcification stages of the hard tissue. Thus, any Vitamin D deficit disturbs the enamel and dentine structure (Schroth et al., 2014).

2.2.2.2.2 Breastfeeding in early life

Breastfeeding is one of the critical early life factors influencing OH development. A Brazilian cohort study was conducted in 2015 at age five with 1,123 participants (Peres et al., 2015b). Using the Pelotas birth cohort's data, the study compared the impacts of exclusive breastfeeding and the predominant one on occlusion. The breastfeeding pattern was reported at birth, 3, 12 and 24 months. Children exclusively breastfed until six months displayed a 41-72% lower moderate to severe malocclusion rate than those who were never breastfed.

Additionally, Peres et al. (2015a), in a metal analysis, included forty-one studies (n = 27 023 individuals), concluded that participants who breastfed for a longer period were 60% less likely to have malocclusions in contrast to those who were breastfed for shorted time, suggesting a dose-response effect of breastfeeding on malocclusions. The reason is that the sucking and squeezing forces stimulate the lips, tongue, facial bone and muscles, which, in turn, encourage normal craniofacial development (Sánchez-Molins et al., 2010). In contrast to bottle feeding, where the nipples of the bottle are less flexible to adapt to the shape of the infant's mouth and consequently interfere with the development of oral structures, causing inadequate space for teething, malalignment, and mouth breathing (Inoue et al., 1995)

Another meta-analysis investigated 63 studies concerning the link between breastfeeding and dental caries (Tham et al., 2015). The results found that breastfeeding for up to 12 months was associated with a reduced level of dental caries, and breastfeeding beyond 12 months was associated with an increased level of dental caries. Another metanalysis recently confirmed the results (Cui et al., 2017). Nevertheless, most of these studies asserted that other confounders could interfere

with such associations as the frequency of tooth brushing and cariogenic food intake. Breastfeeding is considered a protective factor against dental caries among children because of three principles. First, breast milk comprises antibodies and proteins that interfere with bacterial growth, including S. mutans. The primary sugar of breast milk is lactose, which is less metabolised by mutans than sucrose (Arnold et al., 1977). Furthermore, the bottle nipples tend to prevent access of saliva to upper central incisors, exposing them to a prolonged acidic environment compared to natural nipples (Seow, 1998).

2.2.2.3 Family environment

Family structure and parenting style were established as critical predictors of children's psychosocial and physical well-being, mainly in the first three years of life (Vernon-Feagans et al., 2012). However, few studies explored the pathways between these variables and OH.

In a systematic review, Kumar et al. (2016) synthesised the evidence on the association between different family characteristics and dental caries in children aged 6-12 years. Among 48 studies, the results revealed that only parents' SES was associated with children's dental caries level. The results were inconsistent regarding the associations between family structure and their' OH behaviour and children's dental caries rate.

However, Hooley et al. (2012), analysing 55 studies with children aged 1-6 years old, reported a significant association between children living with single or separated parents and poor OH. The authors claimed that parents' economic burden and stress shift their attention away from their children, placing OH as a low priority. Further, the study found that higher birth order and larger family size were associated with higher

dental caries levels. This is because the large family size and high birth order can influence the emotional and economical resources available to the family.

Research supports a possible association between the different dimensions of parenting styles, child behaviour, and dental caries, but there is limited research. Baumrind (1967) divided parenting styles into three types, namely authoritarian, authoritative and permissive, based on warmth and control levels. Researchers have identified a fourth type of parenting style called rejecting-neglecting, explained by low warmth and control (Lamborn et al., 1991). One related study examined the link between the different types of parenting styles and a child's behaviour during a restorative dental visit (Aminabadi and Farahani, 2008). The study found that a caregiver's parenting style influences a child's reaction to restorative treatment. The study also suggested that permissive and authoritarian parenting styles were associated with poorer behaviour during dental treatment than authoritative ones.

Further, Howenstein et al. (2015) studied the association between parenting style, child's behaviour during the first dental visit and dental caries among 132 pairs of parents/child samples in Ohio. The study found that children with an authoritative parent (high responsiveness and high control) had a substantially favourable influence on children's OH behaviour and the lowest rate of dental caries compared to children with authoritarian and permissive parents. Further, Hooley et al. (2012) also determined that permissive parenting generally creates children with temperamental or challenging behaviour who showed higher caries rates. Authors argued that permissive style parents tend to soothe or reward their difficult children with sugary snacks. Sarit et al. (2021) also argued that children raised with permissive parenting experience a higher level of ECC because parents have less control of their children's cariogenic food.

2.2.2.4 Oral health-related behaviours

Socioeconomic and psychosocial pathways have been outlined to evaluate the complex associations between childhood SES and adult OH behaviours. Data from the Dunedin Multidisciplinary Health and Development Study (DMHDS) provided fundamental prospective longitudinal evidence for the possible pathways between childhood SES and OH behaviours from birth to 38 years old among 878 members (Broadbent et al., 2016). Based on structural equation modelling, the study outlined five findings. First, early SES and parental OH beliefs were associated with participants' OH beliefs at ages 15, 18, and 26. Second, positive OH beliefs in early adulthood predicted routine dental check-ups and more frequent tooth brushing at ages 26 and 32. Third, toothbrushing and dental check-ups were linked with the number of untreated dental caries and missing tooth surfaces at age 38. Fourth, high adult SES was associated with favourable OH self-care at ages 26 and 32. Finally, the number of untreated caries lesions and missing teeth was associated with OHRQoL. Consequently, OH in adulthood resulted from intergenerational factors operating since childhood life.

Further, a psychosocial pathway in terms of a sense of coherence was also proposed to explain the relation between SES and OH behaviours in adulthood (Heilmann et al., 2015). Bernabé et al. (2009) investigated the role of a sense of coherence in the association between parental SES and OH behaviours in adulthood among 5,399 Finnish adults. The study found a significant association between adults' sense of coherence and favourable OH-related behaviours like routine dental visits, frequent toothbrushing, less sugar intake, and reduced smoking habits. However, the moderating role of sense of coherence on the link between childhood SES and OH behaviours in adulthood was not supported by the study.

2.2.2.2.5 Dental services

Evidence from life course research supports the long-lasting benefits of dental care visits on OH (Thomson et al., 2010). The study used data from 932 participants from the DMHDS in New Zealand who were followed at ages 15, 18, 26 and 32 years. The results revealed that at any given age, routine dental attendees had better OH, less missing teeth due to caries, and a lower mean of DMFS. The effect was stronger among those with longer routine dental visits. By age 32, routine dental attendees had better better self-report OH and fewer missing teeth and dental caries.

A retrospective study based on data from 13 European countries confirmed the association between early life routine dental visits and chewing ability at 50 years (Listl et al., 2014). The study even suggested that childhood routine dental visits predicted adulthood dental care visiting patterns.

2.2.2.3 Impact of social mobility on OH

Based on the accumulation of risk model, health outcomes could be influenced by reducing and increasing risk via upward and downward social mobility (Heilmann et al., 2015).

Peres et al. (2011) studied the effect of family income trajectories and the number of unsound teeth in a sample of 729 24-years-old from the 1982 Pelotas birth cohort data. The study obtained data regarding family income trajectories at birth, age 15 and 24 years, while the number of unsound teeth was recorded at age 24. The study results indicated that the consistently poor group had the highest level of unsound teeth. The group who experienced upward and downward mobility had more unsound teeth than those of the never poor group. Further, the study revealed a dose-response association between cumulative exposures to poverty and the number of unsound

teeth, parallel with the accumulation life course model. Similar findings were also found previously by Poulton et al. (2002) and Thomson et al. (2004).

In contrast, Pearce et al. (2009) studied the association between the number of retained teeth at age 50 and OHRQoL and social mobility using the Newcastle Thousand Families cohort data. Social class data was collected for the age of 25 and 50 years. Based on constricting four SES mobility groups (stable manual, stable non-manual, upward and downward), the results revealed that maintaining functional dentition, determined if \geq 21 natural teeth (WHO, 1992) to age 50 was only related to social mobility amongst women. In addition, women who remained in the stable-nonmanual group (high social class) were more likely to maintain functional dentition than women who were either in the stable manual group (lower class) upwardly or downwardly mobile group. However, the study did not find any link between social mobility and retained teeth for the men or between social mobility and OHRQoL in this cohort.

Further, a recent metanalysis (Celeste et al., 2021) explored the link between social mobility and tooth loss among middle-aged and older people within eighteen articles from different countries. The results confirmed that poor SES in any stage of life was significantly associated with tooth loss, supporting the accumulation model. However, the high heterogeneity of the included studies prevented the review from concluding which group (downward, upward) may be more detrimental.

2.2.3 Life course framework for oral health

Overall, the Ben-Shlomo and Kuh life course model (2002) has been the most applied framework in dental research to date. However, more recently, Heilmann et al. (2015) has proposed a life course theoretical framework for OH that combines Ben-Shlomo and Kuh's model (2002) with the social determinants of OH and their impact on risk factors of common general and OH diseases (see Figure 3).



Figure 3. Life course framework for oral health (Heilmann et al. 2015)

The model underlines the role of economic, political and social factors at the societal level to shape SES (education, income) in childhood and adulthood. It assumes that childhood SES predicts adult SES considering a country's level of social mobility and the national policies and provision of education, social security, public health and healthcare. Different pathways have been proposed to explain how exposures in early life influence poor OH in adulthood. The first pathway is through the formation of enamel defects due to enamel's abnormal development within a critical time window (Psoter et al., 2005). The development of enamel occurs within a specific time. Thus, it was revealed that exposures during the enamel formation period might cause irreversible damage and become embedded in the physical structure of the dentition (Seow, 2014). Moreover, enamel defects may increase the vulnerability of dental caries due to the chain of risk model (Psoter et al., 2005).

The model assumes that childhood OH status predicts adult OH status for some reason. First, parental SES influences the child's life chances like education and income; then, adulthood SES is significantly linked to common risk factors for OHCs. Second, biofilm colonisation of the oral cavity, irreversible impairment to dental hard tissues and tooth loss in early life can directly affect OH in later life. Additionally, early exposure to psychosocial factors may increase the risk of developing diseases via biological mechanisms (allostatic load). Notwithstanding these pathways, the model also suggests that the trajectories of diseases can be changed according to social trajectories, i.e., they are not fixed and unchangeable. Finally, the model proposes an interrelationship between oral and general health and that both influence a person's quality of life.

2.3 Dental caries life course studies among children and adolescents

Recently, a systematic review has evaluated the link between exposures in early life and throughout the life course and the development of dental caries in children and adolescents (Abreu et al., 2015). The review contained eleven observational studies that predominantly employed Ben-Shlomo and Kuh's (2002) life course model. Further, six studies were found after systematic search and update for Abreu et al (2015), that published between 2012 and July 2021 (Appendix 2) (Birungi et al., 2017, Bernabé et al., 2017, Sun et al., 2017, Priyanka et al., 2019, Silva et al., 2019, Velasco et al., 2018). One more study evaluated social mobility (Peres et al., 2007). The following section briefly summarises the characteristics of these studies (Table 1) and highlights the possible factors associated with dental caries among children and adolescents within the life course approach. Table 1. Life course studies to understanding dental caries indicators and risk factors among children and adolescents (2012-2021)

Author, city, Country	Study design	Participant's age at baseline	Participants' age at the end of the study	Initial sample size	Final Sample Size	Dental caries measures	Life course factors
Peres et al., 2003, Pelotas, Brazil	Cohort/retrospective	6 years	6 years	400	359	dmft≥1	 1.< 8 years of maternal education. Family income < 6-time Brazilian minimum wages at birth. Not attending day-care at 6 years. Sweat consumption of a minimum of one /day at 6 years.
Nicolau et al., 2003, Cianorte, Brazil	Cross-sectional	13 years	13 years	Phase I:764 Phase II:330	652 311	DMFT ≤6 Iow >6 high	 Family income ≤ 5 Brazilian minimum wages. Sconed or higher birth order. Being a girl. High level of paternal punishment. non-brick house at birth. Low birth weight. High birth order ≥ second.
Bedos et al., 2005 Quebec- Canada	Cross-sectional	5-9 years	5-9 years	7382	6303 264 edentulous	DMFT/dmfs ≥1	 Edentulous mother (child's primary teeth). Edentulous mother (child 'permanent teeth).
Peres et al., 2005 Pelotas-Brazil	Cohort/retrospective	6 years	6 years	400	359	dmft ≤1 Iow ≥4 high	 Self-employed. Unemployed < 8 years of parental education at childbirth. Height by age deficit at 12 months Not attending day-care at 6 years. Sweet consumption of a minimum of one /day at 6 years. Teeth brushing < once /day.
Nicolau et al., 2005 Cianorte- Brazil	Cross-sectional	13 years	13 years	652	586	DMFT ≤6 low >6 high	 Maternal education for ≤ 8 years. Residence in a rural area. Being a second or later child. High level of paternal punishment. Height (Being a shorter individual)
Peres et al., 2007 Pelotas, Brazil	Cohort/ retrospective	15 years	15 years	888	875	(DMFT) DMFT=0, high caries DMFT>5	 Always poor= poor DMFT pattern Never were poor= better DMFT pattern. Moved from poverty to non-poverty= poor DMFT pattern Moved from non-poverty to poverty= poor DMFT pattern Girls had fewer caries, better OHPs, and higher use of dental care.

Cinar et al., 2008b Finland- Turkey	Cross-sectional	10-12 years	10-12 years	Turkish=611 Finnish=338	Turkish=459 Finnish=155	DMFT>0	 Lower height in Finnish pre-adolescent. Lower height in Turkish pre-adolescent. Turkish with lower levels of self-esteem. Finnish from single-parent families. Turkish with a higher number of children. Finnish spending less time with mother. Finnish with higher body weight.
Peres et al., 2009 Pelotas-Brazil	Cohort/prospective 6 years duration	6 years	12 years	359	339	DMFT≥1	 Stunting at 12 months and 4 years. 1.3 and 4-19 DMFT at 6 years. Gingival bleeding (28.1-92%) at 12 years.
Zhou et al., 2011 Xinhua-China	Cross-sectional	2 years	2 years	400	394	dmft mean	 Maternal education < 12 Yeats. Visible plaque index > 60%. Streptococcus mutans
Lu et al., 2011 Hong Kong- China	Cohort/prospective 6 years duration	12 years	18 years	638	221	DMFT mean	 DMFT at age 12 positively affected the value at 15 and 18 years old. CPI at age 12 positively affected the value at 15 and 18 years old. The income level at 12 years old directly affected the utilisation of dental services at 12-15 years and 15-18 years and significantly influenced the DMFT at 18 years, but not the CPI.
Zhou et al., 2012 Xinhua-China	Cohort/prospective 2 years duration	8 months	32 months	225	155	dmfs mean	 Maternal education < 12 years. Higher family monthly income at birth (≥450 USD). Enamel Hypoplasia. Z value of height <-2. Visible plaque (≥ 20%) Presence of Streptococcus mutans.
Wong et al., 2012 Hong Kong- China	Cohort/ prospective	3-4 years	5-6 years	465	358	dmft>0	 1. ≤ 9 years of parental education. Using nursing bottles at 4 years. First-time teeth are brushed after 12 months. Snacking one or more times/day at 4 years.
Birungi et al., 2017 Uganda	Cohort/prospective 5 years duration	24 weeks	5 years	863	417 mother- child	dmft >0	1.Exclusive breastfeeding 24 weeks 2.living with both parents
Sun et al., 2017 China	Cross-sectional	5 years	5 years	12692	9722	dmft mean	 Exclusive & predominant breastfeeding Brushing at later life Being a boy Second or third birth order >1 sibling Low income Primary education Brushing <1 Sugar ≥2
Bernabé et al., 2017 Dundee- Scotland	Cohort/ prospective 4 years duration	1 year	4 years	1102	765	dmfs	1. Low birth weight 2. Smoking

Velasco et al., 2018 Bauru, Brazil	Cross-sectional	9-11 years	9-11 yeas	350	350	DMFT EDD index	 1.EDD increase caries risk 2. Maternal education 3. Household income
Priyanka et al., 2019 Guntur-India	Cross-sectional	12 years	12 years	800	800	DMFT mean 0 1-3 >3	 Government school Shorter height Low and medium SES
Silva et al., 2019 Australia	Cohort/prospective 6 years duration	24 gestational weeks	6 years	250 pairs	172 pairs	ICDAS Any caries Advanced caries	1.No difference in caries between types of twins. 2.Maternal obesity, non-fluorinated water, and enamel defects

2.3.1 Study design

Of the 18 studies, ten were cohort studies, and the remaining eight were cross-sectional (Table 1). Three cohort studies were designed retrospectively (Peres et al., 2003, Peres et al., 2005, Peres et al., 2007), and the remaining seven were prospective. Most of the above studies studied the association between early life social, biological, psychosocial, and behavioural factors and dental caries in children and adolescents from 0-18 years old and 2-6 years follow-up periods. In contrast, Peres et al. (2007) investigated the impacts of social mobility on dental caries among adolescents.

2.3.2 Origin of the studies

Generally, most studies were conducted in Brazil, four of which were drawn from a nested birth cohort study in Pelotas in 1993 (Peres et al., 2003, Peres et al., 2005, Peres et al., 2007, Peres et al., 2009). The other two were in the city of Cianorte in Brazil (Nicolau et al., 2005, Nicolau et al., 2003) and one in Bauru (Velasco et al. 2018). Five studies were conducted in China; two in Hong Kong (Lu et al., 2011, Wong et al., 2012), two in Xinhua (Zhou et al., 2012, Zhou et al., 2011) and one included the entire population of China (Sun et al., 2017). One study was carried out in Quebec, Canada (Bedos et al., 2005), another was conducted in both Finland and Turkey (Cinar et al., 2008b), one in Uganda (Birungi et al., 2017), one in Scotland (Bernabé et al., 2017), one in India (Priyanka et al., 2019), and one in Australia (Silva et al., 2019)

2.3.3 Participants' age

Children's ages at baseline ranged from 24 weeks (gestation) to 18-years of age. The final wave of the cohort studies involved participants of different ages based on the duration of the follow-up period (2-6 years) (Table 1).

2.3.4 Oral examination and indices

DMFT measured dentine dental caries as the primary outcome in seven studies and dmft in six studies. Two studies measured DMFT and dmft (Cinar et al., 2008b, Bedos et al., 2005). Three studies assessed dental surfaces instead of the tooth component of the DMF index (Zhou et al., 2012, Bernabé et al., 2017, Bedos et al., 2005). One study used ICDAS (Silva et al., 2019).

It is essential to state that different DMFT/dmft or DMFS/dmfs thresholds were applied in the studies. The most common was the presence or absence of dental caries in eight studies (Table 2). Three studies applied different ranges to indicate low and high levels of dental caries (Table 3), whilst the remaining five studies used the mean value of the DMF index.

Study	Index	The threshold of dental caries
Peres et al., 2003	dmft	≥1
Bedos et al., 2005	DMFS/dmfs	≥1
Peres et al., 2007	DMFT	≥5
Cinar et al., 2008b	DMFT	>0
Peres et al., 2009	DMFT	≥1
Wong et al., 2012	dmft	>0
Bernabé et al., 2017	dmfs	>0
Velasco et al., 2018 DMFT		>0

 Table 2. DMFT's mean threshold used in life course studies

 Table 3. DMFT's mean threshold used to assess severity of dental caries in life course studies

Study	Index	Low	High
Nicolau et al., 2003	DMFT	<mark>≤ 6</mark>	<mark>> 6</mark>
Nicolau et al., 2005	DMFT	<mark>≤ 6</mark>	<mark>> 6</mark>
Peres et al., 2005	dmft	≤1	≥4

Although the DMFT/dmft index is recommended by the WHO for OH surveys and its widely used (WHO, 1997), the adoption of the index in analytic epidemiologic studies has been extensively criticised for measuring dental caries at the cavitated stage only (dentine). In contrast, the condition is present, diagnosable and reversible before this stage (Ismail et al., 2007). Furthermore, the index does not consider the conditions of sealants, disease progression, restorations, and activity conditions. Likewise, it gives equal value to restored, missing, and untreated teeth, making it hard to assess progression or regression changes in teeth status in longitudinal studies (Pitts and Ekstrand, 2013).

2.3.5 Non-clinical data collection

Self-completed questionnaires and interviews were used to collect non-clinical information in the epidemiologic studies of the life course of dental caries (Table 4). The interviews were conducted with adolescents and their parents or with the parents only regarding younger children. Two studies employed maternal self-administered questionnaires (Bedos et al., 2005; Cinar et al., 2008b). The latter also used a self-administered questionnaire to gather children's information.

Study	Child/Adolescent	Parent's data collection
Nicolau et al., 2003	Interview	Interview
Peres et al., 2003	No	Questionnaire
Peres et al., 2005	No	Questionnaire
Bedos et al., 2005	Interview	Questionnaire- maternal- Self-administered
Nicolau et al., 2005	Interview	No
Peres et al., 2007	Interview	No
Cinar et al., 2008b	Questionnaire (self-administered)	Questionnaire- maternal- Self-administered
Peres et al., 2009	Interview	No
Lu et al., 2011	Questionnaire	No
Zhou et al., 2011	No	Questionnaire
Wong et al., 2012	No	Questionnaires (both surveys)
Zhou et al., 2012	No	Questionnaire
Birungi et al., 2017	No	Interview
Sun et al., 2017	No	Questionnaire
Bernabé et al., 2017	No	Questionnaire
Velasco et al., 2018	No	Questionnaire
Priyanka et al., 2019	Interview	No
Silva et al., 2019	No	Questionnaire

Table 4. Use of questionnaires and interviews in dental caries life course studies

Pre-existing data on dental and related information were used in some studies as baseline data of participants. For example, in the nested birth cohort study conducted in Pelotas with 1,993 participants (Peres et al., 2003, Peres et al., 2005, Peres et al., 2007, Peres et al., 2009), or in the 2001 Hong Kong OH Survey (Lu et al., 2011), National OH Survey in China 2005 (Sun et al., 2017) and the PROMISE-EBF trial in Uganda (Birungi et al., 2017). Bedos et al. (2005) used the Study on the Oral Health of Quebec Students Aged 5-6 and 7-8 Years, 1998-99 in Canada. Moreover, the OH data of Finnish pre-adolescents were recovered from Helsinki City Health Department (Cinar et al., 2008b). Bernabé et al. (2017) also used data from the birth cohort of 1993-1994. The remaining studies did not use any pre-existing information.

2.3.6 Dental caries life course factors

Among the 18 studies in Table 1, there were 73 life course factors related to dental caries arranged and analysed at different hierarchical levels. Regardless of the levels used in each study, the factors can be organized into six groups: demographics, socioeconomic characteristics at birth and at follow up, biological, psychological, OH behaviours, and OH status.

Demographic data comprised age, gender, skin colour, birthplace, mother's and father's ages, family structure, family type (single or two-parent) and the number of children. Socioeconomic variables at birth included house building material, the presence of a toilet, house tenure, piped water, car ownership and overcrowding, social class, parent's schooling, household income, parent's employment status, area of residence and the school type. At follow up, they consisted of house material, house tenure, overcrowding, presence of a toilet or piped water, car ownership, household income and parents' level of education.

Biological factors included birth order, birth weight, maternal health thru pregnancy, maternal smoking thru pregnancy, and mother's age at delivery. It also included childhood infectious diseases, feeding habits (breastfeeding, bottle feeding at night), stunting, and wasting conditions at 12 months. Additionally, it included the number of erupted teeth when the child was 12 months old, the number of times admitted to hospital, prolonged use of medication, gestational age, enamel defects, sleeping with a pacifier and shared cutlery with other children.
The psychological factors were related to family life, such as parental levels of punishment and support, mother academic involvement, and self-efficacy and self-esteem. Behavioural factors included oral hygiene practices (when started tooth brushing/ frequency of toothbrushing /help/floss/chewing gum), dental attendance pattern, fluoride use, and dietary habits (stable access to food, snacking frequency and consumption of sweets). Moreover, these variables included day-care centre attending at the age of six months and visits to a dentist. Similarly, what the child drinks regularly, the reason for the last dental visits, and visible dental plaque. Finally, certain studies considered previous dental caries experience, gingival bleeding, S. mutans and the mother's dental status (dentate or edentulous).

2.3.7 Life course models and analysis methods

Most studies utilised the critical period and the accumulation of risk models proposed by Ben-Shlomo and Kuh's (2002) to guide the analysis of the association between life course factors and dental caries. Generally, the studies assumed that risk accumulates independently from early and throughout life and causes dental caries. However, Lu et al. (2011) presumed that social factors cause a chain of risks that eventually trigger dental caries during a child's life. Silva et al. (2019) examined the biological programming theory (critical model).

The studies mainly relied on analysing the associations between those factors and dental caries using hierarchical regression, where variables were introduced in the analysis in a specific sequence of levels. The number of levels varied between the studies from six

(Peres et al., 2009) to three levels (Wong et al., 2012), except for Lu et al. (2011), where the chain of risk model was tested using path analysis.

2.3.8 Summary of the results of the studies

This section summarises the results of the 18 studies that explored the associations between early and current life course factors and dental caries in children and adolescents using the life course approach according to different age groups.

2.3.8.1 Age 1-5 years old

Six studies evaluated the pathways between early life circumstances and ECC development in 1-5-year-olds (Zhou et al., 2012, Zhou et al., 2011, Wong et al., 2012, Birungi et al., 2017, Sun et al., 2017, Bernabé et al., 2017).

Zhou et al. (2012) explored the risk factors of ECC in 225 children aged eight months old from a rural area of southern China. They followed the participants every six months for two years. Mothers with a low level of education (< 12 years), families with higher monthly income (≥450 USD) at childbirth, children with hypoplasia, visible plaque, low height-z value at 0-2 years old and having S. mutans were associated with a high level of ECC in children below the age of three years. Similar findings were also concluded in the previous research undertaken by Zhou et al. (2011) but using a cross-sectional design study. Although this study provided a comprehensive prospective longitudinal design study (Abreu et al., 2015), there was no sample size calculation. Participants were selected from the maternity and children's hospital with relatively small sample size. Additionally, the study experienced a significant drop-out (31.1%) on completion, which restricted the generalizability of the study.

Wong et al. (2012) also explored the early life risk factors associated with ECC lesions in toddlers through a prospective cohort study in Hong Kong. Three hundred and fifty-eight children were recruited between the ages of 3-4 years and were followed for two years. The study revealed that children with parents with \leq nine years of education, night bottle-feeding at four years of age, snacking \geq one time per day at age four, and tooth brushing initiated after 12 months were more prone to dental caries. They found that one-third of the children developed at least one new lesion over the two-year follow-up period, but only with a small increment (0.9). However, the accuracy and reliability of dental caries measures in this study are questionable because different examiners carried out the examination in 2007 and 2009.

Birungi et al. (2017) explored the impact of early social and behavioural factors in developing ECC among five-year-old children in Uganda using a prospective cohort study. The study involved 417 mother-child pairs and followed them at 24 weeks, two and five years of the child's age. The results revealed that exclusive breastfeeding and living with both parents were protective factors for ECC in 5-year-old children.

Further compelling evidence was reported by Bernabé et al. (2017), who conducted a longitudinal study to evaluate the role of early life risk factors on ECC increments over four years among 1102 infants in Scotland. The study found that children born with low birth weight and whose mothers were smokers had sharper increments of dental caries over four years. On the other hand, no association was reported between breastfeeding duration and dental caries. Interestingly, birth weight, breastfeeding, and maternal smoking were not associated with children's baseline dmfs. However, by Wave 4, low birthweight children and those with smoking mothers had 1.9 and 1.7 more tooth surfaces

with caries than their better-off counterparts. Both factors placed children on a trajectory of increasing dental caries over time. This could highlight the importance of focusing on the accumulation of disease over time and explain why previous studies failed to report such a relationship when measuring caries only once in time. However, the study reported selection bias since the participants who continued in the study were healthier and wealthier than those lost to follow up, which could have affected the study's generalizability.

Using a cross-sectional study, Sun et al. (2017) assessed the link between early life factors and dental caries among 9722 pre-school children in China. Interestingly, the study ascertained that formula-fed children who brushed their teeth within the first year of life had lower dental caries than those exclusively breastfed and those who started to have their teeth brushed later in life. However, the study has some limitations as it was a secondary analysis of cross-sectional data collected in 2005. Thus, due to the nature of the study, no causal and temporal association can be implied.

2.3.8.2 Age 5-11 years old

Five studies investigated the association between early social and biological factors and dental caries among children aged 5-11 years old (Peres et al., 2003 and 2005; Bedos et al., 2005; Velasco et al., 2018; Silva et al., 2019).

Both studies undertaken by Peres et al. (2003 and 2005) were conducted retrospectively in Pelotas, Brazil, involving 400 participants aged six years. The most recent study (Peres et al., 2005) was more comprehensive as it included more variables than the previous one (Peres et al., 2003). Peres et al. (2005) found that low social class and father's low level

of education (< 8 years) at the time of the childbirth, child's with stunting condition at 12 months, low frequency of toothbrushing (< once a day) and once a day consumption of sweets at six-years-old were substantial risk factors for dental caries development at the age of 6 years. Interestingly, attending daycare in Pelotas was a fundamental protective factor for developing dental caries at six years old. The authors argued that this was because of the early introduction of OH education and prevention programmes or because the daycare provided a conducive environment to acquire good OH practices (Peres et al., 2005). The study did not report any significant association between access to dental care at 4 and 6 years and dental caries.

From a different perspective, Silva et al. (2019) established a longitudinal study in Australia to evaluate the role of genetic, fetal and environmental factors on dental caries risk. The authors collected non-clinical data among 250 twin pregnancies at 24 weeks through age six when the children undertook oral examination. At follow-up, 345 children were examined for enamel developmental defects (EDD) and dental caries. The study compared the caries rate between monozygotic and dizygotic twins using concordance rate and found no difference in caries between both types of twins. The findings revealed that environmental rather than genetic factors determine the difference in dental caries, despite the biological programming hypothesis. The study concluded that factors like non-fluoridated community water, enamel defect and maternal obesity were important early life determinants of dental caries (Silva et al., 2019).

In a similar approach, Bedos et al. (2005) explored the association between maternal edentulism and dental caries in children aged 5-9 years using a cross-sectional study with a sample of 6,303 mother-child pairs in Quebec. Children with edentulous mothers were

more likely to have dental caries in both deciduous and permanent teeth than those with dentate mothers. These results were independent of sociodemographic factors and OH behaviours. The authors claimed that dental caries was the main cause of maternal edentulism, which increased their children's susceptibility to dental caries. Three hypotheses explained the association: transmission of the mother's genes, the programming hypothesis while pregnant, and the transmission of OH habits and behaviour from mother to child. However, the reasons for edentulousness were not investigated, whether caries or periodontal reasons.

Velasco et al. (2018) studied the association between EDD and dental caries among 350 children aged 9-11 years in Bauru, Brazil. Children with EDD were twice as likely to develop dental caries than those without EDD. However, the possible pathways could not be identified due to the study's cross-sectional design.

2.3.8.3 Age 12-18 years old

This part outlines the findings of six studies that examined the biological, social and behavioural factors related to dental caries in permanent dentition in early life and throughout the child's life course. Peres et al. (2009) conducted a population-based cohort study in Pelotas involving 339 children followed from birth to six and 12 years old. The study identified four risk factors of dental caries at the age of twelve through regression analysis, including low maternal education at the time the child was born, height deficit at 12 months and four years of age, previous experience of dental caries in primary teeth, and gingival bleeding. The authors explained the association between dental caries in the primary and permanent teeth by means of three hypotheses. First, dental caries experience at six years old might result from a life course's social, behavioural and

feeding pattern accumulative effect. Second, the persistence of the cariogenic bacteria over the years may predispose dental caries in both dentitions. Third, it could be a combination of both of these previous explanations.

The authors argued that the association between gingival bleeding and dental caries might result from ineffective tooth brushing, influencing both caries and gingival health. It should be stated that the above findings cannot be generalised to other regions of Brazil because the area where the study was conducted (Pelotas) has a water fluoridation system in 90% of houses. This could also explain the low levels of dental caries (1.2 mean) with a low number of children with caries compared to the country (2.8 mean) (Peres et al., 2009).

While Peres et al. (2009) explored the cumulative effect of life course factors on dental caries, Lu et al. (2011) sought to explore the effect of social factors in triggering a chain of risks that might predict dental caries among 638 adolescents in Hong Kong. The study retrospectively followed participants at birth, then from 12, 15 and 18 years old. In this study, family income directly influenced dental services use, and dental services use positively predicted children's dental caries. Nevertheless, no association was observed between education level and utilisation of dental services. The authors argued that economic rather than educational variables might be a stronger predictor of dental care utilization in Hong Kong, mainly when the government does not fund dental services and people must use costly private clinics (Lu et al., 2011). The study clearly explains that dental services were well suited for the chain of risk model despite the following limitations. First, the study did not mention whether Hong Kong has fluoridated water. Second,

different examiners collected data throughout the study. Third, the follow up after six years resulted in only 50.8% of the population present, possibly resulting in attrition bias.

Family life might significantly impact individual OH development and behaviours (Castilho et al., 2013, Kumar et al., 2016). However, there is limited evidence on the pathways between family characteristics and the development of OHCs. Four cross-sectional studies examined the association between family characteristics and dental caries throughout the life course among adolescents (Nicolau et al., 2005, Cinar et al., 2008a, Priyanka et al., 2019, Nicolau et al., 2003).

Nicolau et al. (2003) conducted their study in two phases. Six hundred and fifty-two adolescents (aged 13 years) were recruited in the first phase, and 311 families were recruited in the second phase in the city of Cianorte in Brazil. There was an association between a high family SES and lower dental caries in both phases. Moreover, adolescents born with low birth weight had a higher level of parental punishment. Being the second or later birth order doubled the risk of having dental caries. In contrast, being taller had a protective effect on dental caries. This finding was confirmed by another study conducted by Nicolau et al. (2005). The study's cross-sectional design restricted the causal inference between variables. Additionally, the study relied on retrospective information, resulting in recall bias. The use of the cut-off points for low DMFT (≤ 6) and high DMFT mean (> 6) complicates the comparison with other studies.

Further evidence came from a study conducted in Turkey (N = 611) and Finland (N = 338) (Cinar et al., 2008a). The study assessed the influence of family type, the number of children, and time spent between mother and child on the association between adolescent health (height-weight), self-esteem, toothbrushing, and dental caries experience. Turkish

adolescents with more than two siblings were more likely to be shorter and to have more dental caries than their counterparts. Alternatively, Finnish adolescents who lived with a single parent were less likely to spend time with their mother and more likely to have dental caries. In this study, Turkish adolescents had a higher rate of dental caries (84%) than Finnish ones (33%).

Nonetheless, it was unclear if the study attempted to compare the two communities or explored the pathways between the proposed association. Another limitation was that the anthropometric measures of the adolescents were obtained by their mothers, which may raise questions about the validity and reliability of those measures.

Finally, a cross-sectional study with 800 adolescents aged twelve studied different biological and family-related factors that might influence their risk of developing caries (Priyanka et al., 2019). The study found that children with shorter height, low and medium income, and attending governmental school were more likely to have dental caries than their counterparts. However, the study did not find family variables associated with dental caries risks, such as family structure, number of siblings, and time spent between children and their mothers. Although the study was recent and claimed to adopt a life course approach, it did not provide any life course model to support its theory and analysis.

2.3.8.4 Impacts of social mobility

Few studies explored the impact of social mobility on dental caries among children and adolescents. Most of the available evidence was measured during adulthood. However, it should be mentioned that at least two measurements are required to assess health

trajectories. Likewise, at least three measurements are required to assess the pattern of these trajectories (Burton - Jeangros et al., 2015).

Notwithstanding this, analysis of the Pelotas birth cohort concerning social trajectories with adolescents (n=875) generated mixed results (Peres et al., 2007). The study investigated the impact of family SES trajectories from birth to adolescence on dental caries, tooth brushing and use of dental services at age 15. The study showed that adolescents from the always poor group had the highest caries rate had a lower frequency of tooth brushing and attending dental services than their correspondent counterparts. However, no variations were observed in dental caries and teeth brushing patterns between adolescents from upward or downward mobility groups and those who lived in poverty. Thus, the upward social improvement did not reverse the unfavourable effect of poverty in early life, and the study suggested that poverty, even in one stage of early life, is formidable enough to cause poor OH in later life.

2.3.9 Conclusion

The limitations of the lifestyle approach to preventing chronic conditions motivated the interest to examine the natural history of risk factors of diseases using the life course approach (Ben-Shlomo and Kuh, 2002).

Recently, the life course approach has been demonstrated to be suitable to study the factors related to OHCs. Studies that adopted this approach concluded that dental caries among children and adolescents result from exposures accumulated early and later in life.

Thus, Heilmann et al. (2015) developed their life course framework specifically to study the factors associated with OH by incorporating the social determinants of OH together

with the critical and accumulation life course model and common risks connected to health and OH in their framework.

After highlighting possible life course factors related to dental caries, the next section summarised the impact of OHCs on children's and adolescents' school outcomes.

2.4 The impact of OHCs on school performance and attendance in children and adolescents

2.4.1 Overview

Education during early life can predict the opportunities throughout the life course (Banerjee, 2016). OHCs can impair children's cognitive ability and hinder their academic school performance in early life (Rebelo et al., 2019), influencing their academic attainment in adult life and future career development (Dornan and Woodhead, 2015). This section summarises the available evidence on the impacts of OHCs on children's and adolescents' school performance and attendance using an umbrella review method.

2.4.2 Aim

To summarise the available systematic reviews that explored the impacts of OHCs on school performance and school attendance among children and adolescents.

2.4.3 Study question

What is the association of clinical and self-reported OH measures with children's and adolescents' school performance and attendance?

2.4.4 Inclusion criteria

2.4.4.1 Type of participants

The scope of the overview included children and adolescents up to 19 years of age participants.

2.4.4.2 Phenomena of interest

The phenomena of interest (outcomes) included school performance, and attendance was measured by school reports of grades and attendance, self-reported measures, such as parent's or adolescent's perceptions, or any other measure included in the systematic reviews.

2.4.4.3 Exposures

The exposures included clinical and subjective OH measures.

2.4.4.4 Context

No restriction was made in terms of country and cultural variables.

2.4.4.5 Type of studies

Any systematic review or meta-analysis of epidemiologic studies exploring the association between OH measures, school performance and school attendance in schoolchildren.

2.4.4.6 Reviews' features

The plethora of systematic reviews began in 1990 (Smith et al., 2011). Hence, the electronic databases were searched from 1990 until March 2019, with no language restriction. The literature search presented at the end of this section was updated to include primary studies until July 2021.

2.4.5 Search strategy

2.4.5.1 Database search and terms

The electronic search was conducted using PubMed, Education database, Scopus and Medline using the following MeSH terms and keywords: ("oral health" or "dental health", or "oral health status" or "dental health status", or "dental caries") and ("academic performance" or "academic achievement" or "school performance" or "educational performance" or "educational achievement" or "educational measurement" or "lost school days" or "school absenteeism" or absenteeism or attendance) and (children or child or adolescent or adolescence) and (review or "systematic review" or "meta-analysis").

2.4.5.2 Review Selection

The initial search was performed by the lead reviewer (SM). Screening of the titles and abstracts was conducted by two reviewers independently (SM and YA).

2.4.6 Review method

2.4.6.1 Assessment of methodological quality

The Assessment of Multiple Systematic Reviews (AMSTAR) checklist tool was used to evaluate the quality of the included systematic reviews (Shea et al., 2007) (see Appendix 3). Two reviewers independently evaluated the quality of the included systematic reviews (SM, YA). The tool comprises eleven criteria (one point each), and the final scores can range from 0 to 11. The quality of the reviews was classified as follows: high (8-11 points), medium (4-7 points) and low quality (0-3).

2.4.6.2 Data extraction and management

The identified reviews were saved in Endnote X7. The duplicated articles were deleted. The Jonna Briggs Institutes (JBI) data extraction form was employed to report the data of the included systematic reviews (see Appendix 4).

2.4.6.3 Search and quality assessment results

Figure 4 illustrates the results of the electronic database search using the PRISMA flow chart (Moher et al., 2009). The electronic search retrieved 178 articles after removing duplicates. One hundred and seventy-four articles did not meet the eligibility criteria after screening the title and abstract. The umbrella review included four systematic reviews (Ribeiro et al., 2018, de Paula and Mialhe, 2013, Ruff et al., 2019, Rebelo et al., 2019). Most of the excluded studies were not systematic or meta-analysis reviews.

According to the AMSTAR, Shea et al. (2007), de Paula and Mialhe (2013), Rebelo et al. (2019) and Ribeiro et al. (2018) were judged to be of moderate quality, while Ruff et al. (2019) was assessed as having high quality. The result is presented in Table 5 below.

 Table 5. The quality assessment results of four systematic reviews using AMSTAR (Shea et al. 2007)

Study	Q1	Q2	Q3	Q4	Q 5	Q 6	Q	Q 8	Q9	Q1	Q11	Overal
							7			0		1
de Paula & Mialhe., 2013	Х	~	~	Х	~	~	Х	Х	Х	Х	~	5
Rebelo et al., 2019	>	~	✓	Х	Х	✓		Х	✓	Х	✓	7
Ribeiro et al., 2018	>	~	✓	✓	Х	✓		✓	Х	Х	Х	7
Ruff et al., 2019	Х	~	✓	✓	~	✓		X	✓	~	✓	8



Figure 4. Prisma flow chart for identifying systematic reviews on the impacts of OHCs on school performance and attendance in children and adolescents (2019)

2.4.7 Summaries of the systematic reviews

Table 6 summarises and compares the four systematic reviews in terms of authors and country, aim, inclusion and exclusion criteria, number and type of the included studies, the use of PRISMA, risk assessment, results and the main recommendations.

2.4.7.1 Population and sample

The target population included in the systematic reviews were schoolchildren and adolescents. Across all the included studies, there were 359,912 children and adolescents aged 3-19 years. Three of the four reviews selected primary studies involving specific age groups. Ribeiro et al. (2018) included three- to nineteen-year-old students, Ruff et al. (2019) comprised students aged five to eighteen and Rebelo et al. (2019) consisted of those aged eighteen years or younger. No specific age group restrictions were identified in the review of de Paula and Mialhe (2013). Such diversity in target groups is related to the selected studies included in each systematic review.

Table 6. Summary of systematic reviews exploring the impacts of OH on school performance and attendance in children and adolescents

Author-year-	Aim	Inclusion criteria	Exclusion	PRISMA	Risk	Results	Conclusions and
de Paula & Mialhe, 2013 (Brazil) 21 studies 14 cross- sectional and 7 ecological studies	To investigate the association between OHCs and school attendance and performance in children & adolescents	(1) English language only (2) Between 1990 and 2013	(1) Other languages	No	No	The significant association found between OH and school performance and absences by	 Longitudinal studies - Multivariate statistics to control confounders - Use reliable and valid criteria
Ribeiro et al., 2018 (US-Brazil) 6 cross- sectional studies	To confirm the association between OHCs & academic performance in children and adolescents.	 (1) Observational studies and clinical trials. (2) OHS with clinical examinations. (3) Academic performance by school records (4) Age range 3-19 years old. (5) No restrictions on time or language. 	(1) Self-reported measures.(2) Letter, review papers, case reports and case studies	Only the flow chart	National Heart, Lung and Blood Institute	 4 papers (high risk of bias) OHS were associated with children's academic performance 2 papers (good quality) did not show an association unless mediated by SES. 	 Longitudinal studies Considering confounders Valid OH measures carious lesions in different stages (e.g., PUFA or CAST). Objective measures of academic performance.
Rebelo et al., 2019 (UK) 18 studies 1 cohort 1 case-control 16 cross- sectional	To explore the possible impact of OH on school performance and attendance in children and adolescents.	 (1) Clinical trials and observational. (2) Aged 18 years or under. (3) Any language. (4) No publication period restrictions to May 2018. 	(1) Letter, review papers, case reports and case studies	Yes + MA	- NOS for cohort studies and case- control. - Modified NOS for cross- sectional studies	Children with - ≥1 decay and poor parental perception of OH have a higher probability of poor school performance attendance	 Longitudinal studies. Reliable school data from school records Valid OHS measures. SES and family structure considered.
Ruff et al., 2019 (US) 14 (10 low risks and 4 high risks 1 cohort 1 case-control 12 cross- sectional)	To evaluate the association between OHS & school attendance and academic performance in children aged 5 through 18 years.	 (1) Observational studies (2) Aged 5 to 18 years (3) Exposures: caries, periodontitis or gingivitis s (4) school performance and attendance (5) Jan1945 - Dec 2017 	(1) Included only articles published in English.	Only flow chart + MA	The modified NOS for non- randomised studies.	Poor OH was significantly associated with increased odds of poor academic performance	 Design and implement targeted interventions for specific conditions. Understanding the best period in which OH immediately influences performance.

2.4.7.2 Exposures

All the systematic reviews included the broad term "oral health" followed by different terms such as status, problem or measures to refer to the exposure. The most common OHCs assessed were dental caries, included in all systematic reviews. Nevertheless, evaluation of other conditions, including oral hygiene, periodontal diseases, dental trauma, malocclusion, dental pain and self-report OH, were assessed in a few studies.

2.4.7.3 Assessment of exposures

Three systematic reviews considered primary studies that assessed exposures using clinical examination (normative needs) or perceived (reported) OH status. The latter can be either parental or self-reported OH status. It should be stated that only the review from Ribeiro et al. (2018) believed the OH information in the included studies was required based on clinical examination alone. DMFT, dmft and DMFS indices were predominantly used to assess dentine dental caries. The descriptions of these indices are provided when describing the included studies.

2.4.7.4 Outcome measure

Three systematic reviews included school performance and school attendance as the outcomes of interest, and the other review considered the former outcome only (Ribeiro et al., 2018).

2.4.7.5 Evaluation of outcome measure

School performance was assessed using either school records (i.e., school grades or marks using standard tests or teacher reports) or self-reported (i.e., questionnaire or interview assessing child's or adolescent's or parents' perception of school performance). Ribeiro et al. (2018) covered only studies that used school records to

evaluate school performance. The other three systematic reviews included both school records and self-reported assessments of school performance. Similarly, school attendance was evaluated using either school records (i.e., school attendance registers) or self-reported measures (i.e., parental or self-reported information by questionnaire or interview regarding the number of missing school days due to dental problems).

2.4.7.6 Registration of protocol

Two of the four systematic reviews were registered at the Prospective International Registration of Systematic Reviews (PROSPERO). Ribeiro et al. (2018), under the reference CRD42016041587 and Rebelo et al. (2019), under the reference CRD42016036909.

2.4.7.7 Design of the included studies

The systematic reviews included observational studies. Although clinical trials were eligible for inclusion, no intervention studies were identified.

2.4.7.8 Search strategy

Ruff et al. (2019) searched for publications in three databases (PubMed, EMBASE and Google Scholar). Alternatively, Rebelo et al. (2019) examined five databases (Medline, SCOPUS, Web of Science, Science Direct, LILACS). de Paula and Mialhe (2013) and Ribeiro et al. (2018) searched four databases. The former explored Medline, ISI, Lilacs and Scielo, whilst the latter used PubMed, Lilacs, Web of Science and Scopus.

Three reviews provided keywords and a search strategy, whereas one review paper (de Paula and Mialhe, 2013) provided only the keywords. All the reviews included a

search of the references lists. Two reviews included the grey literature search, specifically Ribeiro et al. (2018) and Ruff et al. (2019).

2.4.7.9 Search restrictions and justifications

Overall, some restrictions were applied regarding participants' age, language and time. All of the reviews included participants aged 19 years and younger. However, Raff et al. (2018) and Ribeiro et al. (2018) restricted the youngest age to be children aged five and three years, respectively. On the contrary, de Paula and Mialhe (2013) did not restrict the age group. Regarding language restrictions, two reviews, namely de Paula and Mialhe (2013) and Ruff et al. (2019) included only articles published in English. Other reviews did not report language restrictions. Furthermore, Ruff et al. (2019) limited their search to commence from January 1945, as this was necessary to coincide with the beginning of community water fluoridation in the US. Similarly, the studies undertaken by de Paula and Mialhe (2013) included studies published from 1990 to 2013. However, without time limitations, Ribeiro et al. (2018) and Rebelo et al. (2019) unearthed no literature prior to 1990.

2.4.7.10 Study selection and data extraction

Two reviewers selected and extracted the data of the identified studies in two systematic reviewes (de Paula and Mialhe, 2013; Ribeiro et al., 2018). Three reviewers selected and extracted the data from the published paper in Rebelo et al. (2019). Data extraction was completed in duplicate in Ruff et al. (2019). However, in the latter review, it was not clear whether the selection of the studies was made in the same way.

2.4.7.11 Excluded studies

Ruff et al. (2019) excluded fifteen studies after reading the full text, and the reasons for excluding was informed in a flow chart. The fifty excluded articles that were read in full were not reported in Rebelo et al. (2019), and the justifications for exclusion were explained in a flow chart. Similarly, Ribeiro et al. (2018) excluded eight articles whose justifications were shown in the flow chart. In de Paula and Mialhe (2013) review, no such exclusion was done as all studies were read in full and included. Interestingly, four articles were excluded from the Rebelo et al. (2019) and one paper from Ruff et al. (2019), as there was no access to the full text of the manuscripts.

2.4.7.12 Characteristics of the included studies in the systematic reviews

The following details of the characteristics of all the included primary (n=39) studies in the four systematic reviews (see Appendix 5).

2.4.7.12.1 Number of the studies

Fifty-nine studies were included in the four reviews. de Paula and Mialhe (2013) included 21 papers, followed by Rebelo et al. (2019) (18 studies), Ruff et al. (2019) (14 studies) and Ribeiro et al. (2018) (6 studies). Several of these studies were found in more than one systematic review. Hence, the combined list of all the studies included in these systematic reviews after excluding duplicates was 39 articles. Only one study was included in all the reviews (Seirawan et al., 2012); another three were included in three of the four reviews (Jiang et al., 2005, Piovesan et al., 2012, de Paula et al., 2015a). Moreover, 25 primary studies appeared in one only review. Table 7 presents the combined list of the included studies according to the identified systematic reviews.

Table 7. Studies included in the four systematic reviews (Ribeiro et al., 2018, de Paulaand Mialhe, 2013, Ruff et al., 2019, Rebelo et al., 2019)

	Studies included in the previous	de Paula &	Rebelo	Ribeiro et	Ruff et	Total
	systematic reviews	Mialhe., 2013	et al.,	al., 2018	al.,	
1	Gift et al., 1992	X	2013		2013	1
2	Petridou et al., 1996	X			Х	2
3	Milgrom et al., 1998	X				1
4	Astrøm & Okullo., 2003	X				1
5	Viviane & Sandra., 2003			Х		1
6	Egri & Gunay., 2004	Х				1
7	Muirhead & Marcenes., 2004	Х				1
8	Sandra et al., 2005	Х				1
9	Jiang et al., 2005	Х	Х		Х	3
10	David et al., 2006	Х			Х	2
11	Gherunpong et al., 2006	Х				1
12	Muirhead & Locker, 2006	Х				1
13	Bernabé et al., 2007	Х				1
14	Bernabe et al., 2007	Х				1
15	Blumenshine et al., 2008	Х	Х			2
16	Freire et al., 2008	Х				1
17	Pau et al., 2008	X				1
18	Petersen et al., 2008	X				1
19	Pongpichit et al., 2008				Х	1
20	Butani et al., 2009	Х				1
21	Jürgensen & Petersen., 2009	X			Х	2
22	Gradella et al., 2011		Х			1
23	Jackson et al., 2011	Х	Х			2
24	Garg et al., 2012		Х	X		2
25	Guarnizo-Herreno & Wehby., 2012		Х		X	2
26	Piovesan et al., 2012		Х	X	Х	3
27	Seirawan et al., 2012	X	Х	X	Х	4
28	Krisdapong et al., 2013		Х		Х	2
29	Detty & Oza-Frank., 2014				Х	1
30	Osman & Alsayed., 2015			X		1
31	Agaku et al., 2015		Х		Х	2
32	de Paula et al., 2015a		Х	x	Х	3
33	Nasuuna et al., 2016		Х			1
34	Neves et al., 2016		X			1
35	Paula et al., 2016		X			2
36	Sameer et al., 2016		X		X	2
37	Maharani et al., 2017		X			1
38	Kaewkamnerdpong & Krisdapong., 2018		X			1
39	Naavaal & Kelekar., 2018		Х			1
	Grand Total	21	18	6	14	59

2.4.7.12.2 Location of the studies

The study undertaken by Egri and Gunay (2004) gathered data from 44 developing countries. The remaining 38 studies were originated from 17 different countries. Ten studies were conducted in the US, followed by Brazil (9 studies). Four studies were conducted in Thailand, while two were conducted in China, India and Peru. The other ten studies were carried out in different countries. Table 8 illustrates the countries of the primary studies included in the previous systematic reviews. Interestingly, given the context of the present thesis, very few studies have been conducted in Middle Eastern countries.

Country	Author, year
Australia	Nasuuna et al., 2016
Brazil	Viviane & Sandra., 2003, Sandra et al., 2005, Freire et al., 2008,
	Gradella et al., 2011, Piovesan et al., 2012, de Paula et al., 2015a,
	Neves et al., 2016 and Paula et al., 2016
Canada	Muirhead & Locker., 2006
China	Jiang et al., 2005, Petersen et al., 2008
Greece	Petridou et al., 1996
India	David et al., 2006, Garg et al., 2012
Indonesia	Maharani et al., 2017
Laos	Jurgensen & Petersen., 2009
Pakistan	Pau et al., 2008
Peru	Bernabé et al., 2007, Bernabé et al., 2007
Saudi Arabia	Sameer et al., 2016
Sudan	Osman & Alsayed., 015
Thailand	Gherunpong et al., 2006, Pongpichit et al., 2008, Krisdapong et al.,
	2013, Kaewkamnerdpong & Krisdapong, 2018
Turkey	Egri & Gunay., 2004
Uganda	Astrom & Okull., 2003
UK	Muirhead & Marcenes., 2004
US	Gift et al., 1992, Milgrom et al., 1998, Blumenshine et al., 2008, Butani
	et al., 2009, Jackson et al., 2011, Guarnizo-Herreno and Wehby., 2012,
	Seirawan et al., 2012, Detty and Oza-Frank., 2014, Agaku et al., 2015,
	Naavaal & Keleka., 2018

Table 8.	Country of	origin of the	studies	included in	n the systematic	reviews
----------	------------	---------------	---------	-------------	------------------	---------

2.4.7.12.3 Study designs

Table 9 summarises the study design of studies included in all the systematic reviews.

The studies were grouped into four types: longitudinal, case-control, cross-sectional,

and ecological.

Table 9. Study design of the studies included in the systematic reviews

Study design of the included studies	Number of studies	Author, year
Longitudinal studies	2	Pongpichit et al., 2008, Nasuuna et al., 2016
Cross-sectional studies	32	Petridou et al., 1996, Milgrom et al., 1998, Astrøm and Okullo., 2003, Viviane & Sandra., 2003, Sandra et al., 2005, Jiang et al., 2005, David et al., 2006, Gherunpong et al., 2006, Bernabé et al., 2007, Bernabe et al., 2007, Blumenshine et al., 2008, Freire et al., 2008, Petersen et al., 2008, Pau et al., 2008, Butani, et al., 2009, Jürgensen and Petersen., 2009, Gradella et al., 2011, Jackson et al., 2011, Garg et al., 2012, Guarnizo-Herreno and Wehby., 2012, Piovesan et al., 2012, Seirawan et al., 2012, Krisdapong et al., 2013, Detty & Oza-Frank., 2014, Osman & Alsayed., 2015, Agaku et al., 2015, de Paula et al., 2015a, Neves et al., 2016, Sameer et al., 2016, Maharani et al., 2017, Kaewkamnerdpong and Krisdapong., 2018, Naavaal and Kelekar., 2018
Case-Control study	1	Paula et al., 2016,
Ecological studies	4	Gift et al., 1992, Egri and Gunay., 2004, Muirhead and Marcenes., 2004, Muirhead and Locker., 2006
Total	39	

2.4.7.12.4 Sample size

The sample size of the included studies in the four systematic reviews varied significantly, possibly related to the difference in the study design, research resources and availability of participants. The sample size was unknown only in Egri and Gunay (2004) (Table 10). Gift et al. (1992) included the largest sample size (N=109,603), which accounts for approximately 30.5% of the total participants (N=359,912) in all the 39 included studies. Agaku et al. (2015) included 65,680 participants, comprising more than 18.3% of the total sample.

In contrast, thirty-one studies contributed less than 1% of the overall sample, with Viviane and Sandra's (2003) having the least participants (n = 101). Piovesan et al. (2012) also had a small sample size (n=281).

	Studies included in all systematic review	Sample size	Age group
1.	Gift et al., 1992	109.,603	≥2 years
2.	Gradella et al., 2011	765	2-4 years
3.	Seirawan et al., 2012	2.,313	2-16 years
4.	Garg et al., 2012	600	3-5 years
5.	Neves et al., 2016	836	3-5 years
6.	Viviane & Sandra., 2003	101	4 years
7.	Sandra et al., 2005	861	≥4 years
8.	Muirhead & Marcenes., 2004	1.,968	≥ 5 years
9.	Nasuuna et al., 2016	24.,678	4-7 years
10.	Milgrom et al., 1998	828	5-11 years
11.	Osman & Alsayed., 2015	385	5-15 years
12.	Naavaal and Kelekar., 2018	3.,833	5-17 years
13.	Butani et al., 2009	416	6-10 years
14.	Maharani et al., 2017	984	6-7, 10-11 years
15.	Agaku et al., 2015	65.,680	6-17 years
16.	Guarnizo-Herreno & Wehby., 2012	46.,750	6-17 years
17.	Detty & Oza-Frank., 2014	16.,022	8 years
18.	Muirhead & Locker., 2006	59.,266	8-11 years
19.	Paula et al., 2016	1.,149	8-19 years
20.	Pongpichit et al., 2008	1.,211	9-13 years
21.	Bernabé et al., 2007	903	11-12 years
22.	Bernabe et al., 2007	805	11-12 years
23.	Gherunpong et al., 2006	1.,226	11-12 years
24.	Jürgensen & Petersen., 2009	621	11-13 years
25.	Jiang et al., 2005	2.,662	11, 13 and 15 years
26.	Petersen et al., 2008	2.,662	11,13, and 15 years
27.	David et al., 2006	838	≥12 years
28.	Pau et al., 2008	500	≥12 years
29.	de Paula et al., 2015a	515	12 years
30.	Egri & Gunay., 2004		12 years
31.	Kaewkamnerdpong & Krisdapong., 2018	925	12 years
32.	Piovesan et al., 2012	281	12 years
33.	Krisdapong et al., 2013	1.,063	12-15 years
34.	Petridou et al., 1996	360	12-17 years
35.	Astrøm & Okullo., 2003	1.,146	13-19 years
36.	Freire et al., 2008	761	≥15 years
37.	Sameer et al., 2016	480	16-18 years
38.	Blumenshine et al., 2008	2.,871	≤<17 years
39.	Jackson et al., 2011	1.,782	≤<18 years
	Total	359,912	

2.4.7.12.5 Participants' ages

Three articles included the lowest children's age (two years old) (Gradella et al., 2011, Seirawan et al., 2012, Gift et al., 1992). However, the former study did not have any children at this age, and the latter excluded those aged 2-5 years from the analysis due to the lack of school performance records. In their study, Gradella et al. (2011) also included children aged two years and those aged 3-4 years in their analysis. The maximum age group was up to 19 years old (Åstrøm and Okullo, 2003). Other six studies included participants aged up to 17 years.

Twelve years was the most common age group investigated, and this threshold was used as the minimum or maximum age limit in eleven studies (Table 10).

2.4.7.12.6 Exposures

The included studies investigated different OHCs as exposures. Dental caries was the most OHCs investigated, as 23 studies assessed this dental condition. The term "oral health" was used to describe exposure in 14 studies. Dental pain was investigated in ten studies under different terms (oral pain, toothache and dental pain). Three studies described oral hygiene and gingival bleeding on probing, whereas four investigated malocclusion. Moreover, five studies assessed dental trauma (David et al., 2006, Muirhead and Locker, 2006, Jürgensen and Petersen, 2009, Piovesan et al., 2012, Neves et al., 2016).

2.4.7.12.7 Exposure measures

The assessment of the OHCs as the exposure was through clinical examination or perceived OH. The clinical data were obtained either by a clinical examination conducted by the researchers or using a previously recorded dental examination or

dental records. The perceived OH was self-reported either by the child or their parents (see Appendix 5).

Twenty studies assessed OHCs through clinical examination. DMFT index was used to measure dentine dental caries in 14 studies (Petridou et al., 1996, David et al., 2006, Jürgensen and Petersen, 2009, Gradella et al., 2011, Piovesan et al., 2012, Krisdapong et al., 2013, Paula et al., 2016, Kaewkamnerdpong and Krisdapong, 2018, Osman and Alsayed, 2015, Freire et al., 2008, Astrøm and Okullo, 2003, de Paula et al., 2015a, Gherunpong et al., 2006). Additionally, the DMFS index was used to measure dentine dental caries in one study (Petridou et al., 1996), dmft index in two studies (Osman and Alsayed, 2015, Paula et al., 2016), df-t index in one study (Garg et al., 2012), ICDAS in one study (Neves et al., 2016) and the PUFA index in one study (Gradella et al., 2011).

Additionally, other studies adopted specific criteria to define caries. Viviane and Sandra (2003) measured caries in terms of the presence or absence of caries lesions. Sandra et al. (2005) used the criteria established by the Brazilian Ministry of Health survey, and Seirawan et al. (2012) recorded dental caries as cavitated and non-cavitated caries lesions.

Among the studies that investigated dental trauma, only one specified the Traumatic Dental Injury index (TDI) as a measurement tool to assess dental trauma (Neves et al., 2016).

Oral hygiene status was assessed in three studies according to the amount of dental biofilm and dental calculus using the Oral Hygiene Index Simplified (OHI-S) (David et al., 2006, Kaewkamnerdpong and Krisdapong, 2018, Gherunpong et al., 2006). Similarly, two studies used bleeding on probing (BOP) (Paula et al., 2016, de Paula et

al., 2015a) to evaluate gingival health. Additionally, four studies assessed malocclusion using different measurements, like the Index of Orthodontic Treatment Need (IOTN) (Gherunpong et al., 2006), self-perceived malocclusion (Bernabé et al., 2007) and Dental Aesthetic Index (DAI) (de Paula et al., 2015a). Furthermore, Neves et al. (2016) examined malocclusion based on the presence of at least one of the following occlusal deviations: overbite (>2 mm), overjet (>2 mm), anterior open bite, anterior or posterior crossbite. Five studies utilised previously recorded dental data or dental records, including the National Health Interview Survey (NHIS) in the US (Gift et al., 1992, Naavaal and Kelekar, 2018), the WHO Global Oral Data Bank for DMFT among 12-year-olds (Egri and Gunay, 2004), The British Association for the Study of Community Dentistry (BASCD) for 5-year-olds survey in the UK (Muirhead and Marcenes, 2004). Detty and Oza-Frank (2014) evaluated untreated dental caries using the Basic Screening Survey of Ohio 2009-2010. Two other studies used data from school's data screening programs (Milgrom et al., 1998, Muirhead and Locker, 2006)

Seven papers used the parental perception of their child's OHS (Pongpichit et al., 2008, Blumenshine et al., 2008, Butani et al., 2009, Jackson et al., 2011, Agaku et al., 2015, Nasuuna et al., 2016, Guarnizo-Herreño and Wehby, 2012) and another study used parental reporting of dental care (Naavaal and Kelekar, 2018). While self-reported dental health was applied in five cross-sectional studies (Jiang et al., 2005, Bernabé et al., 2007, Bernabe et al., 2007, Pau et al., 2008, Petersen et al., 2008), Paula et al. (2015) applied both parental and child's perception of OH and dental caries.

Finally, toothache or dental pain was investigated in ten studies using self-structured questionnaires (Jürgensen and Petersen, 2009, Pau et al., 2008), Child-OIDP (C-OIDP) (Krisdapong et al., 2013), and CPQ₁₃ (Piovesan et al., 2012), Sameer et al.

(2016). Pongpichit et al. (2008) developed a form to collect dental pain and school absenteeism information. The other three studies used the parental perception of children's toothache in the last 12 months (Maharani et al., 2017, Seirawan et al., 2012, Jackson et al., 2011). Another study utilised previously recorded data (BASCD) (Muirhead and Locker, 2006).

2.4.7.12.8 Outcome

Twenty-two studies examined school performance, while eleven studies used school attendance. The other six studies employed both outcomes.

2.4.7.12.9 Outcome measures

The outcome of interest was assessed using school records or by self-reported measures. Fourteen studies measured overall school performance or subject's grades (Petridou et al., 1996, Muirhead and Marcenes, 2004, Muirhead and Locker, 2006, Detty and Oza-Frank, 2014, Paula et al., 2016, Nasuuna et al., 2016, Maharani et al., 2017, Kaewkamnerdpong and Krisdapong, 2018, Osman and Alsayed, 2015, de Paula et al., 2015a), teacher's report (Viviane and Sandra, 2003, Garg et al., 2012), passing or failing at school (Freire et al., 2008) and the Egri and Gunay (2004) used different educational indicators like adult literacy rate and the student-teacher ratio for primary level.

In contrast, seven studies measured school performance by a self-report questionnaire. Of those specific studies, four used OIDP as a proxy measure of self-reported school performance (Bernabé et al., 2007, Bernabe et al., 2007, Astrøm and Okullo, 2003, Gherunpong et al., 2006), whereas three studies used direct self-reporting (Jiang et al., 2005, David et al., 2006, Petersen et al., 2008). In one study (Blumenshine et al., 2008), school performance was reported by the parents.

School attendance data was obtained from school records in two studies (Pongpichit et al., 2008, Sameer et al., 2016). In seven studies, the parents or guardians enquired about their children's school attendance (Milgrom et al., 1998, Sandra et al., 2005, Butani et al., 2009, Gradella et al., 2011, Neves et al., 2016, Agaku et al., 2015, Naavaal and Kelekar, 2018). Self-reported school attendance in 12-year-old adolescents was used in one study (Jürgensen and Petersen, 2009), whilst both parents and adolescents reported missing school hours in the Gift et al. study (1992).

Combined school performance and school absenteeism were measured in two studies and obtained from school records (Piovesan et al., 2012; Seirawan et al., 2012). Similarly, two studies combined both outcomes using students self-reported measures (Pau et al., 2008, Krisdapong et al., 2013). C-OIDP was used in both studies as a proxy measure of school performance. Additionally, two studies used parents' perceptions of school performance and attendance (Jackson et al., 2011, Guarnizo-Herreño and Wehby, 2012).

2.4.7.12.10 Confounders

Overall, there were 43 confounders controlled for in the 39 studies. However, 15 studies did not control for any confounder. In three studies, the level of confounder control was unknown because of the unavailability of the full text. Three studies controlled more than ten confounders: 11 confounders (Jackson et al., 2011); 14 confounders (Kaewkamnerdpong and Krisdapong, 2018) and 19 confounders (Guarnizo-Herreño and Wehby, 2012). Gender, age and race/ethnicity were the confounders that were controlled the most. Important confounders, such as SES and family structure, were considered and controlled in very few studies despite their

importance in shaping every aspect of the children's development (Amato, 2005) (see Appendix 6).

2.4.7.12.11 Results of the included studies

Among the 39 primary studies, 16 studies examine the association between clinical measures of dental caries and school performance (Petridou et al., 1996, David et al., 2006, Piovesan et al., 2012, Paula et al., 2016, Kaewkamnerdpong and Krisdapong, 2018, Osman and Alsayed, 2015, Freire et al., 2008, Egri and Gunay, 2004, Muirhead and Locker, 2006, Muirhead and Marcenes, 2004, Detty and Oza-Frank, 2014, Seirawan et al., 2012, Astrøm and Okullo, 2003, de Paula et al., 2015a, Garg et al., 2012, Gherunpong et al., 2006). Generally, the results of these studies suggested a negative association between dental caries and school performance, regardless of age and demographic characteristics, except in 3 studies (Piovesan et al., 2012, Seirawan et al., 2012, Paula et al., 2016). The studies argued that psychosocial and socioeconomic factors mediated the association between dental caries and school performance.

In contrast, the ten studies using the parental or child perception of OHS reported an association between OH and school performance (Jiang et al., 2005, Bernabé et al., 2007, Bernabe et al., 2007, Petersen et al., 2008, Blumenshine et al., 2008, Jackson et al., 2011, Nasuuna et al., 2016, Piovesan et al., 2012, de Paula et al., 2015a, Guarnizo-Herreño and Wehby, 2012).

It was observed that most of the studies using self-perceived OH measures did not specify the OHCs related to school outcomes. Some studies tended to specify certain conditions like TDI and malocclusion (Jürgensen and Petersen, 2009, Muirhead and

Locker, 2006, Neves et al., 2016, Piovesan et al., 2012). Nevertheless, only two studies that measure TDI reported nonsignificant associations with school outcomes (Piovesan et al., 2012, Neves et al., 2016). Regarding malocclusion, Neves et al. (2016) reported no link between malocclusion and school absenteeism, while Bernabé et al. (2007) found that only 15.5% of children aged 11-12 years in Peru reported impacts due to self-perceived malocclusion during the last three months.

Further, nine studies examined the association between clinical measures of dental caries and school attendance (Osman and Alsayed, 2015, Sandra et al., 2005, Gradella et al., 2011, Jürgensen and Petersen, 2009, Krisdapong et al., 2013, Milgrom et al., 1998, Neves et al., 2016, Piovesan et al., 2012, Seirawan et al., 2012). Three studies did not report an association between dental caries and missing school days (Seirawan et al., 2012; Krisdapong et al., 2013; Osman and Alsayed, 2015). However, two studies suggested that dental visits were the main factor in school absence because of the oral problem. For example, Gift et al. (1992) reported 51 million school hours lost annually because of dental visits and OH problems in the USA. In contrast, Naavaal et al. (2018) reported 34 million school hours lost annually because of emergency dental care.

Toothache or dental pain was investigated in ten studies and was usually considered the primary mechanism by which OHCs influenced school absence (Jackson et al., 2011, Krisdapong et al., 2013). Seirawan et al. (2012) found that dental pain reported by parents was associated with school absence and grade point average in primary and elementary school children in the US. Similar results were reported by Piovesan et al. (2012). Maharani et al. (2017) conveyed a significant link between toothache and school performance. However, a reanalysis of their data conducted by Quadri et al. (2019) revealed no evidence for such claimed association as the initial result was due to an error when interpreting the results. Despite that, two studies reported the prevalence of missing school days because of dental pain to be 18% in Saudi Arabia (Sameer et al., 2016) and 6% in Thailand (Pongpichit et al., 2008). However, they did not support the claim by testing a formal hypothesis. Pau et al. (2008) has shown that 32% of children reported the impact of pain on going to school and learning in class compared with 67% of those who did not report the impact of dental pain on their school life. Finally, Jürgensen and Petersen (2009) and Muirhead and Locker (2006) also reported an association between dental pain and missing school days.

2.4.7.13 Results and conclusions of the systematic reviews

Both de Paula and Mialhe (2013) and Ribeiro et al. (2018) described the selected studies, but no meta-analysis was undertaken. de Paula and Mialhe (2013) concluded that school performance and school attendance were associated with OHCs. On the contrary, Ribeiro et al. (2018), based on the findings of two good quality papers, suggested no association between poor OH and school performance unless mediated by socioeconomic and psychosocial factors. Thus, the study could not support the association. It is important to mention that Rebelo et al. (2019) and Ruff et al. (2019) justified combining the data in a meta-analysis using a random-effects method.

Consequently, Rebelo et al. (2019) suggested that children with one or more carious teeth or those with a poor parental perception of their child's OH were more likely to experience poor school performance and poor school attendance (Table 11).

Table 11. The result of the relation between OH measures and school outcomes fromRebelo et al. (2019) metanalysis

	Decayed tooth/teeth	Poor parental perception of the child's oral health OH
Poor school performance	↑ 44%	↑ 51%
Poor school attendance	↑ <mark>57%</mark>	↑ 35%

At the same time, Ruff et al. (2019) reported that children and adolescents with poor OH have a higher probability of having low academic performance (OR: 1.52; 95% CI, 1.20-1.83) and high school absenteeism (OR: 1.43; 95% CI, 1.24-1.63), than children with better OH.

2.4.7.14. Risk of bias

The methodological quality assessment was performed on all the systematic reviews, except for de Paula and Mialhe (2013). However, the quality assessment tool and the rating scale varied among these reviews.

Ruff et al. (2019) used the modified NOS for non-randomised studies. The studies were classified as low risk of bias (\geq 4 points) and high risk of bias (<4), giving the cross-sectional and the case-control equal scores. Similarly, Rebelo et al. (2019) used the modified NOS but different scales. Cohort and case-control were rated as low-risk studies if they received a score of 9 stars, those with scores of 6-8 stars were evaluated as moderate, whereas those with less than five stars were considered low quality. The cross-sectional studies had to score 7 to earn a low-risk assessment, 4-6 stars were scored as moderate, and those with three or less were assessed as low quality.

Ribeiro et al. (2018) used the National Heart, Lung and Blood Institute tool with ten items scale. Accordingly, the studies that discussed all the ten items were evaluated as good. In contrast, papers that discussed seven items were considered fair. Likewise, the authors evaluated papers as of poor quality if they did not discuss the validity and reliability of exposure, whilst the use of self-reported outcome measures and no confounders statistical analysis was evaluated. This reflects the variations of their results concerning the quality assessment (see Table 12).

Rebelo et al. (2019) evaluated two studies as low quality and the rest as moderate quality. Ruff et al. (2019) evaluated four studies as having a high risk of bias and the remaining as low risk of bias, while Ribeiro et al. (2018) assessed two studies as having good quality, one was fair and the remaining as being of poor quality.

Authors	Rebelo et al., 2019	Ribeiro et al., 2018	Ruff et al., 2019
Petridou et al., 996			High risk
Viviane and Sandra., 2003		Poor	
Jiang et al., 2005	Moderate		High risk
David et al., 2006			High risk
Blumenshine et al., 2008	Moderate		
Pongpichit et al., 2008			Low risk
Jurgensen & Petersen., 2009			Low risk
Jackson et al., 2011	Moderate		
Gradella et al., 2011	Moderate		
Garg et al., 2012	Low	Poor	
Guarnizo-Herreno and Wehby., 2012	Moderate		Low risk
Piovesan et al., 2012	Moderate	Good	Low risk
Seirawan et al., 2012	Moderate	Fair	Low risk
Krisdapong et al., 2013	Moderate		Low risk
Detty and Oza-Frank., 2014			Low risk
Agaku et al., 2015	Moderate		Low risk
de Paula et al., 2015a	Moderate	Good	Low risk
Osman & Alsayed., 2015		Poor	
Neves et al., 2016	Low		
Paula et al., 2016	Moderate		Low risk
Sameer et al., 2016	Moderate		High risk
Nausana et al., 2016	Moderate		
Maharani et al., 2017	Moderate		
Kaewkamnerdpong and Krisdapong., 2018	Moderate		
Naavaal and Kelekar., 2018	Moderate		

Table 12. The risks of bias results from the three reviews (Rebelo et al., 2019, Ruff etal., 2019, Ribeiro et al., 2018)
2.4.7.15 Publication bias

Rebelo et al. (2019) did not test for publication bias because they argued that the metaanalysis included less than ten studies. On the contrary, using funnel plots, Ruff et al. (2019) tested less than ten studies for publication bias. The result suggested a lack of small studies in the current literature.

2.4.7.16 Funding and conflict of interest

Rebelo et al. (2019) and Ruff et al. (2019) reported no conflict of interest. Only de Paula and Mialhe (2013) described its source of funding.

2.4.7.17 Recommendations

Despite the diversity between the four systematic reviews, they reported similar recommendations. All the reviews agreed with the need for more high-quality longitudinal studies involving different populations and age groups. These should consider using valid measures and control for confounders, such as socioeconomic status, family structure, general health and others. They also suggested the adoption of reliable and valid measures to assess OH. When it comes to dental caries, Ribeiro et al. (2018) recommended the use of indices such as Pulpal involvement, Ulceration, Fistula and Abscess (PUFA) and Caries Assessment Spectrum and Treatment (CAST) to account for the different stages of the caries status. Finally, using reliable data, such as school records of both school performance and school attendance, to assess those outcomes will strengthen the results of such studies.

2.4.8 Discussion

The four systematic reviews covered 39 primary studies, encompassing 359,912 children aged between 3 and 19 years. All the reviews suggested an association between OH and school performance and attendance except for Ribeiro et al. (2018).

Nevertheless, there were many methodological differences among the 39 primary studies, limiting the reviews' generalizability and strength. For example, 92% of the studies were cross-sectional, with primarily low or moderate quality. Similarly, roughly 30% of the primary studies relied on self-reported measures of dental caries rather than a clinical examination. Several studies relied on parents' reports of their child's oral health (Jackson et al., 2011), raising concerns about the validity of these measures (de Almeida et al., 2013).

Although the remaining studies involved a dental clinical examination, most studies relied on data obtained using the DMFT index to assess dentine dental caries (Kaewkamnerdpong and Krisdapong, 2018) or gathered DMFT data from hospital records (Egri and Gunay, 2004). Despite being the most commonly used index worldwide to assess dental caries (Casamassimo et al., 2009), only cavitated teeth are registered through DMFT, limiting a proper assessment of the severity of dental caries (de Souza et al., 2014).

Only one of the primary studies (Gradella et al., 2011), a moderately well-designed cross-sectional study, accounted for the severity of dental caries and their untreated sequela when exploring links between OH, school performance and school attendance. Nevertheless, this is crucial in assessing such an association, as some studies showed no association between cavitated caries and school performance unless dental pain was present (Seirawan et al., 2012).

In addition, 36% of the studies measured overall school performance based on school grades, while the remaining studies relied on self-reported measures. Some of these studies applied the OIDP index as a proxy for school performance (Bernabé et al., 2007), some used adolescent self-reported measures (Petersen et al., 2008), and one study relied on parents' self-reporting (Blumenshine et al., 2008).

Similarly, only four studies measured the latter via school records of the 12 studies that assessed the link between OH and school attendance. Other studies relied on selfreported measures. However, relying on self-reported measures to assess school outcomes might not be sufficient due to the lack of knowledge of the validity and reliability of those measures, which might result in a recall or social desirability bias (Althubaiti, 2016).

Another relevant aspect to mention is that 43% of the studies primarily used in the review by de Paula and Mialhe (2013) did not aim to study the impact of OHCs on school performance or attendance. For example, some studies aimed to explore the prevalence of oral impacts on children's daily performance using QoL instruments like OIDP or C-OIDP (Bernabé et al., 2007, Astrøm and Okullo, 2003, Gherunpong et al., 2006). In their study, Sandra et al. (2005) aimed to evaluate the psychosocial effect of dental caries using the Parental–Caregivers Perceptions Questionnaire (P-CPQ).

The results of these studies were used to make inferences on the association between OHCs and school performance and attendance, considering the QoL instrument as an indirect measure of evaluating children's school outcomes. Nevertheless, they are not designed specifically to evaluate school outcomes, which might introduce measurement biases. Therefore, combining those studies to examine one specific research question may affect the validity of their results.

Furthermore, most studies on this topic involved children in Western societies, limiting their generalisability, especially in Middle Eastern countries, which have a remarkably higher rate of OHCs (Morgano et al., 2010). Although two of the primary studies (Osman and Alsayed, 2015, Sameer et al., 2016) were cross-sectional studies conducted in Middle Eastern countries, both were evaluated as having poor methodological quality, according to Ribeiro et al. (2018) and Ruff et al. (2019), respectively.

Finally, a noteworthy limitation of the previous studies is that 38% of them did not account for relevant confounders. Some studies accounted for SES and demographics, although school outcomes could also be shaped by other family and psychosocial factors (Banerjee, 2016). Thus, the association between OH and school performance could be confounded or mediated by other relevant aspects, and a comprehensive framework should be considered to investigate the association between the variables. However, unfortunately, such a framework was not evident among the 39 primary studies.

Thus, the question as to the role of OH in school performance and attendance and the pathways which may link these remains unclear among children and adolescents.

2.4.9 Conclusion

This review summarised the evidence on the association between OHCs and school performance and attendance from previous systematic reviews. Their findings suggest that OHCs, particularly dental caries, can negatively affect school performance and the attendance of children and adolescents. Nevertheless, the existing evidence remains unclear due to the many methodological issues of the primary studies. These limitations include the use of unstandardised measures, together with poor adjustment

for confounders, both of which mean that it has not been possible to assume a cause and effect relationship between OHCs and school performance and school attendance.

2.4.10 Updates of the systematic reviews

This section presents an update of the primary literature regarding the impacts of OHCs on children's and adolescents' school performance and school attendance. The search was carried out to identify and assess relevant newly published papers from 2018 until July 202. Details regarding search strategy, mesh terms, and Prisma are reported in Appendix 7.

The new search retrieved eight recent observational studies apart from those included in the four systematic reviews. Those studies examined the association between OHCs and school performance and/or school attendance among children and adolescents aged between 5 and 17 years, involving 157,114 participants. The description of the studies, including their main characteristics, primary findings, and limitations, are reported below.

2.4.10.1 Studies' characteristics

The characteristics of the eight studies are described in Table 13. Overall, there were seven cross-sectional studies (Gopalan et al., 2018, Almeida et al., 2018, Guarnizo-Herreno et al., 2019, Darley et al., 2021, Cunha et al., 2019, Karki et al., 2019, Quadros et al., 2021), and one longitudinal study (Ortiz et al., 2021). The studies were conducted in India (Gopalan et al., 2018), Nepal (Karki et al., 2019), and the US (Guarnizo-Herreno et al., 2019), and the rest were from Brazil (Table 13).

The research investigated different OHCs, including dental caries, malocclusion, consequences of untreated dental caries and dental pain. WHO's criteria for classifying

dental caries using the DMFT index (Gopalan et al., 2018, Cunha et al., 2019, Quadros et al., 2021) or the D component of DMFT (Karki et al., 2019), or D and M components of DMFS (Ortiz et al., 2021) was used to evaluate dentine dental caries. Almeida et al. (2018) used the CAST index to measure caries and sequela. Interestingly, two studies also measured sequela of dental caries using the pufa/PUFA index (Karki et al., 2019, Quadros et al., 2021). In the remaining studies, questionnaires were used to assess oral conditions (Guarnizo-Herreno et al., 2019, Darley et al., 2021).

In addition to dental caries, one study assessed malocclusion using the DAI (Ortiz et al., 2021), one assessed periodontal problems, overjet and open bite as dentofacial problems (Cunha et al., 2019), and one assessed oral health by OHI-S (Gopalan et al., 2018).

Perceptions of a child's OH was examined in some studies. Guarnizo-Herreno et al. (2019) used caregivers' (primarily parents) ratings to assess participants' current OH (categorised as excellent, very good and good versus fair and poor). Other studies used OHRQoL of life measures like the Child Perception Questionnaire (CPQ8-10) (Almeida et al., 2018), the C-OIDP (Karki et al., 2019), and the OIDP (Cunha et al., 2019). Among these studies, only Karki et al. (2019) used OHRQoL as an outcome measure (along with school performance and absenteeism).

Additionally, dental pain was included in five studies. In two studies, the parents reported the child's dental pain via questionnaire (Guarnizo-Herreno et al., 2019, Ortiz et al., 2021), while it was recorded in Almeida et al. (2018) during the dental examination. In the two studies, adolescents self-reported dental pain was assessed (Cunha et al., 2019; Darley, 2021).

Psychosocial factors also were considered in three studies. Gopalan et al. (2018) used the Rosenberg self-esteem scale to evaluate children's self-esteem, whereas Ortiz et al. (2021) measured verbal bullying related to OH via a structured questionnaire. Social capital, as indicated by community level of support, crime level, and adolescent level of happiness, was assessed by Cunha et al. (2019).

Overall, all studies used self-administered questionnaires to collect data regarding demographics, SES, and access to dental services, mostly answered by parents, except for Cunha et al. (2019) and Darley et al. (2021), the questionnaire answered by adolescents. However, one study collected children's demographic characteristics from school records (Karki et al., 2019). Two studies did not mention how those data were collected (Guarnizo-Herreno et al., 2019, Quadros et al., 2021). Three studies also collected data regarding the utilisation of dental services (Ortiz et al., 2021, Darley et al., 2021, Cunha et al., 2019).

All eight studies assessed participants' school performance except for Darley et al. (2021). Four of them obtained grades from the school register (Almeida et al., 2018, Ortiz et al., 2021, Karki et al., 2019, Quadros et al., 2021). While Gopalan et al. (2018) included two measures for school performance, one was last year grades of first language and mathematics obtained from the school register, and the second was parent's report of homework completion. On the other hand, Guarnizo-Herreno et al. (2019) used the National Survey of Children's Health (NSCH) data to obtain school performance data originally collected from caregivers via questionnaires. In contrast, Cunha et al. (2019) measured school failure by asking adolescents to inform their year of study, which was converted into years of schooling to differentiate delay and repetition regarding progress according to the expected age.

Darley et al. (2021) measured school absenteeism in the past 12 months based on adolescent's self-report as the outcome measure. School absenteeism was also measured in three studies (Gopalan et al., 2018, Guarnizo-Herreno et al., 2019, Karki et al., 2019), which was recorded for the past academic year from schools' registers except in Guarnizo-Herreno et al. (2019), where participants' caregivers reported school attendance.

2.4.10.2 Results

2.4.10.2.1 School performance

The association between OHCs and school performance was reported in seven studies. Of the six studies that assessed dental caries through clinical examination, two studies (Cunha et al., 2019, Quadros et al., 2021) found a significant association with school performance. Quadros et al. (2021), in a cross-sectional study using path analysis, found a direct positive association between sequela of dental caries and poor school performance, while dental caries was found to influence poor performance via sequela of dental caries among Brazilian adolescents. Similarly, Cunha et al. (2019) found that adolescents with dental caries, tooth loss, and dentofacial problems were more likely to fail at school.

On the other hand, Karki et al. (2019) concluded that dental caries' experience and consequences significantly affect school performance, school attendance, and quality of life among Indian children and adolescents. The adjusted models revealed that the combination of sequela of untreated caries and dental caries was only associated with C-OIDP. Dental caries, sequela, and restored teeth were associated with neither school performance nor attendance. Almeida et al. (2018) and Ortiz et al. (2021) did not find an association between dental caries and school performance. Still, the former

found a significant association between CPQ (8-10) and school performance and the latter reported OIDP to be significantly associated with school failure.

Gopalan et al. (2018) found that irregular homework completion was associated with more dental caries among Indian adolescents. However, children with at least one dental problem, including dental caries, could have more school problems and miss one school day by at least 50%. Further, children with poor/fair reported OH were 80% more likely to have school problems and more likely to miss three school days or six by about 60% and 90%, respectively (Guarnizo-Herreno et al., 2019).

Concerning dental pain and other psychosocial factors and school performance, Ortiz et al. (2021) found dental pain and verbal bullying because of OH to be associated with lower school performance but not with other OH clinical measures (DMFS, DAI). Also, Cunha et al. (2019) found that adolescents feeling insecurity, unhappiness, and dental pain were more likely to fail school. However, Gopalan et al. (2018) found no significant association between self-esteem and OH as measured by OHI-S and DMFT.

Except for Gopalan et al. (2018), which stratified the results by gender, all other studies evaluated and adjusted the analysis for some demographic and socioeconomic factors. The final models showed that only a few of those were significantly associated with the outcomes. Being male predicted poor school performance among Brazilian children and adolescents (Ortiz et al., 2021, Quadros et al., 2021), and a higher maternal educational level and living in a family with a high income was associated with high academic performance among Brazilian adolescents (Ortiz et al., 2021). In contrast, Cunha et al. (2019) found that being female, declaring a non-white ethnic group, and being over 16 years old was associated with school failure.

Finally, two studies investigated the association between access and visit to dental care and school performance, but the results showed no significant link between them (Ortiz et al., 2021, Cunha et al., 2019).

2.4.10.2.2 School absenteeism

Overall, all four studies that evaluated the association between OHCs and absenteeism found those to be significantly associated (Gopalan et al., 2018, Guarnizo-Herreno et al., 2019, Darley et al., 2021, Karki et al., 2019). Gopalan et al. (2018) reported a significant association between school absenteeism, caries, and poor gingival health. Similarly, Darley et al. (2021) reported an association between dental pain, frequency of dental visits and school absenteeism. Likewise, Guarnizo-Herreno et al. (2019) showed that children with reported oral health problems were more likely to report poor school attendance. Even though Karki et al. (2019) stated that dental caries and their sequelae were associated with school absenteeism, the link between both was not supported by the final adjusted models.

2.4.10.3 Discussion

As the above studies are more recent, one would expect them to account for previous shortcomings. Thus, producing more robust and conclusive results regarding the association between OHCs and school performance and school attendance.

In fact, some studies accounted for some of the previous limitations, such as the use of more robust measures to assess dental caries (CAST), its sequelae (pufa/PUFA), as well as the use of school records and the adoption of theoretical models. However, this was not clear in the study of Guarnizo-Herreno et al. (2019). This may have something to do with making the study more comparable to its preceding one

(Guarnizo-Herreño and Wehby, 2012). Another factor that may have contributed to the reduction of the quality of the study was the use of secondary data. However, Darley et al. (2021) applied a theoretical model to support their results using secondary data.

Despite that, definitive conclusions concerning the association between OHCs, and school performance and/or school attendance are still unclear. One aspect that might explain such differences is the diversity of the characteristics between studies, such as differences in the measurement of variables and outcomes, country of origin, and the selection of participants. On the other side, the inherent limitations of these studies were another factor that contributed to the observed disparities.

To illustrate such limitations, Gopalan et al. (2018) examined children for dentine caries (DMFT) and gingival health (OHI-S) in 2014. In contrast, school performance and attendance data were obtained from previous academic year school's records. Additionally, the parents reported data regarding homework completion, but it was not clear whether it was from 2014 or the previous academic year. Thus, there is confusion and concern about temporality. Although the authors claimed that the study aimed to evaluate the impact of school performance and attendance on students' OH (reverse causation); however, the theory behind claimed association was not explained.

Similarly, the Karki et al. (2019) study concluded that untreated dental caries and their consequences significantly affected school performance, school attendance, and students' quality of life. However, in the final adjusted models, it was only the combination of pufa/PUFA and dental caries to grade the severity of untreated dental caries (GUDC) associated with C-OIDP. Simultaneously, no relation was demonstrated between dental caries, its sequelae, or the combination of both (GUDC)

with neither school performance nor school absenteeism. Thus, the study conclusions were not supported by their results.

Concerning Ortiz et al. (2021), the study was designed to be longitudinal with two years of follow-up to collect OHCs measures (D, M and DAI) and other school performance variables. Nevertheless, the methodology was not accurately described, leading the article to different interpretations. According to the authors, OHCs were measured in 2012, and school performance was recorded in 2014 to establish temporality. However, the authors mentioned reassessment of the participants in 2014 and reported different data than in 2012 for demographics and clinical variables in tables. This suggests that two examinations have been conducted. However, the authors did not report any recalibration for their examiners after two years, nor did they report which clinical data were used to relate OHCs to school performance. One may suggest that 2012 records were used for such a correlation to claim temporality. Though such an interpretation does not seem logical, knowing that in 2012 there were 1134 participants, and in 2014 there were 771 participants. Of them, only 644 had complete academic performance records. The authors did not explain how missing data were handled, if there are any. Finally, the authors acknowledged that their analysis was cross-sectional.

Finally, except for Gopalan et al. (2018), all the other studies used some form of regression analysis to consider confounders. The former study used stratification to account for gender. Ideally, the authors should have reported the OR for both stratified and unstratified models, tested homogeneity between them, and reported adjusted estimates or whether the adjusted models should be reported separately, such as for boys and girls (Kahlert et al., 2017). Unfortunately, only the results of the models were

reported (stratified and unstratified). Hence, it was unclear which model fits the data best.

Among seven studies, regression analysis was adjusted for confounders; three did not use theoretical models (Almeida et al., 2018, Guarnizo-Herreno et al., 2019, Karki et al., 2019). The other four studies used theoretical models (Darley et al., 2021, Ortiz et al., 2021, Cunha et al., 2019, Quadros et al., 2021). Two of these studies used hierarchical levels (Cunha et al., 2019; Darley et al., 2021), but no details were given on how the models were developed. In contrast, Quadros et al. (2021) and Ortiz et al. (2021) both explained the grounds of their models and used model fit indices to assess the goodness of fit. Where Ortiz et al. (2021) used multilevel logistic regression analysis, Quadros et al. (2021) used path analysis to measure direct and indirect effects.

The selection of confounders based on fit models is a definitive improvement in the analysis, but the challenge remains if all the confounders were accounted for in the first place. Thus, it is wise to explore other social factors within the child's immediate environment, such as family, school, and peers, that influence their development, including their cognitive and OH (Bronfenbrenner, 1977). Also, exploring different analytical methods may help capture the complex and interrelated pathways between these variables to help reach more valid results.

2.4.10.4 Conclusion

To conclude, many studies have explored the association between OH and school performance and school attendance. However, the nature of the association has not been confirmed to date because of the heterogeneity between studies, data quality, and limited analytical approach.

Table 13. Summaries of recent studies on the impacts of OHCs on school performance and attendance among children and adolescents (2018-2021)

Author-year	Study design	Setting/country	Participant age	Sample size	OH measures	Outcome	Outcome measure	Main findings
Gopian et I., 2018	Cross- sectional	Tiruchengode/ South India	12-15	2014 907 girls 1107 boys	DMFT OHI-S	Self-esteem Absent Performance Homework	Rosenberg (1995) School records Marks for language and mathematics Parents	 Significant link between school absences, caries, and poor OHI Irregular homework completion associated with caries
Almeida et al., 2018	Cross- sectional	Brazil	6-8	374 189 girls 85 boys	Assessment Spectrum and Treatment (CAST) Dental pain CPQ	School performance	Reading and mathematics (Provinha Brasil exam)	 No association between school performance and carious or dental pain. Children with lower school performance had higher CPQ scores.
Karki et al., 2019	Cross- sectional	Nepal	5-6 12-15	1137	dmft/DMFT pufa/PUFA C-OIDP	School performance attendances	School register	 High caries rate and pufa/PUFA has a high impact on C-OIDP, school absenteeism and performance High absent rate associated with poor school performance
Guarnizo- Herreño et al., 2019	Cross- sectional	US	6-17	45711	Parent report of children OH problem Child self- reported OH	School performance	School report if the child has any problem in school	1. At least one dental problem can associate with school problems, and miss at least one school day by at least 50% 2. Poor/fair OH perception can associate with 80% of school problems and link with missing 3 school days or 6 by about 60% and 90%.
Cunha et al., 2019	Cross- sectional	São Paulo- Brazil	Adolescence	5162	Caries Periodontal Open bite overjet OIDP	School failure	Delay and repetition regarding progress according to the expected age	1. Adolescents > 16 years, non-white, female insecurity, unhappiness, dental pain, , caries, tooth losses, dentofacial, periodontal changes were more likely to fail at school

Quadros et al., 2021	Cross- sectional	Manaus-Brazil	12	363	DMFT pufa/PUFA	School grades	School register	1.Dental caries linked with poor school performance via pufa/PUFA2.Being male predicted poor school performance
Ortiz et al., 2021	Longitudinal -2 years follow- up	Brazil	12	1,134 610 girls 524 boys 771 410 girls 361 boys	DMF-S DAI	School performance 2014	Basic Education Development Index (IDEB) School register	 Toothache, verbal bullying and being male had a lower academic performance. High SES was associated with high academic performance.
Darley et al., 2021	Cross- sectional	Brazil	13-17	102,072	Dental pain Dental service	School absenteeism	Self-reported questionnaire by adolescent	1. The link between dental pain, school absenteeism, and increased absenteeism rate with increasing dental visits.

2.5 Predictors of school performance and attendance

2.5.1 Overview

In September 2015, the United Nations (UN) Sustainable Development Assembly for 2030 established seventeen global goals to defeat poverty inequality and improve health, with education being the stepping stone of each of the seventeen goals (United Nations, 2010).

Several individuals, parental, family, and environmental factors besides OHCs can restrict children's access to quality education and limit their academic potential (Banerjee, 2016). Because these variables reflect different environment levels, some researchers have proposed a bioecological model to explain school performance and attendance predictors over time (Melvin et al., 2019, Brownell et al., 2016).

The most commonly used bioecological model to explain predictors of school outcomes is the Bronfenbrenner (2005) model of child development. This model views child development as a multifaceted system of relationships influenced by a nested hierarchy of environments, from the immediate settings of parents, peers and schools to overall cultural values and laws (Figure 5). The connections between the individual and their environment shape their development over time.

The centre of the Bronfenbrenner model focuses on the child's own individual biological and physiological factors (e.g., gender, age, knowledge, attitude, skills, and beliefs). The model has different environments around the child's that might influence the child's attributes and development. These systems are expressed in external layers, including the microsystem, which includes parents, schools, and peers (proximal factors); mesosystem, which includes connections between two systems or two tenets of the child's microsystem (e.g. the association between parents and

teacher); exosystem which includes the contexts that shape components of the microsystem but are not directly related to the child (e.g. parent workplace); a macro system which includes culture, laws, politics, and broader society; chronosystem which includes the effect of time, changes and transitions over time (e.g. divorce) (Bronfenbrenner and Morris, 2006). Further, the model argues that the proximal system (microsystem) influences a child's development more than those in the distal system.

This section briefly explains the predictors of children's school performance and attendance, especially those within the children's immediate environment using the Bronfenbrenner (2005) bioecological model.



Figure 5. The Bronfenbrenner (2005) bioecological model for child's development

2.5.2 Child's characteristics

Bronfenbrenner and Morris (2006) identified individual characteristics as demand, resource, and force attributes. Demand is one of the attributes of the individual that directly appears to others, including age, gender, and ethnicity. Resource attributes refer to individual skills, abilities, disabilities, and experience. Force attributes include behavioural factors that influence individual development.

2.5.2.1 Demand

Demand attributes such as gender and ethnicity have been linked to school performance and attendance. School performances (Spinath et al., 2014, Epstein et al., 1998) and school attendance (Corville-Smith et al., 1998, Gubbels et al., 2019) tend to be higher among females than male students, particularly in countries where equal educational opportunities are in place. The reason for this gap is not well understood. However, some researchers have suggested that individual attributes like intelligence, motivation and self-regulation can explain this variation to some extent (Spinath et al., 2010). However, others found no association between gender and school performance (Chambers and Schreiber, 2004), while the link between gender and school attendance rate might differ by type of absenteeism (Skedgell and Kearney, 2018).

Race has been shown to influence student academic life in the US. African American and Hispanic students showed lower scores at school than White students (Seyfried, 1998). In addition to poverty and fewer resources to invest in their children's education (Banerjee, 2016), both former groups experienced discrimination and were disconnected from their classmates and schools, which provided an unfavourable environment to achieve higher grades (Crosnoe et al., 2004). Also, a report by the US

Department of Education (2016) indicated that Indian, Pacific, and Black students were more likely to miss school classes than their white peers. However, Maynard et al. (2017) found that racial differences were associated with other factors like older age, alcohol and marijuana use, and fighting, which increased the risk for lower academic engagement and poor school performance among those minority groups.

Recent data from the UK shows mixed results when exploring how child poverty and education outcomes varied for different ethnic communities. For example, children from Bangladeshi and Pakistani backgrounds were the most likely to live in low SES conditions among all ethnic groups in the UK; however, they had higher educational outcomes than the national average (Antonopoulos et al., 2021).

2.5.2.2 Resource

A range of resource attributes was also linked to school outcomes, including biological and psychological factors. Below average intelligence (IQ) and cognitive abilities have been considered one of the main predictors of school performance (Roth et al., 2015).

Consistent with the critical period model, some studies have shown that low birth weight influenced cognitive development and abilities in childhood (Farajdokht et al., 2017) and adulthood (Grove et al., 2017). A recent systematic review studied the association between very low birth weight (<1500g=VLBW) and cognitive functions in children (Farajdokht et al., 2017). Evidence from forty-four longitudinal studies found that VLBW babies showed a higher incidence of adverse cognitive outcomes, including low IQ, poor language skills, learning and memory deficits, attention deficit, poor academic performance, and behavioural problems compared with normal birth weight children.

Malnutrition and chronic health conditions are essential resources influencing children's school outcomes. At the age of three, malnutrition has been linked to poor cognitive functions at the age of 11 in Mauritius, alongside anti-social behaviour at the age of 8, 11, and 17 years (Liu et al., 2004). Iron deficiency, anaemia, zinc and vitamins A, B2, B1, B6, D3, and E deficiencies have been linked with long-lasting cognitive development problems (Liu et al., 2004). Disabilities such as hearing impairment (Roberts et al., 2002), visual impairment (Simons, 2005), asthma (Forrest et al., 2013), epilepsy (Schouten et al., 2002), cerebral palsy (Beckung and Hagberg, 2002), sickle cell (Schatz et al., 2001), thalassemia (Nash, 1990), haemophilia (Shapiro et al., 2001), obesity (Caird et al., 2014), and type I diabetes have been linked to poor academic performance (McCarthy et al., 2003). Another group of healthrelated conditions related to genetic disorders that might influence children's school achievement is neurobehavioral disorders, including specific learning disability (SpLD), autism spectrum disorder, attentive deficit hyperactivity disorder (ADHD), and Tourette syndrome (Forrest et al., 2013, Karande and Kulkarni, 2005). The association between most of these health conditions and academic performance is mediated by school absenteeism due to medication or hospital visits (Jones et al., 2009).

2.5.2.3 Force

Force attributes represent many children's behaviours that can influence their school life. For example, self-regulation, defined as the individual's capability to control their thoughts, emotions, and behaviours to maintain goal-oriented action, was positively associated with school performance (Weis et al., 2013) and school attendance (Balkıs et al., 2016). Self-regulated students are more proactive at school because they have goals, develop a creative learning environment, apply effective learning measures and have self-efficacy beliefs (Zimmerman, 2011).

Another example of force attribute is school engagement, known as effective involvement in the learning process inside and outside the classroom (Kuh, 2007). Low levels of student school engagement cause academic struggles, behavioural problems, high rates of absenteeism, and high dropout rates (Fredricks et al., 2004). Engagement might be cognitive, which refers to students' effort to exceed school requirements or invest in their learning process (Metallidou and Vlachou, 2007). Engagement might be emotional in terms of their value and feelings about school, but it is challenging to observe their inner feeling. However, this can also be manifested as fear or enthusiasm for homework and interactions with teachers and school extracurricular activities (Reschly and Christenson, 2006). Engagement can also be a behavioural aspect, manifested by student concentration and participation in in-class activities by asking and answering questions or participating in extracurricular activities (Skinner et al., 2008). However, the three definitions are argued to be correlated; for example, it has been suggested that behavioural engagement is shaped by emotional engagement (Pekrun et al., 2002). Similarly, studies on cognitive engagement relate to student emotions and behaviours toward learning (Zimmerman, 1990).

School dropout is a serious problem related to student disengagement and poor school achievement (Fredricks et al., 2004). Most research focusing on this correlation argue that behavioural engagement is a strong predictor of school dropout. Studies have shown that less engaged students, as reflected by high school absenteeism, make less effort, do not do homework, and do not participate in extra-curricular activities (Ekstrom et al., 1986). Also, students who skip school and are suspended from school are more likely to drop out of school (Connell et al., 1995). These results have been confirmed by a longitudinal study carried out in a high school in Canada (Archambault et al., 2008).

2.5.3 Microsystem/mesosystem

Parents or caregivers, teachers, and peers are part of the children's microsystem. Positive connections between these create a balanced and inductive environment for growth and learning (Bronfenbrenner, 1977).

2.5.3.1 Parents

Important parental factors related to student performance and school attendance include parent-child interaction in terms of early life attachment, children's academic activities, and parenting style. These attributes have been considered effective predictors of student school engagement, motivation to attend school and attain higher school performance (Ansong et al., 2017, Blondal and Adalbjarnardottir, 2014, Gubbels et al., 2019).

The type of association and attachment between parents and children established from birth plays a vital role in predicting children's future social interactions with teachers and peers. They also influence children's desire for learning, curiosity, emotional control, independence, and self-esteem (Sroufe, 2005). Ekman (2004), in his book 'Emotions Revealed', stated that infants' brains are ingrained in only six feelings: fear, joy, sadness, anger, surprise, and disgust. Therefore, infants under three years old should be fostered with the primary caregiver and provided unconditional love, support, safety, and a stable environment to have a healthy life. Attunement or harmony between parent and infant is critical between 6 and 24 months of children's life to instil healthy emotions like empathy, forgiveness and appreciation. Deficits in meeting these needs could impair the development of the infant brain, emotions, maturation, and enthusiasm for schooling and education attainment (Clarke and Younas, 2017).

Parental school involvement generally refers to the active participation of parents in children's education process (Jeynes, 2000, Jeynes, 2017). It refers to different dimensions, including parental expectations of their children's school potential, monitoring and involvement with children's homework and progress, attendance of school meetings about children's progress, and participation and volunteering in school activities (Epstein and Dauber, 1991). Thus, it could consider a microsystem or a mesosystem as it involves dimensions connecting parents to child and parents to school and teacher.

In 2003, 2005, 2012, and 2017, Jeynes performed four meta-analyses to identify the association between parental involvement and school performance in primary and elementary schools. The analyses of 21 studies (1988-1999) suggested that parental involvement generally influenced academic achievement among primary school children despite differences in ethnicity or gender (Jeynes, 2003). Moreover, parental involvement influenced standardised test scores more than cumulative average grades. This result was confirmed in each of the four meta-analyses (Jeynes, 2003, Jeynes, 2005, Jeynes, 2012, Jeynes, 2017). In another meta-analysis, Kim and Hill (2015) examined the same association among kindergarten to 12th-grade children. Overall, the study found an association stronger in longitudinal studies than in cross-sectional ones. Father's and mother's involvement in education was significantly associated with student attainment, although mothers were more involved in children's academic activities than fathers.

The association between poor parental involvement and higher rates of school absenteeism has been confirmed (Kearney, 2008). Mothers with low academic expectations for their children and who are less likely to engage in extracurricular

activities were found to have children with higher absence rates at school (Hancock et al., 2018).

Parenting style is another relevant family characteristic that influences children's school life. Masud et al. (2015) conducted a systematic review to examine the association between adolescent parenting style and school performance. The study reviewed 39 studies, including four longitudinal and 35 cross-sectional studies, published between 1987 and 2013. All primary studies confirmed the association between parenting style and academic performance and concluded that the authoritative style (high responsiveness, high demanding) was the most effective style to enhance children's school performance.

Pinquart (2016) conducted a comprehensive meta-analysis, integrating 308 original papers that assessed parenting dimensions with children's school achievement published between 1974 and 2015. The study found that parent responsiveness, behaviour control, autonomy granting, and authoritative parenting were positively linked with school performance, although the effect was small. On the other hand, harsh parenting, psychosocial control, neglect, authoritarian, and permissive caring were associated with lower school achievement.

In addition, parenting style has been associated with chronic student absenteeism and high school dropout rates (Melvin et al., 2019). Blondal and Adalbjarnardottir (2014) conducted a longitudinal study to examine the association between parenting style and student retention in high school adolescents from Iceland from 14 to 22-years old. The result revealed that adolescents with authoritative parents were more likely to complete high school than non-authoritative parents.

The reason why authoritative parenting provides more advantages to children's school life is that authoritative parents (high responsiveness, high demanding) adopt specific achievement strategies like the high expectation of success, coupled with independent and critical thinking that positively influence their children's behaviours toward learning (Aunola et al., 2000). In contrast, children who experienced neglectful parenting were low academic performance because of maladaptive achievement strategies and behaviours. The permissive parenting style is described by high responsiveness but low control. Children exposed to this parenting style performed worse than children under the authoritative parenting style but better than children under the neglectful style (Pinquart, 2016).

2.5.3.2 Peers

Classmate peers are another tenet of an individual's microsystem that can impact learning engagement, especially during adolescence (Wang and Degol, 2014). The youth period has been described as the wish to 'fit in' period (Hamm et al., 2014). To avoid rejection and embarrassment, adolescents might adopt and display similar academic norms and beliefs to their classmates (Rambaran et al., 2017). Throughout the academic year, being together in school might also influence the similarity of academic norms among peers. Wong et al. (2018) recently explored the influence of peers on student emotional, cognitive, and behavioural engagement in Finland (N=1419; mean age=16). The study found that peers profoundly influenced the three dimensions of engagement. The study also found that adolescents with peers who valued school the most and were committed to learning were more likely to transfer these values to classmates. Also, Ansong et al. (2017) tested the association between parents, peers, teacher support, and student engagement in Ghana in adolescents

aged 12 to 23 years. The study found that classmate support was the strongest predictor of school engagement.

2.5.3.3 Teachers

Teachers are a powerful instrument in the learning process as they can provide formal and informal social support. A constructive teacher-student association depends on warmth, empathy, and time and positively impacts student engagement and academic outcomes (O'connor et al., 2011). Hattie (2009), in his book 'Visible Learning', synthesised 800 meta-analyses relating to school attainment. It was concluded that the quality and nature of a teacher-student association had a greater influence on student achievement than student SES. According to Stipek (2006), students who performed poorly in school had a poor association with their teachers. Positive teacher expectations can raise motivation and engage students in class, increasing academic performance and school attendance (Tyler and Boelter, 2008).

2.5.3.4 Family structure

Family is the first environment where the child interacts. Since birth, family is the main source of the essential needs and amenities like care, food, love, shelter and factors needed for good physical, emotional, and cognitive development (Peet et al., 2015). Many family characteristics like structure, size, and stability can impact children' school life (Bronfenbrenner, 1996).

Studies have found that children raised in two-parent families showed better and more stable educational attainment than those raised in single-parent families (Oberlander and Black, 2011). Growing up in a single-parent family reduced children's educational achievement (Garriga and Martínez-Lucena, 2018) and increased the risk of delay in high school graduation and university attendance (Cavanagh and Fomby, 2012). The explanation of schooling disadvantages according to the type of family structure is based on the conceptual model of parental investment from Coleman (1988); as much as parents invest financial, human, and social resources in their children's education, the better educational attainment for their children. According to Amato (2005), children raised by two parents had higher income, saved more money, were offered more time, supervision and care, and greater help with homework. Importantly, when one of the parents was in deficit in their social role, the second parent could make up the deficit (Amato, 2005). In contrast, upbringing in a single-parent household meant that the parental resources were reduced compared with two biological parents. Also, step-parents were less likely to invest or allocate resources for the education of their stepchildren (Coleman, 1988). However, a longitudinal study showed that the negative influence of the family structure on children school attainment was either decreased or eliminated when family socioeconomic resources were accounted for (Sun and Li, 2011).

In line with the above-mentioned theory, there have been many studies that have reported an inverse association between family size and children's academic achievement in terms of the use of resources (the quality and quantity model) (Becker, 1962, Becker and Lewis, 1973, Becker and Tomes, 1976, Becker and Barro, 1988, Becker, 1993).

This theory has been challenged, especially by (Black et al., 2005), who argued that the amount and quality of studies are insufficient to demonstrate a cause-effect association between education and family size. Consequently, Black et al. (2005) conducted a cohort study involving entire Norway's population aged 16-74 years from 1986-to 2000. The study found a negative association between family size and education attainment, but the effect was attenuated when birth order was considered

or use twin births suggesting association but not a cause-and-effect association. Similar results were found in Israel (Angrist et al., 2010).

However, family instability, such as divorce before school entry, is exceptionally critical in predicting poor academic achievement (Heard, 2007), acquisition of behavioural misconduct (Ryan and Claessens, 2013), alcohol misuse and drugs (Cavanagh and Fomby, 2012), and school absenteeism (Gubbels et al., 2019).

2.5.3.5 Family SES

Previous literature has shown that the higher the SES, the better the school performance (White, 1982, Sirin, 2005, Murphy, 2018) and the better school attendance rate among children (Henry, 2007, Morrissey et al., 2014).

Sirin (2005) investigated the evidence for the period 1990-2000 to determine the effect of family SES on children's school performance. The review concluded that SES influenced student academic scores directly through material resources and indirectly through the provision of social capital necessary for cognitive and social skills development and, therefore, to adapt to the school environment.

The direct influence of a family's SES on children's education outcomes means more resources to invest in their children education (Hsin and Xie, 2017). Such investment might have various forms, including stimulating children's physical and general health and cognitive development with a healthy lifestyle, better housing conditions, and a safe neighbourhood (Evans et al., 2010). These features are less common in children from low-income families. Further, children from a high social class often benefited from private hired tutors, books, computers and other resources that facilitated their learning processes (Sirin, 2005). SES can also determine the quality and type of school children are admitted to (Murphy, 2018).

In contrast, the family can result in harmful health and safety conditions such as malnutrition, inadequate health care and environmental hazards that affect school and cognitive skills (Marmot, 2005, Burnett et al., 2015). Furthermore, living in poverty can result in poor quality housing (including dampness and crowding), increasing children's risk of infectious and respiratory diseases, toxic materials like lead and smoking that can impair brain function (Evans et al., 2010). Thus, such exposures can increase missing days at school or late attendance (Morrissey et al., 2014).

Families with low SES tended to be overwhelmed with extra working hours, leaving children at home to care for themselves and/or relying upon elder children to look after their younger siblings, thus reducing the time available for homework or playing (US Census Bureau, 2006). They tended to experience conflicts and instability in their parent's relationship (Heard, 2007). Slack et al. (2004) revealed that parenting style became more rigorous as income decreased, eventually leading to child abuse and neglect. Hussey et al. (2006) reported that children raised in poverty were 1.5 times more likely to be physically abused, emotionally neglected, and two times more likely to report sexual abuse than well-nurtured children. These figures are also linked with parents alcohol or drug use. Child abuse is one of the important risk factors for children and adolescent depression (Jones et al., 2009) and school failure (Karande and Kulkarni, 2005). Child depression or chronic stress creates unconducive conditions for their learning, distracting their attention and concentration (Erickson et al., 2003) impairing their cognitive, memory and innovations (Lupien et al., 2001). These factors are related to 50% of school absenteeism (Jones et al., 2009).

However, evidence has shown that family socioeconomic conditions during childhood appeared to be more relevant in shaping children's development later (Duncan et al., 2013). Inequality in education starts in the womb (Banerjee, 2016) and persists over

time (Pearce et al., 2016). Each period in childhood, starting from the prenatal period, is considered critical or sensitive to the environment if exposed to adverse social circumstances (i.e., family income), and they can influence a child's educational development in different ways (Shonkoff, 2000). For example, consistent with Barker's hypothesis (Barker, 1990), living in an underprivileged family during the prenatal stage, as indicated by low birth weight and preterm birth, is associated with adverse outcomes across an individual's life. Such relationship includes the influence of social inequalities on cognitive development, learning difficulties and poor school performance in later life (Farajdokht et al., 2017, Martinson and Reichman, 2016). According to Karande and Kulkarmi (2005), up to 33% of children born between 32 and 35 of gestational age, and up to 25% of children born <2000g, are at higher risk of experiencing school difficulties in late childhood even if they do not have any neurological impairments. The association mentioned above is mediated by poor intelligence, and vocabulary. nonverbal problems with mathematics and concentration. A similar finding was reported in the meta-analysis (Allotey et al., 2018). Low birth weight children accounted for 38%-48% of the variation in vision impairment and were associated with lower scores in motor functions, behaviours, reading, mathematics, and spelling in elementary school children, and persisted to high school except for mathematics.

Another causal pathway through which family SES affects school outcomes in early life is via cognitive abilities (Hsin and Xie, 2017). Keane and Wolpin (1997) revealed that the most important period for cognitive development is early to middle childhood. During this period, families with high SES can offer greater material resources to develop children's learning skills, reinforcing their academic performance (Korenman et al., 1995). For example, Weizman and Snow (2001) found that children from a low

SES family experienced limited language skills because of the poor speaking skills of the caregiver, including the use of short sentences and a less comprehensive grasp of the language. Coley (2002) stated that 36% of parents with low SES read to their toddlers every day. In contrast, 62% of high-income parents read to their children on a daily basis.

2.5.4 Conclusion

To conclude, student school performance and attendance is essential for optimising a child's development. Besides OHCs, other aspects of the child's immediate environment, including parental and family factors, can influence these school outcomes. Nevertheless, most of them have not been considered when studying the association of OHCs with school outcomes in studies conducted to date. Yet, given that SES, parental and family factors can influence both OH and school outcomes, it suggests that integrating the micro-mesosystem from the bioecological model to Heilmann et al. (2015) life course framework provides an interdisciplinary framework to understand the associations between OHCs and school performance and attendance. The modification of Heilmann et al. (2015) was explained in the next section.

2.6 Theoretical framework of the present study

2.6.1 Overview

Although two models have been proposed to explore the link between OH status and quality of life measures (Wilson and Cleary, 1995, Locker, 1988), no models have been developed to investigate the possible influence of OH conditions on school outcomes. Yet, school performance and school attendance have been examined as one of the children's quality of life indicators in some OHRQoL measures, such as the Oral Impact on Daily Performance (OIDP) (Gherunpong et al., 2004) and Paediatric Oral Quality of Life POQL (Huntington et al., 2011).

This study adopted Heilmann et al. (2015) (Figure 5) for the OH life course framework (previously explained in Chapter 2, Section 2.2.4 (P.42)) modified to include the Bronfenbrenner bioecological-life course model (2005) (see Figure 6).

Both Heilmann et al. (2015) and the Bronfenbrenner (2005) models emphasise the importance of social, economical, biological, behavioural, and psychological factors on human development over time. However, within the bioecological model, the microsystem is the most influential level (Bronfenbrenner, 2005). The system looks specifically at the child's life course within the context of family and school, considering their interrelationships (mesosystem) and their impacts on children's development, including their educational one. Accordingly, adding it into the Heilmann et al. (2015) framework would provide theoretical explanations of how early life environment (family) is associated with children's OH, school performance, and attendance. Additionally, it completes the Heilmann et al. (2015) framework for OH to explain other risk and protective factors of children's OH from the context of family and their interrelationships.

Further, the Heilmann et al. (2015) framework recognises the interrelationship between oral and general health and their impacts on their quality of life. This pathway enabled this study to test the association of OH status with school performance and school attendance, considered important indicators of children's quality of life.

Thus, integration of the factors from the micro-mesosystem of the bioecological model into Heilmann et al. (2015) framework can explain the association and the pathways between OH and children's school outcomes by considering the most important factors related to both OH and school outcomes from the child's early and current life. The section below provides a further explanation of the modifications.



Figure 6. Life course framework for OH (Source: Heilmann et al, 2015)

2.6.2 Modifications to the Heilmann et al. (2015) framework

The modification of Heilmann et al. (2015) involved four main points (Figure 6):

- The modified framework targeted children OH instead of adult SES; thus, birth SES and middle childhood SES (7-8 years old) were considered instead of childhood and adulthood SES.
- 2. The framework explored the SES, biological, behavioural, parental factors accumulated within four periods of the child's life, considering social patterns, temporal ordering, and interrelationships. The four periods included one critical period (birth) and two sensitive periods (birth-2 years, 3-6 years old) that were explored retrospectively, and the current stage of life was cross-sectionally (Grade 2, 7-8 years old).
- Family-related factors, parenting style, and parent academic involvement were integrated within the common risk factors pathway instead of only considering lifestyle and behavioural factors.
- 4. The final step in the model, whereby the impact of OH on general and quality of life were assumed, was changed to study the association of OHCs with school performance and attendance.

Thus, the framework generally hypothesised that low SES at birth and age 7-8 would shape the children's biological, behavioural, parental and psychosocial resources in such a way as to promote the development of OHCs; hence they would exhibit low school performance and attendance. Conversely, children who experienced high SES at birth at age 7-8 would show little or no evidence of OHCs and exhibit better school

attendance and performance than those living within a low SES family. Further details are discussed in Section 3.9.3.3.2.



Figure 7. Modified from Heilmann et al.'s (2015) life course framework for OH

2.7 Rationale for the study

The rationale for the current study was to identify the associations and the pathways between OHCs and children's school performance and school attendance in the Middle Eastern area that have not been sufficiently explored in the existing literature, using a life course approach. Second, the present study addressed life course factors that could influence OHCs, school performance and school attendance.

The impacts of oral diseases on school performance and school attendance among schoolchildren have been studied extensively in four recent systematic reviews as previously discussed (de Paula and Mialhe, 2013, Ruff et al., 2019, Ribeiro et al., 2018, Rebelo et al., 2019), and in further eight epidemiologic studies recently published. However, a definitive conclusion related to the association could not be reached. This is mainly due to the predominance of cross-sectional design studies, concerns regarding validity and reliability of the exposure and outcome measures, and the flaws and poor methodological quality of some studies, as explained in Sections 2.4.7.

In addition, one common major limitation of the previous studies is that the authors did not account for relevant variables that might confound the association between OHCs and school performance and school attendance. The lack of a theoretical framework to analyse such an association is also noteworthy. Many environmental factors like SES, school and family characteristics may influence children's school performance and attendance (Epstein and Sheldon, 2002, Banerjee, 2016). Those factors can also influence the development of OHCs among children (Abreu et al., 2015). Thus, it would be relevant to investigate the complex association between OHCs and school performance and school attendance, considering the environmental and family factors using one comprehensive framework, such as the life course approach.
The life course framework can be considered well-suited to OH epidemiology, as most dental diseases are chronic conditions with multiple causes and are influenced by social conditions (Nicolau et al., 2003). Recently, different life course models have been applied in dental research to examine the factors related to OHCs (Abreu et al., 2015). However, most of the available studies used regression analysis to evaluate the association between variables based on the life course approach. Whilst such an approach is considered well suited for life course studies, it poses some limitations worth mentioning. First, regression analysis only measures direct effects, whereas indirect effects cannot be measured (Berndt and Williams, 2013). Thus, the complex interrelationship between the life course variables cannot be captured sufficiently well. Second, the variables can act either as predictors or outcomes (Petrocelli, 2003). Thus, it is not possible to test a chain of linear associations between predictors. Third, the regression analyses only measured variables, while SEM incorporates measured and indirectly measured (latent constructs) variables. Thus, a regression cannot estimate the combined effect of directly measured independent variables into indirectly observed constructs. While in SEM, directly observed independent variables do not act alone; instead, conceptually consistent independent variables act together to add to the prediction of dependent variables (Berndt and Williams, 2013).

Finally, all studies exploring the association of OHCs with school performance and school attendance and the studies that explored life course factors of children were predominantly conducted in Western societies, limiting their generalisability, especially in Middle Eastern countries, where the burden of oral diseases is significantly higher (Morgano et al., 2010).

OHCs, academic school performance and school attendance might share common predictors. Exploring those variables through a life course approach might help identify

critical opportunities for tackling them and stimulating protective factors to provide evidence-based interventions, targeting the key life stages (age-graded).

Thus, to the researcher's knowledge, this is the first study conducted in the Middle East to explore the influence of environmental, behavioural, biological, and parent's characteristics on the association between oral diseases and schoolchildren's performance and school attendance using a life course approach. The study accounted for most of the factors related to the development of OHCs, which can influence academic performance and school attendance among children. Furthermore, the study was planned to measure OHCs and perceived oral health status and school outcomes using valid and reliable measures. Also, the study design was time-ordered-cross-sectional to verify the temporal relationship between OHCs and school outcomes. Finally, the study hypotheses were analysed using a robust statistical technique (structural equation modelling) (Berndt and Williams, 2013).

Chapter 3. Methodology

This section explains the population and methods used in this study.

3.1 Study design

The study was a time-ordered-cross-sectional design conducted with 466 schoolchildren in Grade 2 in four government primary schools rated as "good" in Muharraq Governorate in KoB and their parents between 2019 and 2021.

The study is called a time-ordered-cross-sectional because exposures and outcome measures were collected once but at different time points for the same participants (Menard, 2002). This design has been suggested to be appropriate to assume temporal orders between the exposures and the outcome variables (Menard, 2002).

School performance and school attendance data were the outcome variables for children in Grade 2. School performance was measured at the end of the 2019-2020 school year (June 2020), considering children's final cumulative grades. School attendance data were obtained from school records by days of absence for six months (September 2019- February 2020).

The exposures included social, biological, behavioural, parental, and clinical data collected at the beginning of September 2019. Parents completed a questionnaire detailing children's characteristics and environment at birth (retrospectively) and presently. Children were interviewed and dentally examined to assess dental caries, dental trauma, malocclusion, enamel development defects (see Section 3.4 for further details of study measures).

3.2 Geographic location

The study was conducted in the Muharraq Governorate in the KoB. The KoB is a small archipelago located in the Arabian Gulf. It covered a total area of 777.95 km² and comprised a population of 1,501,611 in 2019 (Information & eGovernment Authority-KoB, 2019). The Human Development Index for Bahrain was 0.85 in 2019, ranking the country 42nd out of 189 countries; according to United Nations Development Programme (UNDP) (2020), Bahrain is divided into five health regions: Capital, Muharraq; Central; Southern and Northern. Muharraq is the second-largest city in Bahrain, in terms of population density, with a population of 97,458 (World Population, 2019).

The estimated total number of Grade 2 schoolchildren in Muharraq was 2,480: 1,195 males (48%) and 1,285 females (52%) (Education and Quality Authority-KoB, 2019) 2019. In Muharraq, the mean number of decayed, missing and filling teeth (d₄mft) for six-year-olds was 4.46 from 2011-to 2012 (Naseeb, 2016). This was an increase of 1.01 on the mean of dmft compared to the 2005 oral health survey.

3.3 Selection of population

3.3.1 Target population

The target population consisted of Grade 2 government school children of both genders in the KoB in addition to their parents.

3.3.2 Population source and setting

The population source comprised all children in Grade 2, aged 7-8-years in 2019-2020 and their parents. All participants were attending government primary schools in the Muharraq Governorate.

The government education system in the KoB is segregated by gender. Recently Ministry of Education (MOE) started quality evaluation for each school in KoB. The quality and qualification assurance (QQA) committee applied the assessment, where schools receive a specific quality score: 'outstanding', 'good', 'satisfactory' or 'inadequate'. The QQA examine a school's quality based on three tenets: quality of processes (students' support and guidance, and teaching, learning process), quality of outcomes (students' academic performance and their personal development), and quality assurance of outcomes and processes (leadership, management). Private schools were not included in the study as they have different curricula from public schools.

The list of all students attending primary schools in Muharraq was not available. Thus, a random sampling process involving all schools with different scores was not possible. Therefore, the researcher had to sample the schools as a unit instead of sampling the individuals. However, since the schools were already clustered according to the quality score, the researcher restricted the sample to only those rated as 'good'. Selecting good-rated schools is more logical than selecting the other two extremes of outstanding or inadequate schools. Similarly, it forms a midpoint between the acceptable school ratings of outstanding and satisfactory". Furthermore, it prevents the influence of different school-level factors that may influence school academic performance and school attendance.

3.3.3 Study group

The study group comprised schoolchildren aged 7-8 years old enrolled in Grade 2 from 'good' rated government schools in the Muharraq Governorate in 2019-2020.

3.3.4 Exclusion criteria

Schools were excluded according to the following criteria:

- Primary schools located in governorates other than Muharraq Governorate.
- Primary schools are rated as 'outstanding, 'satisfactory', or 'inadequate'.
- Primary schools with mixed primary-intermediate stages.
- Primary schools with no Grade 2 classes.
- Private schools.

The exclusion criteria for children from the selected schools were:

- Children not enrolled in Grade 2.
- Children less than seven years of age or older than eight years of age.

3.3.5 Sampling and recruitment

The study applied a non-probability sampling technique whereby all schoolchildren in Grade 2 in 'good' rated government primary schools were recruited for the reasons explained earlier.

School's latest ratings were retrieved online from the Ministry of Education (MOE) of the KoB website for the Muharraq governorate: <u>http://www.bqa.gov.bh/En/Reports/SchoolsReports/Pages/default.aspx</u>. There were 23 government primary schools in Muharraq. Of them, nine were rated 'outstanding'; six were rated 'good', six 'satisfactory' and two 'inadequate' in 2018-2019. Most of the available reports were the third quality evaluation reports for the 2018-2019 academic year. Six primary schools in Muharraq Governorate rated as 'good' by Bahrain's MOE were selected (Education and Quality Authority-KoB, 2019), three schools with male students and three schools with female students (see Appendix 8). Of these schools, one school with male students did not have Grade 2 classes and thus was excluded. One school with female students was also excluded because it had mixed primary and intermediate stages. The remaining four schools had 528 children (see Figure 8).

The researcher visited the identified schools in September 2019 to arrange the application of the study with headteachers and children. The parents of all of the children who enrolled in the identified schools and their children in Grade 2 were invited to participate. The researcher visited each classroom to explain to the children the study's objectives and their roles in delivering and returning the informed consent (IC) (Appendix 9) and the questionnaire for their parents in relation to the dental examination. Class headteachers and researchers distributed IC and parents' questionnaires to children during school enrolment to be handled by their parents.

Of the 528 schoolchildren, one female child aged ten years was excluded due to the age criterion. Further, 18 children who participated in the pilot study were excluded from the main study, and 33 students refused to participate. The potential sample size, therefore, was 477 children. Eleven children were also excluded from the analysis because they did not return the parent's questionnaire. Therefore, the final sample consisted of 466 children.



Figure 8. Flowchart of the sampling process

3.3.6 Sample size calculation

To achieve a 5% or less standard error, with a 95% confidence interval in an estimated prevalence of 50% of children reporting excellent school performance, 385 children were minimally required. Likewise, in an estimated prevalence of 50% of children having 100% attendance, 385 children were minimally required. Anticipating a 20% drop-off, the final sample size required was 462. Furthermore, the sample size as regards this study was determined based on the use of the structural equation modelling (SEM) analysis technique. According to Hair et al. (2014), a sample size of at least 200-400 is considered suitable for SEM. Further, the authors also advised that SEM analysis become too sensitive to a sample size of 500, making the model fit difficult. Therefore, the concern for this study was to have a 400-sample size at a minimum.

3.4 Data collection instruments

Different instruments were employed to collect the data, including parents' selfadministered questionnaires, face-to-face interviews, children's clinical and anthropometric examinations, WhatsApp media, and school records. Questionnaires were initially developed in English and subsequently translated into Arabic (the PhD candidate native language). This will be detailed in Section 3.6.1.

 The parent's questionnaire was answered by either mother or father and used to collect data regarding the child's and family's demographics, biology, behaviours, besides socioeconomic and parental factors (Appendix 10). The selection of factors was based on the review of Heilmann et al. (2015) life course framework for OH and exploring factors related to OH and school performance and school attendance, which was discussed in Chapter 3. The questionnaire consisted of five sections and 176 items, with primarily closedended questions.

- The face-face interviews with children collected data regarding the child's Oral Health-Related Quality of Life measures (OHRQoL) and children's perceived support from teachers and peers (Appendix 11).
- A clinical form was developed to collect the child's OH status, height and weight (Appendix 12, 13). The clinical form was based on the guidelines of the dental indices employed.
- Online communication via mobile phone (WhatsApp) was conducted to collect the school performance (outcomes) data at the end of the 2019-2020 academic year.
- School records were used to obtain data on children's absence days (2019-2020).

3.5 Variables

The following sections explain all the variables measured in this study in the parent's questionnaire and children's face-face interviews and clinical examinations. Finally, it explains the outcomes measures. Table 20 summarises the variables, including the instruments, measures and how they were analysed.

3.5.1 Parent's self-administered questionnaire variables

The questionnaire consisted of five sections: demographics, family structure and current SES, family structure and SES at the time of the child's birth, health and OH

behaviours and perceptions, parenting style and parent's academic involvement (Appendix 10).

3.5.1.1 Demographics

The demographic characteristics comprised the child's age, gender, place of birth, ethnicity, and area of residence (Items 5-9).

3.5.1.2 Current life variables

The current life variables included information regarding when the child was in Grade 2 (currently-7-8 years old) (2019-2020).

3.5.1.2.1 Family structure

This includes the family type (Item 10), parent's marital status (Item 11), along with the current number of siblings (Item 12).

3.5.1.2.2 Family SES

SES of the family was assessed according to the educational level of the parents (Items 13, 17), their occupation (Items 15, 19), total household income (Item 21). Housing conditions were assessed in terms of tenure (Item 22), the number of rooms in the household excluding kitchen and toilets (Item 23) and the number of people living in the household (Item 24). Overcrowding was indicated when more than two individuals per room were in the household (Hatton and Martin, 2010).

3.5.1.2.3 Oral health behaviours

Parent's and children's oral health behaviours included frequency of tooth brushing, flossing, sugar consumption, use of mouth wash and dental visits (Items 25-36), use of toothpaste (Item 44), fluoride level (Item 43), as well as spit or rinse procedure (Item

45). To determine the fluoride level, parents were asked to write the name of the fluoride toothpaste used if they did not know the concentration.

3.5.1.2.4 Health status

Children's health status indicators were assessed by the presence of chronic health conditions (Item 56) or neurobehavioural disorders, body mass index (BMI), current use of medication (Item 57) and general health perception reported by their parent (40). Parents' (mother & father) health perceptions (Items 38, 40), smoking status (Items 52, 53), and type of smoking (Items 54, 55) were also obtained.

3.5.1.2.5 Parent's characteristics

This part included the parent's academic involvement and parenting style.

3.5.1.2.5.1 Parent's academic involvement

Assessment of parental academic involvement was conducted using the questionnaire developed by Ames et al. (1993), which had been previously employed (Jeynes, 2005, Jeynes, 2017) and validated with Ghanaian students (Ansong et al., 2017). The instrument consisted of seven Items (163-176) questioning parental involvement in children's schooling. These were answered using a five-point Likert scale from 'never' (1) to 'always' (5).

3.5.1.2.5.2 Parenting style

Where possible, both parents were assessed. The Parenting Style Dimension Questionnaire (PSDQ) was employed to evaluate parenting style (Robinson et al., 1995). The questionnaire consists of 32 items (Items 99-162) that assess features of parenting styles. The scale includes 15 items to assess the authoritative style, 12 items

to assess the authoritarian style, and five items to assess the permissive style, using a five-point Likert scale from 'never' (1) to 'always' (5). To classify the parents' preferred style, the scores were added up and divided by the total number of questions at the end of each section. The highest mean score indicated the preferred parenting style. Validation of the instrument had previously been carried out on a different population (Olivari et al., 2013, Robinson et al., 1995). Robinson et al. (2001), reported the Cronbach's alpha (α) to be acceptable for authoritative (α = .86), authoritarian (α = .82) and low for permissive (α = .64).

3.5.1.3 Early life variables

The children's early life variables included information from childbirth until six years (Items 58-94) (see details below).

3.5.1.3.1 Family structure

Current family structure was measured the same way at birth (Items 58-61).

3.5.1.3.2 Family SES

Current family socioeconomic status (CSES) was assessed similarly to BSES (Items 62-70).

3.5.1.3.3 Maternal health status

This includes variables related to the mother during her pregnancy of the participant; mother's age (Item 71): mother's health status was assessed by asking her to report yes/no if she had the following conditions: gestational diabetes (Item 72), hypertension (Item 73), smoking (Item 75) and postnatal depression (Item 77). Also, a question was asked to determine the type of birth (Item 76).

3.5.1.3.4 Children's birth outcomes and feedings pattern (birth-2 years old)

The birth outcomes variables included gestational age at birth (Item 78) and birth weight (Item 79), and birth order (Item 61). Feeding patterns during infancy included exclusive breastfeeding for the first six months (Item 80), duration of breastfeeding (Item 81), bottle-feeding (Item 82), feeding at night (Item 82), sweet drinks (Item 83), use of pacifier (Item 84) and duration of pacifier use among those who answered yes to the previous question (Item 85).

3.5.1.3.5 Children's oral health behaviours (3-6 years old)

Parents reported early oral health status and behaviours in yes/no questions. This included time of the initiation of tooth brushing (Item 87), adult supervision of tooth brushing (Item 88), the pattern of dental visits (Item 89), experience of dental pain (Item 90) and presence of primary dental caries (Item 91).

3.5.1.3.6 Early learning (3-6 years old)

Early learning variables included children's enrolment in pre-school/nursery (Item 93), previous grades (Grade 1) (Item 95) and previous days of absence (Items 96-98) reported by one of the participant's parents for the academic years (2018-2019). Grade 1 was categorised according to the MOE classification and contained six levels: 'excellent', 'very good ', good', 'average', 'acceptable' and 'failed'. School absence days were classified into; 1-5 days, 6-9 days, \geq 10 or no absences for one academic year. The limit concerning permitted days absent in the KoB is ten days per academic year. Hence, this was the categorisation (MOE-KoB, 2017).

3.5.2 Child's face-face interview variables

3.5.2.1 Teacher and peers' social support

Social support can be defined as "behaviours that, whatever directly or indirectly, communicate to an individual that she or he is valued and cared for by others" (Barnes and Duck, 1994). This includes children at school being supported by their parents (discussed above), in addition to teachers and peers.

The child's assessment of teacher and peer perceived support was conducted using questions adapted from Ansong et al. (2017). The original questions were developed and validated (Torsheim et al., 2000) using a five-point Likert scale with Norwegian school adolescents. However, considering the limited capacity of children to understand the five points relating to the Likert scale (Mellor and Moore, 2014), the Likert scale was replaced by three options in this study: 'no' (0), 'sometimes' (1) and 'yes' (2). Face icons were used to facilitate understanding by the children.

3.5.2.2 Oral health-related quality of life

OHRQoL was collected from the children via face-to-face interviews and measured using the Paediatric Oral Health-Related Quality of Life (POQL) (Huntington et al., 2011). The reasons for choosing this instrument were as follows:

- The POQL has ten items used in clinical and population-based studies, suggesting that it is suitable for application in a school setting. Moreover, little time is required to complete it.
- The POQL covers four functional dimensions: physical, role, social and emotional. Similarly, it contains questions regarding pain and any impact from

the child's perspective – of oral health conditions on school attendance and attention in class, which are in line with the primary aim.

- The scale measures each item in terms of 'how often?' (frequency) and 'how much were you bothered?' (severity) using simple Likert scales of 0-3 and 0-4, respectively.
- Validity was tested and confirmed by face, discriminant and convergent validity (Huntington et al., 2011).
- The internal consistency (Cronbach's alpha) for the total score was 0.83. The test-retest (2-4 weeks) interclass correlation was 0.75 and therefore acceptable.

To measure the overall quality of oral health and health in general, both children and their parents were asked one global question (Rowan, 1994).

POQL (Huntington et al. 2011) scores were calculated by multiplying the frequency response (0-3) by the severity response (0-4). Impact scores were then summed and converted to a percentage (x100/120) to give an overall POQL score ranging from 0 to 100, with higher scores reflecting poorer OHRQoL.

3.5.3 Clinical examination

This included the OH examination and the anthropometric measures.

3.5.3.1 Oral health conditions

3.5.3.1.1 Malocclusion

Dental trauma risk factors related to malocclusion were measured. Malocclusion was recorded in the presence or absence (yes/no) of the following conditions, lip incompetence, overjet and anterior open bite (Soares et al., 2018, Neves et al., 2016). Lip competence was described if the maxillary incisors were covered by the lip, while lip incompetence was defined if a larger part of the crown height was visible in the resting position (Zaleckien et al., 2020).

The incisor relationship was measured according to the WHO (1997) guidelines. The overjet was measured to the nearest whole millimetre as the horizontal distance from the labial incisal edge of the most prominent maxillary incisor to the facial surface of the correspondent lower incisor parallel to the occlusion plane. In contrast, the anterior open bite was classified with no vertical overlap between the maxillary and mandibular anterior teeth. When maxillary teeth were positioned lingually to mandibular teeth, it was defined as an anterior crossbite.

3.5.3.1.2 Dental trauma

Dental trauma was assessed in upper and lower central and lateral incisors. Each tooth was examined and coded according to the criteria used in the UK Children's Dental Health Survey 2013 (UK -CDHS) (O'Brien, 1994) (see Table 14). Thus, dental trauma was reported as present when any type of injury or tooth discolouration was diagnosed or absent (yes/no).

Code	Type of dental trauma
0	= No trauma
1	= Discolouration
2	= Fracture involving enamel
3	= Fracture involving enamel and dentine
4	= Fracture involving enamel, dentine, and pulp
5	= Missing due to trauma
6	= Acid-etch composite restoration
7	= Permanent replacements including crow, denture, bridge pontic

3.5.3.1.3 Dental plaque

Dental plaque was evaluated according to the UK-CDHS (2013) criteria (Anderson et al., 2015). Subsequently, the mouth was divided into six sextants; each was visually examined at the buccal and lingual surfaces. The dental plaque status was coded according to 0 = no visible dental plaque, 1 = visible dental plaque, and 9 = assessment cannot be made (e.g., missing teeth). The average plaque condition was recorded and not the worse area in the segment.

3.5.3.1.4 Enamel Developmental Defect (EDD)

Enamel developmental defects were measured using the FDI modified EDD guidelines (Clarkson and O'Mullane, 1989) presence or absence (yes/no). Only each tooth's buccal or labial surfaces were examined (Ramesh et al., 2011). EDD was recorded as present if any conditions in Table 15 were diagnosed. Three types of defects were considered in this study according to the explanation below:

 In demarcated opacity, the enamel's thickness and morphology are normal, but its translucency is altered due to the occurrence of large organic material remnants between the crystal appetites (Koch et al., 1987). These lesions present with a clear border separating the abnormal and normal enamel. Their colour may range from white to cream and yellow to brown (Seow, 2014)

- Diffuse opacity involves normal enamel thickness with changes in enamel translucency (Seow, 2014), but there are no clearly defined boundaries between the lesion and the surrounding enamel. The lesion tends to be white, appearing opaque with a linear, patchy or confluent distribution. It frequently presents on incisal edges or one-third of the crown.
- Enamel hypoplasia is a malformation of enamel that reduces its thickness.
 Clinically, it is seen as grooves or pits and the partial or total loss of surface enamel (William et al., 2006).

Table 15. Modified FDI	, EDD index (Clarkson	and O'Mullane,	1989)
------------------------	-----------------------	----------------	-------

Code	Type of EDD
0	Normal
1	Demarcated opacities
2	Diffuse opacities
3	Demarcated and diffuse opacities
4	Hypoplasia
5	Hypoplasia and opacities

3.5.3.1.5 Dental caries

The researcher examined dental caries using the International Caries Detection and Assessment System (ICDAS) (Ismail et al., 2007, Pitts and Stamm, 2004). The ICDAS system was chosen because of the evidence related to the high validity, feasibility, reliability, and reproducibility of epidemiological studies. The necessity of noncavitated lesions detection was demonstrated. It enables early, and follow-up detection of dental caries and provides a comprehensive assessment of dental caries according to the stages of a carious lesion, activity and topography (Mendes et al., 2010, Ekstrand et al., 2018, Shoaib et al., 2009, Ismail et al., 2007, Braga et al., 2009, Pitts and Stamm, 2004).

ICDAS is a 'two digits" coding system. The first digit code (ranging from 0 to 9) represents the condition of the current restoration and sealants (see Table 16). The second digit (ranging from 0 to 6) refers to caries lesions severity (Ismail et al., 2007) (see Table 17). Code 1 for caries was not used as no air syringes were available during the examinations at the schools (Neves et al., 2016). Thus, all code 1 were excluded including pits and fissures, smooth surfaces, free smooth surfaces, CARS (caries associated with restorations and sealants); instead, all teeth surfaces, including pits and fissures, have been examined starting from Code 2. The examiners were trained to use the ICDAS as described in Section 3.8.1.1.

Table 16. ICDAS codes for restoration	(Ismail et al.,	2007)
---------------------------------------	-----------------	-------

Code	Criterion
0	Sound, specifically, surface not restored or sealed (use with the codes for primary caries)
1	Sealant, partial
2	Sealant, full
3	Tooth-colour restoration
4	Amalgam restoration
5	Stainless steel crown
6	Porcelain, gold, PFM crown or veneer
7	Lost or broken restoration
8	Temporary restoration
9	Used for the following conditions:
	No examination of the tooth surface, i.e., surface excluded (code 96)
	A tooth is missing because of caries (code 97)
	A tooth is missing for reasons other than caries (code 98)
	Unerupted tooth surface (code 99)

 Table 17. ICDAS codes for caries severity (Ismail et al., 2007)

Code	Criterion
0	Sound tooth surface: no evidence of caries after 5 seconds of air drying
1	The first visual change in enamel: opacity, or discolouration (white or brown) is visible at the entrance to the pit or fissure seen after prolonged air drying
2	The distinct visual change in enamel is visible when wet. The lesion must be visible when dry
3	Localised enamel breakdown (without clinical visual signs of dentinal involvement), seen when wet and after prolonged drying
4	Underlying dark shadow from dentine
5	Distinct cavity with visible dentine
6	Extensive (more than half the surface), distinct cavity with visible dentine

3.5.3.1. 6 Consequences of untreated dental caries

The Pulp-Ulcer-Fistula-Abscess index (PUFA) was used to evaluate the presence or absence (yes/no) of any clinical consequences of untreated dental caries (Monse et al., 2010). As shown in Table 18 below, in the PUFA coding, the uppercase letter denotes the condition in permanent teeth, with the lowercase letter describing the condition in primary teeth. The number of lesions presented in the child's mouth for each of the four forms of odontogenic infection was recorded.

 Table 18. PUFA codes (Monse et al., 2010)

Permanent teeth	Primary teeth	Description
Р	р	pulpal involvement
U	u	ulceration
F	f	fistula
A	а	abscess

3.5.3.2 Anthropometric measures

The child's weight and height were recorded to assess body mass index (BMI) using a high-quality balance beam. The US Centre for Disease Control and Prevention (CDC) growth chart and calculator were used to estimate BMI according to age and gender by percentile. Accordingly, children were classified as underweight if their BMI was < 5th percentile, 5th to < 85th were normal, 85 to < 95 were overweight and > 95 were obese (CDC, 2019).

3.5.4 Outcome variables

School performance in a primary school in the KoB was measured by calculating the average cumulative grades for each academic year out of 100 scale points, then categorised as indicated in Table 19. In this study, the average of 9-10 academic subjects for 2019-2020 was calculated and measured as a numerical scale out of 100 points. School attendance was measured based on the number of missing days at school. The headteacher recorded student attendance every day throughout the whole academic year. Days absent were retrieved from schools' records for the first term (September 2019 to February 2020), as during the second term, students were attending schools remotely because of the COVID-19 pandemic.

Table 19. Grading classifications in the Ko	ding classifications i	Grading	19. (Fable	T
---	------------------------	---------	-------	--------------	---

Grades %	Category
< 50	1 Failed
59-50	2 Acceptable
69-60	3 Average
79-70	4 Good
89-80	5 Very good
100-90	6 Excellent

Table 20. Summary of the measurement and analysis of the study variables

Item number in the	Reference if available	How was it measured?	Type of	How was it analysed?		
questionnaire/variable		Categories	Variable			
Parent's questionnaire						
Demographic						
3. Mother's age			Numerical			
4. Father's age			Numerical			
5. Child's age			Numerical			
6. Gender		2 Male / 1 Female	Nominal			
7. Place of birth		2 Bahrain / 1 Outside Bahrain	Nominal			
8. Ethnicity	(Information &	1 Bahraini	Nominal	1 Asian		
	eGovernment Authority-	2 GCC (Gulf Cooperation Council)		2 Arab		
	KoB, 2019)	3 Arab		3 Bahraini		
		4 Asian				
		5 African				
		6 European				
		7 Others				
9. Area of residence	(Mappr.co, 2014)	1 Muharraq Governorate	Nominal			
		2 Capital Governorate				
		3 Central Governorate				
		4 Southern Governorate				
		5 Northern Governorate				
Current family SES and s	tructure		-			
10. Family type		1 Single parent family (i.e., only one parent and the	Nominal			
		child/children live together)				
		2 Extended Family (i.e., the parents and the child/children				
		live with one or more of the grandparents, uncles, aunts				
		and cousins)				
		3 Nuclear family (i.e., the parents and the children live				
		together)				
11. Marital status		1 Widowed	Nominal			
		2 Divorced				
		3 Married				
12. Number of siblings	(Cinar et al., 2008a)		Numerical			

12.9.17 Lovel of	(Information 8	1 Not advanted	Ordinal	0- not advanted
IS & IT. Level Of		2 Primary asheel education (1.6 years)	Oruman	- not educated
education	Kop 2020b)	2 Finnary school education (1-6 years)		> 9 years of education
	KOB, 2020D)	5 Intermediate school education (7-9 years)		> 9 years of education
		4 Secondary school education (10-12 years)		9 years of compulsory schooling
		5 College/University		IN THE KOB
		6 Post-college		
15 & 19. Employment	(Bureau, 2015)	1 Unemployed	Ordinal	1 Unemployed/retired
status		2 Self-employed		2 Self/private employee
		3 Employed in the private sector		3 Government employee
		4 Military		
		5 Retired		
		6 Government employee (GE) on the salary scale for		
		public jobs		
		7 GE on the salary schedule for specialised jobs		
		8 GE on the salary scales for judges' jobs		
		9 GE on the salary schedule for executive jobs		
		10 GE on the salary scale for academic jobs		
		11 GE on the salary scale for diplomatic jobs		
21. Household Income	(Bureau, 2015)	1 BD<200	Ordinal	1 BD† ≤399
		2 BD 200-399		2 BD 400- 799
		3 BD 400-599		3 BD ≥800
		4 BD 600-799		
		5 BD 800-999		
		6 BD ≥ 1000		
22. Tenure		1 Rented flat	Nominal	1 Living with a parent
		2 Owned flat		2 Rented
		3 Rented house		3 Owned
		4 Owned house		o o miod
		5 Others (please specify)		
23 & 24 Crowding:	(Hatton and Martin 2010)	Crowded = 1 < 2 people per room		
number of people per	(nation and Martin, 2010)	Not crowded = $2 > 2$ people per room		
room				
OH practices and behavi	ours			
25 28 31 Frequency of	(Nicolau et al. 2003)	0 Never	Ordinal	1 Never or weekly
tooth brushing				2 Once or twice
26 20 22 Eroguopov of	4	2 Twice a day		3 More than twice
20, 29, 32. Frequency of		2 I wile a uay 2 More than twice a day		
dental flossing		S more than twice a day		

27, 30, 33. Frequency of		4 Weekly		
mouth wash rinsing				
34, 35, 36. Frequency of				
Sugary snacks				
43-44. Use of fluoride or	(Department of Health-UK,	1 500	Ordinal	
specify type of toothpaste	2021)	2 1000		
		3 1350-1450		
		4 ≥ 5000		
		5 Other		
45. Rinse or spit (for	(Department of Health-UK.	1 Rinse with water	Nominal	1 Rinse
child)	2021)	2 Rinse with mouth wash		2 Spite
,	,	3 Spit the toothpaste only without rinsing with water		
		5 I do not know		
16 51 The nettern of	(Nicelau et al., 2002)	0 De net visit	Nominal	0 Nover/ Emergeney
40-51. The pattern of	(Nicolau et al., 2003)		Nominal	1 Degular visite
dental visits				r Regular visits
		2 Check-up		
		3 Other		
37-42. Perceived OH and	(Rowan, 1994)	0 Poor	Ordinal	
health perception		1 Fair		
		2 Average		
		3 Good		
		4 Excellent		
Health and OH behaviour	ſS			
52,53. Parent's smoking		0 No / 1 Yes	Nominal	
54, 55. Type of smoking		1 Cigarette	Nominal	
		2 Pipe		
		3 E-cigarette		
		4 Other		
56. Child's health	(Karande & Kulkarni,	0 Normal	Nominal	0 No/ 1 Yes
condition	2005)	1 Asthma		
		2 Iron deficiency		
		3 Sickle cell diseases		
		5 Cerebral paley		
		6 Thelessomia		
		7 Type I dishatia Mallitua		
		7 Type Talabetic Mellitus		
		b Disabilities: hearing, mental, visual, speech		

		9 Autism		
		10 ADHD		
		11 Other		
57. Child's medication		0 No / 1 Yes	Nominal	
40, 41. Parents general	(Rowan, 1994)	0 Poor		
health perception		1 Fair		
		2 Average		
		3 Good		
		4 Excellent		
Parent's psychosocial fac	ctors			
99-162. Parenting Style	(Robinson et al., 1995)	1 Permissive	Nominal	
Questionnaire		2 Authoritarian		
		3 Authoritative		
63-170 Parent academic	(Ames, 1993)	Scale 0-28	Numerical	
involvement				
Early life variables		•		
Family structure and SES	at the time of childbirth (e	arly life)		
58. Family type		1 Single parent family	Nominal	
		2 Extended Family		
		3 Nuclear family		
59. Marital status		1 Widowed	Nominal	
		2 Divorced		
		3 Married		
60. Number of siblings	(Cinar et al., 2008a)		Numerical	
61. Birth order	(Nicolau et al., 2003)	1 First	Ordinal	1 First
		2 Second		2 Second or more
		3 Other		
62 & 65. Level of		1 Not educated		0= not educated
education		2 Primary school education (1-6 years)		≤ 9 years of education
		3 Intermediate school education (7-9 years)		> 9 years of education
		4 Secondary school education (10-12 years)		
		5 College/University		
		6 Post-college		

63 & 66. Employment status	(Bureau, 2015)	1 Unemployed 2 Self-employed 3 Employed in the private sector 4 Military 5 Retired	Ordinal	1 Unemployed/retired 2 Self/private employee 3 Government employee
		7 GE on the salary schedule for specialised jobs		
		9 GE on the salary schedule for executive jobs		
		10 GE on the salary scale for academic jobs 11 GE on the salary scale for diplomatic jobs		
		······································		
69. Household Income	(Bureau, 2015) Babraini dinars (BD1 =	1 BD<200 2 BD 200-399	Ordinal	1 BD† ≤399 2 BD 400- 799
	£2.03)	3 BD 400-599		3 BD ≥800
		4 BD 600-799		
		6 BD ≥ 1000		
70. Tenure		1 Rented flat	Nominal	1 Living with a parent
		2 Owned flat		2 Rented
		A Owned house		3 Owned
		5 Other (please specify)		
71. Number of rooms			Numerical	
Mother's demographic c	haracteristics and health du	ring pregnancy		
71. Age			Numerical	
72. Diabetes		0 No / 1 Yes	Nominal	
73. Hypertension		0 No / 1 Yes	Nominal	
74. Other conditions		0 No / 1 Yes	Nominal	
75. Smoking		0 No / 1 Yes	Nominal	
76. Delivery method		1 Caesarean section 2 Normal birth	Nominal	
77. Prenatal depression		0 No / 1 Yes	Nominal	
Child's birth outcomes a	nd feeding patterns from bi	rth-2 years old		•
78. Birth weight	(Kemfang Ngowa et al.,	1 Low birth weight (i.e., less than 2500 g)	Nominal	
	2014)	2 Normal (i.e., equal to or more than 2500 g)		
79. Gestational age	(Kemfang Ngowa et al.,	1 Pre-term birth (i.e., less than 37 completed weeks of	Nominal	
	2014)	gestation)		

		2 Normal (i.e., equal to or more than 37 completed			
		weeks of gestation)			
81-82 Feeding	(Nicolau et al., 2003)	1 Bottle-feeding only			
5		2 Breast and bottle			
		3 Exclusive breastfeeding			
83. Sugar night feeding		1 Yes / 0 No Nominal			
84. Use of pacifier	(Declerck et al., 2008)	1 Yes / 0 No Nominal			
85. Time of using a		1 From birth	Ordinal		
pacifier		2 At 6 months			
		3 After 6 months			
		4 I cannot remember			
86. Duration of using a	(Poyak, 2006)	1 Less than two years	Ordinal		
pacifier		2 Equal or more than two years			
		3 I cannot remember			
		4 Few months			
Child's OH behaviours at	3-6 years old				
87. Start brushing	(Department of Health-UK,	0 Not brushing	Ordinal		
	2021)	1 At more than one year			
		2 At equal to or less than one year			
88. Supervised	(Department of Health-UK,	0 Never	Ordinal	0 No / 1 Yes	
toothbrushing	2021)	1 Occasionally			
		2 Mostly			
		3 Always			
90. Dental pain		0 No / 1 Yes	Nominal		
91. Dental caries in		0 No / 1 Yes	Nominal		
primary teeth					
92. Dental treatment		0 No / 1 Yes	Nominal		
Child's early learning					
95. Final general		1 Failed, 2 Acceptable, 3 Average, 4 Good, 5 Very good,	Ordinal		
academic performance		6 Excellent			
Grade 1					
96. School absence	(MOE-KoB, 2017)	0 No of absences	Ordinal		
Grade 1		1 Days absent from 1-5 days			
		2 Days absent from 6-9 days			
		3 Days absent ≥10			

97. Absence because of		0 No /1 Yes	Nominal			
OH Childle face face interview						
Child's face-face intervie						
Social support	(Torsheim et al., 2000) (Ansong et al.,2017)	Scale 0-12	Numerical			
POQL	(Huntington et al., 2011)	Percentage Number of events	Scale			
General H/OH	(Rowan, 1994)	0 Poor 1 Fair 2 Average 3 Good 4 Excellent	Categoric al Ordinal			
Clinical examination	•					
Lip competent	(Zaleckien et al., 2020)	0 No/ 1 Yes	Nominal	0 No / 1 Yes		
Over jet (positive)	(UK-CDHS, 2013)	Positive overjet	Nominal	≤ 6mm / 0 N0, > 6mm / 1 Yes		
Overbite		0 No / 1 Yes	Nominal	0 No / 1 Yes		
Anterior open bite		0 No / 1 Yes	Nominal	0 No / 1 Yes		
Anterior crossbite		0 No / 1 Yes	Nominal	0 No / 1 Yes		
Dental trauma	(UK-CDHS, 2013) (O'Brien, 1993)	Table 14	Nominal	0 No / 1 Yes		
Dental plaque	(UK-CDHS, 2013)	Average visible plaque condition of each sextant	Nominal	Total areas with plaque/6 total segments		
Enamel developmental defects	FDI modified DDE (Clarkson and O'Mullane, 1989)	Table 15	Nominal	0 No / 1 Yes		
Dental caries	ICDAS (Ismail et al., 2007)	Table 17	Ordinal	Total count of enamel caries, ICDAS caries codes 2-3 (d ₂₋₃) Total count of dentine caries, ICDAS caries codes 4-6 (d ₄₋₆)		
Treated teeth	ICDAS (Ismail et al., 2007)	Table 16	Nominal	Total count of: Extracted teeth because of dental caries = code 97		

				Filled teeth because of caries= ICDAS restoration codes= 3-8	
PUFA	PUFA index (Monse et al.,	Table 18	Nominal	0 No / 1 Yes	
Child BMI (CDC)	(CDC, 2019) CDC (2000)	Height in cm Weight In Kg	Numerical	1 Underweight =< 5th percentile 2 = Normal 5th - <85th percentile 3 = Overweight 85th - <95th percentile 4 = Obese ≥ 95th percentile	
Outcomes					
School performance		By mobile phone: sending an image of a child's grades	Numeric	Out of 100	
School attendance		School records for days absent from September – February 2020	Numeric		

Note: How it was analysed column when left empty means it has been analysed in the same way it was measured.

3.6 Design of study instruments

3.6.1 Questionnaire design and translation

Initial versions of the questionnaires were developed in the English language. To facilitate a cross-cultural and conceptual translation, the forward-translation and back-translation procedure recommended by the WHO (Guillemin et al., 1993) was used to translate the questionnaires, participant information sheets and informed consent forms.

3.6.1.1 Forward-translation

Two bilingual professionals independently performed the first translation from English to Arabic (the primary researcher; a periodontist, YAL) with Microsoft Word Translator 365 (6 April 2019) to emphasise concept equivalence rather than word-to-word translation. The Arabic versions were then compared and revised by both translators.

3.6.1.2 Expert panel

The process involved three professionals: bilinguals in Arabic and English (the primary researcher; a periodontist, YAL and a paediatric dentist, SHK). The panel at this step reviewed and resolved inadequate expressions/concepts between both versions and combined them into one version.

3.6.1.3 Backward-translation

The first Arabic versions were translated back into the English language by two professionals independently. The resulting two back-translated versions were then compared to the original text in the English language, with corrections made

accordingly. Subsequently, the corrected versions were forward translated into the Arabic language.

3.6.1.4 Evaluation

The resulting Arabic documents were evaluated by Mrs MA, a native Arabic speaker (an education specialist in the Scientific Research Directorate at Bahrain's MOE) (15 April 2019). Moreover, the suggested changes were considered and applied. This resulted in the generation of a set of research instruments in Arabic of sufficient quality for pre-testing.

3.6.1.5 Pre-testing

The questionnaires were tested for understanding with twenty parents of different age groups and SES and twenty schoolchildren in August 2020, after which the questionnaires were adjusted accordingly. Slight changes were made to the wording and numbering of items. All changes were documented and applied to derive the final Arabic version.

3.6.1.6 Reliability

Internal consistency measured by Cronbach's alpha (α) and reproducibility measured by interclass correlation coefficients (ICC) were tested for the Arabic questionnaires and discussed in the results section.

3.7 Ethical considerations

3.7.1 Permissions and liaison

Ethical approval to conduct this study was obtained from the University of Sheffield (see Appendix 14), the MOE in Bahrain (see Appendix 15) and the Bahraini Ministry of Health (see Appendix 16).

Permission from Bahrain's MOE was required and granted to facilitate access to the study population, contact parents, collect consent forms from them, and collect clinical data. Permission from Bahrain's Ministry of Health was required and again granted to retrieve each child's birth weight and gestational age from the health records. However, this data was not used as only 76 records were obtained. Finally, informed consent from the children's parents/guardians was obtained before data collection.

Furthermore, it should be noted that initially, it was planned to collect school's performance and attendance data via parent's questionnaires. However, because of the COVID-19 pandemic, the schools in the KoB operated online only. So, communication with parents was only possible by phone. Hence, the researcher had to change the data collection procedures regarding the study outcomes. An ethical amendment for the University of Sheffield and MOE in Bahrain was applied and retrieved (Appendix 17, 18) to contact parents directly for their children's grades and retrieve days absent from school records.

3.7.2 Ethical principles

Ethical principles including honesty, integrity, anonymity, confidentiality, autonomy and non-maleficence were maintained and enforced during the research process. Procedures such as recruitment, gaining participants' consent, data transfer and

storage, in addition to ensuring child safety, were conducted according to strict ethical guidelines.

3.7.2.1 Autonomy and informed consent

Scientific honesty was maintained in acquiring parents informed consent forms. The consent form provided the parents with the following information: the purpose of the study, its benefits and risks, study duration and the researcher's contact number to call should they require any clarification. They were assured of anonymity and confidentiality. Their autonomy was maintained throughout the process by their being given the choice of whether or not to participate and the option to withdraw from the study at any time. Prepaid envelopes were provided with which to return the IC forms. The children were also given the option to withdraw from the study at any time. It is important to mention that the researcher visited the schoolchildren in their classes to explain the research purpose and encourage them to participate.

3.7.2.2 Anonymity

Participants' personal data was pseudonymised for the principal researcher and fully anonymised for the remainder of the research team. The personal data for all participants was collected in sealed envelopes and then transferred by the primary researcher onto an encrypted Excel spreadsheet.

Each participant received a unique study identification number composed of two letters and four numbers. These identification numbers were used on the clinical forms questionnaires and to save parents contact numbers to hide the participant's identity. These were generated using Microsoft Excel 365. The following formula creates a list of random numbers:

(CHAR(RANDBETWEEN(65,90))&CHAR(RANDBETWEEN(65,90))&RANDBETWE EN(1000,9999).

3.7.2.3 Confidentiality

Confidentiality was maintained by storing the collected data without any participant identity information. Personal data was stored in the primary researcher's laptop, protected and encrypted with the assistance of the University of Sheffield. This data was also stored as a backup on the primary researcher's personal space on the University of Sheffield drive. Only the researcher had access to this storage. Moreover, following the 3-2-1-rule for backup, all the anonymised data was shared with the primary researcher's supervisors via Google Drive at the University of Sheffield.

The primary researcher was the only person who had access to the master Excel spreadsheet that matched each participant's personal data to their study identification number. Furthermore, this spreadsheet was stored separately from the self-reported questionnaires containing the parents' telephone numbers, which were kept in a locked cabinet in the primary researcher's office at the University of Bahrain (UOB), to which only the researcher had access. Every Excel spreadsheet used in this study was protected using a password known only to the primary researcher.

3.7.2.4 Non-maleficence

Non-maleficence was maintained so that no harm or risk was expected in collecting the data. Oral examinations were non-invasive and straightforward, performed using only a mirror and, if needed, a dull probe. Examiners underwent training before performing the examinations.

3.8 Conduct

3.8.1 Training

3.8.1.1 Initial ICDAS training and reproducibility assessment

ICDAS training was conducted in June 2019 with a reference examiner (CD), who led a two-day session. Two examiners were trained: the principal researcher (SM) and a periodontist (YA).

On the first day (10 June 2019), the examiners undertook a 90-minute ICDAS elearning programme (<u>https://www.iccms-web.com/content/resources/elearning</u>). **On the second day** (11 June 2019), they participated in four hours of preclinical workshops and discussions with the reference examiner (CD).

For training, reproducibility of the application of the ICDAS codes was first performed using 40 photographs of extracted teeth and 40 extracted teeth. The reference examiner undertook the gold standard assessment ahead of the training. Inter- and intra-reliability tests were performed. An unweighted Kappa coefficient was applied to estimate reliability for the first ICDAS code because the data was nominal (Cohen, 1960) and a weighted Kappa coefficient for the second code because the data was ordinal (Cohen, 1968). There was a substantial agreement of 0.65 and above weighted kappa targeted (Landis and Koch, 1977). The results revealed substantial agreement, as presented below.

	Unweighted Cohen's Kappa value		
	Gold standard	Examiner 1	Examiner 2
Examiner 1	0.85	0.70	0.84
Examiner 2	0.93	0.84	0.92

Table 21. Unweighted Cohen's Kappa for first ICDAS code
Table 22. Weighted Cohen's Kappa for second ICDAS code

	Weighted Cohen's Kappa value				
	Gold standard	Examiner 1	Examiner 2		
Examiner 1	0.92	0.85	0.92		
Examiner 2	0.94	0.92	0.94		

3.8.1.2 Additional training

Prior to conducting the main study, further clinical training and calibration were conducted within the pilot study (September 2019) (see Section 3.8.5, Page 163). The training included revising ICDAS and dental indices criteria and codes with examiners and recorders, followed by calibration.

A senior lecturer in the nursing division (M) trained a nursing student and a research team on measuring the weight and height of children with respect to BMI measures. The scale was calibrated by placing a known standard mass (metal ball= 4 KG) on the scale, where the reading was easily verified.

Two interviewers were also trained by the principal researcher beforehand on administrating and conducting interviews with the children in a friendly manner.

3.8.2 Equipment

The OH examination required minimal equipment such as a dental plain mirror (No.4), a Community Periodontal Index (CPI) probe (disposable and non-disposable), orthodontic ruler, headlights (AUNOL, 400 lumens), cotton rolls, gauze, clinical sheets, toothbrushes, a pen with recording sheets and the personal protective equipment necessary to maintain infection control and reduce cross-contamination, for instance, gloves, masks, waste disposal units and containers for sterilising instrument. All non-

disposable instruments were sterilised at the dental hygiene clinic if required. However, most of the examinations were covered by the use of non-disposable instruments.

In relation to BMI, a high-quality balance beam was required for measuring weight and height (Salter Bracknell HS-200M Physician: Mechanical Standing Height and Weight Scale).

Two desktops, six student desks and six chairs were used to facilitate the examination.

3.8.3 Procedures

The oral health examination, BMI measurements and completion of children's face-toface interviews were conducted on the same day, in that order, in the library or the sports hall of the schools (September-November 2019). Separate stations were provided for each of the three procedures. The collection of the parent's questionnaire occurred from September to December 2019.

3.8.3.1 Dental clinical data collection

Two desktops were arranged end-to-end so that each child could lie on them in a supine position. The examiner was seated behind the child's head and wore a headlight (chargeable). Two student desks, placed to one side, held the OH instruments.

The dental examinations were performed to collect oral health measures in the following order: malocclusion, dental trauma; dental plaque; EDD; ICDAS and PUFA indices. First, the child was examined to identify lip competence and malocclusion conditions in the resting position with the support of the orthodontic ruler with teeth in centric occlusion. Then the incisors were inspected for dental trauma. Subsequently,

the oral cavity was screened visually both buccally and lingually by segment for the dental plaque with the assistance of a dental plain mirror only. The examiners used gauze or a soft disposable toothbrush to wipe plaque off gently before registering EDD, ICDAS and PUFA indices. The teeth were dried with cotton rolls and gauze and examined in the following order: upper right; upper left; lower right; lower left (ICCMS 2018: <u>https://www.iccms-web.com/course/8</u>).

Concerning EDD, any lesions less than 1 mm in diameter were not recorded. Labial of anterior and buccal posteriors was inspected because these are the most common places to discover EDD (Ramesh et al., 2011). Relating to the ICDAS, teeth surfaces were visually inspected from distal, occlusal/incisal, mesial, facial and lingual surfaces to evaluate restoration and dental caries severity following the ICDAS criteria (Ismail et al., 2007). The WHO probe with ball ended equipment was used to check surfaces for discontinuity and remove dental plaque. PUFA examinations involved visual detection only; no instruments were involved. The lips and cheeks were gently retracted to visualise soft tissues (Monse et al., 2010).

3.8.3.2 BMI collection

The scales employed measured both weight and height simultaneously. The student nurse measured height and weight according to CDC (2000) guidelines as follows:

- The children were instructed to remove their shoes and head coverings to ensure accurate measurements.
- Each child stepped onto the scale platform, facing away from the scale. The weight value was recorded to the nearest 0.1 kg.

 For the height measurement, the child remained standing on the platform with their back against the stadiometer rule, with their legs straight, arms at their sides, and shoulders relaxed. The examiner lowered the headpiece until it touched the crown of the child's head firmly. The reading was recorded to the nearest 0.1 cm.

3.8.3.3 Questionnaires

Face-face interviews were conducted in groups of three children in the library or the sports hall of the schools after the dental exam. Parents' questionnaires were sent in September 2019 through their children and collected in a sealed envelope in the same way.

3.8.3.4 School performance and attendance

In June 2020, Grade 2 grades were collected online from parents via phone. The principal researcher contacted all eligible parents by phone through web WhatsApp, with a password-protected computer, and enquired to send a photograph of their child's 2020 Grade 2 school report.

The photos of the grades were collected without any personal identity. The parent's contact numbers were saved in Google drive and identified by the participant's identification number. The research used a new mobile number and linked it to WhatsApp online. Thus, data were not transferred to a mobile device but directly transferred into the password-protected laptop, without any personal details.

Regarding (2019-2020) attendance, the data were obtained from school records in October 2020. Days of absence during the school year from September 2019 to February 2020 were collected.

3.8.4 Personnel

In total, ten individuals participated in the data collection of this study. A nursing faculty was responsible for BMI training; the leading researcher and a periodontist conducted the dental clinical examinations. A nursing student took the anthropometric measures, two dental hygienists acted as recorders, whilst another two dental hygienists conducted the face-to-face interviews. Finally, two more individuals participated as research organisers.

3.8.5 Pilot study

In September 2019, the clinical measures were tested with 18 schoolchildren. The pilot study had several purposes. First, it enabled the examiners to familiarise themselves with the ICDAS and other clinical measures employed, clinical calibration and demonstrated how long it would take to conduct each examination and interview. Additionally, it raised any issues regarding the feasibility and coherence of the clinical examinations and interviews.

The results of the calibration are presented in the Result's Chapter. The pilot study revealed the difficulty associated with the online recording of clinical examinations. Therefore, using the manual exam form was a more time-saving option. Furthermore, the time for conducting individual interviews was very time-consuming. Hence, the decision was made for each interviewer to interview a group of three children at a time. Finally, the pilot study indicated high enamel and dentine caries, with ICDAS prevalence equal to 83.3% among the 18 schoolchildren, with 11.1% having clinical consequences of dental caries. On the other hand, the dmft mean was 9.83 (95%Cl= 5.04/14.63).

3.8.6 Data transfer

The chief researcher transferred the data from the research instruments to the SPSS data file (November 2019-March 2020 & June-November 2020). The study was password-protected. 10% of the data were randomly selected and checked for accuracy.

3.9 Data analysis strategies

The analysis included three stages: descriptive, data screening, and a two-step structural equation modelling (SEM) procedure; 1) confirmatory factor analysis (CFA), and 2) the structural model (SM).

3.9.1 Descriptive

Descriptive analysis was used to answer Objective 1 (P. 4). All descriptive analysis was performed using Statistical Package for Social Sciences (SPSS) versions 25 and 26. The distribution of the variables and reported frequencies, percentages, mean values, standard deviations, and prevalence measures have been calculated. The alpha (Type I error) was set at 5% for all analyses.

Results of the clinical calibration and the reliability questionnaire were also reported. The clinical calibration, which included the inter-and intra-reliability tests, were assessed and targeted 0.65 and above kappa coefficient (substantial agreement) (Landis and Koch, 1977). The reliability of the questionnaires was assessed using the internal consistency test Cronbach's alpha and the test-retest reliability with Pearson's intraclass correlation (ICC). Cronbach's alpha estimate value above 0.70 was considered acceptable (Tavakol and Dennick, 2011), and Pearson with \geq 0.40 was considered a moderate association (Dancey and Reidy, 2007).

3.9.2 Data screening

Before starting the structural equation modelling procedure, data were screened using SPSS (25 and 26) and the software package Analysis of Moment Structures (AMOS), versions (25 and 26) for missing values, outliers, and normality.

To avoid the problem of missing data, when conducting the study, the researcher checked any returned questionnaires for missing data before receiving the completed questionnaires. If any questionnaires had missing information, the children were asked to return them for the parent to complete. Children who failed to return the parent's questionnaire were excluded from the analysis. Data were screened to evaluate the total amount and the pattern of missing data to find the appropriate techniques to deal with missing data (Byrne, 2001). The randomness of the missing data was tested using Little's (1988) Chi-square statistics.

Further, the data were screened for univariate and multivariate outliers. Hair et al. (2014) described outliers as cases with substantially diverse scores from the rest of the observations in a dataset. Kline (2016) defines a univariate outlier "as an extreme value on one variable, whereas a multivariate outlier is a combination of extreme values on two or more variables". It has been suggested that the outliers should be deleted because they can affect model fit estimates and parameter estimates and cause a negative variance (Dillon et al., 1987). However, Hair et al. (2014) advised that the outliers should be retained to ensure the generalisability of the study unless they are genuinely aberrant.

Thus, although the univariate outliers were detected in this study by determining the frequency distributions of the observed data's Z scores, it was considered extreme when it was below or above the mean by (-/+ 3.29 SD) (Kline, 2018). While, the multivariate outlier was examined using the Mahalanobis distance (D²) (Byrne, 2001). Mahalanobis distance (D²) "measures the distance in standard deviation units between each observation compared with the mean of all observations" (Byrne, 2001). In this study, the researcher measured D² using Amos version 25. Observations with a D² measurement with a p-value < 0.001 were considered to be multivariate outliers (Kline, 2016). Despite measuring them, outliers were retained and not deleted.

Normality was assessed by skewness and kurtosis (flat/peaked) tests. According to Hair et al. (2014), data with skewness between +/- 2 and kurtosis between +/ -7 was deemed not to have a severe skew or kurtosis and not severely non-normal. Even when normality was not assumed, the bootstrapping procedure within SEM analysis can handle non-normal data (Byrne, 2001).

3.9.3 Structural equation modelling (SEM)

Stage 3 in the analysis plan included the application of SEM to answer Objective 2-4 (P.4). SEM is a multivariate statistical approach that combines both the measurement and structural models. The approach allows concurrently testing the associations among multiple independent and dependent constructs (Hair et al., 2014). Thus, SEM was applied in this study by first testing the measurement model and then the structural model. The procedure was performed for both steps with maximum likelihood (ML) and 900 bootstrapping estimations using AMOS (25 and 26). The 900 bootstrap samples were resampled to produce less biased standard errors, to report the 95%

Bias-corrected (BC) confidence interval (CI) (Mooney, 1993), to deal with multivariate non-normal data (Byrne, 2001), and to assess mediation effect (Hayes, 2009).

3.9.3.1 Confirmatory factor analysis (CFA)

The first step in building the SEM model was testing the measurement model through CFA. CFA is a process of testing how the measured variables represent a smaller number of constructs. CFA was used to confirm the a priori assumed association between a set of indicator factors (measurement items) and the latent factors (Byrne, 2001). Therefore, CFA was performed in this study by first identifying the model constructs based on the theoretical model (Section 2.6.2) and evaluating the proposed model. Generally, Hair et al. (2014) recommended two methods to evaluate the validity of the measurement model: (1) evaluating the goodness of fit (GOF) criteria indices with model estimates and modification indices, (2) and finding specific evidence of construct validity (loading estimates). In this study, the measurement model was evaluated by CFA by means of three types of indices, namely absolute fit indices, increment fit indices and parsimonious fit indices and the magnitude and significance of loading factors on the correspondence between the indicators and latent variables (Hair et al., 2014). The final step also was completed by examining the modification indices (MI) from the Amos output. The MI examined every possible association not assumed in the model (Byrne 2001). Hair et al. (2014) suggested that if the MI is 4.0 or higher, the fit could be improved by adding additional links or eliminating those items. However, they also recommended that the decision be consistent with the theory, not solely on MI values.

The absolute fit indices are used to evaluate the adequacy of the overall model fit. These indices include the Chi-square test statistic/degree of freedom (χ 2/df) ratio, the root-mean-squared error of approximation (RMSEA), standardised root mean residual (SRMR) and the goodness of fit (GFI). The increment fit indices compare the assumed model with a baseline model and include the comparative fit index (CFI). The parsimonious fit indices are utilised to evaluate which model among a set of competing models is best. This index includes the adjusted goodness-of-fit index (AGFI). A non-significant chi-square indicates that the model is an acceptable representation of the association among the observed variables. However, significant results are expected in studies with a sample size > 250 (Hair et al., 2014). Thus, in this study, Chi-squared/degree of freedom ratio <3 (X²/df) (Hu and Bentler, 1999), SRMR ≤ 0.08 and RMSEA values < 0.06, GFI, CFI and AGFI ≥ 0.90 were used to indicate an acceptable model fit (Byrne, 2001, Vehkalahti, 2011, Hair et al., 2014).

In addition to the GOF criteria, other standardised estimates can also be applied to evaluate the measurement model. Hair et al. (2014) suggested that all factor loadings (standard regression weight) should be statistically significant ($P \le 0.01$) and that their values should be 0.5 or higher or ideally be greater than 0.7 or if less than the variable could be specified for deletion (Hair et al., 2014). In contrast, Bagozzi (1991) proposed a factor loading with 0.3 as an acceptable value. However, lower loading values are expected with a complex theoretical model and sample size exceeding 400-500 (Hair et al., 2014). Thus, in this study, additional consideration was given to the significance of factor loadings, considering the complexity of the theoretical model and the sample size of 466 participants.

3.9.3.2 Structural model (SM)

Following the CFA, the full SM was developed by adding observed variables to examine the hypothesised associations between all variables. The full model fit was assessed using the indices mentioned above in CFA.

A parsimonious model was created by eliminating non-significant paths (p>0.05) from the full SEM's model. The difference in Chi-square and degrees of freedom between full and parsimonious models were calculated to evaluate both models' significant difference (Werner and Schermelleh-Engel, 2010). The parsimonious model was accepted if the difference was not statistically significant (P>0.05). This calculation was performed in an online programme https://www.graphpad.com/guickcalcs/PValue1.cfm.

Then, the direct (the direct path from one variable to the next) and indirect (other variables mediated the link between two variables) paths linking latent and observed variables were estimated (Gunzler et al., 2013). The BC bootstrap 95% CI was used to assess mediation by examining the statistical significance of indirect paths (Hayes, 2009). The estimated indirect paths usually reflect the total indirect effect for all possible mediators on one variable without distinguishing the effect of specific mediators. AMOS, however, can identify the specific mediators with their exact significant indirect paths by using the Plugin and User Defined Estimands functions.

Thus, the online forum <u>http://statwiki.gaskination.com/index.php?title=Plugins</u> was used to guide the installation of Plugin and Estimands function and the estimation of the specific significant indirect paths and mediator with AMOS in this study.

3.9.3.3 Proposed full model for Structural Equation Modelling

The modified Heilmann et al. (2015) (Section 2.6.2) life course framework and literature review Chapter 2 was used to develop the proposed full structural model.

The study sample size restricted the number of included variables in the hypothesised model. The sample size is crucial for SEM, Hair et al. (2014) suggested that if the sample size is 300, then the recommended number of constructs would be seven or fewer with three indicators, and the expected loading factor values would be ≤ 0.45 . They further suggested that if the sample size exceeds 400-500 participants, the number of constructs can be larger than six with three-four indicators each. However, the SEM analysis becomes too sensitive with low factor loadings, and the model will be a challenging fit.

Following that recommendation and based on the proposed theoretical framework, the measurement model was performed with six latent variables, each with 3-4 indicators. Then, twelve observed variables were added to the measurement model to create the full structural model (Figure 9). To confirm that the sample size is sufficient for the proposed SEM model, the sample size was calculated using an online website (<u>https://www.danielsoper.com/statcalc/calculator.aspx?id=89</u>). The result indicated that the required minimum sample size was 236 to detect a minimum effect size of 0.3, the statistical power of 0.95 and the probability level of 0.05, and a sample size of 400 would be required for the model identification.

3.9.3.3.1 Summary of the included variables

Latent variables: Early life

1. Family's SES at the time of the child's birth (BSES) with four indicators: mother's level of education, household income, mother's employment, and father's level of education.

2. Child's birth outcomes with three indicators: birth weight, gestational age, and birth order.

Latent variables: Current life

3. Family's current SES (CSES) with three indicators: mother's level of education, household income, and father's level of education.

4. Mother's oral health practices (OHPs), with three indicators: mother's toothbrushing, mother's flossing, and mother's use of mouthwash.

5. Child's current OHPs with four indicators: child's toothbrushing, child's flossing, child's sugar consumption, and child's use of mouthwash.

6. Parent's characteristics with three indicators: mother's parenting style, father's parenting style, and mother's academic involvement.

Observed variables: Early life

1. Maternal chronic health conditions during pregnancy (CHCs)

2. Exclusive breastfeeding after birth

3. Supervised tooth brushing (TB)

4. Enamel developmental defects in primary teeth (EDD)

Observed variables: Current life

5. Dental plaque

6. Enamel (ds_2_3)

7. Dentine caries (ds_4_6)

8. pufa

9. Filled or extracted primary teeth because of caries (Treated)

10. Current dental pain

- 11. Absent days for Grade 2
- 12. Academic School performance for Grade 2



Figure 9. The proposed full model for the Structural Equation Modelling analysis

3.9.3.3.2 Research hypotheses

The proposed full SEM model aimed to examine the association between OHCs and school outcomes (attendance, performance) using the modified Heilmann et al. (2015) life course framework. This was achieved by; testing the critical and accumulation models to analyse the social, biological, behavioural, and parental pathways that linked early and current life exposures with children's current OHCs; examining the association and the possible mediators that linked OHCs with children's school outcomes. Thus, the following sections explain first the cut-off points of classifying the constructs as high/low, inadequate/inadequate and supportive/unsupportive (Table 23), then the detailed hypotheses testing procedure.

Table 23. Cut-off points for classification of constructs variables

Variables		
1. SES	High SES if	Low SES if
A. Mother's employment status	A. Mother's employment status	A. Mother's employment status
Unemployed/retired	Government employee	 Unemployed/retired
Self/private employee		Self/private employee
Government employee		
B. Mother's level of education at child's birth	B. Mother's level of education at child's birth	B. Mother's level of education at child's birth
Not educated	 >9 years 	Not educated
• ≤9 years		• ≤9 years
 >9 years 		
C. Father's level of education at child's birth	C. Father's level of education at child's birth	C. Father's level of education at child's birth
Not educated	 > 9 years 	 ≤9 years, or not educated
• ≤9 years		
 >9 years 		
D. Family income at child's birth	D. Family income at child's birth	D. Family income at child's birth
 BD[†] ≤399 	• BD ≥800	• BD <800
• BD 400- 799		
• BD ≥800		
2. Birth outcomes	Desirable birth outcomes	Undesirable birth outcomes
A. Child's birth weight	A. Child's birth weight	A. Child's birth weight
 Low birth weight (<2500 g) 	 Normal (≥2500 g) 	 Low birth weight (<2500 g)
 Normal (≥2500 g) 		
B. Child's gestational age	B. Child's gestational age	B. Child's gestational age
 Preterm birth (<37 weeks) 	 Normal (≥37 weeks) 	 Preterm birth (<37 weeks)
 Normal (≥37 weeks) 		
C. Child's birth order	C. Child's birth order	C. Child's birth order
First	First	Second or more
Second or more		

3. Mother's and child's OH practices	Adequate OH practices	Inadequate OH practices
A. Toothbrushing	A. Toothbrushing	A. Toothbrushing
Never or weekly	 > 2 times per a day 	 ≤ 2 times per day/week or never
Once or twice		
More than twice		
B. Mouth wash (MW)	B. Mouth wash (MW)	B. Mouth wash (MW)
Never or weekly	 > 2 times per a day 	 ≤ 2 times per day/ week or never
Once or twice		
More than twice		
C. Flossing (CMF)	C. Flossing	C. Flossing
Never or weekly	 > 2 times per a day 	 ≤ 2 times per day/ week or never
Once or twice		
More than twice		
D. Sugar consumption	D. Sugar consumption	D. Sugar consumption
Never or weekly	 ≤ 2 times per day/week or never 	 > 2 times per a day
Once or twice		
More than twice		
4. Parent's characteristics	Supportive parent's characteristics	Unsupportive e Parent's characteristics
A. Mother's and father's parenting style	A. Parenting style	A. Parenting style (MPS)
Permissive	Authoritative	Permissive
Authoritarian		Authoritarian
Authoritative		
B. Mother academic involvement score	High score of academic involvement	The low score of academic involvement

3.9.3.3.2.1 Early and current life factors linked to OHCs

3.9.3.3.2.1.1 Critical period model

The model assumed that the exposures that happened at the time of the child's birth would associate with the development of OHCs in the current stage of life of Grade 2 children (Heilmann et al., 2015). The accumulation of these factors might influence children's OHCs either independently or cluster by SES (Ben-Shlomo and Kuh, 2002). Therefore, the following hypotheses were tested:

- Children born into a family with high SES would have better OH at the age of 7-8 years (Abreu et al., 2015)(Figure 10a);
- Children born with undesirable birth outcomes were more likely to have OHCs at 7-8 years old (Nicolau et al., 2003) (Figure 10a);
- Children with EDD in primary teeth were more likely to have dental caries (Zhou et al., 2011, Masumo et al., 2013) (Figure 10a);
- Low SES is linked with undesirable birth outcomes and EDD development as a cluster of risks; hence, those factors would associate with worse children's OHCs at 7-8 years of age (Figure 10b).



Figure 10. Life course model (Ben-Shlomo and Kuh, 2002), critical period model; a) independent risks; b) clustered risks

3.9.3.3.2.1.2 Accumulation model with the chain of risks

Chain of risk with additive effect model

Low BSES might influence maternal health during pregnancy, resulting in hypertension and diabetes, which increases the risk of low birth weight and preterm birth (Kramer, 2000). These undesirable pregnancy outcomes may subsequently associate with EDD (Jacobsen et al., 2014). Consequently, EDD would increase dental plaque accumulation, and finally, the dental plaque would increase the risk of developing dental caries (Zhou et al., 2011);

Thus, in this model, each of the factors mentioned above would increase the subsequent exposure and would independently increase the risk of poor OHCs irrespective of later exposure (Ben-Shlomo and Kuh, 2002) (Figure 11a);

Chain of risk with trigger effect model

The model assumed that low BSES have no direct effect on the child's OHCs at 7-8 years of age. Instead, low BSES would trigger the risk of having maternal CHCs during pregnancy. Thus, maternal CHCs can influence a child's birth outcomes, which would trigger the development of EDD. As a result, EDD increases dental plaque accumulation, hence the risk of dental caries. In this model, each risk led to another; only the final link in the chain can influence the development of such chronic conditions (Ben-Shlomo and Kuh, 2002) (Figure 11b);

Risk clustering

Low BSES could lead to a cluster of risk behaviours such as no exclusive breastfeeding (0-2 years old) (Meedya et al., 2010) and unsupervised tooth brushing (3-6 years old) (Declerck et al., 2008) and these behaviours together or independently would increase the risk of worse OHCs in 7-8-year-olds (Figure 11c).



Figure 11. The accumulation of risks models

3.9.3.3.2.1.3 Association between BSES and CSES

The model assumed that BSES predicted the current family's SES. Thus, low current SES might shape a cluster of parental factors associated with the children's oral health practices (OHPs) and OHS (Figure 12). Hence the following hypotheses were tested:

1. A family's low BSES associated with low CSES (Thomson et al., 2004);

- Low CSES associated with inadequate OHPs among mothers (Hooley et al., 2012). At the same time, a mother's OHPs predicted their children's OHPs (Hooley et al., 2012). Therefore, either children's OHPs associated with their OHCs directly or dental plaque would mediate the association between their OHPs and OHCs (Shenkin et al., 2004);
- 3. Low CSES associated with unsupportive parental characteristics like practising a non-authoritative parenting style (Ballantine, 2001, Howenstein et al., 2015) and low mother academic involvement (Andersen et al., 2021). Consequently, parental characteristics are associated with children's OHPs (Howenstein et al., 2015, da Silva et al., 2018). Therefore, either children's OHPs linked with their OHCs directly or dental plaque mediated the association between their OHPs and OHCs.



Figure 12. The relation between BSES and CSES and children's current OH

3.9.3.3.2.2 Association between OHCs and school performance and attendance

Finally, the model assumed adjacent associations between the child's clinical OH and school performance (Osman and Alsayed, 2015) and school attendance (Butani et al., 2009). The association could be direct or mediated by other factors. Thus, the following hypotheses were tested:

- OHCs were directly associated with low school performance and poor school attendance;
- 2. The association of OHCs with low school performance and poor school attendance were mediated by dental pain (Piovesan et al., 2012);
- The association of OHCs with low school performance and poor school attendance were mediated by the consequence of dental caries (Quadros et al., 2021);
- OHCs mediated the association of low SES with low school performance and school attendance (Seirawan et al., 2012);
- 5. The association between OHCs and low school performance was mediated by school attendance (Fornander and Kearney, 2019);
- 6. Biological, social, behavioural, and parental factors in children's early and current life course were also related to their school performance and attendance.

Chapter 4. Results

4.1 Introduction

This chapter presents the results of this study. The results are presented in five main sections.

Section 4.2 shows the final sample size and response rate.

Section 4.3 gives a descriptive breakdown of the collected data in relation to the variables collected at baseline (September 2019).

Section 4.4 summarizes the results of the questionnaires reliability and the clinical calibration.

Section 4.5 screens data for missing, outlier and normality.

Section 4.6 tests the comprehensive interrelationships among the variables using SEM.

4.2 Response rate

Five hundred and ten children and their parents were invited, but 33 refused to participate. A total of 477 children (292 girls and 185 boys) and their parents enrolled on the study in September 2019, when baseline data was collected. However, 11 (2%) of the parents failed to return the completed questionnaires, resulting in their children (five girls and six boys) being excluded from the final sample. Otherwise, data collection was carried out successfully, and the response rate was 98%.

4.3 Descriptive statistics

4.3.1 Socio-demographics

Of the 466 schoolchildren, 287 (61.6%) were females, and 179 (38.4%) were males. Nearly three-quarters (72.5%) of them were Bahraini, while 21% were from Arab countries such as Egypt and 1.5% from other Gulf Cooperation Council Countries (GCC). The remaining 5.2% were from Asia (Pakistani). Almost all children (98.7%) lived within the Muharraq Governorate, only 6 (1.3%) living elsewhere. The mothers' ages ranged from 23-60 years old, with a mean age of 35.8 years (SD = 6.0). The fathers were marginally older, ranging from 24-73 years old, with a mean of 41 years (SD = 7.8) (Table 24).

Independent v	ariable		%		Frequency		
Gender							
Male			38.4	1	79		
Female			61.6	2	87		
Total			100.0	4	66		
Age							
7			91.2	4	25		
8			8.8		41		
Total			100.0	4	66		
Place of birth							
KoB			80.7	3	76		
Outside KoB		19.3		90			
Total		100.0	4	66			
Ethnicity							
Bahraini			72.5	3	38		
GCC			1.3		6		
Arab			21.0		98		
Asian			5.2		24		
Total			100.0	4	66		
Area of reside	nce						
Muharraq Gove	ernorate		98.7	4	60		
Capital Govern	orate		0.2		1		
Southern Gove	rnorate		0.4		2		
Northern Gover	rnorate		0.6		3		
Total			100.0	4	66		
	Mean	Median	Mode	SD	Range		
Father's age	41.0	41	43	7.8	24-73		
Mother's age	35.8	35	32	6.0	23-60		

Table 24. Demographic profile of 466 participants at baseline

4.3.2 Current socioeconomic circumstances

4.3.2.1 Parents' level of education

Nearly two-thirds of the parents had completed more than nine years of formal education. A third of the mothers (33.3%) and nearly half of the fathers (46.1%) had completed their education at secondary school, while another 29.8% of mothers, and 24% of fathers, had graduated from university (see Table 25 below).

4.3.2.2 Parents' employment status

Private sector employee

GE-public jobs salary scale

GE -judges salary scale

Total

GE-specialised jobs salary scale

GE -executive jobs salary scale

GE -academic jobs salary scale

GE -diplomatic jobs salary scale

Military

Retired

No response

Three quarters (74.5%) of the mothers had no salaried occupation, while almost all (98%) of the fathers were employed, over half of them in the military (44%). Of the employed mothers, the majority (14.4%) worked for the government, mostly (11.8%) in less well-paid positions. Similarly, only a third held jobs on higher salary scales than the fathers who worked in government positions (18.7%) (Table 25).

Current SES	N	lother	Father		
Current SES	%	Frequency	%	Frequency	
Level of education					
Uneducated	8.2	38	1.7	8	
Primary	11.8	55	2.1	10	
Intermediate	13.3	62	17.2	80	
Secondary	33.3	155	46.1	215	
College/university	29.8	139	24.0	112	
Postgraduate	3.2	15	8.6	40	
No response	0.4	2	0.3	1	
Total	100.0	466	100.0	466	
Employment status					
Unemployed	74.5	347	1.9	9	
Self-employed	0.4	2	6.4	30	

5.6

2.4

2.4

11.8

1.3

0.0

0.4

0.9

0.0

0.3

100.0

Table 25. Participant's current s	socioeconomic	circumstances
-----------------------------------	---------------	---------------

26

11

11

55

6

0

2

4

0

2

466

12.2

44.0

10.3

18.7

2.4

0.2

1.3

1.1

0.9

0.6

100.0

57

48

87

11

1

6

5

4

3

466

4.3.2.3 Household income

Nearly half (45%) of the participants were from households with a monthly income of less than 600 Bahraini dinars (BD). However, 28.1% of them were from high-income households, earning at least 800 BD per month (Table 26).

4.3.2.4 Housing tenure

Approximately two-thirds of families (65.9%) did not own their homes, and a further 5.6% of them were still living with their parents (Table 26).

4.3.2.5 Housing crowding

Nearly half of the families (45.5%) lived in overcrowded conditions, namely two or more people per room. The number of rooms per home ranged from 1 to 11 (mean no. of rooms = 3.2; SD = 1.4) (Table 26).

 Table 26. Participant's household income and housing conditions

	Household				
		%		Free	quency
Household monthly income					
BD <200 (low)	6.4			30	
BD 200-399 (low)	10.5			49	
BD 400-599 (moderate)	28.1			131	
BD 600-799 (moderate)	26.5			123	
BD 800-999 (high)	13.1			61	
BD ≥1000 (high)	15.0			70	
No response	0.4			2	
Total	100.0			466	
Housing tenure					
Rented flat	51.1			238	
Owned flat	10.1			47	
Rented house	14.8			69	
Owned house	18.4			86	
Other: living with parents	5.6			26	
Total	100.0			466	
Housing crowding					
>2 people per room (overcrowded)	45.5			212	
≤2 people per room (non-crowded)	54.5			254	
Total	100.0			466	
	Mean	Median	Mode	SD	Range
Number of rooms in the home, excluding kitchen and bathrooms	3.2	3	2	1.4	1-11
Number of people living in the home	6.4	6	5	2.3	1-22

4.3.3 Current family structure

Eighty-five per cent of the children lived with both parents, while 10.7% and 4.3% were from extended and single-parent families, respectively. The number of siblings ranged from 0 to 11, with most children (71.9%) having three or fewer siblings (mean no. of siblings = 3.2, SD = 2.0) (Table 27).

Family structure		%	Frequency			
Family type						
Single parent	4.3			20		
Extended	10.7			50		
Nuclear	85.0			396		
Total	100.0			466		
Parent'/s' marital status						
Widowed	1.3			6		
Divorced	3.4			16		
Married	95.3			444		
Total	100.0			466		
	Mean	Median	Mode	SD	Range	
Number of siblings	3.2	3	2	2.0	0-11	

Table 27. Participant's current family structure

4.3.4 Current OH behaviours

Generally, parents reported once or twice daily toothbrushing for themselves and their children. However, according to parents, few children (6.6%) did not brush or brush their teeth weekly. Moreover, few of the children used dental floss (15.7%) or mouthwash (21.5%) (Table 28).

Nearly two-thirds of mothers, fathers, and children (61.2%, 58.2%, and 63.1%, respectively) consumed sugar once or twice daily, and 23% of children consumed sugar more than twice a day (Table 28).

Regarding the pattern of dental attendance, although 24% of parents had regular check-ups, nearly 60% visited the dentist only in an emergency, and 7.2% had never

been to the dentist. Similarly, 67.4% of children visited the dentist only in an emergency, and 12.2% had never been to the dentist (Table 28).

The majority of parents (85%) reported using toothpaste containing fluoride in the concentration 1350-1450 ppm (parts per million). The most popular toothpaste was Sensodyne (14.8%), closely followed by Signal (12.9%) and Colgate (9.6%). Among children, however, the proportion reported to use 1350-1450 ppm fluoridated toothpaste was less than half (43.5%), a quarter (25.5%) used toothpaste with lower fluoride concentrations and 3.9% used toothpaste with no fluoride in it at all. Moreover, there was a 24.5% no-response rate for this question, indicating that 24.5% of the children did not use toothpaste of any sort. The most used brands by children were Signal (7.6) and Colgate (5.7%). Most children (92%) were reported to rinse with water after brushing, with only 1.7% spitting out any remaining toothpaste without rinsing.

Parent's OH ratings were almost similar. Most parents perceived themselves and their children as having good OH. Only 1.5% of both parents reported having poor OH. Parental perception of their children's OH did not match the child's perception; 27.7% of parents perceived their children as having excellent; OH, while 61.8% of children perceived themselves as having excellent OH (Figure 13).

Table 28.	Oral	health	behaviours	of	children	and	parents'	participants

One like a life is a line strange	Mother		Father	•	Child	
Oral health behaviours	%	F	%	F	%	F
Toothbrushing regime						
Never	1.9	9	3.3	15	3.6	17
Weekly	0.9	4	1.1	5	3.0	14
Once a day	15.5	72	20.8	97	34.1	159
Twice a day	59.0	275	52.1	243	50.9	237
More than twice a day	22.1	103	21.0	98	8.4	39
No response	0.6	3	1.7	8	0.0	0
Total	100.0	466	100.0	466	100.0	466
Use of dental floss						
Never	53.0	247	52.1	243	84.3	393
Weekly	17.0	79	14.8	69	4.5	21
Once a day	17.8	83	18.5	86	7.1	33
Twice a day	7.5	35	8.2	38	2.6	12
More than twice a day	4.1	19	4.7	22	1.5	7
No response	0.6	3	1.7	8	0.0	0
Total	100.0	466	100.0	466	100.0	466
Use of mouthwash						
Never	36.5	170	38.4	179	78.5	366
Weekly	15.7	73	15.9	74	0	0
Once a day	25.7	120	23.2	108	17.4	81
Twice a day	12.9	60	12.7	59	4.1	19
More than twice a day	8.6	40	8.1	38	0.0	0
No response	0.6	3	1.7	8	0.0	0
Total	100.0	466	100.0	466	100.0	466
Use of fluoridated					•	
toothpaste		%	F	16	3.9	18
No fluoride	3	3.4		11	15.4	72
500 ppm	2	2.4		22	10.1	47
1000 ppm	4	.7		396	43.5	203
1350-1450 ppm	85	5.0		8	2.6	12
Other	1	.7		13	24.5	114
No response	2	2.8		466	100.0	466
Total	100	0.0				
Spit or rinse (toothpaste)						
Rinse with water					92.1	429
Rinse with mouthwash					3.2	15
Spit (no rinsing)					1.7	8
Do not know					3.0	14
Total					100.0	466
Consumption of sugar						
Never	10.7	50	11.6	54	4.3	20
Weekly	16.7	78	14.2	66	9.6	45
Once a day	43.8	204	36.7	171	35.2	164
Twice a day	17.4	81	21.5	100	27.9	130
More than twice a day	10.8	50	14.4	67	23.0	107
No response	0.6	3	1.6	8	0.0	0
Total	100.0	466	100.0	466	100.0	466
Visits to dentist						
Do not visit	5.6	26	8.8	41	12.2	57
Emergencies only	60.3	281	58.8	274	67.4	314
Regular checkup	24.9	116	23.2	108	20.4	95
Other	84	39	7.3	34		
No response	0.8	4	1.9	9	0.0	0
Total	100.0	466	100.0	466	100	466



Figure 13. OH rating of parents, parent's rating of the child's OH, and child's own rating

4.3.5 Current health: perceptions, behaviours, and status

More than 80% of parents perceived themselves as having good to excellent general health, while 13% of both fathers and mothers reported having average health (Table 29; Figure 14). 29.2% of fathers and 2.8% of mothers reported that they smoked tobacco. Among those fathers, cigarettes were the preferred medium (64.2%), shisha (77%) was the favourite among the mothers.

Parental perception of their children's health did not match the child's perception; 51.9% of parents perceived their children as having excellent general health, while 90.6% of children perceived themselves as having excellent general health (Figure 14).

Table 29. Health	perceptions,	behaviours,	and status o	f participants

	Mother		Fath	ner	Ch	ild
	%	F	%	F	%	F
General health (parents' ratings)						
Poor	0.2	1	1.1	5	0.4	2
Fair	2.2	10	2.1	10	1.5	7
Average	13.1	61	13.1	61	5.6	26
Good	51.3	239	48.5	226	40.6	189
Excellent	32.6	152	33.5	156	51.9	242
No response	0.6	3	1.7	8	0	0
Total	100.0	466	100.0	466	100.0	466
Tobacco smoking						
Non-smoker	96.6	450	70	326		
Smoker	2.8	13	29.2	136		
No response	0.6	3	0.8	4		
Total	100.0	466	100.0	466		
Mode of smoking						
Cigarettes	23.0	3	64.2	87		
Shisha	77.0	10	31.4	43		
Electronic cigarettes	0.0	0	4.4	6		
Total	100.0	13	100.0	136		
Chronic health conditions						
Normal (none)					90.6	422
Asthma					2.8	13
Iron deficiency					1.3	6
Epilepsy					0.2	1
Disability (sight or speech)					0.9	4
Autism					0.2	1
ADHD					0.4	2
Other					3.6	17
Total					100.0	466
Current medication						
Not taking medication					95.9	447
Currently taking medication					4.1	19
Total					100.0	466
BMI percentile						
Underweight (<5th)					7.7	36
Normal BMI (5th-85th)					61.8	288
Overweight (85th-95th)					11.8	55
Obese (≥95th)					18.7	87
Total					100.0	466



Figure 14. Health rating of parents; parent's rating of the child health, and child's own rating

Less than 10% of the children had chronic health conditions. Of them, nearly a third of those children were asthma (2.8% of the total sample). However, 6.8% reported having other conditions like glucose-6-phosphate dehydrogenase (G6PD) deficiency, allergies, leukaemia, eczema, urinary tract infection, and others (Figure 15).



Figure 15. Prevalence and type of chronic health conditions among children

Interestingly, obesity was another condition of concern among children in Muharraq. Nearly one third (31%) of children were overweight or obese, which was true for both genders (see Figure 16).



Figure 16. Prevalence of overweight and obesity by gender

4.3.6 Early life factors

4.3.6.1 Socioeconomic circumstances at the time of the child's birth

4.3.6.1.1 Parents' levels of education

A third of mothers (65.7%) and more than three quarters (77.2%) of the fathers completed more than nine years of study. The most frequent level of education among both fathers (45.3%) and mothers (34.5%) was the attainment of a secondary school certificate (Table 30).

4.3.6.1.2 Parent's types of employment

Three-quarters of the mothers (74.7%) were not in salaried employment when the child was born, contrasting with only 3.2% of the fathers. Almost half of the fathers (46.4%) were in the military, and a further 22.3% were public sector workers, mostly teachers (Table 30).

4.3.6.1.3 Household income

Less than a quarter of the families had a household income of ≤399 BD per month. The majority (77.4%) fell into the moderate- and high-income groups (400 BD per month or more) (Table 30).

4.3.6.1.4 Housing tenure

A quarter (25.3%) of the families owned their house or flat when their child was born. Nearly all the remaining families rented the house, with just a few (7.1%) living with their parents (Table 30).
Table 30. Participant's socioeconomic circumstances at the time of child's birth

Farkelife OFO	Mother		Father		
Early life SES	%	Frequency	%	Frequency	
Level of education					
Uneducated	9.0	42	2.6	12	
Primary school	11.8	55	3.0	14	
Intermediate school	13.6	63	17.2	80	
Secondary school	34.5	161	45.3	211	
College/university	27.7	129	25.1	117	
Postgraduate	3.0	14	6.2	29	
No response	0.4	2	0.6	3	
Total	100.0	466	100.0	466	
Type of employment					
Unemployed	74.7	348	3.2	15	
Self-employed	1.1	5	5.6	26	
Employed in the private sector	7.1	33	13.4	62	
Military	0.8	4	46.4	216	
Retired	0.2	1	3.4	16	
GE - public jobs salary scale	13.4	62	22.3	104	
GE - specialised jobs salary scale	0.8	4	2.6	12	
GE - judges' salary scale	0	0	0	0	
GE - executive jobs salary scale	0	0	0.8	4	
GE - academic jobs salary scale	0	0	1.1	5	
GE - diplomatic jobs salary scale	1.5	7	0.4	2	
No response	0.4	2	0.8	4	
Total	100.0	466	100.0	466	
		Househo	old		
		• /	_		
Users also bell and with he in a sure		%	Fre	quency	
Household monthly income		%	Free	quency	
Household monthly income BD <200 (low)	6	% .9	32	quency	
Household monthly income BD <200 (low) BD 200-399 (low)	6 15	% .9 .7	32 73	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-300 (moderate)	6 15 35	% 5.9 5.7 5.4	32 73 165	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-799 (moderate)	6 15 35 21	% 5.9 5.7 5.4 .2	32 73 165 99	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-799 (moderate) BD 800-999 (high) BD >1000 (high)	6 15 35 21 10	% 5.9 5.7 5.4 .2 5.5	32 73 165 99 49	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-799 (moderate) BD 800-999 (high) BD ≥1000 (high)	6 15 35 21 10 10	% 5.9 5.7 5.4 .2 5.5 5.1	32 73 165 99 49 47	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-799 (moderate) BD 800-999 (high) BD \geq 1000 (high) No response	6 15 35 21 10 10	% 5.9 5.7 5.4 .2 5.5 5.1 2.2	32 73 165 99 49 47 1	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-799 (moderate) BD 800-999 (high) BD ≥1000 (high) No response Total	6 15 35 21 10 10 <i>100</i>	% 5.9 5.4 .2 0.5 0.1 0.2 0.0	Free 32 73 165 99 49 47 47 1 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 <i>0</i> <i>100</i>	% 5.9 5.7 5.4 .2 0.5 0.1 0.2 0.0	Free 32 73 165 99 49 47 1 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 <i>100</i> <i>100</i>	% 5.9 5.7 5.4 .2 5.5 5.1 5.2 5.0 5.4 6	Free 32 73 165 99 49 47 1 466 216 26	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 46 5	% 5.9 5.7 5.4 .2 5.5 5.1 5.2 5.0 5.4 5.6 2	Free 32 73 165 99 49 47 1 466 216 26 99	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 46 5 21	% .9 .7 .4 .2 .5 .1 .2 .0 .0 .0 .4 .6 .2 .7 .4 .6 .2 .7 .4 .2 .5 .1 .2 .0 .7 .4 .2 .5 .1 .2 .0 .7 .4 .2 .2 .2 .2 .2 .2 .0 .7 .4 .2 .2 .2 .0 .7 .4 .2 .2 .0 .7 .4 .2 .2 .2 .0 .7 .4 .2 .2 .0 .7 .7 .7 .7 .7 .7 .7 .7 .7 .7	Free 32 73 165 99 49 47 47 1 466 216 26 99 92	quency	
Household monthly income BD <200 (low) BD 200-399 (low) BD 400-599 (moderate) BD 600-799 (moderate) BD 800-999 (high) BD ≥1000 (high) No response Total Housing tenure Rented flat Owned flat Rented house Owned house Other: living with parents	6 15 35 21 10 10 100 100 46 5 21 19 7	% 5.9 5.7 5.4 .2 5.5 5.1 5.2 5.0 5.4 5.6 .2 5.7 5.1 5.2 5.7 5.1 5.2 5.7 5.1 5.2 5.7 5.1 5.2 5.7 5.1 5.2 5.7 5.1 5.2 5.7 5.1 5.2 5.2 5.2 5.2 5.2 5.2 5.2 5.2	Free 32 73 165 99 49 47 1 466 216 26 99 92 33	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 100 100 46 5 21 19 7 100	% 5.9 5.7 5.4 .2 5.5 0.1 0.2 0.0 5.4 5.6 .2 0.7 1.1 0.0	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 100 100 46 5 21 19 7 100	% .9 .7 .4 .2 .5 .1 .2 .0 .0 .4 .6 .2 .7 .1 .0 .7 .1 .0 .0 .7 .1 .0 .0 .0 .1 .2 .0 .0 .1 .2 .0 .0 .0 .1 .2 .0 .0 .1 .2 .0 .0 .1 .2 .0 .0 .0 .1 .2 .0 .0 .0 .1 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 46 5 21 19 7 100	% 5.9 5.7 5.4 .2 5.5 0.1 0.2 0.0 5.4 5.6 .2 0.7 7.1 0.0	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 46 5 21 19 7 100	% 5.9 5.7 5.4 .2 5.5 0.1 0.2 0.0 5.4 5.6 .2 0.7 1.1 0.0	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 100 70 7 100 76	% 5.9 5.7 5.4 .2 5.5 0.1 0.2 0.0 5.4 5.6 .2 0.7 1.1 0.0 5.8 5.8	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466 358	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 100 100 7 100 7 100 76 23	% 3.9 3.7 3.4 .2 3.5 3.1 3.2 3.0 3.4 3.6 3.2 3.7 3.1 3.0 3.8 3.2 3.8 3.2	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466 358 108	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 100 46 5 21 19 7 100 76 23	% 3.9 3.7 3.4 .2 3.5 3.1 3.2 3.0 3.4 3.5 3.1 3.2 3.2 3.7 3.1 3.0 3.2 3.7 3.1 3.0 3.2 3.7 3.2 3.2 3.2 3.2 3.2 3.2 3.2 3.2	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466 358 108	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 46 5 21 19 7 100 76 23 100	% .9 .7 .4 .2 .5 .1 .2 .0 .0 .4 .6 .2 .7 .1 .0 .0 .8 .8 .2 .0 .0 .0 .0 .0 .0 .0 .0 .0 .0	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466 358 108 466	quency	
Household monthly incomeBD <200 (low)	6 15 35 21 10 10 100 46 5 21 19 7 100 76 23 100	% 3.9 3.7 3.4 2.5 3.1 3.2 3.0 3.4 3.6 3.2 3.7 3.1 3.0 3.4 3.6 3.2 3.7 3.1 3.0 3.2 3.0 3.4 3.5 3.1 3.2 3.0 3.2 3.0 3.4 3.5 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.0 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.2 3.1 3.1 3.2 3.1 3.1 3.1 3.1 3.1 3.1 3.1 3.1	Free 32 73 165 99 49 47 1 466 216 26 99 92 33 466 358 108 466	quency	

4.3.6.2 Family structure

Most of the children (83.9%) were born in a nuclear family, and 15.0% were born in an extended family. The number of siblings ranged from 0 to 14, with the majority of the children (53.9%) having more than two siblings. The children's birth order ranged from the first to the fifteenth, with the largest quadrant (31.5%) being the fourth or later. However, 24.3% were their parents' first child (Table 31).

Table 31.	Participant's	family structure	at the time o	of child's birth

Family structure	%	Frequency
Family type		
Single	1.1	5
Extended	15.0	70
Nuclear	83.9	391
Total	100.0	466
Family marital status		
Widowed	0.4	2
Divorced	2.6	12
Married	97.0	452
Total	100.0	466
Child's birth order		
First	24.3	113
Second	23.2	108
Third	21.0	98
Fourth or later	31.5	147
Total	100.0	466
(Mean: 3.0; Median: 3.0; Mode: 1; SD: 2.0; Range: 1-15)		
Number of siblings		
0	22.7	106
≤2	23.4	109
>2	53.9	251
Total	100.0	466
(Mean: 2.0; Median: 2; Mode: 1; SD: 1.9; Range: 0-14)		

4.3.6.3 Maternal demographic and health conditions

When the child was born, the mother's age ranged from 16 to 53 years (mean age: 28.7 years; SD = 6.0). During pregnancy, about 12.9% of the mothers had diabetes mellitus, and 8.6% had hypertension. Only 1.5% of the mothers smoked during pregnancy. Few mothers (5.2%) reported having other health conditions such as asthma, iron deficiency or a thyroid problem, while 8.4% of them reported having postnatal depression. The majority of mothers (71.2%) delivered their child normally, the rest undergoing caesarean section (Table 32).

Table 32. Mother	's demographic and	d health during pregnancy	1
		0,0,0	

Biological/health factor	%	Frequency
Age		
≤28 years	52.8	246
>28 years	46.6	217
No response	0.6	3
Total	100.0	466
(Mean: 28.7; Median: 28; Mode: 30; SD: 6.3; Range: 16-53)		
Medical condition: Diabetes		
Yes	12.9	60
No	86.9	405
No response	0.2	1
Total	100.0	466
Medical condition: Hypertension		
Yes	8.6	40
No	91.2	425
	0.2	1
	100.0	466
Other medical condition(s)	5.0	24
Tes	5.2	24
	94.0	441
	100.0	166
Total	100.0	400
Type of other medical condition		
Asthma	12.5	3
B12, Fe deficiency	4.2	1
Infection	4.2	1
Iron deficiency	54.1	13
Thyroid	25.0	6
Total	100.0	24
Smoking during pregnancy		
Yes	1.5	7
No	98.3	458
No response	0.2	1
Total	100.0	466
Prenatal depression		
Yes	8.4	39
No	91.4	426
No response	0.2	1
Total	100.0	466
Delivery method	71.2	332
Normal birth	28.6	133
Caesarean section	0.2	1
No response Total	100.0	466

4.3.6.4 Children's birth outcomes and feeding practices

Most children were born with normal gestational age and normal birth weight; 11% were born prematurely, while 11.2% had a low birth weight. Regarding feeding practices, almost a third (27.9%) of the children were exclusively breastfed during the first six months, while 19.1% were bottle-fed only, the remainder (52.8%) being both bottles- and breastfeeding. Nearly a quarter of the parents bottle-fed their child with a sweetened drink at bedtime (Table 33).

Just over a quarter of the children (25.3%) used a pacifier, and the majority of those (81.4%) used the pacifier from birth. Most commonly (55.9% of pacifier users), pacifiers were used for a duration of more than a few months but less than two years, although 21.2% of pacifier users used one for longer than that.

Table 33	. Children's bi	th outcomes a	nd feeding	practices in	n infancy
----------	-----------------	---------------	------------	--------------	-----------

	%	Frequency
Gestational age		
Preterm birth (less than 37 weeks)	11.0	51
Normal (equal to or more than 37 weeks)	88.8	414
No response	0.2	1
Total	100.0	466
Birth weight		
Low birth weight (less than 2500 g)	11.2	52
Normal (equal to or more than 2500 g)	88.6	413
No response	0.2	1
Total	100.0	466
Feeding practices		
Breastfeeding only	27.9	130
Bottle feeding only	19.1	89
Breast and bottle feeding	52.8	246
No response	0.2	1
Total	100.0	466
Nighttime sweet drinks		
Yes	24.9	116
No	74.9	349
No response	0.2	1
Total	100.0	466
Use of pacifier		
Yes	25.3	118
No	74.5	347
No response	0.2	1
Total	100.0	466
Timeframe of using a pacifier		
From birth	81.4	96
Began at 6 months	5.9	7
Began after 6 months	3.4	4
Cannot remember	9.3	11
Total	100.0	118
Duration of using a pacifier		
A few months	11.9	14
Less than two years	55.9	66
Two years or more	21.2	25
Cannot remember	11.0	13
Total	100.0	118

4.3.6.5 Child's OH behaviour at 3-6 years old

Most children (85.4%) had started brushing their teeth older than 12 months. Less than half of the sample (40.6%) always brushed their teeth under parental supervision; only 4.5% had never had their toothbrushing supervised by their parents.

Before enrolling in school, 48.9% of the children experienced dental pain, 57.3% had developed caries in primary teeth, and 47% had undergone dental treatment (Table 34).

Table 34. Children's OH status and OH behaviours at 3-6 years old

Oral health factors or behaviour	%	Frequency
Age when started brushing teeth		
Not brushing	0.4	2
Older than 12 months	85.4	398
12 months or younger	14.2	66
Total	100.0	466
Supervised toothbrushing		
Never	4.5	21
Occasionally	28.3	132
Mostly	26.6	124
Always	40.6	189
Total	100.0	466
Dental pain		
Yes	48.9	228
No	51.1	238
Total	100.0	466
Dental caries in primary teeth		
Yes	57.3	267
No	42.7	199
Total	100.0	466
Dental treatment		
Yes	47.0	219
No	53.0	247
Total	100.0	466

4.3.7 Child's early learning

Most children (72.7%) were enrolled in kindergarten and spent 1-3 years there before entering primary school (Table 35).

At Grade one, nearly four-fifths of the children (77.9%) were rated 'Excellent' for their overall performance as reported by their parents. Over half (55.1%) of the children were absent for 1-5 days during their Grade 1 academic year, while another third (33.3%) attended school every day. Few children (12.4%) were absent from school

because of dental problems. The main reason for this (46.6% of dental-related absences) was dental pain (Table 35).

 Table 35. Children's early learning outcomes

Children's academic performance and attendance	%	Frequency	
Attended kindergarten?			
Yes	72.7	339	
No	27.3	127	
Total	100.0	466	
lf 'Yes', years at kindergarten			
One	37.2	126	
Two	38.6	131	
Three	24.2	82	
Total	100.0	339	
Overall academic performance			
Acceptable	0.4	2	
Average	1.7	8	
Good	4.1	19	
Very good	15.9	74	
Excellent	77.9	363	
Total	100.0	466	
School attendance			
No absences	33.3	155	
Absent for 1-5 days	55.1	257	
Absent for 6-9 days	8.2	38	
Absent for ≥10 days	3.4	16	
l otal	100.0	466	
Any absences because of a dental problem(s)	10.1	50	
Yes	12.4	58	
NO	87.6	408	
lotal	100.0	400	
Reason(s) for absence(s)			
Caries	15.5	9	
Cheek injury	1.7	1	
Surgery	1.7	1	
Extraction	3.4	2	
Infection	10.3	6	
Pain	46.6	27	
Pain, caries, infection	3.4	2	
Pain and treatment	10.3	6	
Ireatment	7.1	4	
lotal	100.0	58	

4.3.8 Child's school outcomes (Grade 2)

Similarly, with Grade one, most of the children were rated as having an excellent overall academic performance for Grade 2 (79%) (Table 36). Figure 17 shows the mean score of school performance among male (mean= 93.8, SD= 6.5) and female (mean= 93.5, SD= 6.5). A Mann-Whitney U test displayed no significant difference (U = 25511, p = 0.133) in school performance between male and female schoolchildren. On the other hand, children have been absent from 1-22 days per the first term of the academic year (2019-2020) (Table 36). The mean score of absence day among females (mean= 1.7, SD= 2.2) was higher than the mean score among males (mean= 1.4, SD= 2.8) (Figure 18). The Mann-Whitney U test displayed a significant difference (U = 21304, p = 0.001) in school absence between male and female schoolchildren.

Children's academic performance and attendance	%		Freque	ncy	
Overall academic performance					
Average	0.9	4			
Good	4.3	20			
Very good	15.8	74			
Excellent	79.0	368			
Total	100.0	466			
School attendance					
No absences	37.6	155			
Absent for 1-5 days	57.3	257			
Absent for 6-9 days	3.6	38			
Absent for ≥10 days	1.5	16			
Total	100.0	466			
	Mean,95%CI	Median	Mode	SD	Range
Absence school days	1.5 1.3/1.8	1	0	2.2	1-22
School performance	93.5 92.9/94.1	96.0	90.0	6.5	65- 100



Figure 17. School performance per gender for the academic year (2019-2020)



Figure 18. School absence days per gender for the first term (2019-2020)

4.3.9 Parent's characteristics and social support

4.3.9.1 Parenting style

Based on the PSDQ (Robinson et al. 1995) questionnaire, most parents' (91.9% of mothers, 88.2% of fathers) answers indicated an authoritative parenting style (Table 37).

4.3.9.2 Parents' academic involvement

The scores from the seven relevant items on the questionnaire were totalled, yielding each parent's overall score for academic involvement. Scores ranged from 0 to 28, with higher scores reflecting higher involvement. The mother's mean academic involvement score (20.7) was higher than the father's mean score (14.7). The median academic involvement score for mothers was 21, compared to a median of 15 for fathers (Table 37).

4.3.9.3 School social support

School social support was calculated by adding the scores from the six relevant items. The total ranges from 0 to 12, with higher scores indicating greater support. The mean social support score among children was 11.3, with SD equal to 1.5 (Table 37).

Table 37. Parent's characteristics and social support of participants

Variables	Mother			Father
	%	Frequency	%	Frequency
Parenting style				
Permissive	3.2	15	5.2	24
Authoritarian	2.6	12	3.2	15
Authoritative	92.1	429	88.2	411
Permissive and Authoritative	0.9	4	1.1	5
Authoritarian and Authoritative	0.2	1	0	0
Permissive, Authoritarian and Authoritative	0.4	2	0	0
No response	0.6	3	2.3	11
Total	100.0	466	100.0	466

	Mean, 95%Cl	Median	Mode	SD	Range
Parental academic involvement – mothers	20.7 20.3/21.2	21	28	5.3	0-28
Parental academic involvement – fathers	14.7 14.0/15.4	15	0	7.5	0-28
Social support from teachers and peers	11.3 11.2/11.5	12	12	1.5	0-12

4.3.10 Paediatric oral health-related quality of life (POQL)

The distribution of POQL scores was positively skewed, indicating that few impacts were reported among the sample (Figure 19). The mean (SD) POQL score was 8.8 (SD: 12.0), the median was 3.8, the first quartile was 0.0, and the third quartile was 13.3.

Most children rate their OH and general health as excellent (61.8% and 90.6%, respectively) (Table 38). The majority (67.6%) of the children reported minimally one impact on their quality of life. In contrast, one third (32.4%) did not report any impact. Pain, difficulty in eating, and crying were the events most reported by the children (Table 39).

Table 38. POQL scores of 466 children (100%) child

	Mean9 5%Cl	Median	Mode	Range	SD	Percentile 25,50,75
POQL overall Impact %	8.8 7.7/9.9	3.8	0	0-78	12.0	0.0 3.8 13.3
POQL by number of events reported	2.2 2.0/2.4	2	0	0-10	2.4	0 2 3
		%			F	
OH perception Poor Faire Average Good Excellent <i>Total</i>		1.5 2.4 16.1 18.2 61.8 <i>100</i>		7 11 75 85 288 <i>4</i> 66		
	% F					
General health perception Poor Faire Average Good Excellent <i>Total</i>	0.6 1.5 2.4 4.9 90.6 100			3 7 11 23 422 466		



Figure 19. POQL scores were distributed among 466 children

Item	Did no happe	ot en	Once while	e in a e	Som times	e of the s	All th	ie times
Dental pain	289	(62.0%)	58	(12.5%)	71	(15.2%)	48	(10.3%)
Difficulty eating	320	(68.7%)	40	(8.6%)	71	(15.2%)	35	(7.5%)
Attention at school	383	(82.2%)	34	(7.3%)	44	(9.4%)	5	(1.1%)
Missing schools	389	(83.4%)	32	(6.9%)	39	(8.4%)	6	(1.3%)
Avoid smiling	368	(79.0%)	32	(6.9%)	56	(12.0%)	10	(2.1%)
Worry less attractive	383	(82.2%)	20	(4.3%)	50	(10.7%)	13	(2.8%)
Unhappy with look	404	(86.7%)	19	(4.1%)	30	(6.4%)	13	(2.8%)
Angry	391	(84.0%)	23	(4.9%)	44	(9.4%)	8	(1.7%)
Worry	384	(82.4%)	30	(6.4%)	33	(7.1%)	19	(4.1%)
Cry	309	(66.3%)	48	(10.3%)	76	(16.3%)	33	(7.1%)
Sum up								
Children reported 0 events	151 (3	32.4%)						
Children reported ≥ 1	315 (6	67.6%)						

Table 39. POQL items scored by 466 children (100%)

4.3.11 Clinical data

Between September and November 2019, 466 students were examined by two examiners, and 10286 teeth were screened (Table 40).

 Table 40. Number of children and teeth examined

Children/teeth	Number
Total children examined	466
Children with permanent teeth present (along with primary)	452
Children with only primary teeth present (no permanent teeth yet)	14
Children examined by examiner 1/ recorder 1 (51.93%)	242
Children examined by examiner 2/ recorder 2 (48.07%)	224
Primary teeth examined	6,644
Permanent teeth examined	3,642
Total teeth examined	10,286

4.3.11.1 Dental caries

The ICDAS results displayed the prevalence of enamel and dentine caries to be greater in the primary dentition (93.1%) than in permanent dentition (45.8%) (Table 41). Total ICDAS mean of enamel and dentine caries per teeth surface was considered (Table 41, 42). Overall, the ICDAS distribution was positively skewed (Figure 20). The mean (SD) overall ICDAS score was 19.5 (13.9), the median was 18.0, the first quartile was 8.0, and the third quartile was 28.0. The mean of ICDAS were slightly higher for male (mean=19.8, SD= 13.7) than female (mean= 19.3, SD= 14) schoolchildren (Figure 21); however according to the Mann-Whitney U test, the difference was not significant (U = 26449, p = 0.589). ICDAS scores 4-6 were converted into dmfs/DMFS (Iranzo et al., 2013) to facilitate comparison between ICDAS and DMF.

Table 41. Dental caries measures a	ccording to ICDAS and dmfs/DMFS
------------------------------------	---------------------------------

Dental caries for primary and permanent dentition				
Number of teeth missing due to caries	225			
Number of children with no present or past caries experience (ICDAS)	31			
Number of children with no present or past caries experience (D ₄ MFS)	48			
Prevalence of dental caries (ICDAS), 95%CI	93.5%, 92/95			
Prevalence of dental caries (D₄MFS/d₄mfs), 95%CI	89.7%,88.6/91.4			
Dental caries (primary teeth)				
No. of children with no present caries experience (ICDAS) in primary teeth	32			
No. of children with no past or present caries experience (ICDAS) in				
primary teeth	32			
No. of children with only dental caries according to ICDAS	434			
No. of children with no past or present caries experience (d ₄ mfs) in primary				
teeth	48			
No. of children with only dental caries according to dmfs	418			
Prevalence of dental caries in primary teeth (ICDAS)	93.1%, 92/95			
Prevalence of dental caries in primary teeth (d₄mfs)	89.7%, 88.6/91.4			
The difference between ICDAS and dmfs prevalence for primary teeth	3.4%			
Dental caries (permanent teeth)				
No. of children with no present caries experience (ICDAS) in permanent				
teeth	245			
No. of children with no past or present caries experience (ICDAS) in				
permanent teeth	245			
No. of children with only dental caries according to ICDAS	207			
No. of children with no past or present caries experience (D ₄ MFS) in				
permanent teeth	384			
No. of children with only dental caries according to D₄MFS	82			
Prevalence of dental caries in permanent teeth (ICDAS)	45.8%, 43/49			
Prevalence of dental caries in permanent teeth (D ₄ MFS)	18.1%, 6/30			
The difference between ICDAS and DMFS prevalence for permanent teeth	27.7%			



Figure 20. Distribution of ICDAS means among 466 children



Figure 21. ICDAS mean per gender among 466 children

Table 42. ICDAS mean and total counts in primary and permanent dentition of 466

 children

	Mean	95%CI	Median	Mode	Range	SD	Percentile 25,50,75
ICDAS mean for primary and permanent surfaces	19.5	18.2/20.8	18	0	0-70	13.9	8.0 18.0 28.0
Total counts of surfaces with ICDAS scores 2 and more in primary surfaces ds (2-6)	17.9	16.7/19.1	17	0	0-68	13.0	7.0 17.0 26.0
Total counts of surfaces with ICDAS scores 2 and more in permanent surfaces DS (2- 6)	1.6	1.3/1.8	0	0	0-20	2.6	0.0 0.0 2.0
Total counts of surfaces with ICDAS scores 2-3 in primary surfaces ds (2-3)	1.6	1.4/1.7	1	0	0-11	1.9	0.0 1.0 2.0
Total counts of surfaces with ICDAS scores 4-6 in primary surfaces ds (4-6)	16.3	15.2/17.5	15	0	0-68	12.9	6.0 15.0 24.3
Total counts of filled primary surfaces related to caries	0.2	0.0/0.2	0	0	0-5	0.6	0.0
Total counts of missed primary surfaces related to caries	2.4	1.0/2.7	0	0	0-25	4.3	0.0 0.0 5.0
Total counts of sound primary surfaces	46.7	45.1/48.2	47	52	0-88	17.1	35.0 47.0 58.0
Total counts of surfaces with ICDAS scores of 2 and more surfaces with fillings (f) and surfaces of extracted teeth (m)= ds (2-6)fm	20.2	19.1/21.7	19	0	0-70	14.0	9.0 19.0 30.0
Total counts of surfaces with ICDAS scores 2-3 in permanent surfaces DS (2-3)	1.2	1.0/1.3	0	0	0-14	2.0	0.0 0.0 2.0
Total counts of surfaces with ICDAS score 4-6 in permanent surfaces DS (4-6)	0.4	0.3/0.6	0	0	0-20	1.5	0.0
Total counts of permanent surfaces filled due to caries (F)	0.0	0.0/0.01	0	0	0-1	0.2	0.0
Total counts of permanent surfaces missed due to caries (M)	0.0	0.0/0.0	0	0	0-5	0.2	.00
Total counts of permanent sound surfaces	33.0	31.9/34.2	36	28	0-52	12.5	27.0 36.0 43.0
Total counts of surfaces with ICDAS scores of 2 and more surfaces with fillings (F) and surfaces of extracted teeth (M)= DS (2-6)FM	1.6	1.4/1.8	0	0	0-20	2.7	0.0 0.0 2.0

4.3.11.2 Clinical consequences of untreated dental caries

PUFA/pufa prevalence was 25.7% (Table 43), primary teeth were 24.7%, and permanent teeth were 1.8%. The Untreated Caries PUFA Ratio was 8.6%, implying that 8.6% of the D+d component evolved into an odontogenic infection. The main component of PUFA/pufa was pulpal involvement (13.7%).

Table 43. Prevalence and most prevalent consequences of untreated caries in primaryand permanent teeth among participants

PUFA/pufa at primary and permanent dentition - child level			
Total number of students with PUFA/pufa	120		
Prevalence of PUFA/pufa	25.7%, 22/30		
Total number of teeth with PUFA/pufa	218		
PUFA/pufa mean. 95%Cl	0.47, 0.4/0.6		
First most prevalent PUFA/pufa condition - (P/p)	13.7%		
Second most prevalent PUFA/pufa condition - (PU/pu)	9.4%		
Common PUFA condition with one code only-child leve	1		
First most prevalent (P/p)	13.7%		
Second most prevalent (U/u)	12.5%		
pufa primary teeth-child level			
No. of students with pufa	115		
Prevalence	24.7%		
Sum of all pufa in primary teeth	211		
pufa mean, 95%Cl	1.2, 1.2/1.3		
First most prevalent condition in pufa (p)	13.1%		
Second most prevalent condition in pufa (pu)	9.4%		
First most prevalent condition- one code pufa (p)	13.1%		
Second most prevalent condition- one code pufa (u)	12.2%		
PUFA permanent teeth-child level			
No. Of students with pufa	8		
Prevalence of PUFA	1.8%		
Sum of all PUFA in permanent teeth (teeth no.)	11		
PUFA mean, 95%Cl	1.0, 1.0/1.03		
First most prevalent condition in PUFA (p)	0.7%		
Second most prevalent condition in PUFA (a)	0.4%		
First most prevalent condition- one code PUFA (A)	0.9%		
Second most prevalent condition- one code PUFA (P)	0.7%		

4.3.11.3 Dental trauma

Only four children presented dental trauma on maxillary and mandibular permanent

incisor teeth (prevalence = 0.9%) (see Table 44).

Table 44. Dental trauma in permanent incisors of children

Dental Trauma Index			
Total number of students with dental trauma on maxillary and mandibular permanent incisor teeth	4		
Prevalence of dental trauma on maxillary and mandibular permanent incisor teeth	0.9%		
Prevalence of dental trauma on maxillary and mandibular permanent incisor teeth (when excluding students with no permanent incisor teeth)	0.9%		
First most prevalent condition at the children level (fractured enamel)	0.7%		
Second most prevalent condition at the children level (discolouration)	0.2%		

4.3.11.4 Malocclusion

Malocclusion conditions that might increase the risk of dental trauma were generally low (Table 45).

 Table 45. Malocclusion conditions among children

Condition	Prevalence	Number of children with the condition
Incompetent lip	10.2%	47
Overjet	3.4%	16
Overbite	2.1%	10
Anterior open bite	6.7%	31
Anterior cross-bite	2.6%	12

4.3.11.5 Enamel development defect (EDD)

The overall prevalence of EDD was 17.2%. EDD was more prevalent in the permanent dentition (17.0%) than in the primary dentition (1.5%), with central incisors and first permanent molars being the most affected teeth. Demarcated opacities were the most common defects identified (14.2%) (Table 46).

 Table 46. EDD prevalence and most common type among children

EDD for primary and permanent dentition				
Total number of children with EDD	80			
Prevalence of EDD	17.2%			
First, most prevalent EDD condition was Demarcated opacities	14.2%			
Second most prevalent EDD condition was Hypoplasia)	2.2%			
EDD for primary teeth				
No. of children with primary tooth EDD	7			
Prevalence of EDD	1.5%			
Sum of all primary teeth with EDD (teeth no.)	16			
First, most prevalent condition was Demarcated opacities	0.9%			
Second most prevalent condition was Hypoplasia	0.6%			
EDD for permanent teeth				
No. of children with permanent tooth EDD	77			
Prevalence of EDD	17.0%			
Sum of all permanent teeth with EDD (teeth no.)	239			
First, most prevalent condition was Demarcated opacities	14.4%			
Second, most prevalent condition was Hypoplasia and Diffused opacities	1.8%			

4.3.11.6 Dental plaque

The prevalence of dental plaque was 84.3%, with a mean of 65.5 (SD=37.7, 95%

CI=62.0/68.9), a median of 83.3, a mode of 100, and a range of 0-100 (Table 47).

 Table 47. Prevalence of dental plaque among children

	Children			
	% of sample Frequency			
Plaque-free (0%)	15.7	73		
100% plaque	40.3	188		
Some plaque, but <50%	13.5	63		
≥50% plaque, but <100%	30.5	142		
Total	100	466		

4.4 Reliability of the study measures

4.4.1 Reliability of the questionnaires

4.4.1.1 Internal reliability

The reliability of the parenting styles and dimensions questionnaire, the parental academic involvement questionnaire, and the child's social support and POQL was tested using Cronbach's alpha. The results revealed acceptable to good internal consistency (see Table 48).

Table 48. Internal consistency of parenting style questionnaires

Questionnaire	Cronbach's alpha
Parenting style	0.89
Parental academic involvement	0.85
School social support	0.70
POQL	0.90
$\alpha \ge 0.9$ Excellent $0.9 > \alpha \ge 0.8$ Good $0.8 > \alpha \ge 0.7$ Acceptable $0.7 > \alpha \ge 0.6$ Questionable $0.6 > \alpha \ge 0.5$ Poor $0.5 > \alpha$ (Tavakol and Dennick, 2011)	
(Tavakol and Dennick, 2011)	

4.4.1.2 Test-retest reliability

The reliability of questionnaires was tested by calculating ICC using Pearson's correlation. The results revealed very strong to moderate reliability (see Table 49).

Table 49. Test-retest results for parenting style questionnaires by ICC

Questionnaire	ICC
Parenting style	0.53
Parental academic involvement	0.94
School social support	0.93
POQL	0.84
1.00 Perfect	
0.8099 Very strong	
0.60-0.80 Strong	
0.40-0.60 Moderate	
0.20040 Weak	
020 Very Weak	
(Dancey and Reidy, 2017)	

4.4.2 Reliability of dental clinical measures

The initial inter-and intra-calibration tests against the gold standard revealed substantial to perfect agreement between the two examiners for ICDAS (Section 3.8.1.1). Further, the calibration during the pilot study showed substantial to almost perfect agreement between and within examiners (Table 50).

	Inter-examiners Kappa coefficient	Intra-examiner 1 Kappa coefficient	Intra-examiner 2 Kappa coefficient
ICDAS restoration code/Unweighted	0.89	0.93	1.00
ICDAS caries code/Weighted	0.77	0.87	0.88
Lip competency	1.00	1.00	1.00
Overjet	0.88	1.00	1.00
Overbite	0.90	0.91	1.00
Open bite	0.61	1.00	1.00
Anterior cross-bite	1.00	1.00	1.00
EDD	0.93	1.00	1.00
PUFA	0.75	0.89	1.00
Dental trauma	0.91	1.00	1.00
Dental plaque	0.84	0.87	0.87
 Kappa < 0: poor agreement, Kappa between 0.0 – 0.20: slight agree Kappa between 0.21 – 0.40: fair agree Kappa between 0.41 – 0.60: moderate Kappa between 0.61 – 0.80: substanti Kappa of 0.81 and above: almost perfect andis and Koch 1977) 	ement, ement, e agreement, al agreement, ect agreement.		

4.5 Data screening

The data were screened for missing values, outliers, and normality before conducting SEM.

4.5.1 Missing values

Though there are no definite cut-off values to evaluate how much considers a large amount of missing information, Cohen et al. (1983) suggested that 5-10% of missing data on a particular variable is not large. According to Hair et al. (2014), if missing data is less than 10% of observations, and the pattern of missing data is random (i.e., missing completely at random = MCAR), any approach to deal with this problem is assumed to produce acceptable results. Appendix 19 presents the frequency and percentage of missing data for the variables involved in the analysis. Thus, 41 cases

showed 0.54% missing values in 20 variables among the 32 included in the analysis. The maximum percentage of missing data was 3.4%, and the lowest was 0.2%.

Although the amount of missing data was low, the randomness of the missing data was tested using Little's (1988) Chi-square statistics. The test revealed a nonsignificant p-value (Little's MCAR test: Chi-Square = 517.370, DF = 19504, Sig. = 1.000), indicating it is safe to assume the data was MCAR. Therefore, the values and the pattern can be considered negligible. However, because SEM requires a dataset without missing values, means and medians were used to replace the missing values (Donner, 1982).

4.5.2 Outliers

Appendix 19 shows the univariate and multivariate analyses. The univariate outliers were detected for the continuous values by determining if the frequency distributions of Z scores of the observed data were below or above the mean by (-, + 3.29 SD) (Kline, 2018). On the other hand, the multivariate outliers were detected by measuring the Mahalanobis D^2 with a p-value < 0.001 (Kline, 2016).

Although the results in Appendix 19 indicates some univariate and multivariate outliers, Hair et al. (2014) suggest that the outliers should be retained to ensure generalisability. Accordingly, the outliers in this study were retained.

4.5.3 Data normality

Appendix 19 shows the normality test results measured by AMOS based on skewness and kurtosis tests. According to Hair et al. (2014), data having skewness between +/-2, and kurtosis between +/ -7, is considered not to have severe skew or kurtosis and not be severely non-normal. The eight variables showed non-normal distribution.

218

Therefore, a bootstrapping procedure was performed with SEM. Bootstrapping is a resampling procedure to create multiple sub-samples from the original data set with multiple parameters estimated (Byrne, 2001). The procedure uses the subsample to draw inferences about the population without assuming normality (Zhu, 1997).

4.6 Structural equation modelling

This section presents the results of the measurement and the structural models for the SEM and formal testing of the study hypotheses.

4.6.1 Measurement model

The measurement model was specified with six latent and 20 observed variables explained in the next section. Table 51 shows the distribution of those variables in relation to school performance and school attendance. The model was evaluated by GOF indices, model estimates and modification indices. **Table 51.** Latent variables distributions according to children's school performance and school absence days

Variables	School performance N %, Mean, SD			School a Me	bsence days ean, SD	Type of variable
	Total N (%)	Mean	SD	Mean	SD	
1. Birth-SES						Latent exogenous
A. Mother's employment status						Observed, endogenous
Unemployed/retired	351(75.3)	93.3	6.5	1.6	2.6	
Self/private employee	38(8.2)	93.2	6.0	2.0	1.7	
Government employee	77(16.5)	94.8	6.0	1.3	2.0	
B. Mother's level of education at child's birth						Observed, endogenous
Not educated	42(9)	91.0	7.2	1.3	2.0	
• ≤9 years	118 (25.3)	92.3	6.3	1.5	3.3	
 >9 years 	306 (65.7)	94.3	6.1	1.5	2.1	
C. Father's level of education at child's birth						Observed, endogenous
Not educated	12 (2.6)	91.7	7.4	2.4	3.0	
• ≤9 years	94 (20.2)	91.5	7.4	3.0	3.1	
 >9 years 	360 (77.2)	94.2	5.9	1.5	2.2	
D. Family income at child's birth						Observed, endogenous
 BD[†] ≤399 	106 (22.7)	91.9	7.3	1.3	1.6	
 BD 400- 799 	264 (56.7)	93.5	6.3	1.7	2.9	
 BD ≥800 	96 (20.6)	95.2	4.8	1.4	1.5	
2. Current SES						Latent exogenous
A. Current family income						Observed, endogenous
 BD ≤399 	81 (17.4)	92.4	7.0	1.9	2.6	
 BD 400- 799 	254 (54.5)	93.1	6.7	1.6	2.8	
 BD ≥800 	131 (28.1)	95.0	5.1	1.2	1.3	
B. Current father's level of education						Observed, endogenous
Not educated	8 (1.7)	90.0	8.1	1.5	1.8	
• ≤9 years	90 (19.3)	91.8	7.1	1.8	3.2	
 >9 years 	368 (79)	94.0	6.1	1.5	2.2	
C. Current mother's level of education						Observed, endogenous
Not educated	38(8.2)	90.9	7.2	1.1	1.5	
• ≤9 years	117(25.1)	92.5	5.9	1.8	3.4	
• >9 years	311(66.7)	94.2	6.3	1.5	2.0	

3. Birth outcomes						Latent exogenous
A. Child's birth weight						Observed, endogenous
 Low birth weight (<2500 g) 	52 (11.2)	93.5	6.5	1.5	3.3	
 Normal (≥2500 g) 	414 (88.8)	93.5	6.4	1.6	2.3	
B. Child's gestational age						Observed, endogenous
 Pre-term birth (<37 weeks) 	51(10.9)	93.2	6.6	1.5	1.9	_
 Normal (≥37 weeks) 	415(89.1)	93.5	6.3	1.6	2.5	
C. Child's birth order						Observed, endogenous
First	113(24.2)	93.5	6.6	1.7	2.6	_
Second or more	353(75.8)	93.5	6.3	1.5	2.4	
4. Mother's OH practices (OHPs)						Latent exogenous
A. Toothbrushing						Observed, endogenous
Never or weekly brushing	16(3.4)	88.6	8.0	1.8	1.9	
Once or twice	347(74.5)	93.5	6.3	1.5	2.4	
More than twice	103(22.1)	94.2	5.8	1.5	2.5	
B. Mouth wash						Observed, endogenous
Never or weekly brushing	246(52.8)	93.7	6.5	1.5	2.7	
Once or twice	180(38.6)	93.5	5.9	1.7	2.2	
More than twice	40(8.6)	92.2	7.4	1.1	1.7	
C. Flossing						Observed, endogenous
Never or weekly brushing	329(70.6)	93.5	6.5	1.5	2.4	· •
Once or twice	118(25.3)	93.6	5.6	1.6	2.5	
More than twice	19(4.1)	93.2	7.8	1.5	1.4	
5. Child's OHPs						Latent exogenous
A. Toothbrushing						Observed, endogenous
Never or weekly brushing	32(6.9)	89.9	7.5	2.0	3.0	_
Once or twice	395(84.7)	93.8	6.2	1.6	2.4	
More than twice	39(8.4)	92.7	6.7	1.0	1.5	
B. Flossing						Observed, endogenous
Never or weekly brushing	413(88.6)	93.6	6.3	1.5	2.4	_
Once or twice	46(9.9)	92.4	6.5	1.9	2.3	
More than twice	7(1.5)	91.5	10.8	1.7	2.2	
C. Mouth wash						Observed, endogenous
Never or weekly brushing	366 (78.5)	93.8	6.2	1.5	2.5	
Once or twice	81(17.4)	92.5	6.8	1.9	2.1	
More than twice	19(4.1)	91.3	7.4	0.9	0.8	
	` ´					

D. Sugar consumption							Observed, endogenous
 Never or weekly brushing 		65(13.9)	92.7	6.7	1.6	2.8	
Once or twice	2	294(63.1)	93.9	6.2	1.5	2.1	
More than twice		107(23)	92.8	6.8	1.6	3.0	
6. Parent's characteristics							Latent exogenous
A. Mother's parenting style							Observed, endogenous
Permissive		15 (3.2)	89.0	8.6	1.6	1.8	
Authoritarian		12 (2.6)	92.8	5.2	1.3	1.8	
Authoritative	4	439 (94.2)	93.7	6.4	1.6	2.5	
B. Father's parenting style							Observed, endogenous
Permissive		24 (5.2)	91.3	6.3	2.2	4.4	
Authoritarian		15 (3.2)	92.8	6.4	1.1	1.5	
Authoritative	4	427 (91.6)	93.6	6.4	1.5	2.3	
C. Mother's academic involvement	4	466(100)	20.8	5.2			Observed, endogenous

4.6.1.1 Goodness of fit indices

Table 52 shows the recommended criteria for GOF indices and summarises the obtained results of the model fit indices.

In detail, the significant Chi-square test (X^2 =332.760, df= 151) at p<0.05 indicated that the model was not a good fit. However, Hair et al. (2014) argued that relying solely on the Chi-square test is not enough and can be misleading, as this test is susceptible to sample size. Thus, another GOF was used to evaluate the specification of the model. The result revealed that the values of SRMR= 0.051, RMSEA= 0.05, GFI= 0.93, CFI= 0.95, and AGFI= 0.91. These values were consistent with the recommended criteria, indicating that the model fits the data.

Table 52.	The recommended	and the obtaine	ed fit indices for	the measuremer	it model
using CFA					

Index	Reference	Recommended criteria	Obtained criteria
<i>χ2/df</i> ratio, p	(Hu and Bentler, 1999)	X ² /df < 3, P >0.05	1.45. P=<0.001
SRMR	(Hair et al., 2014; Byrne, 2001)	≤ .08	0.051
RMSEA	(Hair et al., 2014; Byrne, 2001)	< 0.06	0.05
GFI	(Hair et al., 2014; Byrne, 2001)	≥ 0.90	0.93
CFI	(Hair et al., 2014; Byrne, 2001)	≥ 0.90	0.95
AGFI	(Hair et al., 2014; Byrne, 2001)	≥ 0.90	0.91

4.6.1.2 Model estimates

Table 53 shows the standardised regression weights provided by Beta (β) (factor loadings) related to total effects between variables and the square multilabel correlation (R-square). All variables revealed significant loading (p \leq 0.01) despite being lower than the recommended level of 0.30 in some variables (Bagozzi et al., 1991). However, none of the items with \leq 0.30 was eliminated from the model, as this would violate the 3-4 indicators rule, hence model fit (Hair et al., 2014).

The first latent variable was the family's SES at the time of the child's birth (BSES) with four indicators: mother's level of education, household income, mother's employment, and father's level of education. The mother's level of education showed the highest loading value (0.998), and family income was with the lowest loading value, while the father's level of education presented a medium effect value (0.390).

The second latent variable was the family's current SES (CSES) with three indicators: the mother's level of education, household income, and father's level of education. Similarly, the mother's level of education presented with the highest loading value (0.994). Thus, the results might indicate that maternal related factors might be more important than parental and income in defining family SES.

The third latent variable was the child's birth outcomes with three indicators: birth weight, gestational age, and birth order. Gestational age and birth weight almost had a similar regression weight (0.747 and 0.740, respectively). While birth order had a lower loading value (0.207), it showed a significant loading and deleting it would cause the model to be unidentified.

The fourth latent variable was the mother's oral health practices (OHPs), with three indicators: mother's toothbrushing, mother's flossing, and mother's use of mouthwash.

224

Though all the variables had significant loadings, surprisingly, the use of mouthwash was the factor with the highest loading value (0.791).

The fifth latent variable was the child's current OHPs with four indicators: child's toothbrushing, child's flossing, child's sugar consumption, and child's use of mouthwash. Similarly, mouth wash had the highest loading value (0.840), and sugar consumption had the lowest loading value (-0.195).

The sixth latent variable was the parent's characteristics with three indicators: mother's parenting style, father's parenting style, and mother's academic involvement. Mother parenting style presented with the highest loading (0.708), indicating that the mother related factors were the most important.

Table 53. Standardised regression weights of the measurement model: ß, R-square, BC 95% CI, and bootstrapping SE according to the CFA model

Parameter			Estimate β	BC 95% CI	P-value	R-square	Bootstrapping SE
Father's level of education	<	BSES	.390	.283/0.471	.004	.152	.037
Mother's level of education	<	BSES	.998	.973/1.027	.003	.995	.026
Mother's job	<	BSES	.353	.307/.398	.001	.125	.016
Household income	<	BSES	.187	.104/.290	.001	.035	.016
Mother's level of education	<	CSES	.994	.973/1.013	.002	.988	.020
Father's level of education	<	CSES	.362	.267/.451	.003	.131	.035
Household income	<	CSES	.211	.131/.299	.001	.045	.018
Flossing	<	Mother's OHPs	.296	.142/.418	.003	.088	.041
Use of mouth wash	<	Mother's OHPs	.791	.647/1.096	.001	.626	.195
Teeth-brushing	<	Mother's OHPs	.205	.081/.322	.006	.042	.025
Sugar consumption	<	Child's OHPs	195	298/077	.003	.038	.022
Use of mouth wash	<	Child's OHPs	.840	.724/.951	.003	.706	.097
Flossing	<	Child's OHPs	.478	.322/.605	.002	.229	.066
Teeth-brushing	<	Child's OHPs	.234	.114/.350	.002	.055	.028
Mother parenting style	<	Parent's characteristics	.708	.259/.933	.003	.501	.208
Father parenting style	<	Parent's characteristics	.536	.255/.723	.001	.288	.112
Mother academic involvement	<	Parent's characteristics	.312	.127/.596	.004	.097	.099
Birth weight	<	Birth outcomes	.740	.528/.980	.003	.547	.222
Birth order	<	Birth outcomes	.207	.066/.331	.002	.043	.027
Gestational age	<	Birth outcomes	.747	.521/.974	.003	.557	.198

4.6.1.3 Modification indices (MI)

The final step was done through the evaluation of the MI output. The results revealed four possible relations that the model did not estimate. The MI assumed an association between the father's level of education at birth and the current time. Further, it assumed that the mother's tooth brushing and flossing were also related to child brushing and flossing. Since those relations are sound in the literature (Bernabe et al., 2011, Özbek et al., 2015) and consistent with the theoretical framework of the study, covariances were added between their error terms (Hair et al., 2014) (Figure 22).



Figure 22. Measurement model with six latent variables and 20 observed variables

4.6.2 Parsimonious model

After adding twelve observed variables to the measurement model, the full model was estimated and met all the recommended criteria (Table 54: Figure 23). Table 55 shows the distribution of the 12 observed variables according to children's school outcomes. However, all non-significant paths (p>0.05) were deleted to obtain a parsimonious model described in Section 3.9.3. The fit indices for the parsimonious models met all the pre-established criteria for an acceptable model fit (Table 54). The absolute fit indices, i.e., GFI and RMSEA, were 0.92 and 0.03, respectively. The incremental fit indices, i.e., CFI, were 0.95, and the parsimony fit measure, i.e., AGFI, was 0.91. Which all were within or above the recommended criteria. Additional to these indices, the χ^2 / df ratio was within the threshold level supporting these findings.

The X² difference between the full and parsimonious models was 33.69 (df=46) and was not statistically significant (p=0.911). This suggests that removing the non-significant paths was not relevant to the model, and the parsimonious model was accepted (Figure 21). Solid lines indicate direct effects for the final statistically parsimonious model, and dashed lines indicate indirect effects. The full model can be found in Chapter 8, Appendix 21.

Table 54. Fit indices for the full structural model and the parsimonious model

	χ2/df	SRMR	RMSEA	GFI	CFI	AGFI
Criteria	X ² /df < 3	≤ .08	< 0.06	≥ 0.90	≥ 0.90	≥ 0.90
Full model	1.50	0.052	0.03	0.92	0.95	0.90
Parsimonious model	1.43	054	0.03	0.92	0.95	0.91



Note: Abbreviations stand for EDD (enamel developmental defects), OHPs (oral health practices), PS (parenting style), AI (academic involvement), TB (toothbrushing), CHCs (chronic health conditions)

Figure 23. Structural model of the life course framework for OH as proposed by Heilmann et al. (2015), after mapping latent variables from CFA adding observed variables, and assuming paths between all the variables

 Table 55. Observed variables, according to children's school performance and school absence days

Nominal observed variables	School perfor N %, Mean, SD		School abs Mean, SD	sence days	Type of variable	
	Total N (%)	Mean	SD	Mean	SD	
1. EDD						Observed, endogenous
No EDD	459(98.5)	93.5	6.4	1.5	2.4	
Yes EDD	7(1.5)	93.3	4.0	3.3	3.8	
2. Maternal chronic health conditions (CHCs)						Observed, endogenous
Yes	107(23.0)	93.7	6.2	1.7	3.1	
• No	359((77.0)	93.4	6.4	1.5	2.2	
3. Exclusive breast feeding						Observed, endogenous
• No	336(72.1)	93.4	6.5	1.5	2.5	
Yes	130(27.9)	93.9	6.0	1.7	2.3	
4. Supervised tooth brushing						Observed, endogenous
• No	153(32.8)	93.4	6.5	1.3	2.1	
Yes	313(67.2)	93.5	6.3	1.7	2.6	
5. Child's dental pain						Observed, endogenous
• No	289(62.0)	93.7	6.3	1.6	2.7	
Yes	177(38.0)	93.2	6.4	1.4	1.9	
6. pufa in primary teeth						Observed, endogenous
• No	351(75.3)	93.6	6.1	1.6	2.6	
Yes	115(24.7)	93.3	7.1	1.5	2.0	
Continues observed variables	Mean	Median	Mode		SD	Type of variable
7. Enamel caries in primary teeth (ds_2_3)	1.6	1	0		1.9	Observed, endogenous
8. Dentine caries in primary teeth (ds_4_6)	16.3	15.0	0		12.9	Observed, endogenous
9. Treated primary teeth	2.5	0.0	0		4.3	Observed, endogenous
10. Plaque	65.5	83.3	100		37.7	Observed, endogenous
11. Absence days	1.6	1	0		2.4	Observed, endogenous
12. School performance	93.5	95.4	93.5		6.4	Observed, endogenous


Note: * p < 0.05, ** p < 0.01, β = bootstrapped standardised estimate. Solid line = direct effect; Dash line= indirect effect, the error terms and indicators were deleted to ease the interpretation Abbreviations stand for EDD (enamel developmental defects), OHPs (oral health practices), PS (parenting style), AI (academic involvement)

Figure 24. Significant direct and indirect effects for the parsimonious model

4.6.3.3 Results of hypothesis testing from parsimonious model

This section displays the results of the hypothesis testing from the parsimonious model. First, the section presents the early and current life factors associated with children's OHCs at the current age of life (7-8 years). Second, the section presents the association between OHCs and children's performance and attendance.

Table 56 summarises the results of the direct and indirect associations between the variables, showing only the significant associations. The indirect paths in Table 56 reflect the total indirect effect for all possible mediators, while Table 57 presents the specific significant indirect paths and mediators.

Table 56. Significant direct and indirect relations between the variables in the parsimonious model

Parameter			Total effect	Direct effects	Indirect effects	Bootstrap SE	BC 95% CI
CSES	<	BSES	0.986**	0.986**		0.012	0.957/1.006
Supervised TB	<	BSES	0.175**	0.175**		0.058	0.048/0.278
Child's OHPs	<	BSES	-0.187**		-0.187**	0.062	313/-0.074
Parent's characteristics	<	BSES	0.457*		0.457*	0.144	0.200/0.724
Dentine caries	<	BSES	-0.248**	-0.248**		0.051	-0.344/-0.153
Dental pain	<	BSES	-0.039**		-0.039**	0.015	-0.074/-0.016
School performance	<	BSES	0.136**		0.136**	0.063	0.048/0.280
Dentine caries	<	EDD	-0.086**	-0.086**		0.028	-0.137/-0.026
Dental pain	<	EDD	-0.013**		-0.013**	0.006	-0.028/-0.004
School performance	<	EDD	0.013**		0.013**	0.006	0.004/0.030
Birth outcomes	<	Maternal CHCs	0.161**	0.161**		0.063	0.048/0.291
Child's OHPs	<	Mother's OHPs	0.778**	0.778**		0.082	0.575/0.924
Child's OHPs	<	CSES	-0.190**	-0.190**		0.062	-0.317/-0.076
Parent's characteristics	<	CSES	0.464*	0.464*		0.146	0.202/0.727
School performance	<	CSES	0.099**		0.099**	0.067	0.018/0.257
Enamel caries	<	Plaque	0.094*	0.094*		0.044	0.008/0.180
Treated teeth	<	Plaque	0.121**	0.121**		0.044	0.026/0.202
Dentine caries	<	Plaque	0.364**	0.364**		0.038	0.286/0.433
Dental pain	<	Plaque	0.057**		0.057**	0.018	0.023/0.097
School performance	<	Plaque	-0.044*		-0.044*	0.020	-0.085/-0.006
Dental pain	<	Dentine caries	0.156**	0.156**		0.047	0.064/0.248
School performance	<	Dentine caries	-0.155**	-0.155**		0.052	-0.250/-0.047
School performance	<	Treated teeth	0.100*	0.100*		0.037	0.029/0.171
School performance	<	Parent's characteristics	0.213**	0.213**		0.083	0.058/0.379

Note: **P<0.01, *P<0.05, β = boot strapped standardised estimate, SE = standard error, Bias-corrected 95 % confidence interval = BC 95% CI

Table 57. The specific indirect paths and mediators in the parsimonious model

Indirect Path	β	Unstandardized estimates	95% CI	P-Value
Child's OHPs < CSES <bses< td=""><td>-0.187**</td><td>-0.229</td><td>-0.371/-0.110</td><td>0.008</td></bses<>	-0.187**	-0.229	-0.371/-0.110	0.008
Parent's characteristics < CSES < BSES	0.457**	2. <mark>1</mark> 81	0.687/4936	0.003
School performance < Parent's characteristics < CSES <bses< td=""><td>0.457**</td><td>1.707</td><td>0.451/4.095</td><td>0.003</td></bses<>	0.457**	1.707	0.451/4.095	0.003
School performance < Parent's characteristics < CSES	0.099**	1.551	0.0435/0.003	0.007
Dental pain <dentine bses<="" caries<="" td=""><td>-0.039***</td><td>-0.052</td><td>-0.097/-0.024</td><td>0.001</td></dentine>	-0.039***	-0.052	-0.097/-0.024	0.001
School performance < Dentine caries <bses< td=""><td>0.039**</td><td>0.675</td><td>0.278/1.305</td><td>0.004</td></bses<>	0.039**	0.675	0.278/1.305	0.004
School performance < Parent's characteristics <cses< td=""><td>0.099**</td><td>1.551</td><td>0.435/3.500</td><td>0.003</td></cses<>	0.099**	1.551	0.435/3.500	0.003
School performance < Treated teeth <plaque< td=""><td>0.012**</td><td>0.002</td><td>0.001/0.004</td><td>0.008</td></plaque<>	0.012**	0.002	0.001/0.004	0.008
Dental pain < Dentine caries < Plaque	0.057**	0.001	0.000/0.001	0.001
School performance <dentine <plaque<="" caries="" td=""><td>-0.056**</td><td>-0.010</td><td>-0.016/-0.004</td><td>0.004</td></dentine>	-0.056**	-0.010	-0.016/-0.004	0.004

Note: **P<0.01, *P<0.05, β = boot strapped standardised estimate, Bias-corrected 95 % confidence interval = BC 95% CI

4.6.2.1 Early and current life factors linked to OHCs

This section presents the results of testing different life course models (critical, accumulation) to analyse the early and current social, biological, behavioural, and parental conditions associated with developing OHCs among children aged 7-8 years (Objective 2).

4.6.2.1.1 Critical period hypothesis

Testing the critical period model revealed that only BSES predicted child's OHCs among the critical period factors (Figure 25).

On the one hand, a high family's SES at birth was directly associated with a low level of dentine caries when the child was 7-8 years old (β = -0.248**, SE=0.051, BC 95% CI=-0.344/-0.153), and indirectly associated with less dental pain (β = -0.039**, SE=0.051, BC 95% CI=-0.074/-0.016). However, BSES was not associated with enamel caries, pufa, treated teeth, dental plaque, birth outcomes and EDD (Figure 25a).

On the other hand, none of the paths estimated from birth outcomes to OHCs revealed significant associations (the p-value for the path estimates was > 0.05) (Figure 25b; Appendix 21).

Finally, having high EDD in primary teeth was directly associated with having a low level of dentine caries (β = -0.086^{**}, SE= 0.028, BC 95% CI= -0.137/-0.026) and indirectly linked with a low level of dental pain when the child was 7-8 years old (β = - 0.013^{**}, SE=0.006, BC 95% CI=-0.028/-0.004). In other words, children with EDD were less likely to have dentine caries and, therefore, less likely to experience dental pain (Figure 25d).



Figure 25. The critical period model: the path estimates between BSES (a), birth outcomes (b), EDD (c) and child's OHCs independently and dependently from BSES, and mediation effect of dental caries (d)

4.6.2.1.2 Accumulation model with the chain of risks

The chain of risk model results with additive and trigger effects revealed that OHCs were unnecessary to be a subsequent chain of risks driven by SES (Figure 26).

To clarify, the paths between BSES and maternal CHCs, the path between birth outcomes and EDD, and the path between EDD and plaque were non-significant (Appendix 21). The only significant paths in the chain were between maternal CHCs and a child's birth outcomes (β = 0.161**, SE= 0.063, BC 95% CI= 0.048/0.291) (Figure 26a). In other words, mothers without chronic conditions during pregnancy were more likely to have a baby with a normal weight of ≥2500g, normal gestational age of ≥37 weeks, and ≥second birth order.

This indicated that each risk in the proposed chain did not trigger or add to another (Figure 26a). Rather, some risks have an independent effect on developing OHCs, especially dental caries. To illustrate, the results revealed that BSES and EDD directly affect a child's level of dentine dental caries, as mentioned earlier. Further, dental plaque has a direct positive association with dentine caries (β =0.364**, SE= 0.038, BC 95% CI= 0.286/0.433), enamel caries (β =0.094*, SE=0.044, BC 95% CI= 0.008/0.180), and treated teeth (β =0.121**, SE= 0.044, BC 95% CI= 0.026/0.202), and indirectly related to dental pain through dentine caries (β =0.057**, SE=0.018, BC 95% CI= 0.023/0.097) (Figure 26b).

Similarly, when testing the influence of BSES in developing risk behaviours that would lead to OHCs, the model was not fully supported (Figure 26c). To illustrate, the results revealed that high BSES was directly associated with parents' supervised tooth brushing when the child was three years old (β = 0.175**, SE= 0.058, BC 95% CI=

0.048/0.278), but BSES was not associated with the child's exclusive breastfeeding. Further, the paths between those behaviours and OHCs were not significant.



b The mediation effect of dental caries on the relation of plaque and dental pain



Figure 26. Accumulation of risk models

4.6.2.1.3 Association between BSES and CSES

Figure 27 shows the association between BSES and CSES and the possible behavioural and parental pathways that link CSES and OHCs among children.

The results indicated that high BSES predicted high current SES (β =0.986**, SE= 0.012, BC 95% CI= 0.957/1.006) (Figure 27). In addition, CSES mediated the indirect association between BSES and the child's OHPs (β =-0.187**, SE= 0.062, BC 95% CI=-.313/-0.074) (Figure 26a), as well as the indirect link between BSES and parent's characteristics (β = 0.457*, SE= 0.144, BC 95% CI=0.200/0.724) (Figure 28b). Thus, it means that high BSES and CSES associated with children's inadequate OHPs (brushing, flossing, use of mouth wash frequencies = ≤ 2 times/day, and sugar= > two times/day) and supportive parent's characteristics (parents with authoritative parenting style, high mother's academic involvement).

Analysing the behavioural pathway between family current SES and OHCs, revealed that the family's CSES directly predicted the child's OHPs (β = -0.190**, SE= 0.062, BC 95% CI=-0.317/-0.076), without the mediation effect of mother's OHPs frequencies. However, the result did not find an association between child's OHPs and their current dental plaque or OHCs (Figure 27, Appendix 21). Interestingly, adequate mother's OHPs was directly associated with adequate child's OHPs (β =0.778**, SE= 0.082 0.575/0.924).

Additionally, when testing the parental characteristics pathway between the family's CSES and child's OHCs, the models found that CSES directly associated with the parent's characteristics (β =0.464*, SE= 0.146, BC 95% CI= 0.202/0.727); however, parents' characteristics were not related to the child's OHPs or their OHCs (Appendix 21).



Figure 27. The association between BSES and CSES



Figure 28. The mediation effects of CSES between BSES and child's OHPs (a) and between BSES and parent's characteristics (b)

4.6.2.2 Association between OHCs and school performance

This section presents the results of examining the association between OHCs and school performance (Objective 3). School attendance was removed as no factors were related to it (Appendix 21).

Testing the direct relations between OHCs and school performance revealed that dentine dental caries was directly associated with low school performance (β = - 0.155**, SE= 0.052, BC 95% CI= -0.250/-0.047). Treated teeth was also directly associated with high school performance (β =0.100*, SE= 0.037, BC 95% CI= 0.029/0.171) (Figure 29).



Figure 29. The relation between OHCs and school performance

Further, testing mediations revealed that dental plaque was indirectly associated with poor school performance via dentine caries and treated teeth (β =-0.044**, SE= 0.020, BC 95% CI= -0.085/-0.006) (Figure 30).



Figure 30. The mediation effect of treated teeth and dentine caries on the link between plaque and school performance

On the other hand, no significant associations were found between pufa, enamel caries, dental pain, and school performance. Thus, pufa, and dental pain were not significant mediators between OHCs and school performance. Although the results revealed that if dentine caries increased among children, they were more likely to experience dental pain (β =0.156**, SE= 0.047, BC 95% CI= 0.064/0.248); however, such an association was not related to children's school performance.

Similarly, school attendance was not a significant mediator between OHCs and school performance. The path between OHCs and school attendance was neither significant nor between school attendance and school performance (Appendix 21).

Finally, testing the mediation effect of OHCs between SES and school performance indicated that dentine caries mediated the positive association between BSES and

children's school performance. Thus, high BSES was indirectly associated with higher school performance via dentine caries (0.039**, 95% CI= 0.278/1.305) (Figure 31).



Figure 31. The mediation effect of dental caries on the link between BSES and school performance

4.6.2.3 The association between other life course factors and school performance

Besides the link with BSES via dentine caries, the results also revealed high CSES linked with high school performance via parent's characteristics (0.099^{**} , SE= 0.067, BC 95% CI= 0.018/0.257) (Figure 32) (Objective 4). In the same time parent's characteristics linked directly with better children's school performance (β = 0.213^{**} , SE= 0.083, BC 95% CI= 0.058/0.379). Thus, BSES and CSES predicted parents with supportive characteristics like authoritative parenting styles and mothers with high academic involvement. In return, parents with supportive characteristics were associated with high school performance among children (Figure 33).



Figure 32. The mediation effect of parent's characteristics in the link between CSES and school performance



Figure 33. The association between social and parental factors with children school performance

4.6.3 Summary of the results

To summarise, a total of 466 children aged 7-8 years old, along with their parents, were enrolled in this study in September 2019 (response rate = 98%). Most participants were Bahraini (72.5), some (22.5%) were from other Arab countries, and 5% were Asian. Half of the families received a moderate monthly income (400-600 BD). Further, 74.5% of mothers were unemployed, while 98% of the fathers were employed. Only 14.6% of children lived with single or extended families, while the remaining children lived with both parents.

Regarding OHCs, dental caries and clinical consequences of dental caries were generally more prevalent in the primary teeth than in the permanent ones. The prevalence of dental caries and PUFA/pufa were 93.4% and 26.5%, respectively. Unlike that, EDD was more prevalent in permanent teeth, with a prevalence of 17.0%. The majority of children (67.6%) reported at least one impact of OH on their quality of life.

Testing the association of OHCs with school performance and school attendance demonstrated that social factors in early life were significant predictors of poor OH and low school performance among children aged 7-8 years old. Also, the results revealed that dentine dental caries was directly associated with low school performance. Conversely, treated teeth and parents' characteristics were directly associated with high school performance.

Chapter 5. Discussion

5.1 Introduction

This study examined the associations between OHCs and school performance and attendance among schoolchildren aged 7-8 years old in Muharrag, KoB, using OH's Heilmann et al. (2015) life course framework. The four main objectives of the present research were: firstly, to describe the prevalence of OHCs in Muharrag city among children attending good, rated schools; secondly, to analyse the relevant social, biological, behavioural, and parental pathways that linked early and current life exposures with children's current OHCs through testing different life course models (critical and accumulation models); thirdly, to examine the association of OHCs with school performance and attendance among schoolchildren, including the direct and indirect links and possible mediators; fourthly, to explore the possible early and current life's course social, biological, behavioural and parental factors that might associate with children school outcomes besides OHCs. It was hypothesised that low SES at birth and age 7-8 would shape the children's biological, behavioural, parental, and psychosocial resources in such a way as to promote the development of OHCs; hence they would exhibit low school performance and attendance. Conversely, children who experienced high SES at birth at age 7-8 would show little or no evidence of OHCs and would exhibit better school attendance and performance than those living within a low SES family.

5.2 Summary of the key findings

The prevalence of enamel and dentine caries and their clinical consequences were 93.4% and 25.7%, respectively. The second most prevalent condition was enamel defect, with 17.2%.

Regarding the early life course factors linked with OHCs, the only significant factor was low SES at birth. High BSES was directly associated with a low level of dentine caries and indirectly with a low level of dental pain among children in Grade 2 in Muharraq. At age 7-8, a high level of the dental plaque was an important factor associated with a high level of enamel and dentine caries, dental pain, and treated teeth. On the other hand, none of the early and current life courses behavioural and parental factors was associated with OHCs.

It was found that dentine dental caries was negatively and directly linked with children's school performance, while dental plaque was negatively and indirectly linked with school performance. Conversely, treated teeth were directly associated with high school performance. Parents' SES was also significantly related to children's school performance via dentine caries and parental factors. On the other hand, none of the family SES and parental nor OHCs were associated with children's school attendance.

5.3 Discussion of Objective 1

To describe the prevalence of OHCs among 7-8 years old schoolchildren of the good, rated schools in the city of Muharraq

5.3.1 Prevalence of dental caries

Although there is widespread published research on dental caries among diverse populations and age groups worldwide, there is no published data from KoB. This study is the first to comprehensively assess dental caries through clinical examination by ICDAS, their sequela (PUFA/pufa) and consider their impact on children's quality of life (measured using the POQL) in KoB.

ICDAS was used because it is a validated measure of caries experience. The system can detect and assess the severity and progress of caries lesions ranging from the first visible signs of dental caries in enamel to an excessive cavity with visible dentine (Ismail et al., 2007). The system is compatible with life course theories, as it acknowledges that the dental caries process is not limited to the presence or absence of cavitated lesions; instead, it is the result of different developmental stages. Understanding these stages over time can provide valuable information not only for caries diagnosis, prognosis and the clinical management of dental caries ranging from preventive to operative care but also regarding the determinants of caries at different stages of life and proposing the best timing of interventions necessary to reverse the disease process (Pitts and Ekstrand, 2013).

The Kappa index for the ICDAS examination was substantial in this study; the weighted and unweighted kappa values were>0.65, indicating an acceptable level of agreement between and within examiners, similar to other studies (Braga et al., 2009, Honkala et al., 2011). This indicates that ICDAS presented good reproducibility and accuracy in detecting dental caries (Jablonski-Momeni et al., 2008, Ekstrand et al., 2018).

The study found that the prevalence of ICDAS code >1 was 93.4%, greater in the primary dentition (93.0%) than in the permanent dentition (45.8%). The means for enamel caries were 1.5 in the primary dentition and 1.2 in the permanent dentition, while for dentine caries, the means were 16.3 and 0.43, respectively among the total sample. Only 6.5% of the examined children might be a caries-free, however, because the exclusion of Code 1, this should be interpreted with caution. The prevalence of clinical consequences of untreated dental caries (PUFA/pufa \geq 1) was 25.7. This prevalence confirmed the high incidence of severe caries in this world region. Among the limited number of studies that have measured dental caries in primary teeth using ICDAS, the prevalence shown in this study is unusually high. For example, in India, Arangannal et al. (2016) found that the prevalence of dental caries among 2,796 schoolchildren aged 5-15 was 68.8%, and almost 65% among those aged 7-8 years. A study conducted in Brazil using ICDAS≥3 found the prevalence to be 47.9% among 836 children aged 3-5 years old (Neves et al., 2016). Nevertheless, a similar prevalence of high caries has been reported in Estonia (Honkala et al., 2011) and Brazil (Mendes et al., 2010) using the same measure.

Honkala et al. (2011) argued that the total caries experience is usually higher in mixed dentition since the primary teeth are exposed to risk factors like sugar for a longer period. Additionally, the longer exposure time can explain why the primary dentition has a higher rate of dentine caries than the permanent dentition. Some might also argue that the prevalence might be higher because of the inclusion of the active and non-active lesions in contrast to DMFT, which only account for active dentine lesions (Potlia et al., 2016).

The downside of using the ICDAS system was that it was difficult to compare the prevalence of caries among countries in the same region since most studies have used the DMF/dmf index, following the WHO criteria. Even though the ICDAS system has been adopted by many recently, comparing results was also challenging because there is inconsistency in reporting and presenting the dental caries experience (EISalhy et al., 2019). A current systematic review found that among 126 papers reporting dental caries findings using ICDAS, 44 different measuring methods were used. For example, total counts of combined caries scores for surfaces or teeth, total counts of every score, and central tendency and dispersion measures. Despite the difficulty in comparing different epidemiological studies, this range of options demonstrates the system's flexibility, enabling it to accommodate the needs of different ICDAS researchers.

This study counted the total ICDAS measure to summarise caries progression or regression among the population (ElSalhy et al., 2019). Additionally, it categorised dental caries into enamel and dentine to examine how different types of caries can influence children's school outcomes and to help in determining categories of treatment needed (ElSalhy et al., 2019). The flexibility of the system facilitated comparisons with WHO-reported data. The ICDAS II established code four or three as an equivalence cut-off point between the ICDAS and WHO indices (Iranzo-Cortés et al., 2013). Thus, the system exhibits high adaptability to different uses in epidemiologic research.

To compare caries prevalence among Muharraq schoolchildren with other regions of the world, in this study, ICDAS scores 4-6 were converted into dmfs/DMFS (Honkala et al., 2011). The prevalence of dental caries using dmfs/DMFS was 89.7% in primary teeth and 45.8% in permanent teeth, whereas the mean of dmfs/DMFS was 6.3 and 0.3, respectively. Thus Muharraq city is falling behind in meeting the global goal of

2020 to reduce the "proportion of children and adolescents who have dental caries experience in their primary or permanent teeth" (Hobdell et al., 2003).

Comparing the results of this study with those of the last OH survey (2011) in Muharraq, there is a trend toward a greater prevalence of dental caries despite the country's efforts to improve OH (Naseeb, 2016). In 2011 the prevalence of dental caries in primary teeth among children aged 6 in Muharraq was 82%, with a mean equal to 4.5. The corresponding mean in the current study shows a 1.8 increase, at 6.3. This, however, could be attributed to the difference between age group, as the sample in this study were two years old younger than the previous OH survey. Although caution must be taken when comparing the prevalence of dental caries with other areas, our sample was restricted to children attending good quality schools, which might limit the generalisability of the study's findings to other schools.

In contrast to the studies in Sections 2.3 and 2.4 of the literature reviews, Muharraq exhibited the highest prevalence of dentine dental caries in the primary dentition. The reported prevalence and dmft means are as follows, 60% (3.5); 66.9% (3.6) and 39% (1.6) in Brazil; China; and Uganda, respectively (Peres et al., 2005, Sun et al., 2017, Birungi et al., 2017); in Scotland, only dmfs was recorded (3.6) (Bernabé et al., 2017); and in Sudan, only a low dmft \geq four prevalence was recorded (51%) (Osman and Alsayed, 2015).

Bahrain, Qatar, Saudi Arabia, Kuwait, the United Arab Emirates (UAE), and Oman together, these countries constitute the GCC (At-Twaijri, 1990). The GCC population shares similar SES, culture, traditions, religion, and language; these similarities were one of the bases for creating the GCC (At-Twaijri, 1990). However, the prevalence and the means of dmft of caries in the primary dentition in Muharraq showed a slightly

higher trend than in other cities in the GCC countries. For example, Kowash et al. (2017) reported that the prevalence and the mean of dentine dental caries were 74.1% and 3.1, respectively, in the primary dentition among 540 children aged five years old in Ras Al Khaimah in the United Arab of Emirates (UAE). In contrast, Farooqi et al. (2015) reported that the prevalence and the mean of dental caries were 78% and 3.7, respectively, among 397 children aged 6-9 years old in Dammam city in Saudi Arabia. A similarly high prevalence and mean of dmft in the primary dentition was observed in some Asian countries, such as 90% (6.8), 84.5%, and 77.7% (6.2) in Indonesia, China, and Thailand, respectively (Adiatman et al., 2016, Li et al., 2017, Krisdapong et al., 2014).

In contrast to findings from other Mediterranean regions and African countries, the caries level among primary teeth was found to be highest in Muharraq, the prevalence and means recorded in other countries being as follows: 62% (4.4); 70%(4.2); 76.4% (3.3); 52.5% (2.3); and 20.2% (0.9), in Yemen, Syria, Jordan, Sudan, and Tanzania, respectively (Alshoraim et al., 2018, Qadri et al., 2012, Rajab et al., 2014, Elidrissi and Naidoo, 2016, Mwakayoka et al., 2017), respectively.

The figures for Arab children and the Middle East were considerably greater than those recorded for children in the Western region. The mean dmft in Germany, the UK, Scotland, and Sweden, was 2.9, 0.8, 1.3, and 2.0, respectively (Lee et al., 2021, Jones et al., 2017, Koch et al., 2017). However, it was close to the dft mean in the US (4.3) (Dye et al., 2015).

The data above shows a geographically disproportional distribution of dental caries in the primary dentition and that the rate of dental caries was higher in Muharraq and GCC than in the developed countries and the African region. The trends in dental

caries could be attributed to differences in sample sizes, varying age ranges of the participating children and geographical location. However, the decrease in dental caries among the developed countries was judged to be due to progress in prevention measures and high access to dental care, despite the inconsistent results showing that dental caries continues to be more prominent among the disadvantaged population in many of these countries (Kassebaum et al., 2015). In contrast, it has been observed that developing countries, including KoB, have recently started to show an upward trend in caries prevalence due to an increase in sugar consumption, insufficient fluorides exposures and limited access to OH services (Kassebaum et al., 2015, Lagerweij and van Loveren, 2015).

The WHO data indicates that dental caries has been declining worldwide, with the greatest reduction observed in the high- and middle-income countries. In contrast, the decline is less explicit in low-income countries (Lagerweij and van Loveren, 2015). However, the WHO lists a few exceptions where caries prevalence has risen, such as the Gambia and Saudi Arabia (WHO, 2013). In this sense, KoB and the whole GCC need to be included in this list of exceptions, and further investigation is needed because the GCC countries are considered high-income (World Bank, 2020) but experience a high rate of dental caries.

KoB is thus considered by the World Bank (2020) to be a high-income country. The gross domestic product (GDP) per capita for KoB was about 22,402 in 2020, which places KoB in the top 50. The HDI value for 2019 was 0.852. This value ranks Bahrain in the 42nd position out of 189 countries and as a very high human development area in the UNDP's. Thus, the assumption that high-income countries have a better standard of OH is not met in KoB (Watt et al., 2019). Similarly, Korea and Hong Kong have very high HDI scores, but the prevalence of caries is high (>50%) compared with

Europe (Chen et al., 2019). It might be that the HDI reflects the development level of the whole country and that the SES of a specific city or district cannot be reflected in this index.

Other reasons for variations in findings may be due to marked differences in health services and sugar consumption, as indicated above. Oral health services in Bahrain are incorporated within primary health care, through 27 primary health centres, under the MOH umbrella. Dentists, dental hygienists, and dental assistants are the main dental workforce in MOH. The government has funded free OH services for all (MOH-KoB, 2014), while non-Bahraini pay a nominal amount, which the Government greatly subsidies. Since 1986, there has been an OH survey in KoB every five years, collaborating with the WHO. However, the survey tends to measure only disease prevalence; neither determinants nor impact of diseases on people's lives have been measured. Thus, unfortunately, it leads to curative dental care rather than preventive.

Identifying social inequality in Bahrain will not help eliminate oral diseases; instead, it will help determine policies targeting the disadvantaged group. Also, there was no data to explain the relation between OH and SES in Bahrain before this study. An OH survey in 2005 showed that children attending private schools demonstrate less dental caries than those attending government-run schools, but no further investigations have been taken.

The available services and workforces in MOH are struggling to meet the growing population's needs (MOH-KoB, 2014). The Organization for Economic Cooperation and Development (OECD) suggested a 20,000 average population per health centre as a benchmark for KoB. Thus, although the number of health centres has increased to 27 since 2006, most health services are serving well above this number (MOH-KoB,

2014). As suggested by Health Information Directorate, the acceptable dentist to population ratio for KoB was 1/5000 (Ahmed et al., 2000). However, only 110 dentists with a few specialities and 60 dental hygienists are working with the MOH currently (MOH-KoB, 2019). The numbers are below the suggested ratio; as a result, accessibility and availability are hindered. Moreover, the MOH operates a salary-based remuneration system with no incentives, leading to early retirement and the movement of many specialists to the private sector, making many services unaffordable for a large section of the population, especially non-Bahrainis.

With regard to OH promotion strategies implemented by dental hygiene services, in KoB, dental hygienists used a targeted prevention approach, applying fissure sealant and fluoride varnish to all government-school children. Although this would appear to be a practical method of saving resources for high-risk individuals (Rose et al., 2008), its effectiveness in reducing dental caries is not evident given the high prevalence of caries among KoB schoolchildren, nor has the impact of the fluoride and fissure sealant programmes ever been evaluated to determine their implications or efficiency. Another strategy used by dental hygienists in Bahrain is to promote positive oral health behaviours through OH education, distributing toothpaste and toothbrushes among government school pupils from Grade 1. However, there is little instruction about using the correct amount of fluoride in toothpaste or the concept of 'brush and spit' rather than rinsing the mouth with water after brushing. It was found that 92% of the children in the current study rinse out the toothpaste with water; thus, most of the fluoride, if used, is being diluted and not used effectively.

Another reason why clinical preventive interventions such as topical fluorides and fissure sealant are ineffective in controlling caries in Bahraini children, despite their proving highly effective in other regions (Li et al., 2020), is maybe high sugar

consumption. Although children spend most of their time at school, there are no policies to reduce sugary snacks at schools — the schools continue to serve children sugary drinks and snacks, which counteract OH care. This study found that 71.9% of the schoolchildren surveyed consume \geq 1 sugary snack daily. Similarly, a previous study on KoB among schoolchildren found the mean daily intake of total sugars to be 101.3 g and 89.1 g for boys and girls, respectively (Gharib and Rasheed, 2011). This value is considered extremely high compared to the recommended intake of sugar (< 5% of the total energy intake) (WHO, 2015). The evidence suggests that caries continues to progress if the individual's total energy intake of free sugars is above 10% (Moynihan and Kelly, 2014). Even in the presence of an optimal level of fluoride, evidence showed that free sugar consumption as low as 2–3% of total energy intake is enough to increase the risk of caries (Sheiham and James, 2015).

Although this study did not capture the link between sugar and the rate of dental caries, life course evidence confirms the accumulative effects of sugar intake on caries increment (Broadbent et al., 2008); and, specifically, that progression of dental caries from age 6 to age 18 is constantly and positively linked with high patterns of sugar consumption throughout the life course (Peres et al., 2016). In order to understand this association in KoB, longitudinal studies would be needed.

Additionally, this study showed that families with high SES consumed more sugar than families with low SES. This is because the growth of the KoB economy is accompanied by a substantial shift from traditional healthy foods to diets rich in fat, sugar and red meat (Musaiger, 2000). The sugar consumption rate in KoB among children is alarming, putting them at risk of future dental caries and diabetes, obesity, and other chronic conditions (Sheiham and Watt, 2000). Concomitantly, this study found that 31% of the schoolchildren were overweight or obese across both genders. Thus,

adopting a common risk approach in early life can help to prevent most of these conditions.

Although dental caries levels showed a declined trend in the primary dentition among children in different parts of the world, caries being a rapidly progressive condition of childhood has shifted—the concern now related to their slow progression and cumulative nature that manifest in older age (Whelton, 2004). This is illustrated clearly through the Dunedin study, wherein a cohort of 1,037 was tracked from 5 up to the age of 38 (Broadbent et al., 2013). Roughly 40% of this cohort were classified as low caries, 45% as moderate caries, and 15% as high caries. Interestingly, the annual increase in caries remained steady throughout the 32 years. This indicates that the risk of caries is not only a concern among children but that it continues throughout life.

Therefore, it can be argued that KoB, with its high level of caries in children's primary teeth, should be considerably more proactive in its prevention strategies, especially given the evidence linking caries in primary teeth with caries in permanent teeth and later life risk as a dose-response relationship (Peres et al., 2009). Accordingly, it needs to be recognised that the preventive efforts in KoB should be not only more effective and more closely monitored than at present, but that they should begin earlier, at pre-school age, in a kindergarten setting and with parental participation.

5.3.2 Prevalence of EDD

Using the modified EDD index (Clarkson and O'Mullane, 1989), the prevalence of EDD was found to be greater in permanent teeth (17%) than in primary teeth (1.5%) in this study, in line with other research (Seow et al., 2011, Slayton et al., 2001). This may be because primary teeth presented with more caries experience (89.7%) or had been exfoliated already.

Compared to other evidence using the modified EDD index, in the same age group (7-8 years), the present study indicated a lower prevalence than studies in Australia (58%) (Seow et al., 2011) and Spain (52%) (Robles et al., 2013), whereas it was higher than a study conducted in Libya (2.9%) (Fteita et al., 2006).

Some studies examined the prevalence of EDD in terms of Molar Incisor Hypomineralization (MIH) using the criteria of the European Academy of Paediatric Dentistry (EAPD), whereby only first molars and incisors are examined (Weerheijm et al., 2003).

Interestingly, the present findings revealed that the first molars and incisors in permanent dentition were the most affected sites by EDD (Section 4.3.11.5), thus facilitating comparisons with studies that examined MIH using the EAPD criteria. Accordingly, at 14.4%, the prevalence of MIH in the current study was similar to the global mean (13.1% to 14.2%) (Schwendicke et al., 2018) but lower than results reported prevalence of 20%, 20.2%, 19.8%, and 17.7% in Sudan (Abdalla et al., 2021), Iran (Ahmadi et al., 2012), Japan (Saitoh et al., 2018) and Nigeria (Oyedele et al., 2015), respectively. Higher prevalence of MIH was reported to be 25.1%, 26.7%, 27.2%, and 33.2% in Saudi Arabia (Hazim et al., 2018), in Lebanon (Elzein et al., 2020), in Dubai (Hussain et al., 2018), and in Sweden (Jälevik et al., 2018), respectively. Egypt, in contrast, showed a very low prevalence (2.3%) (Saber et al., 2018).

In the present study, demarcated opacities were the most dominant type amongst the affected teeth. This result was compatible with Dubai, Sweden, and Sudan (Hussain et al., 2018, Jälevik et al., 2018, Abdalla et al., 2021). In addition, this study revealed that the maxillary teeth were more affected by MIH than the mandibular teeth, which

is consistent with the results from Dubai and Sudan (Hussain et al., 2018, Abdalla et al., 2021), but not with those of Lebanon and Saudi Arabia (Elzein et al., 2020, Hazim et al., 2018) where a similar pattern of MIH was found in both upper and lower jaws. The present study also found that MIH affected the mandibular and maxillary molars more than the incisors, which mirrored the findings from Dubai (Hussain et al., 2018) but contradicted those for Sudan (Abdalla et al., 2021).

The association between environmental, biological factors and EDD was not supported in the current study. Although the causes of EDD and MIH are not very clear, it is generally agreed that EDD is caused by various factors, including genetic and environmental such as malnutrition and exposure to chemicals like fluoride (Heilmann et al., 2015). What is clear is that EDD provides valuable information regarding the timing of teeth development and can be used as an indicator for early life disadvantages (Heilmann et al., 2015). Another relevant point is that EDD has been linked to dental caries, and it has been used in life course research as an early life stressor in the chain of risk model (Peres et al., 2005). Thus, an early diagnosis of EDD might aid in the prevention of dental caries in early and later life.

5.4 Discussion of Objective 2

To analyse the social, biological, behavioural, and parental pathways that link early and current life exposures with children's current OHCs by testing the critical and the accumulation life course models considering social determinants of OH.

5.4.1 Early and current life factors linked to OHCs

The study analysed the relevant social, biological, behavioural, and parental pathways that linked early and current life exposures with children's current OHCs by testing the critical and accumulation life course models considering social determinants of OH.

Regarding the critical period model, the study indicated that socioeconomic factors were more important than biological and behavioural factors in determining the children's current OH. These results do not support the argument that dental caries experience is "programmed" in early life, as is the case for the development of chronic conditions (Barker, 1998). In accordance with previous studies (Peres et al., 2005, Sun et al., 2017, Zhou et al., 2012, Zhou et al., 2011), but not with Nicolau et al. (2003), the study found no association between LBW, PB, and dental caries. This result mirrors the finding of a current meta-analysis (Occhi-Alexandre et al., 2020), which noted that 28 out of 36 observational studies showed no association between LBW, PB, and dental caries. Bernabé et al. (2017) argued that the inconsistent results might be because the association was assessed within a cross-sectional study rather than a longitudinal study; thus, the accumulation of the diseases over time could not be measured. However, Zhou et al. (2011, 2012) found no association between LBW and caries in their cross-sectional and longitudinal studies. In the current study, the low prevalence of LBW (11.2%) and PB (10.9%) children may hinder the detection of a statistical association between LBW, preterm and dental caries.

Heilmann et al. (2015) suggested that low birth weight is linked to childhood caries through EDD. The current study did not support this hypothesis, and most of the studies previously mentioned in Section 2.3 did not evaluate this pathway. The reason might be related to the use of regression analysis, which only allows the study of associations between blocks of independent factors and one dependent factor; the interrelationships between the set of independent variables cannot be assessed (Berndt and Williams, 2013). Further, Peres et al. (2005) argued that the stunting condition is more related to the severity of dental caries than birth weight because it

reflects long-term cumulative nutrition deficiency, which might affect the formation of the organic dental matrix EDD, then caries.

In this study, such a pathway was evaluated using SEM, but the association was still not supported. This could be explained by more substantial evidence for the relation between impaired foetal growth and EDD in infants with very LBW (<1500 grams) (Jacobsen et al., 2014). Thus, it is possible that such an association existed but was masked because a very LBW was not recorded in this study.

Although it is now evident that EDD increases susceptibility to caries (Costa et al., 2017), unlike Zhou et al. (2011; 2012), the study found that dentine caries was reduced if there were EDD present. This may be because the prevalence of EDD in primary teeth was only 1.5%, such that the association could not be captured. Furthermore, the most prevalent EDD found in this study was demarcated opacities, while evidence suggests that hypoplasia is the EDD most frequently associated with early childhood caries (ECC) (Matee et al., 1994). In this study, conditions were measured at only one point in the child's life; thus, the EDD detected is expected to contribute to an increased risk of future caries.

Although mothers' exclusive breastfeeding has previously been found to be socially patterned and associated with the level of dental caries among children (Birungi et al., 2017), the current study did not find this to be the case; this result is in accordance with (Bernabé et al., 2017). The reason that exclusive breastfeeding was not socially patterned might be that 74% of the total population of Bahrain is Muslim (Information & eGovernment Authority-KoB, 2020a), and in Islam, women are advised to breastfeed until their child is two years old. Thus, this behaviour seems to be influenced by religion or maybe beliefs and cultural norms (Osman et al., 2009). The

association between breastfeeding and dental caries is not agreed upon in the literature. Some research suggested that breast milk is cariogenic (Bowen and Lawrence, 2005), while other studies suggested that breastfeeding can positively influence oral health outcomes, including a lower occurrence of malocclusion (Peres et al., 2015) and dental caries (Cui et al., 2017). In this study, the effect of breastfeeding might be masked by the effect of visible dental plaque and SES.

The influence of early life SES on dental caries in primary teeth has been well established from life course framework studies (Abreu et al., 2015). The results of this study further confirm this association. However, most previous studies focused on one indicator of SES, either occupation or education, or considered each parent's SES separately. For example, Zhou et al. (2011) found that maternal SES-related factors at a child's birth are more important than paternal SES-related factors affecting caries in later life. In contrast, Peres et al. (2005) found the father's level of education at the child's birth to be a crucial predictor of dental caries in 6-year-old children. Other studies (Peres et al., 2003; Zhou et al., 2012) concluded that family income in early life was associated with a child's dental caries throughout their life course. The differences in assessing which variables can determine the family's SES position might be related to the analytical method. Most of the studies mentioned above used regression analysis, where the researcher cannot utilise multiple indicators to indicate SES; regression analysis does allow the inclusion of blocks of variables, but it cannot estimate the combined effects of observed independent variables into a construct (Berndt and Williams, 2013). In this study, SEM was used, and thus education, occupation, and income variables did not act alone as they would in regression analysis. Instead, in SEM, those variables were used as indicators of SES latent variables to predict other dependent variables. Thus, the CFA and the SEM models

found that the mother's and father's level of education, mother's employment status, and household income at the child's birth SES significantly represented the family's SES at the child's birth. However, the mother's level of education showed the highest loading of the SES construct, confirming that maternal education was a powerful marker for SES.

In their cross-sectional study of adolescents in Brazil, Nicolau et al. (2005) identified a clear pathway by which SES in early life and throughout the life course may influence the acquisition of risk behaviour related to dental caries, such as a high level of sugar consumption and a low level of daily tooth brushing. Lu et al. (2011) found the chain of risks model well suited to explain the relation between early SES, utilisation of dental services, and dental caries in adolescents in Hong Kong.

However, this study found that SES at birth and through life shaped some behaviours such as the child's toothbrushing being supervised, the child's current OH practices (brushing, flossing, mouth wash use, sugar consumption), and the parents' parenting styles. However, contrary to the hypothesis, none of these behavioural and psychosocial factors accounted for the SES disparities in caries experience.

The findings show that OHPs had no significant association with OHCs, which is in line with Zhou et al.'s (2011) conclusions, contrary to those of Wong et al. (2012). While a diet rich in extrinsic sugars has been shown to be cariogenic for children (Marshall et al., 2005), different viewpoints regarding the association between the consumption of extrinsic sugars and dental caries have been reported. In a systematic review, Burt et al. (2001) suggested that the relation between sugar and dental caries was less affinitive than previously because of the prevalent use of fluoridated toothpaste. On the other hand, Habibian et al. (2001) recommended that the

association depends on consumption's total frequency. Compared with more than four times (Nicolau et al., 2003) and seven times per day (Habibian et al., 2001), the current research found no association between the intake of sweets more than twice a day and dental caries. In this study, 74.5% of parents reported that their children brushed once and twice, 22% reported that their children brushed more than twice a day, and 32% reported that they didn't supervise their children's toothbrushing habits. However, the mean visible plaque among the children was 65.48%, indicating that the children were not brushing their teeth effectively. Accordingly, the relation between OHPs and dental caries was challenging to detect, whereas the relation between visible plaque and enamel and dentine caries, treated teeth, and dental pain was significant. Although toothbrushing, if accurately adapted, can clean plaque, the toothbrushing habits of student in this research was not significantly related to OHCs. Thus, the result is perhaps due to the ineffective toothbrushing behaviours of these schoolchildren. The results, echoing Zhou et al. (2011), indicate that visible plaque is a more reliable indication of current toothbrushing habits.

Another relevant finding was that the family's SES at the time of the child's birth predicted its SES when the child reached middle childhood. This might support the cumulative risk model; however, using a cross-sectional design study enabled the examination of the link between family SES trajectories from birth to middle childhood and OHCs and OHPs. Interestingly, the study found that children's OHPs were shaped directly by their parents' current SES and mother's OHPs and indirectly by early-life SES. However, the most controversial finding was that if the parents' SES increased, the frequency of OHPs decreased, and sugar consumption increased; although this result is significant, it was found to have only a small effect (β = -0.19). This finding contradicts life course evidence (Peres et al., 2007) and refutes the common belief

that disadvantaged people are more prone to adopt unhealthy OH self-care (Sabbah et al., 2009). Most of the life course studies described earlier did not highlight the possible pathways between family SES and children's OHPs. However, evidence from a recent life course study among adults explained such pathways (Broadbent et al., 2016). The study suggested that early SES and parental OH beliefs were associated with the study participants' OH beliefs at different life stages.

Positive OH beliefs in early adulthood predict people's OHPs (Broadbent et al., 2016), and thus, the association between SES and OHPs could be mediated by OH beliefs. However, although OH beliefs were not examined in the current study, some of its findings suggest that parents might have been less aware of the importance of primary teeth. For example, 79.4% of parents reported that their children never visited the dentist or that they visited the dentist for treating primary teeth only in an emergency; this despite dental services for all citizens of KoB being provided free of charge. Eightyfive per cent of parents reported that they started brushing their children's teeth after 12 months of age. Moreover, although most parents (85%) in this study reported using toothpaste containing fluoride themselves, there was a 24% no-response rate for the question asking them what type of toothpaste their children used. This may indicate that for 24% of parents, this question was irrelevant since their children did not use toothpaste of any sort, or possibly that they did not know the answer, being unaware of the importance of using fluoridated toothpaste at this stage.

Given this, it, therefore, seems essential to explore how parents' OH knowledge and beliefs might shape children's OHCs and OHPs in this region, particularly in cases where SES have an adverse effect on those outcomes. It is worth mentioning here that KoB is considered a high-income country by the World Bank (2020), with a largely unexamined culture and set of beliefs, especially in relation to OH. Moreover, although

GCC countries, including the KoB, have experienced economic growth, it has been argued that the GCC continue to share some features of developing countries, such as poor low health literacy rates and health profiles (Alkhamis et al., 2014). This combination of economic growth and ignorance concerning health matters has resulted in many Arab countries experiencing increased consumption of refined sugars compared with other developing countries, mainly Africa (Petersen, 2004).

A few studies have used SEM to measure the association between SES, behavioural factors, and dental caries. One of these was conducted by Polk et al. (2010) and produced results in line with the current study. Polk et al.'s cross-sectional study of 530 9th- and 11th-grade Pennsylvania students, and their parents, was designed to determine differences in OHPs and the extent to which preventive measures accounted for the SES difference in caries experience across the sample. Using SEM, the study showed that lower SES was associated with severity and rate of dental caries. However, OHPs and preventive measures did explain the variations in dental caries defined by SES.

Different studies have examined the effect of the family's oral hygiene-related behaviours on children's OHPs and have found similar results (Özbek et al., 2015, Castilho et al., 2013, Hooley et al., 2012). In parallel with this, the current study found mothers' OHPs to be stronger predictors (β = 0.778) than parents' SES of children's OHPs. The results indicate that a higher frequency of mothers' OHPs was noticeably correlated with a higher frequency of toothbrushing, flossing, MW use, and a low frequency of sugar consumption among their children. This confirms the importance of socialisation to OH behaviours or the modelling activity whereby children replicate their parents' behaviours (Bandura, 1986). Parental modelling was found to be an effective measure to facilitate important behaviours among children, such as OHPs
(Blinkhorn, 1978) but has rarely been studied. The current study underlined that mothers could play a key role in establishing favourable OHPs among children and transmitting OHPs to them.

5.5 Discussion for Objective 3

To examine the association between OHCs and school performance and school attendance among schoolchildren, including the direct and indirect (mediated) pathways.

5.5.1 The association between OHCs and school performance

The results show that children with dentine caries and dental plaque were more likely to perform less well at school. The association between dentine caries and school performance were not mediated by dental pain, consequences of untreated dental caries, or school attendance. In contrast, dentine caries mediated the association between early-life SES and low school performance. Also, children whose caries were treated were more likely to have a higher school performance. No association was found between OHCs and school absenteeism.

The finding that children with more dentine caries were more likely to have a lower school performance is consistent with previous studies (Detty and Oza-Frank, 2014, Paula et al., 2016, Viviane and Sandra, 2003, Osman and Alsayed, 2015, Cunha et al., 2019, Garg et al., 2012). However, no association was found between dentine caries and school absence. These findings are in accordance with Osman & Alsayed (2015), who found that in a cross-sectional study of 385 Sudanese students aged from 5 to 15 years old, there was no significant association between absenteeism and

dental caries contrasting school performance. Similar results were also reported by Krisdapong et al. (2013) and Seirawan et al. (2012).

Despite the general agreement on the negative impact of dental caries on children's quality of life and school outcomes, especially school performance (Rebelo et al., 2019), the question remains: How do dental caries influence school outcomes? Different pathways have been anticipated to describe the association.

The first compelling argument is that the impact of a cavitated lesion, at least on quality of life, is slight when not linked with dental pain or clinical consequences of untreated dental caries (Leal et al., 2012). In parallel with this, Seirawan et al. (2012) argued that the presence of a cavitated lesion is not linked with a child's academic performance unless a toothache is also present. A similar conclusion was reached by Piovesan et al. (2012), who showed that children with higher scores for oral symptoms (including dental pain) – as measured by the CPQ₃₃ – were statistically associated with lower grades in the Portuguese skills examination, which suggests that psychosocial factors mediated the relation between dental pain and performance. On the other hand, two studies (Krisdapong et al., 2013, Jackson et al., 2011) argued that dental pain is the primary mechanism by which OHCs influence school absence first, then school performance.

The current study, using SEM, attempted to test these interlinked pathways. However, although dentine caries was found to predict dental pain, the finding that dental pain was not significantly associated with school outcomes suggests that dental pain did not necessarily play a pivotal role in these associations. These results are consistent with previous cross-sectional studies (Almeida et al., 2018, Maharani et al., 2017). However, one explanation of dental pain's lack of significance in the current study may

be that dental pain was measured only once, at the beginning of Grade 2, and thus the nature of the pain (chronic or acute), its duration, and whether it occurred at a critical time (for example, during midterm or final exams), were not examined. The inclusion of such factors could cause the association to change, and, as such, future studies are necessary to assess these associations longitudinally. A second potential explanation is that children with dental pain, as opposed to a medical illness, could attend and perform well in school unless they had severe symptoms. The study found that 10% of children reported constant pain, while 62% reported no pain because of OH within the POQL index. This explanation is corroborated by the study conducted by Pongpichit et al. (2008), which found that, compared with absence from school due to dental care and medical reasons, absence due to dental pain was minimal. The current study found that 66.7% of children missed at least one day for non-dental related reasons, while 12.4% reported absence of at least one day per academic year when they were in Grade 1.

Conversely, two studies suggested that dental visits resulting from oral problems such as dental pain are the main factors for school absence. Gift et al. (1992) and Naavaal et al. (2018) reported 52 million and 34 million hours respectively of school lost annually in the US because of dental visits. Recently, Darley et al. (2021) reported similar findings in Brazil, concluding that there was an association between dental pain and school absenteeism and an increase in absenteeism level according to the number of dental visits.

The conflicting results might also be attributed to how dental pain was measured. In the current study, dental pain was measured through the quality-of-life index (POQL) reported by children, similar to previous studies (Seirawan et al., 2012, Maharani et al., 2017). This measure did not indicate whether dental caries caused the pain; it

might, for instance, have been a result of exfoliation since the children were transitioning from deciduous to permanent dentition (Garg et al., 2012). Almeida et al.'s (2018) cross-sectional study of 374 Brazilian 6- to 8-year-old children avoided this bias by collecting dental pain data using the CPQ8-10 instrument and confirming it with clinical examination. However, it did not establish an association between school performance and dental pain or carious lesions.

Regarding the mediator role of the clinical consequences of untreated caries on the association between caries lesion and school outcomes, this was not observed in the current study, contradicting the findings of a recent study by (Quadros et al., 2021). This association was examined in only four studies, as described in Section 2.4: Quadros et al. (2021), Almeida et al. (2018), Karki et al. (2019) and Gradella et al. (2011). Of the four, only Quadros et al. (2021), through path analysis, found dental caries to be associated with children's school performance via consequences of untreated dental caries. Almeida et al. (2018) did not find an association between caries, their sequela and school performance, while Karki et al. (2019), despite concluding that untreated dental caries and their consequences significantly affected school performance and attendance, their final adjusted models demonstrated no relation with either school performance or absenteeism. Gradella et al. (2011) concluded that the consequences of untreated caries was associated with toothache and eating certain foods among pre-school children but was not associated with absences. Quadros et al. (2021) recorded the prevalence of untreated dental caries to be 19.6%, while in this study, the prevalence was higher (24.7%) but was not associated with either school performance or absences. This slightly unanticipated result may be due to some dental caries sequelae like ulcer, fistula, and even fully open necrotic pulp, not resulting in dental symptoms. Other differences may be related

to differences in age group, sample size, method of analysis, and the number of variables considered.

Thus, the argument that dental pain and the consequences of untreated dental caries predict school absenteeism then school performance was not supported in the present study, despite using a robust analytical approach and considering many of the risks related to both outcomes. It is worth mentioning that most of the studies that reported a relation between school attendance and performance found unauthorised, chronic absence, or school truancy, to be seriously associated with poor school performance over time and subsequent dropout, which was primarily observed in middle and high school children (Melvin et al., 2019, Klein et al., 2020). Chronic school absence is determined as missing at least 15% of days of a school year for any reason (Allison and Attisha, 2019). The reasons behind chronic absence are mainly the consequences of poverty, such as poor mental and physical health, poor living conditions, violence, abuse, alcohol and drug use (Gennetian et al., 2018, Melvin et al., 2019).

In light of this, the evidence in Section 2.4 suggests that OHCs may not cause chronic absence at school, as among the five studies that obtained absence from school records, the reported prevalence of absence because of OH was consistently below 15%. Pongpichit et al. (2008), for instance, reported only 0.5% OH-related absence per academic year among children aged 9-13 years old in Thailand, while Sameer et al.'s (2016) survey of 16–18-year-olds in Saudi Arabia found that just 13% of them missed more than three days. Osman and Alsayed (2015) reported that 1.7% of their cohort of children aged between 5 and 15 years old in Sudan missed around 7-9 days for OH reasons, and Seirawan et al. (2012) reported that 5.5% of their sample of US children aged 6–16 missed school because of dental issues. In the current study, the prevalence of Grade 2 children being absent for eleven or more days (i.e., for longer

than the authorised allowance of ten days) over three months for any reason was found to be 1.5%. Since these children were young and thus still dependent on their parents, they fall within the age group wherein problematic absence is difficult to observe (Malcolm et al., 2003).

A psychosocial pathway is also suggested to explain the association between pain and sequela of dental caries and school outcomes. Ortiz et al. (2021) argued that children with toothache who report verbal bullying because of OH would suffer from social exclusion. The resulting emotional disturbance may impact their social behaviours, sleep patterns (Perazzo et al., 2020), school attendance and reduced attentiveness in class. Similarly, Barasuol et al. (2017) argued that sequela of dental caries can cause aesthetic concerns and that children can be bullied because of this, which in turn affects their socialising process at school.

A current United Nations Educational, Scientific and Cultural Organization (UNESCO) (2019) report revealed the prevalence of school bullying among 11- to 15-year-olds in 144 countries in 2019, reporting that 32% of students worldwide had been bullied by their peers at school one or more days at the time of data-gathering in the past month. The Middle East region had the third-highest prevalence of bullying (after Sub-Saharan Africa and North Africa), with 41.1% of schoolchildren being bullied at least once over the same period. The report indicated that children who are appraised as different in any way are more likely to be bullied and that physical appearance is the main reason for bullying.

The current study did not examine the influence of bullying; however, in the POQL index, the effect of OHCs on the children's social behaviour was measured. It was found that 79% reported that they did not avoid smiling, 82.2% reported that they did

not worry about being less attractive, 86% reported that they were not unhappy with their physical appearance, and 82% reported that they were not angry about their teeth. This suggests that bullying because of OHCs at this young age may be less likely; however, it should be further examined in the future.

Another clear pathway suggested by Piovesan et al. (2012), Paula et al. (2016), and Seirawan et al. (2012) were that socioeconomic factors play an integral role between OHCs and school outcomes. However, regression analysis did not test the indirect pathways in the suggested relation. In parallel with their argument, this study found that early life disadvantages were associated directly with dental caries and indirectly with children's school performance, suggesting that dental caries played a mediation role. This result supports the social origin hypothesis and the critical period and chain of risk life course models. Based on this, children who experience low SES at the time of their birth were more likely to have dentine caries and then low school performance, regardless of their current SES. Further, dentine caries in primary teeth result from disadvantages in early life and are considered critical early life exposures that can restrain academic potential in middle childhood.

During the last decades, the literature has frequently revealed that exposure to early life risk factors, like family instability, maltreatment, and poverty, can negatively influence children's academic performance (Farajdokht et al., 2017, Anthony et al., 2014, Hussey et al., 2006). However, this is perhaps the first life course study to illustrate that dentine caries in the primary dentition is also an important early life risk for education attainment. This study's results are consistent with the hypothesis that inequalities in OH (Peres et al., 2011) and inequality in education start in the womb (Banerjee, 2016) and persist over time.

Another promising finding was that children with treated teeth were more likely to have better academic performance. This contradicts the Brazilian case-control study with 1,149, 8–10-year-old students, which found that dental caries, even if treated, were risk indicators of poor school performance (Paula et al., 2016) but is consistent with Petridou et al.'s (1996) cross-sectional study, which concluded that better school performance was linked with better dental health among Greek adolescents aged 12 to 17. Research in this area is limited, but some evidence has been found that after dental caries treatment, the OHRQoL of children improves, as reflected in the mean of Scale of Oral Health Outcome for 5-Year-Old Children (SOHO-5) (Dantas et al., 2015), CPQ₈₋₁₀ (de Paula et al., 2015b), and POQL (Cunnion et al., 2010). This emphasises the relevancy of children accessing dental care at school to mitigate the negative impact of dental caries on their school outcomes and quality of life by early prevention and treatment interventions when needed.

This study also found a dental plaque indirectly predicted children's academic performance, in line with Maharani et al. (2017), who found that among children aged 10-11 years, the odds of having lower grades were twice as high for those with substantial plaque. However, our results clarified that the relation was mediated by the dentine caries and treated teeth conditions. This finding confirms the importance of reinforcing toothbrushing programmes in schools, as they are considered the simplest individual preventive measure to control plaque, reduce caries in later life, and provide effective long-term maintenance of OH (Raviteja et al., 2017). It has been observed that childhood is a good time to establish healthy habits that last for life as children seem to be amenable, and their personality and mental maturations are starting to develop (Ceyhan et al., 2018).

It can be argued that Bahraini culture plays a significant role in explaining some of the differences between the findings of this study and those of previous studies. A technical argument that might be important is that most of the available evidence has been gleaned from Western societies, whereas this study's data is from the Eastern Arab region. These two societal groupings differ significantly in their systems of and approach to education; the education system in the Eastern Arab region is collectivism-oriented (Haj-Yahia, 1997), while individualism is the most distinctive feature of Western societies (Triandis et al., 1988). Collectivistic societies are mainly group-oriented; to succeed in a group, individuals need to give an opinion congruent with the prevailing religion and values despite what they believe. In this environment, the goal of learning is to become productive members of society; children are raised to be persistent in attaining high educational results, so they work hard at school and do enormous amounts of homework. The education system is teacher-centred, and students must submit to the authority of the teachers. The education process is mainly based on book learning and memorisation (Hofstede, 2001, Kaur and Noman, 2015).

Conversely, in individualistic societies, individuals are considered more important than groups. The system encourages critical thinking and creativity. Students are encouraged to study according to their aptitude and desire for learning. Further, students are active in the learning process; they can express their feelings and opinions and argue freely with their teachers; unlike the collectivistic system, mistakes are considered a favourable chance to attain maturity and develop their personality (Hofstede, 2001, Kaur and Noman, 2015). Thus, grades do not matter as much as they do in the eastern educational system.

These issues demonstrate how much Bahraini society values education to gain higher grades and work hard towards them, which might dissuade children from talking freely

to their parents about their dental pain and OH problems. This is reflected in the findings of this study. To illustrate, it was found that 61.6% of mothers were highly involved in their children's education process, and 79.2% of children scored 'Excellent' in their final grades with no failures. This raises some interesting questions regarding the value of OH against the value of the education process. In KoB, is having enamel caries, dental pain, or an oral ulcer an adequate reason for missing a school day or performing less well academically? Do parents know or believe in the connection between OH and general well-being?

Another cultural factor that might result in Bahraini schoolchildren avoiding mentioning dental problems to their parents is fear of the dentist. Unfortunately, there is no available data on KoB to support this argument. However, since KoB and Saudi Arabia (GCC) are considered homogeneous communities (At-Twaijri, 1990), recent evidence from Saudi Arabia can depict the situation in KoB. Alshuaibi et al. (2021) found that the prevalence of dental fear among 816 Saudi schoolchildren aged 7-11 was 50-70%, which is much higher than the 6-20% recorded for European children (Luoto et al., 2009). However, the high prevalence was attributed to the children's young age. The difference in prevalence between Arab and developed countries may be ascribed to the differences in culture and organisation of healthcare delivery services. In developing countries, curative measures are emphasised more than preventive measures so that children utilise healthcare services mainly when in pain, while in developed countries, preventive measures are the most common service (Alshoraim et al., 2018).

Consequently, in KoB, children usually view dentists as merchants of pain because they visit them only for dental treatment, not routine check-ups like in the west; moreover, parents use "going to the dentist" to symbolise fear, scare or punish their

children. The current study found that over 60% of the parents and children visited the dentist only in an emergency. Thus, because of dental treatment anxiety, children may not communicate their OH problem or utilise the dental service (Shafi et al., 2015), and they are unlikely to take time off school to seek dental care, even though that care is available for free. In the POQL interviews, 83.5% of the children in this study reported that they did not miss school because of OH issues.

Inconsistent results among the different studies might also be clarified by heterogeneity in study design, participant's age, measures of school outcomes, analysis method, geographical region, sample size, and the number of variables included. Furthermore, such wide-ranging variations make comparisons between studies difficult.

5.6 Discussion for Objective 4

To explore the early and current life course social, biological, behavioural and parental factors associated with children's school outcomes in addition to OHCs.

5.6.1 Other predictors of school performance

In addition to the two models, critical period and chain of risk, this study also showed that the accumulation of risks could influence children's school performance. Children who experienced favourable SES at the time of their birth and currently were more likely to exhibit high academic performance. The study found that supportive parental characteristics such as practising an authoritative parenting style and having a mother who is highly involved with school life mediated the association between SES and academic performance in the same way as dental caries. This means that families with high SES in early and current life are most likely to provide a supportive

environment for children; hence, they have higher academic achievement than their counterparts.

These findings are consistent also with the bioecological theory of Bronfenbrenner (2006). The theory acknowledged that child development is shaped by the interactions between SES and key social contexts, including families and schools. Thus, children's education development is determined not only by family SES but also by family-child interaction in terms of parenting and family-school interaction in terms of their academic involvement.

Scholars have long claimed that parenting style varies according to family SES; different parenting styles determine different emotions and settings at home (Darling and Steinberg, 1993), with reasonable evidence confirming this link. Bronfenbrenner (1958, p. 420) concluded that "parent-child relationships in the middle-class are consistently reported as more acceptant and equalitarian, while those in the working-class are oriented toward maintaining order and obedience".

Similarly, the positive link between family SES and academic involvement is wellestablished (Al-Matalka, 2014, Benner et al., 2016).

Therefore, it is now widely accepted that parenting style and academic involvement mediate family SES and children's academic performance (Wang et al., 2021, Zhang et al., 2020). Several explanations are given for this indirect effect. In a systematic review, Conger and Donnellan (2007) suggested that SES influences the family's functions through various stress and investment processes. Thus, a parent with low SES is often traumatised by economic worries, overwhelmed with extra working hours, lacking in material and psychological resources, and has little time for rearing children, all of which result in low responsiveness and warmth toward their children and the

embracing of an authoritarian parenting style. Conversely, a parent with high SES is likely to experience lower life stressors and therefore be more capable of investing financial, social, and human resources in the development of their children, exhibiting greater responsiveness and warmth toward their children and adopting an authoritative parenting style (Xie and Li, 2018).

Likewise, in explaining the relation regarding academic involvement, parents with low SES, struggling to support their families, are often too busy working to take the time necessary to help their children in school. Thus, it is difficult for these parents to academically support their children or attend school events and activities (Barg, 2019). There is also the factor that if the parents themselves have a low level of education, they may not provide effective involvement because of limited educational levels and low self-efficacy (Tazouti and Jarlégan, 2019). Providing useful advice on academic matters and strategies for success and fostering independent and critical thinking that can positively influence their children's learning behaviours (Aunola et al., 2000) maybe beyond their capabilities, causing them to inhibit their children's academic outcomes.

The current study did not support the association between children's early and current environmental exposures and school attendance. However, robust evidence establishes beyond doubt that children's SES background is a major predictor of school absenteeism, although the relation is multifaceted and complex (Gennetian et al., 2018, Brownell et al., 2016, Klein et al., 2020). According to Bronfenbrenner's (1977), children from lower SES families confront several individuals and environmental obstacles that disturb their well-being and educational outcomes. The current study did not capture the association between SES background and school attendance for several reasons. Firstly, most problematic absenteeism was reported

among older children, while this study's sample consisted of young children still dependent on their parents. Secondly, absenteeism becomes more of a concern when it is chronic or exceeds the authorised level of absence (15%); the prevalence of children in the sample missing 1-5 days was 57%. Thirdly, maybe because school absence was measured only for the first term of school, thus the period was not enough to capture any related factors. Finally, it can be argued that the schools' strategies to deal with absence played a significant role in their absenteeism rates. For example, schools in the UK are required by law to register children's attendance twice a day; if parents fail to ensure the regular attendance of their children, especially those of compulsory school age, they will face penalty or prosecution (Department for Education, 2016). In KoB, it is the same situation; the student can have an authorised absence for ten days; however, if the student misses more than ten days without reporting sensible reasons, the family might face prosecution because attendance at elementary school is compulsory by law (Information & eGovernment Authority-KoB, 2020b). Such upstream strategies could monitor and tackle the absenteeism problem and enforce pupils' attendance and education. In addition to the cultural issue described above, this might explain why attendance was not found to be predicted by any of the factors examined in this study.

Thus, this study applied a bioecological life course framework that examined OH's pathways and academic outcomes. This is important as it provides evidence that interdisciplinary frameworks can be applied to study how multiple exposures during an individual's early life and subsequent life course can influence their development. It may help design policies and interventions targeting children and their families to improve OH and school outcomes.

5.7 Limitations and strengths

This study has some limitations that should be accounted for when interpreting the findings, as in any research.

The first limitation is that the study was cross-sectional (due to COVID-19). Thus, no causal associations can be assured, and the observed associations might be considered exploratory (Kate Ann, 2006). Furthermore, mediation effects might be over or underestimated in this study when assuming longitudinal effects were evaluated using cross-sectional data (Maxwell and Cole, 2007). Thus, the results of the present should be interpreted with caution. However, Shrout (2011) argued that there still will be some situations where analyses of cross-sectional studies can reflect the mediation effect. Such situations include using well-established theories that describe the causal direction and the temporal process (Shrout, 2011).

In this study, all hypothesised relations were based on reviewing the literature and using the Heilmann et al. (2015) framework. Also, the study design was time-orderedcross-sectional to verify the temporal relationship between OHCs and school outcomes; OHCs were measured at the beginning of the academic year, while school outcomes were measured at the end of the academic year. Additionally, the study used SEM, which provided a rigorous basis for causal inference and provided an appropriate method for assessing mediation (Pearl, 2009). These advantages would not compete with the use of the longitudinal design study; however, it may have these advantages over a pure cross-sectional design for the causal analysis purpose (Menard, 2002). Ideally, this and similar questions can be addressed by means of a birth cohort or prospective research in which children's life trajectories, school outcomes and OHCs are measured repeatedly.

Another possible limitation is the selection of only 'Good' rated schools and sampling them as a unit instead of sampling the individuals. The restrictions were made to reduce the effects of the school's environment heterogeneity and control the possible effect of contextual confounders on the link between OHCs and school performance and attendance; however, it may limit the generalisability of the findings to schools rated differently on the quality continuum. This limitation could have been overcome by including other differently rated schools, but unfortunately, with the consequences of the COVID-19 pandemic and limited time and resources, this could not be achieved. However, the school's 'Good' rating by the QQA in KoB is for overall effectiveness and capacity evaluation; thus, it can be argued that the rating reflects the school environment more than anything else.

Thirdly, early life factors such as birth weight and gestational age were obtained retrospectively and were self-reported by parents, which means they may be subject to recall bias. Efforts were made to obtain this information from the children's health records, but unfortunately, the records were not available. However, a recent study evaluating the accuracy and validity of early life information reported by parents has yielded very positive results. Moreno-Galarraga et al. (2021) collected parent-reported birth data (birth weight, birth length and gestational age) on 241 children in Spain when those children were 4-6 years old then compared it with the medical records. The ICCs calculated for birth weight, and birth length revealed, respectively, 'very good' (0.95)

and 'good' (0.78) levels of agreement, and the weighted kappa for gestational age was substantial (0.90) (Moreno-Galarraga et al., 2021).

Fourthly, school attendance was measured only for the first term of 2019-2020 because the second term was affected by COVID-19, and children could not enrol in schools. This may have resulted in underestimates of the absence days. Further, the absences and the absences related to dental reasons in Grade 2 were unknown. Thus, the short period and the lack of this information may explain why the study did not find any factor related to school attendance.

Fifthly, Code 1 for ICDAS, which referred to detecting the first visual changes of dental caries after 5 seconds of air drying, was not considered because the examination was conducted at schools, and no portable air syringes were available. This might contribute to uncertainty regarding the number of free caries in children and teeth. Alternatively, it might cause underestimating dental caries and overestimating caries-free surfaces.

Lastly, one important limitation of using SEM is that SEM analysis requires deleting variables during model estimation to get a good model fit, resulting in biased parameter estimates (Tomarken and Waller, 2005). This study's first model was constructed with most of the variables presented in Chapter 4 (see Appendix 20), which involved 12 latent variables with 54 observed variables. However, the model was not identified in AMOS. A model with this number of variables would require a larger sample size (2129 participants) to yield an acceptable model fit (Hair et al., 2014). Thus, although the study included most of the relevant factors to both OH and school outcomes based on theoretical framework and literature, unfortunately, factors such as child's health conditions, POQL, dental services, family structure were not

included because they yielded insignificant loadings. However, Tomarken and Waller (2005) argued that researchers rarely recognise omitted variables because if a model fits well, it must contain all the necessary variables concerned in the hypothesised model.

Despite these limitations, this study has several strengths. Firstly, it is research that is driven by theory and framework. The life course framework provides multilevel models that explain social determinants and their influence on the common risk factors involved in the development of OHCs and shared in predicting the school outcomes and interplay between OHCs and school outcomes from early life until the current stage of the child's life. Thus, this study is the first to apply a bioecological life course framework that examines the pathways connected to SES, OH, and school outcomes. Further, it is the first study to consider biological, social, behavioural, and parental factors that interplay between OHCs and school outcomes. The study adds knowledge regarding factors associated with children's educational potential and concludes that dental caries was a significant early life factor linked with children's school performance.

The second most significant strength of this study was its use of SEM. Given the complexity of the life course framework, SEM provided multiple advantages over regression. The first is its breadth of capability. SEM is a multivariate statistical method that provides a process whereby to test a theoretical network of directional and nondirectional associations between independent (exogenous) variables and one or more dependent (endogenous) variables (Kline, 2016). However, in regression, only one dependent variable can be tested (Berndt and Williams, 2013). Accordingly, it was possible to simultaneously test different OHCs concerning different school outcomes. The second advantage is that SEM can examine and interpret both direct and indirect

effects; thus, it was possible to test different well-established life course models and different types of pathways that might connect OHCs with school outcomes. In SEM, the third advantage is that the combined effect of multiple indicators for each construct can be utilised efficiently, unlike regression, where researchers are restricted by including blocks of predictors (Berndt and Williams, 2013). This enabled the discovery that fathers' and mothers' levels of education are essential life course predictors. These three advantages facilitated the testing of many risk factors along the life course of children that have not previously been considered, especially those related to family (SES, parenting style, academic involvement).

The third strength of this study is that it utilised reliable (and multiple) measures to study the relation and the pathways between OHCs and school outcomes. For example, the use of ICDAS and pufa/PUFA measures enabled an assessment of different types of dental caries (enamel, dentine), their stages, and their severity on children's school outcomes. Besides evaluating the effect of clinical OH issues, the study also examined their psychosocial impact with respect to dental pain. This produced the finding that asymptomatic enamel caries does not affect school performance, whereas dentine caries does. Also, the study concluded that dentine dental caries was associated directly with school outcomes without the mediation effect of dental pain or severity of caries. Unlike previous studies (Section 2.4), 43.6% of them focused on dentine dental caries in the DMFT/dmft index. Despite being the most commonly used index worldwide to measure caries (Casamassimo et al., 2009), only cavitated teeth are recorded through DMFT, limiting a proper assessment of the severity of dental caries (de Souza et al., 2014).

In addition, nearly 30% of them relied on self-reported measures of OHCs (Nasuuna et al., 2016, Pongpichit et al., 2008) rather than a clinical examination to study how

specific OH problems might affect children's school outcomes. The question requires quantitative research design; the fundamental philosophy of quantitative research design is positivism (Creswell and Creswell, 2018). The positivist philosophy presumes the use of objectives and scientific measures to develop knowledge. The scientific method requires measurement and a scientific basis for research. It involves the formulation and testing of hypotheses, and this requires mathematical and statistical analysis (Creswell and Creswell, 2018). The analysis of OHCs from parents' perceptions, or students' self-reports, might have subjective interpretations that could create a bias (de Paula and Mialhe, 2013). Further, self-reporting of dental caries and periodontal conditions was revealed to be invalid as a clinical evaluation for an adult population (51- to 86-year-olds) (Pitiphat et al., 2002). This causes serious concern regarding validity when the target population is children since the questionnaires are usually completed by parents, who might have a different perception of their children's OHCs (Ribeiro et al., 2018).

Another reliable measure used in this study is the school records, which supplied data on the children's school outcomes. Previous studies (Section 2.4), in contrast, have mainly relied on self-reported school outcomes (Blumenshine et al., 2008, Petersen et al., 2008), and some of these studies applied the QoL index as a proxy for school performance (Bernabé et al., 2007, Astrøm and Okullo, 2003). Relying on self-reported measures to assess school outcomes might not be sufficient due to the lack of knowledge of the validity and reliability of those measures, which might result in recall bias or social desirability bias (Althubaiti, 2016). Plus, considering the QoL instrument as an indirect measure of evaluating children's school outcomes might generate measurement bias because they are not specifically designed to evaluate them (de Paula and Mialhe, 2013).

Fourthly, the study reached the required sample size of 466 children and their parents with a very high response rate of 98%. This is important to confirm that the results represent the target sample, and the higher the response, the greater the likelihood that the sample will reflect the population (Meterko et al., 2015).

Finally, the study provides OH evidence from an area where research is scarce; it provides a different picture of children's OH from studies conducted from a Western point of view. The first comprehensive life course study conducted in the Arab-Mideast region assesses the relation between OHCs and school outcomes.

Chapter 6. Conclusions and recommendations

6.1 Conclusions

This research has contributed to current knowledge by applying a bioecological life course framework to examine the different possible pathways connecting OH with school outcomes within the context of children's early and current life experiences. It was found that children who experienced high SES at birth had a low rate of dentine caries, hence exhibited high school performance, supporting the critical period and chain of risk models. Similarly, the study found that low SES in early life and currently were associated with low school performance via parent's characteristics, supporting the accumulation of risk and bioecological models. More importantly, the study found that SES in early life was a common factor associated with dentine dental caries, dental pain, and school performance in children's current life course, supporting the theory of the social determinants. On the other hand, dental treatment could be a protective factor linked positively with school performance.

Therefore, it is concluded that low SES, dentine caries, and non-supportive parental characteristics are important indicators of children's low school performance.

The primary conclusions drawn are:

- Enamel and dentine dental caries and the clinical consequences of untreated dental caries were highly prevalent in Muharraq, with prevalences of 93.5% and 25.7%, respectively.
- 2. EDD was more prevalent in permanent dentition, with a prevalence of 17.0%.
- 3. Most children (67.6%) reported at least one impact of OH on their quality of life.

- 4. High SES in early life was associated with a low level of dentine caries and dental pain among children in Muharraq.
- Children in Muharraq with a high level of dental plaque were more likely to have OHCs.
- Children with more dental plaque and dentine caries were more likely to report dental pain.
- Dentine caries and dental plaque were negatively associated with children's school performance.
- 8. Dental treatment was positively linked with children's school performance.
- Dentine caries was directly linked with poor school performance without the mediation effect of either dental pain, clinical consequences of dental caries, or school attendance.
- 10. Family SES was associated with high children's school performance through parental characteristics, both at birth and currently.
- 11.None of the social, behavioural, biological, parental, or OH conditions was associated with school attendance.

6.2 Recommendations

The study findings indicated that SES, especially in early life, was related to both dental caries and school performance. SES was associated with children's school performance via parent characteristics (parenting style, academic involvement) and dentine caries. Further, dental caries and dental plaque were negatively linked with school performance, while treated teeth were linked positively with school performance. Thus, we provided recommendations to tackle social inequalities and dental caries early in life.

6.2.1 Recommendation for social determinants in early life

It is recommended, therefore, to tackle inequality in early life and throughout the individual's life course. Looking through the life course lens means identifying opportunities within the key life stages from early into older age to reduce risk factors and enhance protective factors through evidence-based interventions (Public Health England, 2019).

Public Health England (PHE) has produced a series of evidence-based papers covering several recommended interventions to tackle inequalities at people's important life stages (Donkin, 2014). Review 1 focuses on early intervention strategies, implemented from conception to early adulthood, to prevent poor outcomes in later life among the disadvantaged population. The review emphasises the evidence on two main topics: parenting and home-to-school transition programmes to reduce the impact of social determinants in early life. Parenting programmes aim to improve parent-child interaction and stimulate supportive parenting. One of the key pieces of evidence relates to children's centres. The research argues that children's centres could positively improve children's and families' well-being, with particular attention

given to disadvantaged families to reduce inequalities in child's development and schooling (Donkin, 2014). One effective support programme is the USA's Head Start programme (Kose, 2021). The programme provides comprehensive services to low-income children and families in terms of early education, health, nutrition, and parenting. Evaluations of the programme have shown improvement in children's cognitive, emotional and social development, higher emotional child-parent engagement, constant attention, substantial test scores, and decreased aggressive behaviours (Kose, 2021, Donkin, 2014).

Review 2 focuses on children aged 5-18 years old and building resilient children and families through schools (Allen, 2014). Resilience is the "ability to bounce back" from hardship and have control over things (Allen, 2014). It is believed that resilience aids in increasing proactivity and reducing risk factors in schools to promote well-being (Shi et al., 2021). From 2010 to 2018 in the UK, the Children in Need's Families and Schools Together (FAST) programme effectively bolstered resilience among families and children through schools (Lord et al., 2018). In this programme, families attended eight weeks of workshops within schools to reduce stress and address certain social conflicts. Assessment of the programme has shown a 20% reduction in family conflict, an increase in family-child relationships by 14%, and reductions in emotional disturbance and conduct problems among children by 20% and 18%, respectively (Allen, 2014). Such programmes might be good models to follow whereby to introduce measures to reduce inequality and promote health, both oral and general, in KoB within a family context.

6.2.2 Recommendation for OH promotion

To prevent dental caries and reduce inequalities, strategies should be intersectoral, population base health promotion and common risk factor approaches when possible (Watt et al., 2019), and targeted interventions for the disadvantaged population and priority groups like pregnant women and young children (Public Health England, 2019). Those strategies should target transitions through evidence-based interventions at critical life stages (Department of Health, 2010). Using the Ottawa Charter's five principles for OH promotion (WHO, 1986), the recommendations below could be suitable for KoB at a national level:

6.2.2.1 Public health policies

- Policies to reduce sugar consumption and meet the WHO recommendations: children and adults to reduce their intake of free-sugar to less than 5-10% of total energy intake (WHO, 2015). Such policies would require the industry to reduce the sugar content in food products (Watt et al., 2019). Alternatively, the introduction of sugar taxes on sugary beverages. Data from Mexico shows that sugar taxes decrease sales and consumption and the number of overweight individuals, especially in a low-income population (Colchero et al., 2017).
- Policy to introduce a water fluoridation system. Research in this area has shown that water fluoridation is a cost-effective, safe public health strategy for reducing tooth decay and inequality in a population (Treasure et al., 2002, Belotti and Frazão, 2021).

6.2.2.2 Creating supportive environments

- Restricting TV promotion of unhealthy food and beverages was the most costeffective population-based measure to reduce obesity in children aged 5–14 years (Magnus et al., 2009). Such a measure is also likely to reduce tooth decay rates.
- Eliminate taxes on OH products and make fluoride toothpaste available and affordable (Kwan and Petersen, 2010).

6.2.2.3 Strengthening community actions

Life course data underlines the vital role of families as fundamental societal units that establish and foster individual and community health (Hanson et al., 2019). Initial research indicates that orienting a family-based health promotion, and disease prevention are promising measures despite evidence for this being scarce, which can be attributed to a lack of initiatives, funding, and policies; within the concept of the life course, such an integration can be pivotal (Thomas et al., 2017).

However, the effectiveness of family-centred OH promotion in reducing ECC was recently tested in Hong Kong through a Randomised Control Trial (RCT) (Yu et al., 2021). At baseline, a total of 580 pregnant women and their partners were recruited, and 436 toddlers were followed up when they became three years old (test, n = 228; control, n = 208; follow-up rate, 75.2%). Oral health education (OHE) was given to the test group via behavioural and educational counselling, while the control group received OHE brochures only. Data regarding feeding habits, OHPs, and OH of the toddlers was obtained by parent's self-administered questionnaires and oral examination annually thru home visits. The results revealed that family-centred OHE for parents thru behavioural and educational counselling was more effective in

unleashing good feeding practices and parental toothbrushing habits and reducing dental caries than the distribution of OHE brochures alone.

The family-centred OH promotion has also been delivered via Maternal Child Health Care (MCH) in Australia via the Lift-the-Lip programme (Heilbrunn-Lang et al., 2020). The programme follows families and babies from birth, especially those with low SES; it involves lifting the top lip of a child to check for early signs of dental caries, along with distribution of toothbrushes and toothpaste and a referral system to dental care. The results revealed that MCH provides an ideal context to enable family-centred OH promotion measures and the adoption of good OH behaviours early (Heilbrunn-Lang et al., 2020).

6.2.2.4 Developing personal skills

Toothbrushing to control dental plaque level, spit instead of rinse, proper use of fluoride toothpaste, with the optimum level required for each stage of life, should be reinforced at a national level (Gregory, 2014, Department of Health-UK, 2021).

Research has shown that social media has a similar effect on advertising sugary snacks and foods high in sugar, fat, and salt to traditional media (TV) (Coates et al., 2019). Online social media are readily available, and most individuals access social media daily (Cortesi and Gasser, 2015). Therefore, social media could provide a good platform for a population-based intervention to promote OH, increasing people's awareness regarding the importance of toothbrushing, the proper use of fluoride toothpaste, and the optimum level required for each stage of life. Webb et al. (2010) assessed the effectiveness of Internet-based measures in promoting positive health behaviour via a meta-analysis. The study concluded that approaches based on

behavioural theory and those with many behaviour change measures were most effective (Webb et al., 2010).

Further, a recent RCT examined the effectiveness of a theory-based programme applied on an online social media programme (Telegram) to endorse good OH behaviour among Iranian teenagers (Scheerman et al., 2020). The results revealed significant improvements in adolescent toothbrushing habits, plaque, and CPI scores.

6.2.2.5 Reorienting OH care toward prevention

- OH care in KoB needs to shift its goal of dental care from treatment to prevention (Vernazza et al., 2021). Dental care can be reformed by: adopting multisectoral and patient-centred approaches; training OH professionals with the relevant skill mix; increasing the number of dental health professionals so that dental care is accessible to and available for most of the population; changing the salary-based remuneration system to an incentive-based system that encourages preventative and maintenance care of OH, in the same way, that the US pay-for-performance system has shown an improvement in the quality of healthcare (Watt et al., 2019).
- Evaluation of dental caries in KoB should shift from reporting the presence or absence of cavitated lesions to evaluating the different stages of caries development (ICDAS), their severity (pufa/PUFA) and their impact on people's quality of life. Understanding these stages over time can be valuable for early caries detection, prevention, diagnosis, prognosis, and management, including the formulation of optimally timed interventions to reverse the disease process (Pitts and Ekstrand, 2013).

6.2.3 Targeted OH promotion for pregnant and new mothers

Policies should be introduced integrating OH promotion programmes within MCH, or start home visits by dental health professionals or midwives, to provide early life OH promotion interventions targeting pregnant mothers, new mothers and babies in low SES areas or areas with high caries rates (Rogers, 2011, Feldens et al., 2007, Public Health England, 2019). Below are examples of OH interventions that could be implemented within this scope:

- Reinforcing exclusive breastfeeding and providing suitable complementary feeding practices. The Family Nurse Partnership programme in the USA offers planned home visits from early pregnancy until the child is two. Research conducted over 30 years in the USA has shown positive outcomes among low SES families in parents' breastfeeding, smoking, cognitive and emotional behaviours, and children's behaviours (Public Health England, 2019).
- Distribution of fluoride toothpaste and toothbrushes to pregnant women and young babies. In the UK, the three-monthly posting of OHE leaflets, fluoride toothpaste, and brushes to caregivers of children at high risk of dental caries in non-fluoridated areas decreased caries levels by 16% when the children reached five years of age (Davies et al., 2002). In Leeds (UK), regular dental visits by dental health educators providing OH materials, diet and OH instructions in a low SES and non-fluoridated population with a high caries rate resulted after three years in significant progress in feeding habits and toothbrushing frequency, with a decrease in the high consumption of sweet food and drinks, and lower caries level (Kowash et al., 2006).

- Developing toothbrushing and supervised toothbrushing for mothers and babies (Kowash et al., 2006) (El Fadl et al., 2016, Liu et al., 2020). In Israel, training for nurses in providing toothbrushes, toothpaste, and OHE to mothers increased the routine brushing of 6-12-month-old infants' teeth (Sgan-Cohen and Vered, 2005). A recent RCT study in Hong Kong also found that providing family-centred, behavioural, and educational counselling to pregnant women during their MCH visits improved their OH and decreased plaque, bleeding, and gingivitis (Liu et al., 2020).
- Applying priority access for dental care for referring pregnant women who are in need of clinical intervention. In Sydney (Australia), MCH referral of pregnant women to dental care programmes has shown a considerable increase in the use of dental care, women's OH knowledge, and OH outcomes (bleeding, dental plaque, clinical attachment loss, DFT) (George et al., 2018).
- Reducing the transmission of S. mutants from mothers to their babies by providing mouth rinses or varnishes (fluoride and/or chlorhexidine) and xylitol chewing gum to pregnant women and mothers of newborn babies (Rogers, 2011, Twetman, 2013).

6.2.4 Targeted OH promotion for preschool- and school-age children

Dental hygienists in KoB already implement schools-based OH promotion programmes based on WHO (2003) recommendations in all government-run schools, whereby children receive dental screening and OH education, including demonstrations of toothbrushing and applying sealant and fluoride varnish. Nurseries and schools should be used to the optimum as they are considered a vital environment for promoting OH among children, families, teachers, and the whole community from different SES backgrounds (Bramantoro et al., 2021, Kwan et al., 2005). Accordingly, the following are recommended to be applied in KoB:

- A comprehensive re-evaluation of the currently implemented OH promotion intervention programmes. WHO, UNESCO, UNICEF, the World Bank and Educational International launched a joint initiative – FRESH (Focusing Resources on Effective School Health) – to monitor and evaluate school health programmes (FRESH, 2013). This framework could be adapted in KoB for continuous evaluation.
- Interventions should start at preschool and continue to older ages (Bramantoro et al., 2021).
- Engagement of parents, teachers, schoolchildren and dental health professionals in planning and implementing OH interventions (Public Health England, 2014).
- Policies should be initiated to create a healthy school environment; for example, replacing sugary snacks with healthy choices (Rogers, 2011), establishing protocols to control bullying and violence, and tighter smoking control, especially in older age children (WHO, 2003).
- Dental health education should be integrated into the school curriculum, the message being delivered by teachers. It should be age-appropriate for children's cognitive abilities and the primary health risks they are exposed to at each stage of development, thus enabling children to develop a life-long commitment to maintaining their OH (Public Health England, 2014). As shown in the findings, mothers in KoB are highly involved with their children's early life;

thus, such a curriculum might influence teachers', children's and families' OH knowledge.

- Introduction of supervised and school-based toothbrushing programmes. The teacher-supervised toothbrushing and fluoride toothpaste distribution (1,400 ppm) programmes in a non-fluoridated deprived area of London (UK) contributed to a drop in dental caries (compared to control schools) of 11–21% over the 21 months (Jackson et al., 2005). Similar results were recently found in Israel and Chennai (Natapov et al., 2021, Samuel et al., 2020).
- Increase fluoride exposure, especially in low SES schools. Since KoB does not have a water fluoridation system, an alternative could be implementing regular school-based fluoride rinsing or providing fluoridated milk (Yeung et al., 2015, DivyaLalitha et al., 2019).
- Referrals and priority access to dental care for children with low SES who need dental treatment (Hong et al., 2020).

6.2.5 Recommendations for better school outcomes

- There should be monitoring by school faculties of OH-related absences and complaints to provide data necessary for evaluation and improvement (Pongpichit et al., 2008).
- The existing education system, centred on memorising, achieving grades, and a teacher-centred approach, should be reconsidered. Children in early-life learning should be allowed a voice and free expression to boost their selfesteem and foster in them the ability to control things around them (Butler et al., 2021). Two review papers from PHE examined successful interventions that

ensure children are given the best start in life, improve early school outcomes and reduce inequality:

- Review 1 considers parenting and home-to-school transition interventions targeting children from conception onwards (Donkin, 2014);
- Review 2 summarises the evidence of the possible measures taken in schools to build resilience for all schoolchildren, families, and teachers in partnership, targeting children aged five to eighteen years (Allen, 2014).
- There should be policies to initiate programmes that enhance skills/behaviours that promote secure attachment, positive parenting practices and parent-child relationships. For example, "Helping Parents to Parent" (Clarke and Younas, 2017), an evidence-based report from the UK's Social Mobility Commission, highlights examples of successively applied interventions from different countries that aim to foster more authoritative parenting styles so that parents become more responsive and sensitive towards their children's emotional, physical, and cognitive needs. The programmes target early-life and preschool-and school-age children and are either universal or focused on low SES groups. They are delivered by teachers, nurses or social workers via schools or health centres, home visiting, or using individual face-to-face sessions.
- It should be government policy that schools promote the academic involvement of parents or caregivers, especially those with lower SES, to enhance positive school outcomes for school-age children. The UK report entitled "How Can Schools Support Parents' Engagement in their Children's Learning?" (Axford et

al., 2019) synthesises the best available international evidence on parental engagement in children's learning. The evidence focuses on effective parenting practices associated with positive learning outcomes at different stages of children's development and what schools and early years settings can do to support parents' involvement, particularly for children from disadvantaged backgrounds.

Considering the scarcity of comparable interventional studies from KoB and the Middle Eastern region, I suggest these UK reviews and reports be studied to help determine ways in which inequality in Bahrain can be reduced, academic performance improved, positive parenting practices promoted, and parents or caregivers encouraged to involve themselves in the different stages of Bahraini children's academic lives.

6.2.6 Recommendations for future research

6.2.6.1 Descriptive studies

Update this cross-sectional OH survey in KoB to:

- Evaluate ECC, prevalence, impacts and indicators, targeting preschool children and their families.
- Assess OHCs' prevalence, their stages, severity, impact on QoL, relation to SES and other chronic conditions, and their indicators (Watt et al., 2019).
- Involve parents and assess their OH knowledge and beliefs.

6.2.6.2 Longitudinal studies

Design and execute a prospective birth cohort study with a mixed-method approach in KoB, with OH as one of its focus points from its onset. In KoB, no birth cohort study has ever been conducted, so this would be an opportunity for comprehensive research. Such a project will require good collaboration with experts in this field, a national-level plan to include other disciplines interested in this type of research, and the securing of sustainable, preferably government, funding (Costello and Angold, 2007). The following topics could be studied:

- Inequalities in health, OH, and education with full transparency across the life course (Public Health England, 2019).
- Development of OHCs, their trajectories, risks and protective factors over the life course (Halfon et al., 2018).
- The impacts of family transitions, relations, structure, religion, and culture on the OHCs and school outcomes.
- Pathways involving SES, biological, psychosocial, behavioural, and parental conditions whereby to link OHCs with school outcomes in different stages of life (Halfon et al., 2018).
- Ways in which OHCs like EDD could function as adverse early life indicators for general health and OH in later life (Heilmann et al., 2015).
- Ways in which early caries lesions could be a marker of social deprivation in KoB; this research may help in targeting vulnerable populations with specific needs.
• Ways in which OHCs could be used as a marker for poor school performance and attendance.

6.2.6.3 Interventional studies

Interventional studies are often tailored to evaluate the effectiveness of preventive measures (Thiese, 2014). RCTs are the most common type of interventional study (Machin et al., 2021). The following topics could be studied using an RCT framework:

- The effects of toothbrushing, fluoride varnish and fissure sealant application programmes, and dental treatment, on schoolchildren (Tickle et al., 2016).
- The effects of interventions applied within the family on OH. For example, early childhood interventions include a parent-supervised toothbrushing programme (Natapov et al., 2021).
- The effects of parent's involvement interventions on children's education.

Chapter 7. References

- ABDALLA, H. E., ABUAFFAN, A. H. & KEMOLI, A. M. 2021. Molar incisor hypomineralization, prevalence, pattern and distribution in Sudanese children. *BMC Oral Health*, 21, 1-8.
- ABREU, L. G., ELYASI, M., BADRI, P., PAIVA, S. M., FLORES-MIR, C. & AMIN, M. 2015. Factors associated with the development of dental caries in children and adolescents in studies employing the life course approach: a systematic review. *European Journal* of Oral Sciences, 123, 305-311.
- ADIATMAN, M., YUVANA, A. L., NASIA, A. A., RAHARDJO, A., MAHARANI, D. A. & ZHANG, S. 2016. Dental and gingival status of 5 and 12-year-old children in Jakarta and its satellite cities. *Journal of Dentistry Indonesia*, 23, 5-9.
- AGAKU, I. T., OLUTOLA, B. G., ADISA, A. O., OBADAN, E. M. & VARDAVAS, C. I. 2015. Association between unmet dental needs and school absenteeism because of illness or injury among U.S. school children and adolescents aged 6– 17 years, 2011– 2012. *Preventive Medicine*, 72, 83-88.
- AHMADI, R., RAMAZANI, N. & NOURINASAB, R. 2012. Molar incisor hypomineralization: a study of prevalence and etiology in a group of Iranian children. *Iranian journal of pediatrics*, 22, 245-251.
- AHMED, A. A., FATEHA, B. & BENJAMIN, S. 2000. Demand and supply of doctors and dentists in Bahrain, 1998-2005. *Eastern Mediterranean Health Journal*, 6, 6-12.
- AISLING, A. G., KAREN, L. L., GOIURI, A., FIONNUALA, M. M. & EILEEN, R. G. 2016. Nutrition During Pregnancy Impacts Offspring's Epigenetic Status— Evidence from Human and Animal Studies. *Nutrition and Metabolic Insights*, 2015(S1), 41-47.
- AL-MATALKA, F. I. M. 2014. The influence of parental socioeconomic status on their involvement at home. *International Journal of Humanities and Social Science*, 4, 146-154.
- ALKHAMIS, A., HASSAN, A. & COSGROVE, P. 2014. Financing healthcare in Gulf Cooperation Council countries: a focus on Saudi Arabia. *The International journal of health planning and management,* 29, e64-e82.
- ALLEN, M. 2014. Local action on health inequalities: Building children and young people's resilience in schools. *London: Public Health England*.
- ALLISON, M. A. & ATTISHA, E. 2019. The link between school attendance and good health. *Pediatrics*, 143, 1-13.
- ALLOTEY, J., ZAMORA, J., CHEONG-SEE, F., KALIDINDI, M., ARROYO-MANZANO, D., ASZTALOS, E., POST, J., MOL, B., MOORE, D., BIRTLES, D., KHAN, K. & THANGARATINAM, S. 2018. Cognitive, motor, behavioural and academic performances of children born preterm: a meta- analysis and systematic review involving 64 061 children. *BJOG: an international journal of obstetrics and gynaecology*, 125, 16-25.
- ALMEIDA, R. F., LEAL, S. C., MEDONCA, J. G. A., HILGERT, L. A. & RIBEIRO, A. P. D. 2018. Oral health and school performance in a group of schoolchildren from the Federal District, Brazil: Oral health and its relationship with school performance of schoolchildren from Brazil. *Journal of public health dentistry*, 78, 306-312.
- ALSHORAIM, M. A., EL-HOUSSEINY, A. A., FARSI, N. M., FELEMBAN, O. M., ALAMOUDI, N. M. & ALANDEJANI, A. A. 2018. Effects of child characteristics and dental history on dental fear: cross-sectional study. *BMC oral health*, 18, 1-9.
- ALSHUAIBI, A. F., ALDARWISH, M., ALMULHIM, A. N., LELE, G. S., SANIKOMMU, S. & RAGHUNATH, R. G. 2021. Prevalence of dental fear and anxiety and its triggering factors in the dental office among school-going children in Al Ahsa. *International journal of clinical pediatric dentistry*, 14, 286-292.
- ALTHUBAITI, A. 2016. Information bias in health research: Definition, pitfalls, and adjustment methods. *Journal of multidisciplinary healthcare*, 9, 211-217.

AMATO, P. R. 2005. The impact of family formation change on the cognitive, social, and emotional well- being of the next generation. *The Future of children,* 15, 75-96.

AMES, C. 1993. Parent Involvement: The Relationship between School-to-Home Communication and Parents' Perceptions and Beliefs. Report No. 15. 1-42.

- AMINABADI, N. & FARAHANI, R. 2008. Correlation of parenting style and pediatric behavior guidance strategies in the dental setting: preliminary findings. *Acta Odontol Scand*, 66, 99-104.
- ANDERSEN, S. C., GREGERSEN, M. K., NIELSEN, H. S. & THOMSEN, M. K. 2021. Parent Involvement, Socioeconomic Status and Reading Performance. *Scandinavian journal* of educational research, 65, 1279-1294.
- ANDERSON, T., THOMAS, C., RYAN, R., DENNES, M. & FULLER, E. 2015. Children's Dental Health Survey 2013. Technical Report: England, Wales and Northern Ireland. Final V1.0 ed.
- ANGGRAENI, M. D., PUNTHMATHARITH, B. & PETPICHETCHIAN, W. 2020. A causal model of breastfeeding duration among working muslim mothers in semarang city, central java province, indonesia. *Walailak journal of science and technology*, 17, 1010-1023.
- ANGRIST, J., LAVY, V. & SCHLOSSER, A. 2010. Multiple experiments for the causal link between the quantity and quality of children. *Journal of Labor Economics*, 28, 773-824.
- ANSONG, D., OKUMU, M., BOWEN, G., WALKER, A. & EISENSMITH, SR. 2017. The role of parent, classmate, and teacher support in student engagement: Evidence from Ghana. *Int. J. Educ. Dev.*, 54, 51-58.
- ANTHONY, C. J., DIPERNA, J. C. & AMATO, P. R. 2014. Divorce, approaches to learning, and children's academic achievement: A longitudinal analysis of mediated and moderated effects. *Journal of School Psychology*, 52, 249-261.
- ANTONOPOULOS, V., DUNKLEY, N., RADIA, A., SANTIAGO, L. & WILLIAMS, R. 2021. *Child poverty and education outcomes by ethnicity* [Online]. UK: Office of National Statistics.

https://www.ons.gov.uk/economy/nationalaccounts/uksectoraccounts/compendium/e conomicreview/february2020/childpovertyandeducationoutcomesbyethnicity? cf chl managed tk =CqjU805bq .t2RC85KTAwXI7QtWfwPvXWXE6Q6JGMik-

1642848106-0-gaNycGzNCL0 [Accessed 26 October 2021].

- ARANGANNAL, P., MAHADEV, S. K. & JAYAPRAKASH, J. 2016. Prevalence of dental caries among school children in Chennai, based on ICDAS II. *Journal of Clinical and Diagnostic Research*, 10, ZC09-ZC012.
- ARCHAMBAULT, I., JANOSZ, M., FALLU, J.-S. & PAGANI, L. S. 2008. Student engagement and its relationship with early high school dropout. *J Adolesc*, 32, 651-670.
- ARNOLD, R. R., COLE, M. F. & MCGHEE, J. R. 1977. A Bactericidal Effect for Human Lactoferrin. *Science*, 197, 263-265.
- ASTRØM, A. N. & OKULLO, I. 2003. Validity and reliability of the Oral Impacts on Daily Performance (OIDP) frequency scale: a cross- sectional study of adolescents in Uganda. *BMC oral health*, 3, 1-9.
- AT-TWAIJRI, M. I. 1990. Language effects in cross-cultural management research: An empirical investigation. *International Journal of Value-Based Management*, 3, 137-146.
- AUNOLA, K., STATTIN, H. & NURMI, J.-E. 2000. Parenting styles and adolescents' achievement strategies. *Journal of Adolescence*, 23, 205-222.
- AXFORD, N., BERRY, V., LLOYD, J., MOORE, D., ROGERS, M., HURST, A., BLOCKLEY, K., DURKIN, H. & MINTON, J. 2019. How can schools support parents' engagement in their children's learning? Evidence from research and practice. Education Endowment Foundation.
- BAGOZZI, R. P., YI, Y. & PHILLIPS, L. W. 1991. Assessing construct validity in organizational research. *Administrative science quarterly*, 421-458.
- BAI, Y. K., MIDDLESTADT, S. E., JOANNE PENG, C. Y. & FLY, A. D. 2009. Psychosocial factors underlying the mother's decision to continue exclusive breastfeeding for 6 months: an elicitation study. *Journal of Human Nutrition and Dietetics*, 22, 134-140.

BALKIS, M., ARSLAN, G. & DURU, E. 2016. The school absenteeism among high school students: Contributing factors. *Educational sciences : theory & practice,* 16, 1819-1831.

BALLANTINE, J. 2001. Raising Competent Kids: The Authoritative Parenting Style. For Parents Particularly. *Childhood Education*, 78, 46-47.

BALTES, P. B. 1987. Theoretical Propositions of Life-Span Developmental Psychology: On the Dynamics Between Growth and Decline. *Developmental psychology*, 23, 611-626.

BANDURA, A. 1986. Social foundation of thought and action, Englewood cliffs, NJ: prentice Hall.

BANERJEE, P. A. 2016. A systematic review of factors linked to poor academic performance of disadvantaged students in science and maths in schools. *Cogent Education*, 3, 1-17.

BANK, W. 2020. *Bahrain economic profile* [Online]. World data bank. Available: Data (worldbank.org) [Accessed 23 September 2021].

BARASUOL, J. C., SOARES, J. P., CASTRO, R. G., GIACOMIN, A., GONÇALVES, B. M., KLEIN, D., TORRES, F. M., BORGATTO, A. F., RAMOS-JORGE, M. L. & BOLAN, M. 2017. Untreated dental caries is associated with reports of verbal bullying in children 8-10 years old. *Caries research*, 51, 482-488.

BARG, K. 2019. Why are middle-class parents more involved in school than working-class parents? *Research in Social Stratification and Mobility*, 59, 14-24.

BARKER, D. 2001. Fetal and infant origins of adult disease. *Mon.schr. Kinderheilkd.*, 149, S2-S6.

BARKER, D. J. P. 1990. The fetal and infant origins of adult disease : The womb may be more important than the home. *British Medical Journal*, 301, 1111-1111.

BARKER, D. J. P. 1998. *Mothers, babies and health in later life,* Edinburgh, Churchill Livingstone.

BARKER, D. J. P. & OSMOND, C. 1986. Infant mortality, childhood nutrition, and ischaemic heart disease in England and Wales. *The Lancet*, 327, 1077-1081.

BARKER, D. J. P. & OSMOND, C. 1987. Death rates from stroke in England and Wales predicted from past maternal mortality. *British medical journal (Clinical research ed.)*, 295, 83-86.

BARNES, M. & DUCK, S. 1994. Everyday Communicative Contexts for Social Support. In: BURLESON, B. R., ALBRECHT, T. L. & SARASON, I. G. (eds.) Communication of social support: Messages, interactions, relationships, and community. Washington, DC: Sage Publications, Inc.

BAUMRIND, D. 1967. Child care practices anteceding three patterns of preschool behavior. *Genetic psychology monographs*, 75, 43-88.

BECKER, G. S. 1962. Investment in Human Capital: A Theoretical Analysis. *Journal of Political Economy*, 70, 9-49.

BECKER, G. S. 1993. A treatise on the family (enlarged edition), Harvard University Press.

BECKER, G. S. & BARRO, R. J. 1988. A reformulation of the economic theory of fertility. *The quarterly journal of economics*, 103, 1-25.

BECKER, G. S. & LEWIS, H. G. 1973. On the Interaction between the Quantity and Quality of Children. *Journal of Political Economy*, 81, S279-S288.

BECKER, G. S. & TOMES, N. 1976. Child Endowments and the Quantity and Quality of Children. *Journal of Political Economy*, 84, S143-S162.

BECKUNG, E. & HAGBERG, G. 2002. Neuroimpairments, activity limitations, and participation restrictions in children with cerebral palsy. *Developmental Medicine & Child Neurology*, 44, 309-316.

BEDOS, C., BRODEUR, J., ARPIN, S. & NICOLAU, B. 2005. Dental caries experience: A twogeneration study. *Journal of dental research*, 84, 931-936.

BELOTTI, L. & FRAZÃO, P. 2021. Effectiveness of water fluoridation in an upper middleincome country: systematic review and meta-analysis. *International journal of paediatric dentistry [Preprint]*, 1-11.

- BEN-SHLOMO, Y. & KUH, D. 2002. A life course approach to chronic disease epidemiology: conceptual models, empirical challenges and interdisciplinary perspectives. *International Journal of Epidemiology*, 31, 285-293.
- BENGTSSON, T. & LINDSTRÖM, M. 2000. Childhood misery and disease in later life: the effects of mortality in old age of hazards experienced in early life, southern Sweden, 1760–1894. *Population studies*, 54, 263-277.
- BENNER, A. D., BOYLE, A. E. & SADLER, S. 2016. Parental involvement and adolescents' educational success: The roles of prior achievement and socioeconomic status. *Journal of youth and adolescence*, 45, 1053-1064.
- BERNABE, E., FLORES-MIR, C. & SHEIHAM, A. 2007. Prevalence, intensity and extent of Oral Impacts on Daily Performances associated with self-perceived malocclusion in 11-12-year-old children. *BMC Oral Health*, 7, 6.
- BERNABE, E., SUOMINEN, A. L., NORDBLAD, A., VEHKALAHTI, M. M., HAUSEN, H., KNUUTTILA, M., KIVIMAKI, M., WATT, R. G., SHEIHAM, A. & TSAKOS, G. 2011. Education level and oral health in Finnish adults: evidence from different lifecourse models. *Journal of Clinical Periodontology*, 38, 25-32.
- BERNABÉ, E., KIVIMÄKI, M., TSAKOS, G., SUOMINEN-TAIPALE, A. L., NORDBLAD, A., SAVOLAINEN, J., UUTELA, A., SHEIHAM, A. & WATT, R. G. 2009. The relationship among sense of coherence, socio- economic status, and oral health- related behaviours among Finnish dentate adults. *European Journal of Oral Sciences*, 117, 413-418.
- BERNABÉ, E., MACRITCHIE, H., LONGBOTTOM, C., PITTS, N. B. & SABBAH, W. 2017. Birth weight, breastfeeding, maternal smoking and caries trajectories. *Journal of dental research*, 96, 171-178.
- BERNABÉ, E., TSAKOS, G. & SHEIHAM, A. 2007. Intensity and extent of Oral Impacts on Daily Performances by type of self- perceived oral problems. *European Journal of Oral Sciences*, 115, 111-116.
- BERNDT, A. E. & WILLIAMS, P. C. 2013. Hierarchical regression and structural equation modeling: two useful analyses for life course research. *Family & community health*, 36, 4-18.
- BIRUNGI, N., FADNES, L. T., KASANGAKI, A., NANKABIRWA, V., OKULLO, I., LIE, S. A., TUMWINE, J. K. & ÅSTRØM, A. N. 2017. Assessing causal effects of early life- course factors on early childhood caries in 5- year- old Ugandan children using directed acyclic graphs (DAGs): A prospective cohort study. *Community Dentistry and Oral Epidemiology*, 45, 512-521.
- BLACK, R. E., VICTORA, C. G., WALKER, S. P., BHUTTA, Z. A., CHRISTIAN, P., DE ONIS, M., EZZATI, M., GRANTHAM-MCGREGOR, S., KATZ, J., MARTORELL, R. & UAUY, R. 2013. Maternal and child undernutrition and overweight in low- income and middleincome countries. *The Lancet*, 382, 427-451.
- BLACK, S. E., DEVEREUX, P. J. & SALVANES, K. G. 2005. The More the Merrier? The Effect of Family Size and Birth Order on Children's Education. *The Quarterly Journal of Economics*, 120, 669-700.
- BLAKEMORE, S. J., BURNETT, S. & DAHL, R. E. 2010. The role of puberty in the developing adolescent brain. *Human Brain Mapping*, 31, 926-933.
- BLINKHORN, A. S. 1978. Influence of social norms on toothbrushing behavior of preschool children. *Community dentistry and oral epidemiology*, 6, 222-226.
- BLOMBERG, M., BIRCH TYRBERG, R. & KJØLHEDE, P. 2014. Impact of maternal age on obstetric and neonatal outcome with emphasis on primiparous adolescents and older women: a Swedish Medical Birth Register Study. *British Medical Journal Open, 4*, e005840-e005840.
- BLONDAL, K. S. & ADALBJARNARDOTTIR, S. 2014. Parenting in Relation to School Dropout Through Student Engagement: A Longitudinal Study: Parenting in Relation to School Dropout. *Journal of marriage and family,* 76, 778-795.

- BLUMENSHINE, S. L., VANN, W. F., GIZLICE, Z. & LEE, J. Y. 2008. Children's School Performance: Impact of General and Oral Health. *Journal of Public Health Dentistry*, 68, 82-87.
- BOGIN, B. 1997. Evolutionary hypotheses for human childhood. *American Journal of Physical Anthropology*, 104, 63-89.
- BOWEN, W. H. & LAWRENCE, R. A. 2005. Comparison of the Cariogenicity of Cola, Honey, Cow Milk, Human Milk, and Sucrose. *Pediatrics*, 116, 921-926.
- BRAGA, M., OLIVEIRA, L., BONINI, G., BÖNECKER, M. & MENDES, F. 2009. Feasibility of the International Caries Detection and Assessment System (ICDAS-II) in Epidemiological Surveys and Comparability with Standard World Health Organization Criteria. *Caries Research*, 43, 245-249.
- BRAMANTORO, T., SANTOSO, C. M. A., HARIYANI, N., SETYOWATI, D., ZULFIANA, A. A., NOR, N. A. M., NAGY, A., PRATAMAWARI, D. N. P. & IRMALIA, W. R. 2021. Effectiveness of the school-based oral health promotion programmes from preschool to high school: A systematic review. *PloS one*, 16, e0256007-e0256007.
- BROADBENT, J. M., PAGE, L. A. F., THOMSON, W. M. & POULTON, R. 2013. Permanent dentition caries through the first half of life. *British dental journal*, 215, E12-E12.
- BROADBENT, J. M., THOMŠON, W. M. & POULTON, R. 2008. Trajectory patterns of dental caries experience in the permanent dentition to the fourth decade of life. *Journal of dental research*, 87, 69-72.
- BROADBENT, J. M., ZENG, J., FOSTER PAGE, L. A., BAKER, S. R., RAMRAKHA, S. & THOMSON, W. M. 2016. Oral Health– related Beliefs, Behaviors, and Outcomes through the Life Course. *Journal of Dental Research*, 95, 808-813.
- BRONFENBRENNER, U. 1958. Socialization and social class through time and space In EE, Maccoby, T. M, Newcomb, EL Hartley,(Eds), Readings in social psychology New york Holt, Rinehart and Winston. Inc.
- BRONFENBRENNER, U. 1977. Toward an experimental ecology of human development. *American Psychologist,* 32, 513-531.
- BRONFENBRENNER, U. 1996. *The ecology of human development [electronic resource] : experiments by nature and design,* Cambridge, MA, Cambridge, MA : Harvard University Press, 1996.
- BRONFENBRENNER, U. 2005. *Making human beings human: Bioecological perspectives on human development*, sage.
- BRONFENBRENNER, U. & MORRIS, P. A. 2006. The Bioecological Model of Human Development. *Handbook of child psychology: Theoretical models of human development, Vol. 1, 6th ed.* Hoboken, NJ, US: John Wiley & Sons Inc.
- BROWNELL, M. D., EKUMA, O., NICKEL, N. C., CHARTIER, M., KOSEVA, I. & SANTOS, R.
 G. 2016. A population-based analysis of factors that predict early language and cognitive development. *Early Childhood Research Quarterly*, 35, 6-18.
- BUREAU, B. S. C. S. 2015. *Salary Scales and Allowances, Payroll Basic* [Online]. Bahrain. Available: <u>https://www.csb.gov.bh/en/salary-scales-and-allowances/payroll-basic.html</u> [Accessed March 2019].
- BURNETT, R., KYU, H. H., THOMAS, B. A., ABUBAKAR, I., ABU-RMEILEH, N. M. E., ALBITTAR, M. I., ALEMAN, A. V., ALSHARIF, U., ANDERSON, B. O., ARSENIJEVIC, V. S. A., BAKFALOUNI, T., BALALLA, S., BARQUERA, S., RUVALCABA, C. B., BEKELE, T., BELL, M. L., BHUTTA, Z. A., BORNSTEIN, N. M., BRENNER, H., BRODAY, D. M., BRUGHA, T. S., BUI, L. N., CAMPOS-NONATO, I. R., COLOMAR, M., COOPER, C., COOPER, L. T., DAVIS, A., DEGENHARDT, L., DERIBE, K., DEVRIES, K. M., DHERANI, M. K., DUAN, L., ENDRES, M., ESTEGHAMATI, A., FAY, D. F. J., GOENKA, S., GONZÁLEZ-CASTELL, D., GUGNANI, H. C., HAVMOELLER, R., HEDAYATI, M. T., HOFFMAN, H. J., HU, G., JENSEN, P. N., KARAM, N. E., KARTHIKEYAN, G., KHADER, Y. S., KHANG, Y.-H., KHONELIDZE, I., KNIBBS, L. D., KNUDSEN, A. K., DEFO, B. K., LALLUKKA, T., LARSON, H. J., MAHDI, A. A., MANGALAM, S., MARAPE, M., MARCENES, W., MASHAL, M. T., MCKEE, M., MEANEY, P. A., MEKONNEN, W., MENSAH, G. A., MISGANAW, A., MOHAMMAD,

K. A., MOTURI, W. N., MUKAIGAWARA, M., MULLANY, E. C., NEUPANE, S. P., NISAR, M. I., NOLLA, J. M., OROZCO, R., PARRY, C. D., PAVLIN, B. I., PEREZ-RUIZ, F., PERVAIZ, A., POPOVA, S., QUISTBERG, D. A. A., RAHIMI, K., ROCA, A., RONFANI, L., ROOM, R., RUHAGO, G. M., SANABRIA, J. R., SANCHEZ, L. M., SEPANLOU, S. G., SERDAR, B., SINDI, S., SKIRBEKK, V., STURUA, L., TALONGWA, R. T., TOPOUZIS, F., TRAEBERT, J., TRASANDE, L., TUZCU, E. M., VIJAYAKUMAR, L., WAN, X., WILLIAMS, T. N., WOLDEYOHANNES, S. M., YOUNOUSSI, Z. & ZAKI, M. E. 2015. Global, regional, and national comparative risk assessment of 79 behavioural, environmental and occupational, and metabolic risks or clusters of risks in 188 countries, 1990- 2013: A systematic analysis for the Global Burden of Disease Study 2013. *The Lancet (British edition)*, 386, 2287-2323.

- BURT, B. A. & PAI, S. 2001. Sugar consumption and caries risk: a systematic review. *Journal of dental education,* 65, 1017-1023.
- BURTON JEANGROS, C., CULLATI, S., SACKER, A. & BLANE, D. 2015. A Life Course Perspective on Health Trajectories and Transitions, Cham, Cham: Springer.
- BUTANI, Y., GANSKY, S. A. & WEINTRAUB, J. A. 2009. Parental perception of oral health status of children in mainstream and special education classrooms. *Special Care in Dentistry*, 29, 156-162.
- BUTLER, Y. G., PENG, X. & LEE, J. 2021. Young learners' voices: Towards a learnercentered approach to understanding language assessment literacy. *Language testing*, 38, 429-455.
- BYRNE, B. M. 2001. *Structural equation modeling with AMOS : basic concepts, applications, and programming,* Mahwah, N.J. ; London, Mahwah, N.J. ; London : Lawrence Erlbaum Associates, 2001.
- CAIRD, J., KAVANAGH, J., O'MARA-EVES, A., OLIVER, K., OLIVER, S., STANSFIELD, C. & THOMAS, J. 2014. Does being overweight impede academic attainment? A systematic review. *Health education journal*, 73, 497-521.
- CASAMÁSSIMO, P. S., THIKKURISSY, S., EDELSTEIN, B. L. & MAIORINI, E. 2009. Beyond the dmft: The Human and Economic Cost of Early Childhood Caries. *Journal of the American Dental Association*, 140, 650-657.
- CASELLI, G. 1990. Measurement and analysis of mortality-new approaches. 229.
- CASTILHO, A. R. F. D., MIALHE, F. L., BARBOSA, T. D. S. & PUPPIN-RONTANI, R. M. 2013. Influence of family environment on children's oral health: a systematic review. *Jornal de Pediatria*, 89, 116-123.
- CAUFIELD, P., LI, Y. & BROMAGE, T. 2012. Hypoplasia- associated Severe Early Childhood Caries - A Proposed Definition. *Journal of dental research*, 91, 544-550.
- CAVANAGH, S. E. & FOMBY, P. 2012. Family Instability, School Context, and the Academic Careers of Adolescents. *Sociology of Education*, 85, 81-97.
- CDC. 2019. BMI Percentile Calculator for Child and Teen [Online]. US: Centres for Diseases Control and Prevention. Available: <u>https://www.cdc.gov/healthyweight/bmi/calculator.html</u> [Accessed 3 March 2019].
- CELESTE, R. K., DARIN-MATTSSON, A., LENNARTSSON, C., LISTL, S., PERES, M. A. & FRITZELL, J. 2021. Social Mobility and Tooth Loss: A Systematic Review and Metaanalysis. *Journal of dental research* [Online].
- CEYHAN, D., AKDIK, C. & KIRZIOGLU, Z. 2018. An educational programme designed for the evaluation of effectiveness of two tooth brushing techniques in preschool children. *European journal of paediatric dentistry*, 19, 181-186.
- CHAMBERS, E. A. & SCHREIBER, J. B. 2004. Girls' academic achievement: varying associations of extracurricular activities. *Gender and education*, 16, 327-346.
- CHEN, K. J., GAO, S. S., DUANGTHIP, D., LO, E. C. M. & CHU, C. H. 2019. Prevalence of early childhood caries among 5-year-old children: A systematic review. *Journal of investigative and clinical dentistry*, 10, e12376.
- CHEN, X.-K., WEN, S. W., FLEMING, N., DEMISSIE, K., RHOADS, G. G. & WALKER, M. 2007. Teenage pregnancy and adverse birth outcomes: a large population based retrospective cohort study. *International journal of epidemiology*, 36, 368-373.

- CHOWDHURY, R., SINHA, B., SANKAR, M. J., TANEJA, S., BHANDARI, N., ROLLINS, N., BAHL, R. & MARTINES, J. 2015. Breastfeeding and maternal health outcomes: a systematic review and meta-analysis. *Acta Paediatr*, 104, 96-113.
- CINAR, A. B., KOSKU, N., SANDALLI, N. & MURTOMAA, H. 2008a. Individual and maternal determinants of self-reported dental health among Turkish school children aged 10-12 years. *Community Dental Health*, 25, 84-88.
- CINAR, A. B., MURTOMAA, H. & TSEVEENJAV, B. 2008b. The Life-course Approach in Assessment of Dental Health: A Cross Sectional Study among Finnish and Turkish Pre-adolescents. *European journal of dentistry*, 2, 153-160.
- CLARKE, B. & YOUNAS, F. 2017. Helping Parents to Parent: 20 February 2017. In: SOCIAL MOBILITY COMMISSION (ed.).
- CLARKSON, J. & O'MULLANE, D. 1989. A modified DDE Index for use in epidemiological studies of enamel defects. *Journal of Dental Research*, 68, 445-450.
- COATES, A. E., HARDMAN, C. A., HALFORD, J. C. G., CHRISTIANSEN, P. & BOYLAND, E. J. 2019. Food and Beverage Cues Featured in YouTube Videos of Social Media Influencers Popular With Children: An Exploratory Study. *Frontiers in psychology*, 10, 2142-2142.
- COHEN, J. 1960. A Coefficient of Agreement for Nominal Scales. *Educational and Psychological Measurement*, 20, 37-46.
- COHEN, J. 1968. Weighted kappa: Nominal scale agreement provision for scaled disagreement or partial credit. *Psychological Bulletin,* 70, 213-220.
- COHEN, P., WEST, S. G. & AIKEN, L. S. 1983. Applied multiple regression/Correlation Analysis for the Behavioral Sciences, Psychology Press New York, USA.
- COLCHERO, M. A., RIVERA-DOMMARCO, J., POPKIN, B. M. & NG, S. W. 2017. In Mexico, evidence of sustained consumer response two years after implementing a sugarsweetened beverage tax. *Health Affairs*, 36, 564-571.
- COLEMAN, J. 1988. Social capital in the creation of human capital. *American Journal of Sociology*, 94, S95-120.
- COLEY, R. J. 2002. An Uneven Start: Indicators of Inequality in School Readiness. Policy Information Report. ERIC.
- CONGER, R. D. & DONNELLAN, M. B. 2007. An interactionist perspective on the socioeconomic context of human development. *Annu Rev Psychol*, 58, 175-199.
- CONNELL, J. P., HALPEM-FELSHER, B. L., CLIFFORD, E., CRICHLOW, W. & USINGER, P. 1995. Hanging in There: Behavioral, Psychological, and Contextual Factors Affecting Whether African American Adolescents Stay in High School. *Journal of adolescent research*, 10, 41-63.
- CORTESI, S. & GASSER, U. 2015. Youth online and news: A phenomenological view on diversity. *International journal of communication*, 9, 1425-1448.
- CORVILLE-SMITH, J., RYAN, B. A., ADAMS, G. R. & DALICANDRO, T. 1998. Distinguishing Absentee Students from Regular Attenders: The Combined Influence of Personal, Family, and School Factors. *Journal of youth and adolescence*, 27, 629-640.
- COSTA, F. S., SILVEIRA, E. R., PINTO, G. S., NASCIMENTO, G. G., THOMSON, W. M. & DEMARCO, F. F. 2017. Developmental defects of enamel and dental caries in the primary dentition: A systematic review and meta-analysis. *Journal of dentistry*, 60, 1-7.
- COSTELLO, J. & ANGOLD, A. 2007. Measurement and design for life course studies of individual differences and development. *Epidemiological methods in life course research*, 27-61.
- CRESWELL, J. W. & CRESWELL, J. D. 2018. *Research design : qualitative, quantitative & mixed methods approaches,* Los Angeles, SAGE.
- CROSNOE, R., JOHNSON, M. K. & ELDER JR, G. H. 2004. School Size and the Interpersonal Side of Education: An Examination of Race/Ethnicity and Organizational Context. *Social science quarterly*, 85, 1259-1274.

- CUI, L., LI, X., TIAN, Y., BAO, J., WANG, L., XU, D., ZHAO, B. & LI, W. 2017. Breastfeeding and early childhood caries: a meta-analysis of observational studies. *Asia Pacific journal of clinical nutrition*, 26, 867-880.
- CUNHÁ, I. P. D., PEREIRA, A. C., MENEGHIM, M. D. C., FRIAS, A. C. & MIALHE, F. L. 2019. Association between social conditions and oral health in school failure. *Revista de saúde pública* [Online], 53.
- CUNNION, D. T., SPIRO, A., JONES, J. A., RICH, S. E., PAPAGEORGIOU, C. P., TATE, A., CASAMASSIMO, P., HAYES, C. & GARCIA, R. I. 2010. Pediatric oral health-related quality of life improvement after treatment of early childhood caries: a prospective multisite study. *Journal of dentistry for children*, 77, 4-11.
- DA SILVA, A. N., ALVARES DE LIMA, S. T. & VETTORE, M. V. 2018. Protective psychosocial factors and dental caries in children and adolescents: a systematic review and meta-analysis. *International journal of paediatric dentistry*, 28, 443-458.
- DABELEA, D., HANSON, R. L., LINDSAY, R. S., PETTITT, D. J., IMPERATORE, G., GABIR, M. M., ROUMAIN, J., BENNETT, P. H. & KNOWLER, W. C. 2000. Intrauterine exposure to diabetes conveys risks for type 2 diabetes and obesity: a study of discordant sibships. *Diabetes*, 49, 2208-2211.
- DANCEY, C. P. & REIDY, J. 2007. *Statistics without maths for psychology*, Pearson education.
- DANTAS, L. R., GOMES, M. C., DANTAS, L. R., CRUZ-DA-SILVA, B. R., DE F. PERAZZO, M., SIQUEIRA, M. B. L. D., PAIVA, S. M. & GRANVILLE-GARCIA, A. F. 2015. The impact of dental treatment on oral health-related quality of life among preschool children. *Journal of public health*, 23, 327-331.
- DARLEY, R. M., KARAM, S. A., COSTA, F. D. S., CORREA, M. B. & DEMARCO, F. F. 2021. Association between dental pain, use of dental services and school absenteeism: 2015 National School Health Survey, Brazil. *Epidemiol Serv Saude*, 30, e2020108e2020108.
- DARLING, N. & STEINBERG, L. 1993. Parenting style as context: An integrative model. *Psychological bulletin*, 113, 487-496.
- DARNTON-HILL, I., NISHIDA, C. & JAMES, W. P. T. 2004. A life course approach to diet, nutrition and the prevention of chronic diseases. *Public health nutrition*, 7, 101-121.
- DAVID, J., ÅSTRØM, A. N. & WANG, N. J. 2006. Prevalence and correlates of self-reported state of teeth among schoolchildren in Kerala, India. *BMC Oral Health*, 6, 10.
- DAVIES, G. M., WORTHINGTON, H. V., ELLWOOD, R. P., BENTLEY, E. M., BLINKHORN, A. S., TAYLOR, G. O. & DAVIES, R. M. 2002. A randomised controlled trial of the effectiveness of providing free fluoride toothpaste from the age of 12 months on reducing caries in 5-6-year-old children. *Community dental health*, 19, 131-136.
- DAY, F., BULIK-SULLIVAN, B., HINDS, D., FINUCANE, H., MURABITO, J., TUNG, J., ONG, K. & PERRY, J. 2015. Shared genetic aetiology of puberty timing between sexes and with health- related outcomes. *Nature Communications*, 6, 8842.
- DE ALMEIDA, M. E. L., TEIXEIRA, A. K. M., ALENCAR, C. H., PAIVA, S. M. & DE ABREU, M. H. N. G. 2013. Agreement between parents and adolescents on dental fluorosis: A population-based study. *Brazilian oral research*, 27, 91-96.
- DE PAULA, J. S., AMBROSANO, G. M. B. & MIALHE, F. L. 2015a. Oral Disorders, Socioenvironmental Factors and Subjective Perception Impact on Children's School Performance. *Oral health & preventive dentistry*, 13, 219-226.
- DE PAULA, J. S. & MIALHE, F. L. 2013. Impact of oral health conditions on school performance and lost school days by children and adolescents: What are the actual pieces of evidence? *Brazilian Journal of Oral Sciences*, 12, 189-198.
- DE PAULA, J. S., SARRACINI, K. L. M., MENEGHIM, M. C., PEREIRA, A. C., ORTEGA, E. M. M., MARTINS, N. S. & MIALHE, F. L. 2015b. Longitudinal evaluation of the impact of dental caries treatment on oral health-related quality of life among schoolchildren. *European journal of oral sciences*, 123, 173-178.

- DE SOUZA, A. L., LEAL, S. C., BRONKHORST, E. M. & FRENCKEN, J. E. F. M. 2014. Assessing caries status according to the CAST instrument and WHO criterion in epidemiological studies. *BMC Oral Health*, 14, 119-119.
- DECLERCK, D., LEROY, R., MARTENS, L., LESAFFRE, E., GARCIA-ZATTERA, M. J., BROUCKE, S. V., DEBYSER, M. & HOPPENBROUWERS, K. 2008. Factors associated with prevalence and severity of caries experience in preschool children. *Community Dentistry and Oral Epidemiology*, 36, 168-178.
- DEL GIUDICE, M., ANGELERI, R. & MANERA, V. 2009. The juvenile transition: A developmental switch point in human life history. *Developmental Review*, 29, 1-31.
- DELANEY, L., MCGOVERN, M. & SMITH, J. P. 2011. From Angela's ashes to the Celtic tiger: Early life conditions and adult health in Ireland. *Journal of Health Economics*, 30, 1-10.
- DEPARTMENT FOR EDUCATION. 2016. School Attendance: Guidance for Maintained Schools, Academies, Independent Schools and Local Authorities. Available: <u>https://www.gov.uk/government/publications/school-attendance</u>.
- DEPARTMENT OF HEALTH UK 2007. Review of the health inequalities infant mortality PSA target. Department of Health Publications.
- DEPARTMENT OF HEALTH-UK. 2021. *Delivering better oral health: an evidence-based toolkit for prevention: a review* [Online]. UK: Department of health. Available: <u>https://www.gov.uk/government/publications/delivering-better-oral-health-an-</u>evidence-based-toolkit-for-prevention [Accessed January 24 2021].
- DETTY, A. M. R. & OZA-FRANK, R. 2014. Oral health status and academic performance among O hio third- graders, 2009- 2010. *Journal of Public Health Dentistry*, 74, 336-342.
- DICKENS, C. 1949. A tale of two cities [1859], Gawthorn.
- DILLON, W. R., KUMAR, A. & MULANI, N. 1987. Offending estimates in covariance structure analysis: Comments on the causes of and solutions to Heywood cases. *Psychological bulletin*, 101, 126-135.
- DIVYALALITHA, N., RAJMOHAN, M., PRABU, D., BHARATHWAJ, V. & MANIPAL, S. 2019. School Based Fluoride Mouth Rinse Program and its Effect on Dental Caries - A Systematic Review. *International Journal of Pharmaceutical Sciences Review and Research*, 58, 154-157.
- DONKIN, A. 2014. Local Action of Health Inequalities: Good Quality Parenting Programmes and the Home to School Transition. *Public Health England, UCL Institute for Health Equality*.
- DONNER, A. 1982. The Relative Effectiveness of Procedures Commonly Used in Multiple Regression Analysis for Dealing with Missing Values. *The American statistician*, 36, 378-381.
- DORNAN, P. & WOODHEAD, M. 2015. How inequalities develop through childhood: Lifecourse evidence from young lives cohort study. Italy: Innocenti Discussion Paper No. 2015-01, UNICEF Office of Research.
- DUIJSTER, D., JONG-LENTERS, M., RUITER, C., THIJSSEN, J., LOVEREN, C. & VERRIPS, E. 2015. Parental and family-related influences on dental caries in children of Dutch, Moroccan and Turkish origin. *Community Dentistry and Oral Epidemiology*, 43, 152-162.
- DUNCAN, G., KALIL, A. & ZIOL-GUEST, K. 2013. Early childhood poverty and adult achievement, employment and health. *Family matters (Melbourne, Vic.)*, 93, 27-35.
- DURMUŞ, B., KRUITHOF, C. J., GILLMAN, M. H., WILLEMSEN, S. P., HOFMAN, A., RAAT, H., EILERS, P. H. C., STEEGERS, E. A. P. & JADDOE, V. W. V. 2011. Parental smoking during pregnancy, early growth, and risk of obesity in preschool children: The Generation R Study. *American journal of clinical nutrition*, 94, 164-171.
- DYE, B. A., THORNTON-ÉVANS, G., LI, X. & IAFOLLA, T. J. 2015. Dental caries and sealant prevalence in children and adolescents in the United States, 2011-2012. *In:* STATISTICS, N. C. F. H. (ed.) *NCHS Data Brief.* US Department of Health and Human Services, Centers for Disease Control and and Prevention.

- EDUCATION AND QUALITY AUTHORITY-KOB. 2019. School's quality review reports [Online]. Bahrain's Ministry of Education: Education and Training Quality Authority. Available: <u>http://www.bqa.gov.bh/En/Reports/SchoolsReports/Pages/default.aspx</u> [Accessed 6 April 2019].
- EGRI, M. & GUNAY, O. 2004. Association between some educational indicators and dental caries experience of 12-year-old children in developing countries: An ecological approach. *Community Dental Health,* 21, 227-229.
- EKMAN, P. 2004. Emotions revealed. *Bmj*, 328.
- EKSTRAND, K. R., GIMENEZ, T., FERREIRA, F. R., MENDES, F. M. & BRAGA, M. M. 2018. The International Caries Detection and Assessment System - ICDAS: A Systematic Review. *Caries Research*, 52, 406-419.
- EKSTROM, R., GOERTZ, M., POLLACK, J. & ROCK, D. 1986. Who drops out of high school and why? Findings from a national study. *Teachers College Record*, 87, 356-373.
- EL FADL, R. A., BLAIR, M. & HASSOUNAH, S. 2016. Integrating maternal and children's oral health promotion into nursing and midwifery practice A systematic review. *PLoS One*, 11, e0166760-e0166760.
- ELDER, G. H. 1974. *Children of the Great Depression: Social Change in Life Experience,* Chicago, University of Chicago Press.
- ELDER, G. H. 1998. *The life course and human development,* Hoboken, NJ, US, John Wiley & Sons Inc.
- ELDER, G. H., JOHNSON, M. K. & CROSNOE, R. 2003. The Emergence and Development of Life Course Theory. *In:* MORTIMER, J. T. & SHANAHAN, M. J. (eds.) *Handbook of the Life Course.* Springer, Boston, MA.
- ELDER, G. H., VAN NGUYEN, T. & CASPI, A. 1985. Linking Family Hardship to Children's Lives. *Child Development*, 56, 361-375.
- ELFORD, J., SHAPER, A. G. & WHINCUP, P. 1992. Early life experience and cardiovascular disease-- ecological studies. *Journal of Epidemiology and Community Health*, 46, 1-8.
- ELIDRISSI, S. M. & NAIDOO, S. 2016. Prevalence of dental caries and toothbrushing habits among preschool children in Khartoum State, Sudan. *International dental journal,* 66, 215-220.
- ELSALHY, M., ALI, U., LAI, H., FLORES-MIR, C. & AMIN, M. 2019. Caries reporting in studies that used the International Caries Detection and Assessment System: A scoping review. *Community dentistry and oral epidemiology*, 47, 92-102.
- ELZEIN, R., CHOUERY, E., ABDEL-SATER, F., BACHO, R. & AYOUB, F. 2020. Molar incisor hypomineralisation in Lebanon: prevalence and clinical characteristics. *European Archives of Paediatric Dentistry*, 21, 609-616.
- EPSTEIN, D., ELWOOD, J., HEY, V. & MAW, J. 1998. *Failing Boys?: Issues in Gender and Achievement*, Open University Press.
- EPSTEIN, J. L. & DAUBER, S. L. 1991. School Programs and Teacher Practices of Parent Involvement in Inner- City Elementary and Middle Schools. *The Elementary School Journal*, 91, 289-305.
- EPSTEIN, J. L. & SHELDON, S. B. 2002. Present and Accounted for: Improving Student Attendance Through Family and Community Involvement. *The Journal of educational research (Washington, D.C.),* 95, 308-318.
- ERICKSON, K., DREVETS, W. & SCHULKIN, J. 2003. Glucocorticoid regulation of diverse cognitive functions in normal and pathological emotional states. *Neuroscience and Biobehavioral Reviews*, 27, 233-246.
- ERIKSON, E. H. 1980. *Identity and the life cycle*, New York, W.W. Norton & Co.
- ERIKSSON, J. G., FORSÉN, T., TUOMILEHTO, J., OSMOND, C. & BARKER, D. J. 2001. Early growth and coronary heart disease in later life: longitudinal study. *British Medical Journal*, 322, 949-953.
- EVANS, G. W., RICCIUTI, H. N., HOPE, S., SCHOON, I., BRADLEY, R. H., CORWYN, R. F. & HAZAN, C. 2010. Crowding and Cognitive Development: The Mediating Role of Maternal Responsiveness Among 36- Month- Old Children. *Environment and Behavior*, 42, 135-148.

- FARAJDOKHT, F., SADIGH-ETEGHAD, S., DEHGHANI, R., MOHADDES, G., ABEDI, L., BUGHCHECHI, R., MAJDI, A. & MAHMOUDI, J. 2017. Very low birth weight is associated with brain structure abnormalities and cognitive function impairments: A systematic review. *Brain and Cognition*, 118, 80-89.
- FAROOQI, F. A., KHABEER, A., MOHEET, I. A., KHAN, S. Q., FAROOQ, I. & ARREJAIE, A. S. 2015. Prevalence of dental caries in primary and permanent teeth and its relation with tooth brushing habits among schoolchildren in Eastern Saudi Arabia. Saudi medical journal, 36, 737-742.
- FAYED, A. A., WAHABI, H., MAMDOUH, H., KOTB, R. & ESMAEIL, S. 2017. Demographic profile and pregnancy outcomes of adolescents and older mothers in Saudi Arabia: analysis from Riyadh Mother (RAHMA) and Baby cohort study. *BMJ Open,* 7, e016501-e016501.
- FEATHERMAN, D. L. 1981. The life-span perspective in social science research.
- FEKETE, K., BERTI, C., TROVATO, M., LOHNER, S., DULLEMEIJER, C., SOUVEREIN, O., CETIN, I. & DECSI, T. 2012. Effect of folate intake on health outcomes in pregnancy: a systematic review and meta- analysis on birth weight, placental weight and length of gestation. *Nutrition journal*, 11, 75.
- FELDENS, C. A., VÍTOLO, M. R. & DRACHLER, M. D. L. 2007. A randomized trial of the effectiveness of home visits in preventing early childhood caries. *Community dentistry and oral epidemiology*, 35, 215-223.
- FORNANDER, M. J. & KEARNEY, C. A. 2019. Family environment variables as predictors of school absenteeism severity at multiple levels: Ensemble and classification and regression tree analysis. *Frontiers in psychology*, 10, 2381-2381.

FORREST, C. B., BEVANS, K. B., RILEY, A. W., CRESPO, R. & LOUIS, T. A. 2013. Health and School Outcomes During Children's Transition Into Adolescence. *Journal of Adolescent Health*, 52, 186-194.

- FORSDAHL, A. 1977. Are poor living conditions in childhood and adolescence an important risk factor for arteriosclerotic heart disease? *British Journal of Preventive & Social Medicine*, 31, 91-95.
- FREDRICKS, J. A., BLUMENFELD, P. C. & PARIS, A. H. 2004. School Engagement: Potential of the Concept, State of the Evidence. *Review of Educational Research*, 74, 59-109.
- FREIRE, M., SHEIHAM, A. & NETUVELI, G. 2008. Relationship between Height and Dental Caries in Adolescents. *Caries Research*, 42, 134-40.
- FRESH. 2013. Focusing Resources on Effective School Health [Online]. UNESCO, UNICEF,

 WHO,
 The
 World
 Bank.
 Available:

 <u>https://healtheducationresources.unesco.org/sites/default/files/resources/FRESH_M</u>

 <u>%26E_CORE_INDICATORS.pdf</u> [Accessed 22 November 2021].
- FRIDLIZIUS, G. 1989. The deformation of cohorts: nineteenth-century mortality decline in a generational perspective. *The Scandinavian economic history review*, 37, 3-17.
- FTEITA, D., ALI, A. & ALALUUSUA, S. 2006. Molar-incisor hypomineralization (MIH) in a group of school-aged children in Benghazi, Libya. *European archives of paediatric dentistry*, 1, 92-95.
- FURSTENBERG, F. F., COOK, T. D., ECCLES, J. & ELDER, G. H. 2000. *Managing to make it: Urban families and adolescent success*, University of Chicago Press.
- GARG, N., ANANDAKRISHNA, L. & CHANDRA, P. 2012. Is there an Association between Oral Health Status and School Performance? A Preliminary Study. *International journal of clinical pediatric dentistry*, 5, 132-135.
- GARRIGA, A. & MARTÍNEZ-LUCENA, J. 2018. Growing Up in A Single Mother Family and Student's Tardiness: A Cross-National Study Exploring the Moderating Role of Family Resources. *Journal of Divorce & Remarriage*, 59, 1-24.
- GENNETIAN, L. A., RODRIGUES, C., HILL, H. D. & MORRIS, P. A. 2018. Stability of income and school attendance among NYC students of low-income families. *Economics of education review*, 63, 20-30.
- GEORGE, A., DAHLEN, H. G., BLINKHORN, A., AJWANI, S., BHOLE, S., ELLIS, S., YEO, A., ELCOMBE, E. & JOHNSON, M. 2018. Evaluation of a midwifery initiated oral

health-dental service program to improve oral health and birth outcomes for pregnant women: A multi-centre randomised controlled trial. *International journal of nursing studies*, 82, 49-57.

- GEORGE, L. K. 1993. Sociological Perspectives on Life Transitions. *Annual review of sociology*, 19, 353-373.
- GHARIB, N. & RASHEED, P. 2011. Energy and macronutrient intake and dietary pattern among school children in Bahrain: A cross-sectional study. *Nutrition journal,* 10, 62-62.
- GHERUNPONG, S., TSAKOS, G. & SHEIHAM, A. 2004. Developing and evaluating an oral health-related quality of life index for children; the CHILD-OIDP. *Community dental health*, 21, 161-169.
- GHERUNPONG, S., TSAKOS, G. & SHEIHAM, A. 2006. A sociodental approach to assessing dental needs of children: concept and models. *International journal of paediatric dentistry*, 16, 81-88.
- GIELE, J. Z. & ELDER, G. H. 1998. *Methods of Life Course Research: Qualitative and Quantitative Approaches*, Thousand Oaks: SAGE Publications.
- GIFT, H., REISINE, S. & LARACH, D. 1992. The Social Impact of Dental Problems and Visits. *American Journal of Public Health*, 82, 1663-1668.
- GILBERT, B. B. 1993. The evolution of national insurance in Great Britain : the origins of the welfare state, Gregg Revivals; New edition.
- GOPALAN, T., ASOKAN, S., JOHN, J. & GEETHA PRIYA, P. 2018. School absenteeism, academic performance, and self-esteem as proxy measures of oral health status: A cross-sectional study. *Journal of the Indian Society of Pedodontics and Preventive Dentistry*, 36, 339-346.
- GRADELLA, C. M. F., BERNABÉ, E., BÖNECKER, M. & OLIVEIRA, L. B. 2011. Caries prevalence and severity, and quality of life in Brazilian 2- to 4- year- old children. *Community Dentistry and Oral Epidemiology*, 39, 498-504.
- GREGORY, S. 2014. Delivering Better Oral Health: what the third edition offers. *Dental Nursing*, 10, 682-686.
- GROVE, B. J., LIM, S. J., GALE, C. R. & SHENKIN, S. D. 2017. Birth weight and cognitive ability in adulthood: A systematic review and meta-analysis. *Intelligence (Norwood)*, 61, 146-158.
- GUARNIZO-HERRENO, C. C., LYU, W. & WEHBY, G. L. 2019. Children's Oral Health and Academic Performance: Evidence of a Persisting Relationship Over the Last Decade in the United States. *The Journal of pediatrics*, 209, 183-189.e2.
- GUARNIZO-HERREÑO, C. C. & WEHBY, G. L. 2012. Children's Dental Health, School Performance, and Psychosocial Well-Being. *The Journal of pediatrics*, 161, 1153-1159.e2.
- GUBBELS, J., VAN DER PUT, C. E. & ASSINK, M. 2019. Risk Factors for School Absenteeism and Dropout: A Meta-Analytic Review. *Journal of youth and adolescence*, 48, 1637-1667.
- GUILLEMIN, F., BOMBARDIER, C. & BEATON, D. 1993. Cross- cultural adaptation of healthrelated quality of life measures: Literature review and proposed guidelines. *Journal of Clinical Epidemiology*, 46, 1417-1432.
- GUNZLER, D., CHEN, T., WU, P. & ZHANG, H. 2013. Introduction to mediation analysis with structural equation modeling. *Shanghai Archives of Psychiatry*, 25, 390-394.
- HABIBIAN, M., ROBERTS, G., LAWSON, M., STEVENSON, R. & HARRIS, S. 2001. Dietary habits and dental health over the first 18 months of life. *Community dentistry and oral epidemiology*, 29, 239-246.
- HAIR, J. F., BLACK, W. C., BABIN, B. J. & ANDERSON, R. E. 2014. *Multivariate data analysis,* England, Harlow : Pearson Education Limited.
- HAJ-YAHIA, M. M. 1997. Culturally sensitive supervision of Arab social work students in Western universities. *Social work (New York)*, 42, 166-174.

- HALES, C. N., BARKER, D. J., CLARK, P. M., COX, L. J., FALL, C., OSMOND, C. & WINTER, P. D. 1991. Fetal and infant growth and impaired glucose tolerance at age 64. *British Medical Journal*, 303, 1019-1022.
- HALFON, N., FORREST, C. B., LERNER, R. M. & FAUSTMAN, E. M. 2018. Handbook of Life Course Health Development [electronic resource]. First Edition ed.: Cham : Springer International Publishing : Imprint: Springer.
- HAMM, J. V., FARMER, T. W., LAMBERT, K. & GRAVELLE, M. 2014. Enhancing Peer Cultures of Academic Effort and Achievement in Early Adolescence: Promotive Effects of the SEALS Intervention. *Developmental Psychology*, 50, 216-228.
- HANCOCK, K. J., MITROU, F., TAYLOR, C. L. & ZUBRICK, S. R. 2018. The Diverse Risk Profiles of Persistently Absent Primary Students: Implications for Attendance Policies in Australia. *Journal of education for students placed at risk*, 23, 53-69.
- HANSON, C. L., CRANDALL, A., BARNES, M. D., MAGNUSSON, B., NOVILLA, M. L. B. & KING, J. 2019. Family-focused public health: supporting homes and families in policy and practice. *Frontiers in public health*, 7, 59-59.
- HANSON, M. A., BARDSLEY, A., DE-REGIL, L. M., MOORE, S. E., OKEN, E., POSTON, L., MA, R. C., MCAULIFFE, F. M., MALETA, K., PURANDARE, C. N., YAJNIK, C. S., RUSHWAN, H. & MORRIS, J. L. 2015. The International Federation of Gynecology and Obstetrics (FIGO) recommendations on adolescent, preconception, and maternal nutrition: "Think Nutrition First". *International Journal of Gynecology and Obstetrics*, 131, S213-S213.
- HATTIE, J. 2009. Visible learning [electronic resource] : a synthesis of over 800 meta-analyses relating to achievement. London ; New York : Routledge.
- HATTON, T. J. & MARTIN, R. M. 2010. The effects on stature of poverty, family size, and birth order: British children in the 1930s. *Oxford Economic Papers*, 62, 157-184.
- HAYES, A. F. 2009. Beyond Baron and Kenny: Statistical Mediation Analysis in the New Millennium. *Communication monographs*, 76, 408-420.
- HAZIM, R., MOHAMMAD, A.-M. & MOHAMMED, H. 2018. The prevalence of molar-incisor hypomineralization in primary schoolchildren aged 7-9 years in Qassim Region of Saudi Arabia. *Journal of interdisciplinary dentistry*, 8, 44-48.
- HEARD, H. E. 2007. The Family Structure Trajectory and Adolescent School Performance: Differential Effects by Race and Ethnicity. *Journal of Family Issues*, 28, 319-354.
- HEILBRUNN-LANG, A. Y., CARPENTER, L. M., DE SILVA, A. M., MEYENN, L. K., LANG, G., RIDGE, A., PERRY, A., COLE, D. & HEGDE, S. 2020. Family-centred oral health promotion through Victorian child-health services: a pilot. *Health promotion international*, 35, 279-289.
- HEILMANN, A., TSAKOS, G. & WATT, R. G. 2015. Oral health over the life course. A life course perspective on health trajectories and transitions, 39-59.
- HENRY, K. L. 2007. Who's Skipping School: Characteristics of Truants in 8th and 10th Grade. *J Sch Health*, 77, 29-35.
- HERTZMAN, C. & POWER, C. 2003. Health and Human Development: Understandings From Life-Course Research. *Developmental Neuropsychology*, 24, 719-744.
- HOBDELL, M., PETERSEN, P. E., CLARKSON, J. & JOHNSON, N. 2003. Global goals for oral health 2020. *International dental journal*, 53, 285-288.
- HOCHBERG, Z. E. 2010. Evo- Devo of Child Growth III: Premature Juvenility as an Evolutionary Trade- Off. *Hormone Research in Paediatrics*, 73, 430-437.
- HOEK, H. W., BROWN, A. S. & SUSSER, E. 1998. The Dutch Famine and schizophrenia spectrum disorders. *Social Psychiatry and Psychiatric Epidemiology*, 33, 373-379.
- HOFSTEDE, G. 2001. Culture's consequences: Comparing values, behaviors, institutions and organizations across nations, Sage publications.
- HONG, C. L., BROADBENT, J. M., THOMSON, W. M. & POULTON, R. 2020. The Dunedin Multidisciplinary Health and Development Study: oral health findings and their implications. *Journal of the Royal Society of New Zealand*, 50, 35-46.

- HONKALA, E., RUNNEL, R., HONKALA, S., OLAK, J., VAHLBERG, T., SAAG, M. & MÄKINEN, K. K. 2011. Measuring dental caries in the mixed dentition by ICDAS. *International journal of dentistry*, 2011, 1-6.
- HOOLEY, M., SKOUTERIS, H., BOGÁNIN, C., SATUR, J. & KILPATRICK, N. 2012. Parental influence and the development of dental caries in children aged 0–6 years: A systematic review of the literature. *Journal of Dentistry*, 40, 873-885.
- HOWENSTEIN, J., KUMAR, A., CASAMASSIMO, P. S., MCTIGUE, D., COURY, D. & YIN, H. 2015. Correlating parenting styles with child behavior and caries. *Pediatric dentistry*, 37, 59-64.
- HSIN, A. & XIE, Y. 2017. Life- course changes in the mediation of cognitive and non-cognitive skills for parental effects on children's academic achievement. *Social science research*, 63, 150-165.
- HU, L. T. & BENTLER, P. M. 1999. Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural equation modeling: a multidisciplinary journal*, 6, 1-55.
- HUNTINGTON, N. L., SPETTER, D., JONES, J. A., RICH, S. E., GARCIA, R. I. & SPIRO III, A. 2011. Development and validation of a measure of pediatric oral health-related quality of life: the POQL. *Journal of Public Health Dentistry*, 71, 185-193.
- HUSSAIN, G., AL-HALABI, M., KOWASH, M. & HASSAN, A. 2018. The prevalence and severity of molar incisor hypomineralization and molar hypomineralization in Dubai, UAE. *Journal of Dentistry for Children*, 85, 102-107.
- HUSSEY, J. M., CHANG, J. J. & KOTCH, J. B. 2006. Child maltreatment in the United States: prevalence, risk factors, and adolescent health consequences. *Pediatrics (Evanston)*, 118, 933-942.
- HUXLEY, R., OWEN, C. G., WHINCUP, P. H., COOK, D. G., RICH-EDWARDS, J., SMITH, G. D. & COLLINS, R. 2007. Is birth weight a risk factor for ischemic heart disease in later life? *American journal of clinical nutrition*, 85, 1244-1250.
- HUXLEY, R. R., SHIELL, A. W. & LAW, C. M. 2000. The role of size at birth and postnatal catch-up growth in determining systolic blood pressure: a systematic review of the literature. *Journal of hypertension*, 18, 815-831.
- INFORMATION & EGOVERNMENT AUTHORITY-KOB. 2019. About Bahrain [Online]. Bahrain: EGovernment Portal. Available: https://www.bahrain.bh/wps/portal/!ut/p/a1/04_Sj9CPykssy0xPLMnMz0vMAfGjzOI9

A3MDI0sjLwsfEyNDRx9wkzMnA28DU3CzIEKIpEVGBgbuxo4-

puZB5iHWRqZWRgQp9 AyNfZ0NMEqN X18DA0clpyCfEzdnAwN2YSP04gCNB-4NT8 TD9aNQIWHxBVgBPmdCFOB2R0FuaGhEhWcmAC4pSgY!/dl5/d5/L2dBISEv Z0FBIS9nQSEh/ [Accessed 24 March 2019 2019].

- INFORMATION & EGOVERNMENT AUTHORITY-KOB. 2020a. *Bahrain open data portal* [Online]. Bahrain: General directorate of statistics and population registry. Available: <u>https://data.gov.bh/en/ResourceCenter</u> [Accessed 14 June 2021 2021].
- INFORMATION & EGOVERNMENT AUTHORITY-KOB. 2020b. Compulsory Education [Online]. Bahrain: EGovernment Portal. Available: https://www.bahrain.bh/wps/portal/!ut/p/a1/nZJbU8IwEIX Cjz0sc22hTb4VhIEGS4Oi NK8MEIJLxqS0kbQf2-AccZxRNC8beaczbd7gghalCLptsioLpSkYI-

TYHk7gcD1sDfA_tyHaBJ0p50euH1oG0H8VQC-39sLwvvwseMFGC7zgzfqunct4x-NACJ8PR0-3HQB-

v45 xMiiCRSIzpHMc UtISVpmLJpQWCylUhs0ZJM15boKQoJG USa6UaHBZKSH WXOp9hzIpVihOAgZBmII74UFot3gnsCnzXJvxdith2AOcsE iEyc6O GMyyP1mb0dB L8t5ig4zREb0PAkiekw--

PkgwvCLJ43GxKZSJTU_E2jxT8zMW9lQrHDF4wjyXyclVLxlFe8cl4rc51rXdZXFliw2-2cTKlMcCdRzuuLBT-ZclUbnO9aVK7n8zX23-

1BLx2PbcLaYjuMms0PILP8nQ!!/dl5/d5/L2dBISEvZ0FBIS9nQSEh/ [Accessed 22 May 2019].

INOUE, N., SAKASHITA, R. & KAMEGAI, T. 1995. Reduction of masseter muscle activity in bottle- fed babies. *Early Human Development*, 42, 185-193.

- INSKIP, H., CROZIER, S., GODFREY, K., BORLAND, S., COOPER, C. & ROBINSON, S. 2009. Women's compliance with nutrition and lifestyle recommendations before pregnancy: general population cohort study. *British Medical Journal*, 338, 586-589.
- IRANZO-CORTÉS, J. E., MONTIEL-COMPANY, J. M. & ALMERICH-SILLA, J. M. 2013. Caries diagnosis: Agreement between WHO and ICDAS II criteria in epidemiological surveys. *Community dental health*, 30, 108-111.
- ISMAIL, A. I., SOHN, W., TELLEZ, M., AMAYA, A., SEN, A., HASSON, H. & PITTS, N. B. 2007. The International Caries Detection and Assessment System (ICDAS): an integrated system for measuring dental caries. *Community Dentistry and Oral Epidemiology*, 35, 170-178.
- JABLONSKI-MOMENI, A., STACHNISS, V., RICKETTS, D. N., HEINZEL-GUTENBRUNNER, M. & PIEPER, K. 2008. Reproducibility and accuracy of the ICDAS-II for detection of occlusal caries in vitro. *Caries research*, 42, 79-87.
- JACKSON, R. J., NEWMAN, H. N., SMART, G. J., STOKES, E., HOGAN, J. I., BROWN, C. & SERES, J. 2005. The effects of a supervised toothbrushing programme on the caries increment of primary school children, initially aged 5–6 years. *Caries research*, 39, 108-115.
- JACKSON, S., VANN, W., KOTCH, J., PAHEL, B. & LEE, J. 2011. Impact of Poor Oral Health on Children's School Attendance and Performance. *American Journal of Public Health*, 101, 1900-6.
- JACOB, C. M., BAIRD, J., BARKER, M., COOPER, C. & HANSON, M. 2017. The importance of a life course approach to health: chronic disease risk from preconception through adolescence and adulthood. *In:* WORLD HEALTH ORGANIZATION (ed.).
- JACOBSEN, P. E., HAUBEK, D., HENRIKSEN, T. B., ØSTERGAARD, J. R. & POULSEN, S. 2014. Developmental enamel defects in children born preterm: a systematic review. *European journal of oral sciences*, 122, 7-14.
- JEYNES, W. 2012. A Meta- Analysis of the Efficacy of Different Types of Parental Involvement Programs for Urban Students. *Urban Education*, 47, 706-742.
- JEYNES, W. H. 2000. The Effects of Several of the Most Common Family Structures on the Academic Achievement of Eighth Graders. *Marriage & Family Review*, 30, 73-97.
- JEYNES, W. H. 2003. A Meta- Analysis: The Effects of Parental Involvement on Minority Children's Academic Achievement. *Education and Urban Society*, 35, 202-218.
- JEYNES, W. H. 2005. A meta-analysis of the relation of parental involvement to urban elementary school student academic achievement. *Urban Education*, 40, 237-269.
- JEYNES, W. H. 2017. A Meta- Analysis: The Relationship Between Parental Involvement and Latino Student Outcomes. *Education and Urban Society*, 49, 4-28.
- JIANG, H., PETERSEN, P. E., PENG, B., TAI, B. & BIAN, Z. 2005. Self- assessed dental health, oral health practices, and general health behaviors in Chinese urban adolescents. *Acta Odontologica Scandinavica*, 63, 343-352.
- JONES, C. M., DAVIES, G. M., MONAGHAN, N., MORGAN, M. Z., NEVILLE, J. S. & PITTS, N. B. 2017. The caries experience of 5 year-old children in Scotland in 2013-2014, and in England and wales in 2014-2015. Reports of cross-sectional dental surveys using BASCD criteria. *Community dental health*, 34, 157-162.
- JONES, R., HOARE, P., ELTON, R., DUNHILL, Z. & SHARPE, M. 2009. Frequent medical absences in secondary school students: survey and case–control study. *Archives of disease in childhood*, 94, 763-767.
- JÄLEVIK, B., SZIGYARTO-MATEI, A. & ROBERTSON, A. 2018. The prevalence of developmental defects of enamel, a prospective cohort study of adolescents in Western Sweden: a Barn I TAnadvarden (BITA, children in dental care) study. *European Archives of Paediatric Dentistry*, 19, 187-195.
- JÜRGENSEN, N. & PETERSEN, P. 2009. Oral health and the impact of socio- behavioural factors in a cross sectional survey of 12- year old school children in Laos. *BMC Oral Health*, 9, 29.
- KAEWKAMNERDPONG, I. & KRISDAPONG, S. 2018. Oral diseases associated with condition- specific oral health- related quality of life and school performance of Thai

primary school children: A hierarchical approach. *Community Dentistry and Oral Epidemiology*, 46, 270-279.

- KAHLERT, J., GRIBSHOLT, S. B., GAMMELAGER, H., DEKKERS, O. M. & LUTA, G. 2017. Control of confounding in the analysis phase–an overview for clinicians. *Clinical epidemiology*, 9, 195-204.
- KANNISTO, V., CHRISTENSEN, K. & VAUPEL, J. W. 1997. No increased mortality in later life for cohorts born during famine. *American journal of epidemiology*, 145, 987-994.
- KARANDE, S. & KULKARNI, M. 2005. Poor school performance. *The Indian Journal of Pediatrics*, 72, 961-967.
- KARKI, S., PÄKKILÄ, J., LAITALA, M. L., HUMAGAIN, M. & ANTTONEN, V. 2019. Influence of dental caries on oral health-related quality of life, school absenteeism and school performance among Nepalese schoolchildren. *Community dentistry and oral epidemiology*, 47, 461-469.
- KASSEBAUM, N. J., BERNABÉ, E., DAHIYA, M., BHANDARI, B., MURRAY, C. J. L. & MARCENES, W. 2015. Global Burden of Untreated Caries: A Systematic Review and Metaregression. *Journal of Dental Research*, 94, 650-658.
- KASSEBAUM, N. J., SMITH, A. G. C., BERNABÉ, E., FLEMING, T. D., REYNOLDS, A. E., VOS, T., MURRAY, C. J. L., MARCENES, W., ABYU, G. Y., ALSHARIF, U., ASAYESH, H., BENZIAN, H., DANDONA, L., DANDONA, R., KASAEIAN, A., KHADER, Y. S., KHANG, Y. H., KOKUBO, Y., KOTSAKIS, G. A., LALLOO, R., MISGANAW, A., MONTERO, P., NOURZADEH, M., PINHO, C., QORBANI, M., BLANCAS, M. J. R., SAWHNEY, M., STEINER, C., TRAEBERT, J., TYROVOLAS, S., UKWAJA, K. N., VOLLSET, S. E. & YONEMOTO, N. 2017. Global, Regional, and National Prevalence, Incidence, and Disability-Adjusted Life Years for Oral Conditions for 195 Countries, 1990–2015: A Systematic Analysis for the Global Burden of Diseases, Injuries, and Risk Factors. *Journal of dental research*, 96, 380-387.
- KATE ANN, L. 2006. Study design III: Cross-sectional studies. *Evidence-Based Dentistry*, 7, 24-25.
- KAUR, A. & NOMAN, M. 2015. Exploring Classroom Practices in Collectivist Cultures Through the Lens of Hofstede's Model. *Qualitative Report*, 20, 1794-1811.
- KAWACHI, I., SUBRAMANIAN, S. V. & ALMEIDA-FILHO, N. 2002. A glossary for health inequalities. *Journal of Epidemiology and Community Health*, 56, 647-652.
- KEANE, M. P. & WOLPIN, K. I. 1997. The Career Decisions of Young Men. *Journal of Political Economy*, 105, 473-522.
- KEARNEY, C. A. 2008. An Interdisciplinary Model of School Absenteeism in Youth to Inform Professional Practice and Public Policy. *Educational psychology review*, 20, 257-282.
- KEMFANG NGOWA, J. D., DOMKAM, I., NGASSAM, A., NGUEFACK-TSAGUE, G., DOBGIMA PISOH, W., NOA, C. & KASIA, J. M. 2014. References of birth weights for gestational age and sex from a large cohort of singleton births in Cameroon. *Obstetrics* and gynecology international, 2014, 361451–8.
- KIM, S. W. & HILL, N. E. 2015. Including Fathers in the Picture: A Meta-Analysis of Parental Involvement and Students' Academic Achievement. *Journal of Educational Psychology*, 107, 919-934.
- KLEIN, M., SOSU, E. M. & DARE, S. 2020. Mapping inequalities in school attendance: The relationship between dimensions of socioeconomic status and forms of school absence. *Children and youth services review,* 118, 105432-12.
- KLINE, R. B. 2016. *Principles and practice of structural equation modeling,* New York, New York : The Guilford Press, 2016.
- KLINE, R. B. 2018. Response to Leslie Hayduk's review of Principles and Practice of Structural Equation Modeling, 4th edition. *Canadian studies in population*, 45, 188-195.
- KOCH, G., HALLONSTEN, A. L., LUDVIGSSON, N., HANSSON, B. O., HOIST, A. & ULLBRO, C. 1987. Epidemiologic study of idiopathic enamel hypomineralization in permanent teeth of Swedish children. *Community dentistry and oral epidemiology*, 15, 279-285.

- KOCH, G., HELKIMO, A. N. & ULLBRO, C. 2017. Caries prevalence and distribution in individuals aged 3–20 years in Jönköping, Sweden: trends over 40 years. *European* archives of paediatric dentistry, 18, 363-370.
- KORENMAN, S., MILLER, J. E. & SJAASTAD, J. E. 1995. Long-term poverty and child development in the United States: Results from the NLSY. *Children and youth services review*, 17, 127-155.
- KOSE, E. 2021. Public Investments in Early Childhood Education and Academic Performance: Evidence from Head Start in Texas. *The Journal of human resources*, 419.
- KOWASH, M. B., ALKHABULI, J. O., DAFAALLA, S. A., SHAH, A. & KHAMIS, A. H. 2017. Early childhood caries and associated risk factors among preschool children in Ras Al-Khaimah, United Arab Emirates. *European archives of paediatric dentistry*, 18, 97-103.
- KOWASH, M. B., TOUMBA, K. J. & CURZON, M. E. J. 2006. Cost-effectiveness of a longterm dental health education program for the prevention of early childhood caries. *European archives of paediatric dentistry*, 7, 130-135.
- KRAMER, M. S. 2000. Invited commentary: association between restricted fetal growth and adult chronic disease: Is it causal? Is it important? *American journal of epidemiology*, 152, 605-608.
- KRISDAPONG, S., PRASERTSOM, P., RATTANARANGSIMA, K. & SHEIHAM, A. 2013. School absence due to toothache associated with sociodemographic factors, dental caries status, and oral health-related quality of life in 12- and 15-year-old Thai children. *Journal of Public Health Dentistry*, 73, 321-328.
- KRISDAPONG, S., SOMKOTRA, T. & KUEAKULPIPAT, W. 2014. Disparities in Early Childhood Caries and its impact on oral health–related quality of life of preschool children. *Asia-Pacific Journal of Public Health*, 26, 285-294.
- KUH, D. & BEN-SHLOMO, Y. 2004. The life course and adult chronic disease: An historical perspective with particular reference to coronary heart disease. A Life Course Approach to Chronic Disease Epidemiology. Oxford: Oxford University Press.
- KUH, D., BEN-SHLOMO, Y., LYNCH, J., HALLQVIST, J. & POWER, C. 2003. Life course epidemiology. *Journal of Epidemiology and Community Health*, 57, 778-783.
- KUH, D., BEN-SHLOMO, Y. & SHLOMO, Y. B. 2004. A life course approach to chronic disease epidemiology [electronic resource], Oxford, Oxford : Oxford University Press, 2004.
- KUH, G. D. 2007. How to Help Students Achieve. Chronicle of Higher Education, 53, 1-1.
- KUMAR, S., TADAKAMADLA, J., KROON, J. & JOHNSON, N. W. 2016. Impact of parentrelated factors on dental caries in the permanent dentition of 6– 12- year- old children: A systematic review. *Journal of Dentistry*, 46, 1-11.
- KWAN, S. & PETERSEN, P. E. 2010. Oral health: equity and social determinants. *Equity, social determinants and public health programmes*, 159-176.
- KWAN, S. Y. L., PETERSEN, P. E., PINE, C. M. & BORUTTA, A. 2005. Health-promoting schools: An opportunity for oral health promotion. *Bulletin of the World Health Organization*, 83, 677-685.
- LAGERWEIJ, M. D. & VAN LOVEREN, C. 2015. Declining caries trends: are we satisfied? *Current oral health reports*, 2, 212-217.
- LAMBORN, S. D., MOUNTS, N. S., STEINBERG, L. & DORNBUSCH, S. M. 1991. Patterns of Competence and Adjustment among Adolescents from Authoritative, Authoritarian, Indulgent, and Neglectful Families. *Child development*, 62, 1049-1065.
- LANDIS, J. R. & KOCH, G. G. 1977. The measurement of observer agreement for categorical data. *Biometrics*, 33, 159-174.
- LEAL, S. C., BRONKHORST, E. M., FAN, M. & FRENCKEN, J. E. 2012. Untreated cavitated dentine lesions: impact on children's quality of life. *Caries research*, 46, 102-106.
- LEE, A. K., AIGNER, A., SCHMID, T. & KURTH, T. 2021. Spatio-temporal trends in caries: A study on children in Berlin-Mitte. *Clinical and experimental dental research*, 7, 196-204.

- LEE, C. & RYFF, C. D. 2016. Early parenthood as a link between childhood disadvantage and adult heart problems: A gender-based approach. *Social Science & Medicine*, 171, 58-66.
- LEON, D. A., LITHELL, H. O., VÅGERÖ, D., KOUPILOVÁ, I., MOHSEN, R., BERGLUND, L., LITHELL, U.-B. & MCKEIGUE, P. M. 1998. Reduced fetal growth rate and increased risk of death from ischaemic heart disease: cohort study of 15 000 Swedish men and women born 1915-29. *British Medical Journal*, 317, 241-245.
- LI, F., JIANG, P., YU, F., LI, C., WU, S., ZOU, J., XU, X., YE, L., ZHOU, X. & ZHENG, L. 2020. Comparison between Fissure Sealant and Fluoride Varnish on Caries Prevention for First Permanent Molars: a Systematic Review and Meta-analysis. *Scientific reports*, 10, 2578-2578.
- LI, Y., NAVIA, J. M. & CAUFIELD, P. W. 1994. Colonization by mutans streptococci in the mouths of 3- and 4- year- old Chinese children with or without enamel hypoplasia. *Archives of Oral Biology*, 39, 1057-1062.
- LI, Y., WULAERHAN, J., LIU, Y., ABUDUREYIMU, A. & ZHAO, J. 2017. Prevalence of severe early childhood caries and associated socioeconomic and behavioral factors in Xinjiang, China: a cross-sectional study. *BMC oral health,* 17, 1-10.
- LIM, S. S., VOS, T., FLAXMAN, A. D., DANAEI, G., SHIBUYA, K., ADAIR-ROHANI, H., ALMAZROA, M. A., AMANN, M., ANDERSON, H. R., ANDREWS, K. G., ARYEE, M., ATKINSON, C., BACCHUS, L. J., BAHALIM, A. N., BALAKRISHNAN, K., BALMES, J., BARKER-COLLO, S., BAXTER, A., BELL, M. L., BLORE, J. D., BLYTH, F., BONNER, C., BORGES, G., BOURNE, R., BOUSSINESQ, M., BRAUER, M., BROOKS, P., BRUCE, N. G., BRUNEKREEF, B., BRYAN-HANCOCK, C., BUCELLO, C., BUCHBINDER, R., BULL, F., BURNETT, R. T., BYERS, T. E., CALABRIA, B., CARAPETIS, J., CARNAHAN, E., CHAFE, Z., CHARLSON, F., CHEN, H., CHEN, J. S., CHENG, A. T.-A., CHILD, J. C., COHEN, A., COLSON, K. E., COWIE, B. C., DARBY, S., DARLING, S., DAVIS, A., DEGENHARDT, L., DENTENER, F., DES JARLAIS, D. C., DEVRIES, K., DHERANI, M., DING, E. L., DORSEY, E. R., DRISCOLL, T., EDMOND, K., ALI, S. E., ENGELL, R. E., ERWIN, P. J., FAHIMI, S., FALDER, G., FARZADFAR, F., FERRARI, A., FINUCANE, M. M., FLAXMAN, S., FOWKES, F. G. R., FREEDMAN, G., FREEMAN, M. K., GAKIDOU, E., GHOSH, S., GIOVANNUCCI, E., GMEL, G., GRAHAM, K., GRAINGER, R., GRANT, B., GUNNELL, D., GUTIERREZ, H. R., HALL, W., HOEK, H. W., HOGAN, A., HOSGOOD, H. D., HOY, D., HU, H., HUBBELL, B. J., HUTCHINGS, S. J., IBEANUSI, S. E., JACKLYN, G. L., JASRASARIA, R., JONAS, J. B., KAN, H., KANIS, J. A., KASSEBAUM, N., KAWAKAMI, N., KHANG, Y.-H., KHATIBZADEH, S., KHOO, J.-P. & KOK, C. 2012. A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990-2010: a systematic analysis for the Global Burden of Disease Study 2010. The Lancet, 380, 2224-2260.
- LISTL, S., GALLOWAY, J., MOSSEY, P. A. & MARCENES, W. 2015. Global economic impact of dental diseases. *Journal of dental research*, 94, 1355-1361.
- LISTL, S., WATT, R. G. & TSAKOS, G. 2014. Early life conditions, adverse life events, and chewing ability at middle and later adulthood. *American journal of public health,* 104, e55-e61.
- LITTLE, R. J. A. 1988. Missing-data adjustments in large surveys. *Journal of Business & Economic Statistics*, 6, 287-296.
- LIU, J., RAINE, A., VENABLES, P. H. & MEDNICK, S. A. 2004. Malnutrition at Age 3 Years and Externalizing Behavior Problems at Ages 8, 11, and 17 Years. *American Journal* of *Psychiatry*, 161, 2005-2013.
- LIU, L. P., OZA, S. M., HOGAN, D. P., PERIN, J. P., RUDAN, I. P., LAWN, J. E. P., COUSENS, S. P., MATHERS, C. P. & BLACK, R. E. P. 2015. Global, regional, and national causes of child mortality in 2000–13, with projections to inform post-2015 priorities: an updated systematic analysis. *Lancet*, 385, 430-440.

- LIU, P., WEN, W., YU, K. F., GAO, X., LO, E. C. M. & WONG, M. C. M. 2020. Effectiveness of a family-centered behavioral and educational counselling approach to improve periodontal health of pregnant women: a randomized controlled trial. *BMC Oral Health*, 20, 284-284.
- LOCKER, D. 1988. Measuring oral health: a conceptual framework. *Community Dent Health*, 5, 3-18.
- LORD, P., STYLES, B., MORRISON, J., WHITE, R., ANDRADE, J., BAMFORD, S., LUSHEY, C., LUCAS, M. & SMITH, R. 2018. Families and Schools Together (FAST).
- LU, H. X., WONG, M. C., LO, E. C. & MCGRATH, C. 2011. Trends in oral health from childhood to early adulthood: a life course approach. *Community dentistry and oral epidemiology*, 39, 352-360.
- LUOTO, A., LAHTI, S., NEVANPERÄ, T., TOLVANEN, M. & LOCKER, D. 2009. Oral-healthrelated quality of life among children with and without dental fear. *International Journal* of *Paediatric Dentistry*, 19, 115-120.
- LUPIEN, S. J., KING, S., MEANEY, M. J. & MCEWEN, B. S. 2001. Can poverty get under your skin? Basal cortisol levels and cognitive function in children from low and high socioeconomic status. *Development and psychopathology*. NEW YORK: Cambridge University Press.
- LUYCKX, V. A., PERICO, N., SOMASCHINI, M., MANFELLOTTO, D., VALENSISE, H., CETIN, I., SIMEONI, U., ALLEGAERT, K., VIKSE, B. E., STEEGERS, E. A., ADU, D., MONTINI, G., REMUZZI, G. & BRENNER, B. M. 2017. A developmental approach to the prevention of hypertension and kidney disease: a report from the Low Birth Weight and Nephron Number Working Group. *The Lancet,* 390, 424-428.
- LYNCH, J. & SMITH, G. D. 2005. A life course approach to chronic disease epidemiology. *Annual review of public health,* 26, 1-35.
- MACHIN, D., FAYERS, P. M. & TAI, B. C. 2021. *Randomised clinical trials: design, practice and reporting*, Newark: John Wiley & Sons, Incorporated.
- MAGNUS, A., HABY, M. M., CARTER, R. & SWINBURN, B. 2009. The cost-effectiveness of removing television advertising of high-fat and/or high-sugar food and beverages to Australian children. *International journal of obesity*, 33, 1094-1102.
- MAHARANI, D. A., ADIATMAN, M., RAHARDJO, A., BURNSIDE, G. & PINE, C. 2017. An assessment of the impacts of child oral health in Indonesia and associations with self-esteem, school performance and perceived employability. *BMC oral health,* 17, 65-65.
- MALCOLM, H., WILSON, V., DAVIDSON, J. & KIRK, S. 2003. Absence from school: A study of its causes and effects in seven LEAs. *The SCRE Centre University of Glasgow.* Glasgow.
- MAPPR.CO. 2014. Counties and Regions Maps, Governorates of Bahrain [Online]. Available: <u>https://www.mappr.co/counties/bahrain/</u> [Accessed March 2019].
- MARGOLIS, R. 2008. The effects of early childhood diseases on young adult health in Guatemala. *In:* POPULATION AGING RESEARCH CENTER (PARC) (ed.). The University of Pennsylvania.
- MARK, S. T., JÉAN-PHILIPPE, C., KRISTI, B. A., SALOMÉ, A., JOEL, D. B., LOUISE, C., MARY, D., GUY, F., GARY, S. G., CASEY, E. G., REUT, G., KATHERINE, J., IAN, J., XANNE, J., ALEJANDRA JARAMILLO, G., NICHOLAS, K., CLAIRE, L., JOANNA, M., ANTHONY, D. O., VERONICA, J. P., MARY-ELLEN, R., JOHN, J. R., MARGARET, S., JOHN, C. S., BRIAN, W. T. & VALERIE, C. 2017. Canadian 24-Hour Movement Guidelines for the Early Years (0–4 years): An Integration of Physical Activity, Sedentary Behaviour, and Sleep. *BMC Public Health*, 17, 1-32.
- MARMOT, M. G. 2005. Social determinants of health inequalities. *The Lancet (British edition),* 365, 1099-1104.
- MARMOT, M. G., SHIPLEY, M. J. & ROSE, G. 1984. Inequalities in death-- specific explanations of a general pattern? *The Lancet (British edition)*, 323, 1003-1006.
- MARMOT, M. G., STANSFELD, S., PATEL, C., NORTH, F., HEAD, J., WHITE, I., BRUNNER, E., FEENEY, A. & SMITH, G. D. 1991. Health inequalities among British civil servants: the Whitehall II study. *The Lancet (British edition)*, 337, 1387-1393.

- MARSHALL, T. A., BROFFITT, B., EICHENBERGER-GILMORE, J., WARREN, J. J., CUNNINGHAM, M. A. & LEVY, S. M. 2005. The Roles of Meal, Snack, and Daily Total Food and Beverage Exposures on Caries Experience in Young Children. *Journal of public health dentistry*, 65, 166-173.
- MARTINS, L. G. T., PEREIRA, K. C. R., COSTA, S. X. S., TRAEBERT, E., LUNARDELLI, S. E., LUNARDELLI, A. N. & TRAEBERT, J. 2016. Impact of dental caries on quality of life of school children. *Pesquisa Brasileira em Odontopediatria e Clinica Integrada,* 16, 307-312.
- MARTINSON, M. L. & REICHMAN, N. E. 2016. Socioeconomic Inequalities in Low Birth Weight in the United States, the United Kingdom, Canada, and Australia. *American journal of public health*, 106, 748-754.
- MASON, J., PEARCE, M. S., WALLS, A. W. G., PARKER, L. & STEELE, J. G. 2006. How Do Factors at Different Stages of the Lifecourse Contribute to Oral- health- related Quality of Life in Middle Age for Men and Women? *Journal of Dental Research*, 85, 257-261.
- MASUD, H., THURASAMY, R. & AHMAD, M. 2015. Parenting styles and academic achievement of young adolescents: A systematic literature review. *Quality & quantity*, 49, 2411-2433.
- MASUMO, R., BARDSEN, A. & AASTROEM, A. 2013. Developmental defects of enamel in primary teeth and association with early life course events: a study of 6- 36 month old children in Manyara, Tanzania. *BMC Oral Health*, 13, 21-21.
- MATEE, M., VAN'T HOF, M., MASELLE, S., MIKX, F. & HELDERMAN, W. V. P. 1994. Nursing caries, linear hypoplasia, and nursing and weaning habits in Tanzanian infants. *Community dentistry and oral epidemiology*, 22, 289-293.
- MAXWELL, S. E. & COLE, D. A. 2007. Bias in Cross-Sectional Analyses of Longitudinal Mediation. *Psychological methods*, 12, 23-44.
- MAYNARD, B. R., VAUGHN, M. G., NELSON, E. J., SALAS-WRIGHT, C. P., HEYNE, D. A. & KREMER, K. P. 2017. Truancy in the United States: Examining temporal trends and correlates by race, age, and gender. *Children and youth services review*, 81, 188-196.
- MCCALL, S. J., BHATTACHARYA, S., OKPO, E. & MACFARLANE, G. J. 2015. Evaluating the social determinants of teenage pregnancy: a temporal analysis using a UK obstetric database from 1950 to 2010. *Journal of Epidemiology and Community Health*, 69, 49-54.
- MCCARTHY, A. M., LINDGREN, S., MENGELING, M. A., TSALIKIAN, E. & ENGVALL, J. 2003. Factors associated with academic achievement in children with type 1 diabetes. *Diabetes care*, 26, 112-117.
- MCGEE, M., BAINBRIDGE, S. & FONTAINE-BISSON, B. 2018. A crucial role for maternal dietary methyl donor intake in epigenetic programming and fetal growth outcomes. *Nutrition reviews*, 76, 469-478.
- MCGOVERN, M. E. 2013. Still unequal at birth: Birth weight, socio-economic status and outcomes at age 9. *Economic and Social Review*, 44, 53-84.
- MEEDYA, S., FAHY, K. & KABLE, A. 2010. Factors that positively influence breastfeeding duration to 6 months: A literature review. *Women and Birth,* 23, 135-145.
- MELLOR, D. & MOORE, K. A. 2014. The use of Likert scales with children. *Journal of pediatric psychology*, 39, 369-379.
- MELVIN, G. A., HEYNE, D., GRAY, K. M., HASTINGS, R. P., TOTSIKA, V., TONGE, B. J. & FREEMAN, M. M. 2019. The Kids and Teens at School (KiTeS) Framework: An Inclusive Bioecological Systems Approach to Understanding School Absenteeism and School Attendance Problems. *Frontiers in education (Lausanne),* 4.
- MENARD, S. 2002. Longitudinal Research. Thousand Oaks, California: SAGE Publications, Inc.
- MENDES, F. M., BRAGA, M. M., OLIVEIRA, L. B., ANTUNES, J. L. F., ARDENGHI, T. M. & BÖNECKER, M. 2010. Discriminant validity of the International Caries Detection and Assessment System (ICDAS) and comparability with World Health Organization criteria in a cross-sectional study. *Community Dentistry and Oral Epidemiology*, 38, 398-407.

- METALLIDOU, P. & VLACHOU, A. 2007. Motivational beliefs, cognitive engagement, and achievement in language and mathematics in elementary school children. *International Journal of Psychology*, 42, 2-15.
- METERKO, M., RÉSTUCCIA, J. D., STOLZMANN, K., MOHR, D., BRENNAN, C., GLASGOW, J. & KABOLI, P. 2015. Response rates, nonresponse bias, and data quality: Results from a national survey of senior healthcare leaders. *Public Opinion Quarterly*, 79, 130-144.
- MILGROM, P., MANCL, L., KING, B., WEINSTEIN, P., WELLS, N. & JEFFCOTT, E. 1998. An Explanatory Model of the Dental Care Utilization of Low- Income Children. *Medical Care*, 36, 554-566.
- MOE-KOB. 2017. *Permitted absence days at school of Bahrain* [Online]. KoB: Ministry of Education. Available: <u>https://www.moe.gov.bh/shownews.aspx?id=17538&lan=en</u> [Accessed 14 June 2021].
- MOH-KOB. 2014. Bahrain's Health Agenda. Health Improvement Strategy 2015-2018. *Policy* [Online]. Available:

https://www.moh.gov.bh/Content/Files/Health Improvement Strategy(2015-2018).pdf.

MOH-KOB. 2019. Medical and Allied Health Manpower by Category, Nationality and Sex. 2021. Available:

https://www.moh.gov.bh/Content/Files/Publications/statistics/HS2019/PDF/CH-05human%20resources 2019.pdf.

- MOHER, D., LIBERATI, A., TETZLAFF, J. & ALTMAN, D. G. 2009. Preferred reporting items for systematic reviews and meta- analyses: the PRISMA statement. *Annals of internal medicine*, 151, 264-269.
- MONSE, B., HEINRICH-WELTZIEN, R., BENZIAN, H., HOLMGREN, C. & VAN PALENSTEIN HELDERMAN, W. 2010. PUFA – An index of clinical consequences of untreated dental caries. *Community Dentistry and Oral Epidemiology*, 38, 77-82.
- MORENO-GALARRAGA, L., ÁLVAREZ-ZALLO, N., OLIVER-OLID, A., MIRANDA-FERREIRO, G., MARTÍNEZ-GONZÁLEZ, M. Á. & MARTÍN-CALVO, N. 2021. Parentreported birth information: birth weight, birth length and gestational age. Validation study in the SENDO project. *Gaceta sanitaria*, 35, 224-229.
- MORGANO, S. M., DOUMIT, M., SHAMMARI, K. F. A., AL-SUWAYED, A., AL-SUWAIDI, A., DEBAYBO, D. & AL-MUBARAK, S. 2010. Burden of oral disease in the Middle East: Opportunities for dental public health. *International Dental Journal*, 60, 197-199.
- MORRISSEY, T. W., HUTCHISON, L. & WINSLER, A. 2014. Family Income, School Attendance, and Academic Achievement in Elementary School. *Developmental psychology*, 50, 741-753.
- MOYNIHAN, P. J. & KELLY, S. A. M. 2014. Effect on caries of restricting sugars intake: systematic review to inform WHO guidelines. *Journal of dental research*, 93, 8-18.
- MUIRHEAD, V. & MARCENES, W. 2004. An ecological study of caries experience, school performance and material deprivation in 5- year- old state primary school children. *Community Dentistry and Oral Epidemiology*, 32, 265-270.
- MUIRHEAD, V. E. & LOCKER, D. 2006. School Performance Indicators as Proxy Measures of School Dental Treatment Needs: A Feasibility Study. *Journal of Public Health Dentistry*, 66, 269-272.
- MURPHY, S. 2018. School location and socioeconomic status and patterns of participation and achievement in senior secondary mathematics. *Mathematics Education Research Journal*, 1-17.
- MUSAIGER, A. O. 2000. The state of nutrition in Bahrain. Nutrition and health, 14, 63-74.
- MWAKAYOKA, H., MASALU, J. R. & NAMAKUKA KIKWILU, E. 2017. Dental Caries and Associated Factors in Children Aged 2-4 Years Old in Mbeya City, Tanzania. *Journal* of dentistry (Shiraz), 18, 104-111.
- NAAVAAL, S. & KELEKAR, U. 2018. School hours lost due to acute/unplanned dental care. *Health Behavior and Policy Review,* 5, 66-73.

NASEEB, A. A. A. 2016. National Oral Health Survey of Oral Health Status of Bahraini School Children Aged 6, 12 and 15 Year Old. *Dental news (Beirut, Lebanon),* 23, 12-29.

- NASH, K. B. 1990. A Psychosocial Perspective: Growing Up with Thalassemia, a Chronic Disorder. *Annals of the New York Academy of Sciences*, 612, 442-450.
- NASUUNA, E., SANTORO, G., KREMER, P. & SILVA, A. M. 2016. Examining the relationship between childhood health conditions and health service utilisation at school entry and subsequent academic performance in a large cohort of Australian children. *Journal of Paediatrics and Child Health*, 52, 750-758.
- NATAPOV, L., DEKEL, D., PIKOVSKY, V. & ZUSMAN, S. P. 2021. Dental health of preschool children after two-years of a supervised tooth brushing program in Southern Israel. *Israel journal of health policy research*, 10, 1-42.
- NEVES, É., FIRMINO, R., PERAZZO, M., GOMES, M., MARTINS, C., PAIVA, S. & GRANVILLE-GARCIA, A. 2016. Absenteeism among preschool children due to oral problems. *Zeitschrift für Gesundheitswissenschaften*, 24, 65-72.
- NICOLAU, B., MARCENES, W., ALLISON, P. & SHEIHAM, A. 2005. The life course approach: explaining the association between height and dental caries in Brazilian adolescents. *Community Dentistry and Oral Epidemiology*, 33, 93-98.
- NICOLAU, B., MARCENES, W., BARTLEY, M. & SHEIHAM, A. 2003. A life course approach to assessing causes of dental caries experience: The relationship between biological, behavioural, socio-economic and psychological conditions and caries in adolescents. *Caries Research*, 37, 319-326.
- NICOLAU, B., NETUVELI, G., KIM, J. W. M., SHEIHAM, A. & MARCENES, W. 2007. A lifecourse approach to assess psychosocial factors and periodontal disease. *Journal of Clinical Periodontology*, 34, 844-850.
- NOTKOLA, V., PUNSAR, S., KARVONEN, M. J. & HAAPAKOSKI, J. 1985. Socioeconomic conditions in childhood and mortality and morbidity caused by coronary heart disease in adulthood in rural Finland. *Social science & medicine*, 21, 517-523.
- O'BRIEN, M. 1994. Children's dental health in the United Kingdom 1993 : a survey carried out by the Social Survey Division of OPCS, on behalf of the United Kingdom health departments, in collaboration with the Dental Schools at the Universities of Birmingham and Newcastle, London, London : HMSO, 1994.
- OBERLANDER, S. & BLACK, M. 2011. African American Adolescent Mothers' Early Caregiving Involvement and Childrens' Behavior and Academic Performance at Age 7. Journal of Clinical Child and Adolescent Psychology, 40, 756-764.
- OCCHI-ALEXANDRE, I. G. P., CRUZ, P. V., BENDO, C. B., PAIVA, S. M., PORDEUS, I. A. & MARTINS, C. C. 2020. Prevalence of dental caries in preschool children born preterm and/or with low birth weight: A systematic review with meta-analysis of prevalence data. *International journal of paediatric dentistry*, 30, 265-275.
- OKUBO, M. 1981. *Increase in mortality of middle-aged males in Japan,* Tokyo, Japan, Nihon University, Population Research Institute.
- OLIVARI, M. G., TAGLIABUE, S. & CONFALONIERI, E. 2013. Parenting Style and Dimensions Questionnaire: A Review of Reliability and Validity. *Marriage & Family Review*, 49, 465-490.
- ORTIZ, F. R., ARDENGHI, T. M., PAIVA, S. M., MARONEZE, M. C. & PORDEUS, I. A. 2021. Impact of Oral Conditions and Subjective Factors on Academic Performance. *Pesquisa brasileira em odontopediatria e clínica integrada,* 21.
- OSMAN, H., EL ZEIN, L. & WICK, L. 2009. Cultural beliefs that may discourage breastfeeding among Lebanese women: a qualitative analysis. *International Breastfeeding Journal*, 4, 12-12.
- OSMAN, K. & ALSAYED, M. 2015. Prevalence of Dental Caries and its Impact on the Academic performance of Sudanese Basic school children, AL-Sahafa Residential Area (2013-2014). *Journal of American science*, 11, 195-203.
- OYEDELE, T. A., FOLÁYAN, M. O., ADEKOYA-SOFOWORA, C. A., OZIEGBE, E. O. & ESAN, T. A. 2015. Prevalence, pattern and severity of molar incisor hypomineralisation

in 8-to 10-year-old school children in Ile-Ife, Nigeria. *European archives of paediatric dentistry*, 16, 277-282.

- O'CONNOR, E. E., DEARING, E. & COLLINS, B. A. 2011. Teacher- Child Relationship and Behavior Problem Trajectories in Elementary School. *American Educational Research Journal*, 48, 120-162.
- PANETH, N. & SUSSER, M. 1995. Early origin of coronary heart disease (the "Barker hypothesis"). *British Medical Journal*, 310, 411-4.
- PATTON, G. C. & VINER, R. 2007. Pubertal transitions in health. *The Lancet (British edition),* 369, 1130-1139.
- PAU, A., KHAN, S. S., BABAR, M. G. & CROUCHER, R. 2008. Dental pain and care- seeking in 11– 14- yr- old adolescents in a low- income country. *European Journal of Oral Sciences*, 116, 451-457.
- PAULA, J. S., LISBOA, C. M., CASTRO MENEGHIM, M., PEREIRA, A. C., AMBROSANO, G. M. B. & MIALHE, F. L. 2016. School performance and oral health conditions: analysis of the impact mediated by socio- economic factors. *International Journal of Paediatric Dentistry*, 26, 52-59.
- PEARCE, A., SAWYER, A. C. P., CHITTLEBOROUGH, C. R., MITTINTY, M. N., LAW, C. & LYNCH, J. W. 2016. Do early life cognitive ability and self-regulation skills explain socio-economic inequalities in academic achievement? An effect decomposition analysis in UK and Australian cohorts. *Social Science & Medicine,* 165, 108-118.
- PEARCE, M. S., STEELE, J. G., MASON, J., WALLS, A. W. G. & PARKER, L. 2004. Do Circumstances in Early Life Contribute to Tooth Retention in Middle Age? *Journal of dental research*, 83, 562-566.
- PEARCE, M. S., THOMSON, W., WALLS, A. & STEELE, J. 2009. Lifecourse Socio- economic Mobility and Oral Health in Middle Age. *Journal of dental research*, 88, 938-941.
- PEARL, J. 2009. Causality, Cambridge University Press.
- PEDHAZUR, E. J. 1991. *Measurement, design, and analysis : an integrated approach,* Hillsdale, N.J. ; Hove : Lawrence Erlbaum Associates.
- PEET, E., MCCOY, D., DANAEI, G., EZZATI, M., JARVELIN, M.-R., PILLAS, D. & FINK, G. 2015. Early Childhood Development and Schooling Attainment: Longitudinal Evidence from British, Finnish and Philippine Birth Cohorts. *PLoS One,* 10, e0137219-e0137219.
- PEKRUN, R., GOETZ, T., TITZ, W. & PERRY, R. P. 2002. Academic Emotions in Students' Self-Regulated Learning and Achievement: A Program of Qualitative and Quantitative Research. *Educational psychologist*, 37, 91-105.
- PERAZZO, M. F., GOMES, M. C., NEVES, É. T. B., FIRMINO, R. T., BARROS, A. A., SILVA, L. C., MARTINS, C. C., PAIVA, S. M. & GRANVILLE-GARCIA, A. F. 2020. Self-Perceptions of the impact of oral problems on the social behavior of preschoolers. JDR Clinical & Translational Research, 5, 342-348.
- PERES, K. G., CASCAES, A. M., NASCIMENTO, G. G. & VICTORA, C. G. 2015a. Effect of breastfeeding on malocclusions: a systematic review and meta- analysis. *Acta Paediatrica*, 104, 54-61.
- PERES, K. G., CASCAES, A. M., PERES, M. A., DEMARCO, F. F., SANTOS, I. S., MATIJASEVICH, A. & BARROS, A. J. D. 2015b. Exclusive Breastfeeding and Risk of Dental Malocclusion. *Pediatrics*, 136, E60-E67.
- PERES, M. A., BARROS, A. J., PERES, K. G., ARAUJO, C. L. P. & MENEZES, A. M. B. 2009. Life course dental caries determinants and predictors in children aged 12 years: a population-based birth cohort. *Community Dentistry and Oral Epidemiology*, 37, 123-133.
- PERES, M. A., LATORRE, M. D. R. D. D. O., SHEIHAM, A., PERES, K. G., BARROS, F. C., HERNANDEZ, P. G., MAAS, A. M. N., ROMANO, A. R. & VICTORA, C. G. 2005. Social and biological early life influences on severity of dental caries in children aged 6 years. *Community Dentistry and Oral Epidemiology*, 33, 53-63.
- PERES, M. A., LATORRE, M. D. R. D. O., SHEIHAM, A., PERES, K. G., BARROS, F. C., HERNANDEZ, P. G., MAAS, A. M. N., ROMANO, A. R. & VICTORA, C. G. 2003. Effects of Social and biological factors on dental caries in 6-year-old children: a cross

sectional study nested in a birth cohort in Southern Brazil. *Revista Brasileira de Epidemiologia*, 6, 293-306.

- PERES, M. A., MACPHERSON, L. M. D., WEYANT, R. J., DALY, B., VENTURELLI, R., MATHUR, M. R., LISTL, S., CELESTE, R. K., GUARNIZO-HERREÑO, C. C. & KEARNS, C. 2019. Oral diseases: a global public health challenge. *The Lancet*, 394, 249-260.
- PERES, M. A., PERES, K. G., DE BARROS, A. J. D. & VICTORA, C. G. 2007. The relation between family socioeconomic trajectories from childhood to adolescence and dental caries and associated oral behaviours. *Journal of Epidemiology and Community Health*, 61, 141-145.
- PERES, M. A., PERES, K. G., THOMSON, W. M., BROADBENT, J. M., GIGANTE, D. P. & HORTA, B. L. 2011. The influence of family income trajectories from birth to adulthood on adult oral health: findings from the 1982 Pelotas birth cohort. *American journal of public health*, 101, 730-736.
- PERES, M. A., SHEIHAM, A., LIU, P., DEMARCO, F. F., SILVA, A. E., ASSUNCAO, M. C., MENEZES, A. M., BARROS, F. C. & PERES, K. G. 2016. Sugar Consumption and Changes in Dental Caries from Childhood to Adolescence. *Journal of dental research*, 95, 388-394.
- PETERSEN, P. E. 2003. The World Oral Health Report 2003: continuous improvement of oral health in the 21st century the approach of the WHO Global Oral Health Programme. *Community dentistry and oral epidemiology*, 31, 3-24.
- PETERSEN, P. E. 2004. Challenges to improvement of oral health in the 21st century The approach of the WHO Global Oral Health Programme. *International Dental Journal*, 54, 329-343.
- PETERSEN, P. E., JIANG, H., PENG, B., TAI, B. J. & BIAN, Z. 2008. Oral and general health behaviours among Chinese urban adolescents. *Community Dentistry and Oral Epidemiology*, 36, 76-84.
- PETRIDOU, E., ATHANASSOULI, T., PANAGOPOULOS, H. & REVINTHI, K. 1996. Sociodemographic and dietary factors in relation to dental health among Greek adolescents. *Community Dentistry and Oral Epidemiology*, 24, 307-311.
- PETROCELLI, J. V. 2003. Hierarchical Multiple Regression in Counseling Research: Common Problems and Possible Remedies. *Measurement and evaluation in counseling and development*, 36, 9-22.
- PINQUART, M. 2016. Associations of Parenting Styles and Dimensions with Academic Achievement in Children and Adolescents: A Meta-analysis. *Educational Psychology Review*, 28, 475-493.
- PIOVESAN, C., ANTUNES, J. L. F., MENDES, F. M., GUEDES, R. S. & ARDENGHI, T. M. 2012. Influence of children's oral health-related quality of life on school performance and school absenteeism. *Journal of Public Health Dentistry*, 72, 156-163.
- PITIPHAT, W., GARCIA, R. I., DOUGLASS, C. W. & JOSHIPURA, K. J. 2002. Validation of Self-reported Oral Health Measures. *Journal of public health dentistry*, 62, 122-128.
- PITTS, N. & EKSTRAND, K. 2013. International Caries Detection and Assessment System (ICDAS) and its International Caries Classification and Management System (ICCMS) – methods for staging of the caries process and enabling dentists to manage caries. *Community Dentistry and Oral Epidemiology*, 41, e41-e52.
- PITTS, N. B. & STAMM, J. W. 2004. International Consensus Workshop on Caries Clinical Trials (ICW-CCT)—Final Consensus Statements: Agreeing Where the Evidence Leads. *Journal of Dental Research*, 83, 125-128.
- POLK, D. E., WEYANT, R. J. & MANZ, M. C. 2010. Socioeconomic factors in adolescents' oral health: are they mediated by oral hygiene behaviors or preventive interventions? *Community dentistry and oral epidemiology*, 38, 1-9.
- PONGPICHIT, B., SHEIHAM, A., PIKHART, H. & TSAKOS, G. 2008. Time Absent from School due to Dental Conditions and Dental Care in Thai Schoolchildren. *Journal of Public Health Dentistry*, 68, 76-81.

- POTLIA, I., KUMAR, P. G. N., PRASHANT, G. M., SUSHANTH, V. H., IMRANULLA, M., MALLICK, S. & RUBEL, M. 2016. Comparison of the caries status using DMFT, ICDAS II and cast index system. *Int. J. Adv. Res.*, 4.
- POULTON, R., CASPI, A., MILNE, B. J., THOMSON, W. M., TAYLOR, A., SEARS, M. R. & MOFFITT, T. E. 2002. Association between children's experience of socioeconomic disadvantage and adult health: a life-course study. *Lancet*, 360, 1640-1645.
- POYAK, J. 2006. Effects of pacifiers on early oral development. International journal of orthodontics (Milwaukee, Wis.), 17, 13-6.
- PRIYANKA, G. G., KOTE, S., SRAVANTHI, K. V., CHETHAN, M., ANAND, A., JAIN, M. & SINGH, S. 2019. Life course approach in the assessment of association between dental caries and health capital with family-related characteristics among 12-year-old school children. *Journal of family medicine and primary care*, 8, 2506-2510.
- PSOTER, W., GEBRIAN, B., PROPHETE, S., REID, B. & KATZ, R. 2008a. Effect of early childhood malnutrition on tooth eruption in Haitian adolescents. *Community Dentistry and Oral Epidemiology*, 36, 179-189.
- PSOTER, W. J., REID, B. C. & KATZ, R. V. 2005. Malnutrition and dental caries: a review of the literature. *Caries research*, 39, 441-447.
- PSOTER, W. J., SPIELMAN, A. L., GEBRIAN, B., ST. JEAN, R. & KATZ, R. V. 2008b. Effect of childhood malnutrition on salivary flow and pH. *Archives of Oral Biology*, 53, 231-237.
- PUBLIC HEALTH ENGLAND 2014. Local authorities improving oral health: commissioning better oral health for children and young people: an evidence-informed toolkit for local authorities.
- PUBLIC HEALTH ENGLAND. 2019. *Health matters: Prevention a life course approach* [Online]. UK. Available: <u>https://www.gov.uk/government/publications/health-matters-life-course-approach-to-prevention/health-matters-prevention-a-life-course-approach</u> [Accessed 17 November 2021 2021].
- QADRI, G., NOURALLAH, A. & SPLIETH, C. H. 2012. Early childhood caries and feeding practices in kindergarten children. *Quintessence International,* 43, 503-510.
- QUADRI, M. F. A. & AHMAD, B. 2019. Is there evidence for the impact of poor oral health on school performance? *BMC Oral Health*, 19, 143-143.
- QUADROS, L. N., REBELO, M. A. B., DE QUEIROZ, A. C., PEREIRA, J. V., VETTORE, M. V. & REBELO VIEIRA, J. M. 2021. Clinical consequences of untreated dental caries and school performance in low-income adolescents. *International journal of paediatric dentistry*, 31, 619-626.
- RAJAB, L. D., PETERSEN, P. E., BAQAIN, Z. & BAKAEEN, G. 2014. Oral health status among 6-and 12-year-old Jordanian schoolchildren. *Oral health & preventive dentistry*, 12, 99-107.
- RAMBARAN, J. A., HOPMEYER, A., SCHWARTZ, D., STEGLICH, C., BADALY, D. & VEENSTRA, R. 2017. Academic Functioning and Peer Influences: A Short- Term Longitudinal Study of Network– Behavior Dynamics in Middle Adolescence. *Child Development*, 88, 523-543.
- RAMESH, G., NAGARAJAPPA, R., RAGHUNATH, V. & MANOHAR, R. 2011. Developmental defects of enamel in children of Davangere District and their relationship to fluoride levels in drinking water. *Asia-Pacific Journal of Public Health*, 23, 341-348.
- RAVITEJA, N. V. K., PRASAD, M. G. S., AMBATI, N. R., SAUJANYA, K., KUMAR, K. V. K. S., DIVYA, D. V. & SUNDEEP, R. V. 2017. Evaluation of Mechanical Plaque Removal Effectiveness of Toothbrush and Its Modifications in Intellectually Disabled Children. *Journal of Biomedical and Pharmaceutical Research*, 6, 107-112.
- REBELO, M. A. B., REBELO VIEIRA, J. M., PEREIRA, J. V., QUADROS, L. N. & VETTORE, M. V. 2019. Does oral health influence school performance and school attendance? A systematic review and meta-analysis. *International journal of paediatric dentistry*, 29, 138-148.

- RESCHLY, A. L. & CHRISTENSON, S. L. 2006. Prediction of Dropout Among Students With Mild Disabilities: A Case for the Inclusion of Student Engagement Variables. *Remedial and special education*, 27, 276-292.
- RIBEIRO, A. P. D., ALMEIDA, R. F., MEDONCA, J. G. A. & LEAL, S. C. 2018. Oral Health and Its Effect on the Academic Performance of Children and Adolescents. *Pediatric dentistry*, 40, 12-17.
- RICH-EDWARDS, J. W., STAMPFER, M. J., MANSON, J. E., ROSNER, B., HANKINSON, S. E., COLDITZ, G. A., WILLETT, W. C. & HENNEKENS, C. H. 1997. Birth weight and risk of cardiovascular disease in a cohort of women followed up since 1976. *British Medical Journal*, 315, 396-400.
- ROBERTS, J. E., BURCHINAL, M. R. & ZEISEL, S. A. 2002. Otitis media in early childhood in relation to children's school- age language and academic skills. *Pediatrics,* 110, 696-706.
- ROBINSON, C. C., MANDLECO, B., OLSEN, S. F. & HART, C. H. 1995. Authoritative, Authoritarian, and Permissive Parenting Practices: Development of a New Measure. *Psychological Reports*, 77, 819-830.
- ROBINSON, C. C., MANDLECO, B., OLSEN, S. F. & HART, C. H. 2001. The parenting styles and dimensions questionnaire (PSDQ). *Handbook of family measurement techniques*, 3, 319-321.
- ROBLES, M.-J., RUIZ, M., BRAVO-PEREZ, M., GONZÁLEZ, E. & PEÑALVER, M.-A. 2013. Prevalence of enamel defects in primary and permanent teeth in a group of schoolchildren from Granada (Spain). *Medicina oral, patología oral y cirugía bucal,* 18, e187-e193.
- ROGERS, J. G. 2011. *Evidence-based oral health promotion resource*, Prevention and Population Health Branch, Department of Health-Victoria.
- ROSE, G. A., KHAW, K.-T. & MARMOT, M. 2008. *Rose's strategy of preventive medicine: the complete original text*, Oxford University Press, USA.
- ROSEBOOM, T. J., VAN DER MEULEN, J. H. P., RAVELLI, A. C. J., OSMOND, C., BARKER, D. J. P. & BLEKER, O. P. 2001. Effects of prenatal exposure to the Dutch famine on adult disease in later life: an overview. *Molecular and Cellular Endocrinology*, 185, 93-98.
- ROTH, B., BECKER, N., ROMEYKE, S., SCHÄFER, S., DOMNICK, F. & SPINATH, F. M. 2015. Intelligence and school grades: A meta-analysis. *Intelligence (Norwood)*, 53, 118-137.
- ROWAN, K. 1994. Global questions and scores. *Measuring health and medical outcomes*. Routledge.
- RUFF, R. R., SENTHI, S., SUSSER, S. R. & TSUTSUI, A. 2019. Oral health, academic performance, and school absenteeism in children and adolescents: A systematic review and meta-analysis. *Journal of the American Dental Association*, 150, 111-121.e4.
- RYAN, R. M. & CLAESSENS, A. 2013. Associations Between Family Structure Changes and Children's Behavior Problems: The Moderating Effects of Timing and Marital Birth. *Developmental Psychology*, 49, 1219-1231.
- SABBAH, W., TSAKOS, G., SHEIHAM, A. & WATT, R. G. 2009. The role of health-related behaviors in the socioeconomic disparities in oral health. *Social science & medicine*, 68, 298-303.
- SABER, F., WALY, N. & MOHEB, D. 2018. Prevalence of molar incisor hypomineralisation in a group of Egyptian children using the short form: a cross-sectional study. *European Archives of Paediatric Dentistry*, 19, 337-345.
- SAITOH, M., NAKAMURA, Y., HANASAKI, M., SAITOH, I., MURAI, Y., KURASHIGE, Y., FUKUMOTO, S., ASAKA, Y., YAMADA, M. & SEKINE, M. 2018. Prevalence of molar incisor hypomineralization and regional differences throughout Japan. *Environmental health and preventive medicine*, 23, 55-55.

- SAMEER, S., AMMAR AHMED, S. & MOHAMMAD, A. 2016. School Absenteeism due to Toothache among Secondary School Students Aged 16– 18 Years in the Ha'il Region of Saudi Arabia. *Pain Research and Treatment*, 2016, 7058390–7058394.
- SAMUEL, S. R., ACHARYA, S. & RAO, J. C. 2020. School Interventions–based Prevention of Early-Childhood Caries among 3–5-year-old children from very low socioeconomic status: Two-year randomized trial. *Journal of public health dentistry*, 80, 51-60.
- SANDRA, F., VIVIANE, C. & JIMMY, P. 2005. The psychosocial effects of severe caries in 4year- old children in Recife, Pernambuco, Brazil. *Cadernos de Saúde Pública*, 21, 1550-1556.
- SARI, N., AMALIA, R. & SUPARTINAH, A. 2021. The Association Between Parenting Styles and Feeding Styles on Early Childhood Caries. *Scientific Dental Journal,* 5, 63-67.
- SCHATZ, T. J., BROWN, M. R., PASCUAL, R. J., HSU, R. L. & DEBAUN, R. M. 2001. Poor school and cognitive functioning with silent cerebral infarcts and sickle cell disease. *Neurology*, 56, 1109-1111.
- SCHEERMAN, J. F. M., HAMILTON, K., SHARIF, M. O., LINDMARK, U. & PAKPOUR, A. H. 2020. A theory-based intervention delivered by an online social media platform to promote oral health among Iranian adolescents: a cluster randomized controlled trial. *Psychology & health*, 35, 449-466.
- SCHLUCHTER, D. M. 2003. Publication bias and heterogeneity in the relationship between systolic blood pressure, birth weight, and catch- up growth a meta analysis. *Journal of Hypertension*, 21, 273-279.
- SCHOLL, T. O., HEDIGER, M. L. & ANCES, I. G. 1990. Maternal growth during pregnancy and decreased infant birth weight. *The American journal of clinical nutrition*, 51, 790-793.
- SCHOUTEN, A., OOSTROM, K. J., PESTMAN, W., PETERS, A. C. B. & JENNEKENS-SCHINKEL, A. 2002. Learning and memory of school children with epilepsy: a prospective controlled longitudinal study. *Developmental Medicine & Child Neurology*, 44, 803-811.
- SCHROEDER, D. G. 2001. Malnutrition. *Nutrition and health in developing countries.* Springer.
- SCHROTH, R. J., LAVELLE, C., TATE, R., BRUCE, S., BILLINGS, R. J. & MOFFATT, M. E. K. 2014. Prenatal vitamin D and dental caries in infants. *Pediatrics (Evanston),* 133, e1277-e1284.
- SCHWENDICKE, F., ELHENNAWY, K., REDA, S., BEKES, K., MANTON, D. J. & KROIS, J. 2018. Global burden of molar incisor hypomineralization. *Journal of dentistry*, 68, 10-18.
- SCOTT, J., LANDERS, M., HUGHES, R. & BINNS, C. 2001. Factors associated with breastfeeding at discharge and duration of breastfeeding. *Journal of Paediatrics and Child Health*, 37, 254-261.
- SEIRAWAN, H., FAUST, S. & MULLIGAN, R. 2012. The Impact of Oral Health on the Academic Performance of Disadvantaged Children. *American Journal of Public Health*, 102, 1729-1734.
- SEOW, W. K. 1998. Biological mechanisms of early childhood caries. *Community Dentistry and Oral Epidemiology*, 26, 8-27.
- SEOW, W. K. 2014. Developmental defects of enamel and dentine: challenges for basic science research and clinical management. *Australian dental journal*, 59, 143-54.
- SEOW, W. K., FORD, D., KAZOULLIS, S., NEWMAN, B. & HOLCOMBE, T. 2011. Comparison of enamel defects in the primary and permanent dentitions of children from a low-fluoride district in Australia. *Pediatric dentistry*, 33, 207-212.
- SEYFRIED, S. F. 1998. Academic Achievement of African American Preadolescents: The Influence of Teacher Perceptions. *American journal of community psychology*, 26, 381-402.
- SGAN-COHEN, H. D. & VERED, Y. 2005. A clinical trial of the meridol toothbrush with conical filaments: evaluation of clinical effectiveness and subjective satisfaction. *The Journal of clinical dentistry*, 16, 109-113.

- SHAFI, S., ALASMRI, A., MUSTAFA, A., ALSHAHRANI, A. S. S., ALASMRI, H. & BIJLE, M. N. A. 2015. An assessment of dental anxiety in nonclinical setting among Saudi Arabian children using Abeer Children Dental Anxiety Scale. *Journal of Dental Research and Review*, 2, 172-174.
- SHAPIRO, A. D., DONFIELD, S. M., LYNN, H. S., COOL, V. A., STEHBENS, J. A., HUNSBERGER, S. L., TONETTA, S. & GOMPERTS, E. D. 2001. Defining the impact of hemophilia: the Academic Achievement in Children with Hemophilia Study. *Pediatrics*, 108, e105-e105.
- SHEA, B. J., GRIMSHAW, J. M., WELLS, G. A., BOERS, M., ANDERSSON, N., HAMEL, C., PORTER, A. C., TUGWELL, P., MOHER, D. & BOUTER, L. M. 2007. Development of AMSTAR: a measurement tool to assess the methodological quality of systematic reviews. *BMC medical research methodology*, 7, 10-10.
- SHEIHAM, A. 2006. Dental caries affects body weight, growth and quality of life in pre-school children. *British dental journal*, 201, 625-626.
- SHEIHAM, A. & JAMES, W. P. T. 2015. Diet and Dental Caries: The Pivotal Role of Free Sugars Reemphasized. *Journal of dental research*, 94, 1341-1347.
- SHEIHAM, A. & WATT, R. G. 2000. The Common Risk Factor Approach: a rational basis for promoting oral health. *Community Dentistry and Oral Epidemiology*, 28, 399-406.
- SHENKIN, S. D., STARR, J. M. & DEARY, I. J. 2004. Birth Weight and Cognitive Ability in Childhood: A Systematic Review. *Psychological Bulletin*, 130, 989-1013.
- SHI, Q., LIEW, J., ETTEKAL, I. & WOLTERING, S. 2021. Childhood resilient personality trajectories and associations with developmental trajectories of behavioral, socialemotional, and academic outcomes across childhood and adolescence: A longitudinal study across 12 years. *Personality and individual differences*, 177, 110789-110789.
- SHOAIB, L., DEERY, C., RICKETTS, D. N. J. & NUGENT, Z. J. 2009. Validity and reproducibility of ICDAS II in primary teeth. *Caries Research*, 43, 442-448.
- SHONKOFF, J. P. 2000. From neurons to neighborhoods [electronic resource] : the science of early child development, Washington, D.C., Washington, D.C. : National Academy Press, c2000.
- SHROUT, P. E. 2011. Commentary: Mediation Analysis, Causal Process, and Cross-Sectional Data. *Multivariate behavioral research*, 46, 852-860.
- SILVA, M. J., KILPATRICK, N. M., CRAIG, J. M., MANTON, D. J., LEONG, P., BURGNER, D. P. & SCURRAH, K. J. 2019. Genetic and Early-Life Environmental Influences on Dental Caries Risk: A Twin Study. *Journal of the American academy of pediatrics*, 143, e20183499.
- SIMONS, K. 2005. Amblyopia characterization, treatment, and prophylaxis. *Survey of ophthalmology*, 50, 123-166.
- SIRIN, S. R. 2005. Socioeconomic status and academic achievement: A meta-analytic review of research. *Review of Educational Research*, 75, 417-453.
- SKEDGELL, K. & KEARNEY, C. A. 2018. Predictors of school absenteeism severity at multiple levels: A classification and regression tree analysis. *Children and youth services review*, 86, 236-245.
- SKINNER, E., FURRER, C., MARCHAND, G. & KINDERMANN, T. 2008. Engagement and Disaffection in the Classroom: Part of a Larger Motivational Dynamic? *Journal of Educational Psychology*, 100, 765-781.
- SKOGEN, J. C. & OVERLAND, S. 2012. The fetal origins of adult disease: a narrative review of the epidemiological literature. *Journal of the Royal Society of Medicine Short Reports,* 3, 59-7.
- SLACK, K. S., HOLL, J. L., MCDANIEL, M., YOO, J. & BOLGER, K. 2004. Understanding the Risks of Child Neglect: An Exploration of Poverty and Parenting Characteristics. *Child Maltreatment*, 9, 395-408.
- SLAYTON, R. L., WARREN, J. J., KANELLIS, M. J., LEVY, S. M. & ISLAM, M. 2001. Prevalence of enamel hypoplasia and isolated opacities in the primary dentition. *Pediatric dentistry*, 23, 32-36.

- SMITH, G. C. S. & PELL, J. P. 2001. Teenage pregnancy and risk of adverse perinatal outcomes associated with first and second births: Population based retrospective cohort study. *British Medical Journal*, 323, 476-479.
- SMITH, G. D. 2007. Life-course approaches to inequalities in adult chronic disease risk: Boyd Orr Lecture. *Proceedings of the Nutrition Society,* 66, 216-236.
- SMITH, V., DEVANE, D., BEGLEY, C. M. & CLARKE, M. 2011. Methodology in conducting a systematic review of systematic reviews of healthcare interventions. *BMC medical research methodology*, 11, 15-15.
- SOARES, T. R. C., MAGNO, M. B., JURAL, L. A., LOUREIRO, J. M., CHIANCA, T. K., ANDRADE RISSO, P. & MAIA, L. C. 2018. Risk factors for traumatic dental injuries in the Brazilian population: A critical review. *Dental Traumatology*, 34, 445-454.
- SPINATH, B., ECKERT, C. & STEINMAYR, R. 2014. Gender differences in school success: what are the roles of students' intelligence, personality and motivation? *Educational research (Windsor)*, 56, 230-243.
- SPINATH, B., HARALD FREUDENTHALER, H. & NEUBAUER, A. C. 2010. Domain-specific school achievement in boys and girls as predicted by intelligence, personality and motivation. *Personality and individual differences*, 48, 481-486.
- SROUFE, L. A. 2005. Attachment and development: A prospective, longitudinal study from birth to adulthood. *Attachment & Human Development*, 7, 349-367.
- STANNER, S. A., BULMER, K., ANDRÈS, C., LANTSEVA, O. E., BORODINA, V., POTEEN, V. V. & YUDKIN, J. S. 1997. Does malnutrition in utero determine diabetes and coronary heart disease in adulthood? Results from the Leningrad siege study, a cross sectional study. *British Medical Journal*, 315, 1342-1348.
- STEIN, C. E., FALL, C. H. D., KUMARAN, K., OSMOND, C., BARKER, D. J. P. & COX, V. 1996. Fetal growth and coronary heart disease in South India. *Lancet*, 348, 1269-1273.
- STEPHENSON, J., KUH, D., SHAWE, J. A., LAWLOR, D., SATTAR, N. A., RICH-EDWARDS, J., HANSON, M., INSKIP, H. M. & NELSON, S. M. 2011. Why Should we Consider a Life Course Approach to Women's Health Care?
- STIPEK, D. 2006. No Child Left Behind Comes to Preschool. *The Elementary School Journal*, 106, 455-466.
- SUCKLING, G. W. 1989. Developmental defects of enamel--historical and present-day perspectives of their pathogenesis. *Advances in dental research,* 3, 87-94.
- SUN, X., BERNABÉ, E., LIU, X., GALLAGHER, J. E. & ZHENG, S. 2017. Early life factors and dental caries in 5-year- old children in China. *Journal of Dentistry*, 64, 73-79.
- SUN, Y. & LI, Y. 2011. Effects of Family Structure Type and Stability on Children's Academic Performance Trajectories. *Journal of Marriage and Family*, 73, 541-556.
- SUSSER, M. & LEVIN, B. 1999. Ordeals for the Fetal Programming Hypothesis: The Hypothesis Largely Survives One Ordeal but Not Another. *British Medical Journal*, 318, 885-886.
- SÁNCHEZ-MOLINS, M., GRAU CARBÓ, J., LISCHEID GAIG, C. & USTRELL TORRENT, J.
 M. 2010. Comparative study of the craniofacial growth depending on the type of lactation received. *European Journal of Paediatric Dentistry*, 11, 87-92.
- TAVAKOL, M. & DENNICK, R. 2011. Making sense of Cronbach's alpha. *International journal* of medical education, 2, 53-55.
- TAZOUTI, Y. & JARLÉGAN, A. 2019. The mediating effects of parental self-efficacy and parental involvement on the link between family socioeconomic status and children's academic achievement. *Journal of Family Studies*, 25, 250-266.
- THAM, R., BOWATTE, G., DHARMAGE, S., TAN, D., LAU, M., DAI, X., ALLEN, K. & LODGE, C. 2015. Breastfeeding and the risk of dental caries: a systematic review and metaanalysis. *Acta Paediatrica*, 104, 62-84.
- THIESE, M. S. 2014. Observational and interventional study design types; an overview. *Biochemia medica*, 24, 199-210.
- THOMAS, P. A., LIU, H. & UMBERSON, D. 2017. Family relationships and well-being. *Innovation in aging*, 1, 1-11.

- THOMSON, W. M., POULTON, R., MILNE, B. J., CASPI, A., BROUGHTON, J. R. & AYERS, K. M. S. 2004. Socioeconomic inequalities in oral health in childhood and adulthood in a birth cohort. *Community Dentistry and Oral Epidemiology*, 32, 345-353.
- THOMSON, W. M., WILLIAMS, S. M., BROADBENT, J. M., POULTON, R. & LOCKER, D. 2010. Long-term Dental Visiting Patterns and Adult Oral Health. *Journal of Dental Research*, 89, 307-311.
- TICKLE, M., O'NEILL, C., DONALDSON, M., BIRCH, S., NOBLE, S., KILLOUGH, S., MURPHY, L., GREER, M., BRODISON, J., VERGHIS, R. & WORTHINGTON, H. V. 2016. A randomised controlled trial to measure the effects and costs of a dental caries prevention regime for young children attending primary care dental services: The Northern Ireland caries prevention in practice (NIC-PIP) trial. *Health technology assessment (Winchester, England)*, 20, vii-96.
- TOMARKEN, A. J. & WALLER, N. G. 2005. Structural equation modeling: Strengths, limitations, and misconceptions. *Annual review of clinical psychology*, 1, 31-65.
- TORSHEIM, T., WOLD, B. & SAMDAL, O. 2000. The Teacher and Classmate Support Scale: Factor Structure, Test- Retest Reliability and Validity in Samples of 13-and 15- Year-Old Adolescents. *School Psychology International*, 21, 195-212.
- TREASURE, E. T., CHESTNUTT, I. G., WHITING, P., MCDONAGH, M., WILSON, P. & KLEIJNEN, J. 2002. The York Review A systematic review of public water fluoridation: a commentary. *British dental journal*, 192, 495-497.
- TRIANDIS, H. C., BONTEMPO, R., VILLAREAL, M. J., ASAI, M. & LUCCA, N. 1988. Individualism and Collectivism: Cross-Cultural Perspectives on Self-Ingroup Relationships. *Journal of personality and social psychology*, 54, 323-338.
- TRIVEDI, D. 2018. Cochrane Review Summary: Support for healthy breastfeeding mothers with healthy term babies. *Primary health care research & development,* 19, 529-530.
- TWETMAN, S. 2013. Prevention of Early Childhood Caries (ECC) Review of literature published 1998–2007. *European archives of paediatric dentistry*, 9, 12-18.
- TYLER, K. & BOELTER, C. 2008. Linking Black Middle School Students' Perceptions of Teachers' Expectations to Academic Engagement and Efficacy. *Negro Educational Review*, 59, 27-44.
- UK PARLIAMENT DEPARTMENT OF HEALTH. 2010. *Healthy Lives, Healthy People: Our strategy for public health in England* [Online]. TSO. Available: <u>https://www.gov.uk/government/publications/healthy-lives-healthy-people-our-</u> strategy-for-public-health-in-england [Accessed 20 February 2018].
- UNITED NATIONS 2010. The Millennium Development Goals Report.
- UNITED NATIONS DEVELOPMENT PROGRAMME. 2020. *Human development report 2020* [Online]. Available: <u>https://report.hdr.undp.org/</u> [Accessed 9 June 2021].
- UNITED NATIONS EDUCATIONAL SCIENTIFIC AND CULTURAL ORGANIZATION 2019. Behind the numbers: Ending school violence and bullying, Paris, UNESCO.
- US CENSUS BUREAU 2006. Income Climbs, Poverty Stabilizes, Uninsured Rate Increases. US Census Bureau Washington, DC.
- VEHKALAHTI, K. 2011. A Beginner's Guide to Structural Equation Modeling, Third Edition by Randall E. Schumacker, Richard G. Lomax. *International Statistical Review*, 79, 286-287.
- VELASCO, S. R. M., DA SILVA BASTOS, R., SILVA, R. M. & DE LIMA NAVARRO, M. F. 2018. Association Between Presence of Enamel Defects, Dental Caries and Socioeconomic Conditions on Brazilian Children. *Pesquisa Brasileira em Odontopediatria e Clínica Integrada*, 18, 1-8.
- VERNAZZA, C. R., BIRCH, S. & PITTS, N. B. 2021. Reorienting Oral Health Services to Prevention: Economic Perspectives. *Journal of Dental Research*, 100, 576-582.
- VERNON-FEAGANS, L., GARRETT-PETERS, P., WILLOUGHBY, M. & MILLS-KOONCE, R. 2012. Chaos, poverty, and parenting: Predictors of early language development. *Early Childhood Research Quarterly*, 27, 339-351.
- VINER, R. M., ROSS, D., HARDY, R., KUH, D., POWER, C., JOHNSON, A., WELLINGS, K., MCCAMBRIDGE, J., COLE, T. J., KELLY, Y. & BATTY, G. D. 2015. Life course

epidemiology: recognising the importance of adolescence. *Journal of Epidemiology and Community Health*, 69, 719-720.

- VIVIANE, C. & SANDRA, F. 2003. School performance of pre- school children with severe caries. *Health Sciences*, 25, 129-134.
- WAALER, H. T. 1984. Height. Weight and Mortality The Norwegian Experience. *Acta Medica Scandinavica*, 215, 1-56.
- WADDINGTON, C. H. 2012. The epigenotype. 1942. *International journal of epidemiology*, 41, 10-13.
- WALTERS, D., KAKIETEK, J. J., EBERWEIN, J. D., PULLUM, T. & SHEKAR, M. 2016. Breastfeeding in the 21st century. *The Lancet*, 387, 2087-2087.
- WAN, A. K. L., SEOW, W. K., PURDIE, D. M., BIRD, P. S., WALSH, L. J. & TUDEHOPE, D.
 I. 2003. A Longitudinal Study of Streptococcus mutans Colonization in Infants after Tooth Eruption. *Journal of Dental Research*, 82, 504-508.
- WANG, J., CHEN, C. & GONG, X. 2021. The impact of family socioeconomic status and parenting styles on children's academic trajectories: A longitudinal study comparing migrant and urban children in China. *New directions for child and adolescent development*, 2021, 81-102.
- WANG, M.-T., KIURU, N., DEGOL, J. L. & SALMELA-ARO, K. 2018. Friends, academic achievement, and school engagement during adolescence: A social network approach to peer influence and selection effects. *Learning and Instruction*, 58, 148-160.
- WANG, M. T. & DEGOL, J. 2014. Staying Engaged: Knowledge and Research Needs in Student Engagement. *Child Development Perspectives*, 8, 137-143.
- WATT, R. G., DALY, B., ALLISON, P., MACPHERSON, L. M. D., VENTURELLI, R., LISTL, S., WEYANT, R. J., MATHUR, M. R., GUARNIZO-HERREÑO, C. C., CELESTE, R. K., PERES, M. A., KEARNS, C. & BENZIAN, H. 2019. Ending the neglect of global oral health: time for radical action. *Lancet*, 394, 261-272.
- WEBB, T. L., JOSEPH, J., YARDLEY, L. & MICHIE, S. 2010. Using the internet to promote health behavior change: A systematic review and meta-analysis of the impact of theoretical basis, use of behavior change techniques, and mode of delivery on efficacy. *Journal of medical Internet research,* 12, e4-e4.
- WEERHEIJM, K. L., DUGGAL, M., MEJÀRE, I., PAPAGIANNOULIS, L., KOCH, G., MARTENS, L. C. & HALLONSTEN, A.-L. 2003. Judgement criteria for Molar Incisor Hypomincralisation (MIH) in epidemiologic studies: A summary of the European meeting on MIH held in Athens, 2003. *European Journal of Paediatric Dentistry*, 4, 110-113.
- WEIS, M., HEIKAMP, T. & TROMMSDORFF, G. 2013. Gender differences in school achievement: The role of self-regulation. *Frontiers in psychology*, *4*, 442-442.
- WEISNER, T. S. 1996. The 5 to 7 transition as an ecocultural project. *The five to seven year shift: The age of reason and responsibility.* University of Chicago Press.
- WEIZMAN, Z. O. & SNOW, C. E. 2001. Lexical Input as Related to Children's Vocabulary Acquisition: Effects of Sophisticated Exposure and Support for Meaning. *Developmental Psychology*, 37, 265-279.
- WERNER, C. & SCHERMELLEH-ENGEL, K. 2010. Introduction to structural equation modeling with LISREL. *Goethe University, Frankfurt. Retrieved on February*, 5, 2011.
- WEST-EBERHARD, M. J. 2003. *Developmental plasticity and evolution,* Oxford ; New York, Oxford ; New York : Oxford University Press, 2003.
- WHELTON, H. 2004. Overview of the impact of changing global patterns of dental caries experience on caries clinical trials. *Journal of Dental Research*, 83, 29-34.
- WHITE, K. R. 1982. The relation between socioeconomic status and academic achievement. *Psychological bulletin,* 91, 461-481.
- WHO. Ottawa Charter for health promotion. The 1st International Conference on Health Promotion, 1986 1986 Ottawa. World Health Organization.
- WHO 1992. Recent advances in oral health: WHO Technical Report Series. In: ORGANIZATION, W. H. (ed.). Geneva.
- WHO 1997. Oral health surveys: Basic methods. Geneva: World health organization.

- WHO 2003. Oral health promotion: an essential element of a health-promoting school. World Health Organization.
- WHO 2013. WHO's oral health country-area-profile-programme (CAPP) database. Geneva.
- WHO. 2014. Health for the world's adolescents [Online]. Geneva, Switzerland: World Health Organization. Available: <u>https://www.who.int/publications/i/item/WHO-FWC-MCA-14.05</u> [Accessed 26 January 2021].
- WHO 2015. *Guideline: sugars intake for adults and children*, Geneva: World Health Organization.
- WILLIAM, V., MESSER, L. B. & BURROW, M. F. 2006. Molar incisor hypomineralization: review and recommendations for clinical management. *Pediatric dentistry*, 28, 224-232.
- WILSON, I. B. & CLEARY, P. D. 1995. Linking clinical variables with health-related quality of life: a conceptual model of patient outcomes. *Jama*, 273, 59-65.
- WONG, M. C. M., LU, H. X. & LO, E. C. M. 2012. Caries increment over 2 years in preschool children: a life course approach. *International Journal of Paediatric Dentistry*, 22, 77-84.
- WORLD BANK. 2020. *Bahrain economic profile* [Online]. World data bank. Available: <u>https://data.worldbank.org/country/bahrain?view=chart</u> [Accessed 23 September 2021].
- WORLD POPULATION. 2019. *Population of Cities of Bahrain review* [Online]. Available: <u>http://worldpopulationreview.com/countries/bahrain-population/cities/</u> [Accessed 24 March 2019].
- XIE, S. & LI, H. 2018. Does tiger parenting work in contemporary China? Exploring the relationships between parenting profiles and preschoolers' school readiness in a Chinese context. *Early child development and care*, 188, 1826-1842.
- YAJNIK, C., DESHPANDE, S., JACKSON, A., REFSUM, H., RAO, S., FISHER, D., BHAT, D., NAIK, S., COYAJI, K., JOGLEKAR, C., JOSHI, N., LUBREE, H., DESHPANDE, V., REGE, S. & FALL, C. 2008. Vitamin B 12 and folate concentrations during pregnancy and insulin resistance in the offspring: the Pune Maternal Nutrition Study. *Diabetologia*, 51, 29-38.
- YEUNG, C. A., CHONG, L. Y. & GLENNY, A. M. 2015. Fluoridated milk for preventing dental caries. *Cochrane Database of Systematic Reviews*, 9, CD003876-CD003876.
- YU, K. F., WEN, W., LIU, P., GAO, X., LO, E. C. M. & WONG, M. C. M. 2021. Effectiveness of Family-Centered Oral Health Promotion on Toddler Oral Health in Hong Kong. *Journal of dental research*, 2203452110366-220345211036663.
- YUANYUAN, L. 2018. Epigenetic Mechanisms Link Maternal Diets and Gut Microbiome to Obesity in the Offspring. *Frontiers in Genetics*, 9, 342-342.
- ZALECKIEN, V., PECIULIEN, V., ALEKSEJUNIEN, J., DRUKTEINIS, S., ZALECKAS, L. & BRUKIEN, V. 2020. Dental Trauma Experience, Attitudes and Trauma Prevention in 11-to 13-Year-Old Lithuanian Schoolchildren. *Oral health & preventive dentistry*, 18, 373-378.
- ZHANG, F., JIANG, Y., MING, H., REN, Y., WANG, L. & HUANG, S. 2020. Family socioeconomic status and children's academic achievement: The different roles of parental academic involvement and subjective social mobility. *British journal of educational psychology*, 90, 561-579.
- ZHOU, Y., LIN, H. C., LO, E. C. M. & WONG, M. C. M. 2011. Risk indicators for early childhood caries in 2-year-old children in southern China. *Australian Dental Journal*, 56, 33-39.
- ZHOU, Y., YANG, J., LO, E. & LIN, H. 2012. The Contribution of Life Course Determinants to Early Childhood Caries: A 2- Year Cohort Study. *Caries Research*, 46, 87-94.
- ZHU, W. 1997. Making Bootstrap Statistical Inferences: A Tutorial. *Research quarterly for exercise and sport,* 68, 44-55.
- ZIMMERMAN, B. J. 1990. Self-Regulated Learning and Academic Achievement: An Overview. *Educational psychologist*, 25, 3-17.

- ZIMMERMAN, B. J. 2011. Motivational sources and outcomes of self-regulated learning and performance. *Handbook of self-regulation of learning and performance*. New York, NY, US: Routledge/Taylor & Francis Group.
- ÅSTRØM, A. N. & OKULLO, I. 2003. Validity and reliability of the Oral Impacts on Daily Performance (OIDP) frequency scale: A cross- sectional study of adolescents in Uganda. *BMC Oral Health*, 3, 1-9.
- ÖZBEK, C., ESER, D., BEKTAS-KAYHAN, K. & ÜNÜR, M. 2015. Comparison of the tooth brushing habits of primary school age children and their parents. *Journal of Istanbul University Faculty of Dentistry*, 49, 33-40.

Chapter 8. Appendices

Appendix 1 Published paper

Mohamed SAS, Baker SR, Deery C, Vettore MV. Are oral health conditions associated with schoolchildren's performance and school attendance in the Kingdom of Bahrain? A life-course perspective. Int J Paediatr Dent. 2021 May 10. doi: 10.1111/ipd.12803. Epub ahead of print. PMID: 33971047.

ORIGINAL ARTICLE

AEDIATRIC DENTISTRY WILEY

Are oral health conditions associated with schoolchildren's performance and school attendance in the Kingdom of Bahrain? A life-course perspective

Seham A. S. Mohamed¹ | Sarah R. Baker¹ | Christopher Deery¹ | Mario V. Vettore²

Abstract

¹Unit of Oral Health, Dentistry and Society, School of Clinical Dentistry, University of Sheffield, Sheffield, UK ²Department of Health and Nursing Sciences, University of Agder, Kristiansand, Norway

Correspondence

Seham A. S. Mohamed, School of Clinical Dentistry, University of Sheffield, Sheffield, UK. Email: sehamdh@hotmail.com

Funding information The PhD research was funded by the University of Bahrain (Kingdom of Bahrain). Background: The link between oral diseases and school performance and school attendance remains unclear among Middle Eastern children.

Aim: To investigate the relationship of oral conditions with schoolchildren's school performance and attendance using the life-course approach.

Design: A cross-sectional study was conducted with 466 schoolchildren aged 7-8 years from Kingdom of Bahrain (KoB) and their parents. Questionnaire data on children's current and at-birth environmental characteristics were completed by their parents. Children's oral health measures, including ICDAS (International Caries Detection and Assessment System), PUFA (pulp, ulcer, fistula, abscess), and DDE (developmental defects of enamel) indices, were the exposure variables. School performance and school attendance data obtained from the school register were the outcome variables. The data were analysed using multivariate ordinal logistic regression. Results: The odds of excellent school performance were significantly lower for children with untreated dentinal caries (OR = 0.98; 95% CI: 0.96-0.99). Children with caries-treated teeth showed greater odds of excellent school performance (OR = 1.41; 95% CI: 1.15-1.74). Disease Control and PreventionNone of the dental conditions were significantly associated with children's school attendance. A permissive parental style was associated with poor school attendance (OR = 2.63; 95% CI: 1.08-6.42). Conclusion: Dental caries was associated with poor school performance but not with school attendance. Treated caries was associated with good school performance.

KEYWORDS

caries, children, life course, oral health, school attendance, school performance

1 | INTRODUCTION

In September 2015, the United Nations (UN) General Assembly adopted Agenda 2030, which agreed 17 Global

Sustainable Development Goals (GSDGs) to tackle poverty and reduce health inequalities. Education was the stepping stone for the GSDGs, and the specific goal related to education was to 'ensure inclusive and equitable

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made. © 2021 The Authors. International Journal of Paediatric Dentistry published by BSPD, IAPD and John Wiley & Sons Ltd

Int J Paediatr Dent. 2021;00:1-17.

wileyonlinelibrary.com/journal/ipd 1

Appendix 2. Search strategies for life course dental caries studies among children and adolescents (2012-2019)

1. Search date

From 2012-to 2021

2. Database

Scopus, MEDLINE, PubMed, and Google Scholar

3. Eligibility

The inclusion criteria were studies that explored the relationship between life course and dental caries with a clear definition of the life course theoretical framework, studies with children and adolescents up to 18-years old, and studies from any country but those published in the English language. The exclusion criteria included studies with adults over 18 years old, children and teenagers under 18 years (where it is impossible to analyse data for adults only), and studies published in languages other than English.

The search was carried out by the lead researcher (S.M).

4. Mesh terms

Mesh terms adopted from Abreu et al. (2015) to achieve consistency

Medline/Ovid

 1 - life event.mp. OR exp life change events/ OR life course.mp. OR life-course.mp. OR lifecourse.mp. OR life span.mp. OR life-span.mp. OR lifespan.mp. OR life time.mp. OR life-time.mp. OR lifetime.mp.
 2 - dental caries/ OR dental fissures/ OR root caries/ OR dental decay.mp. OR dental fissure.mp. OR tooth caries.mp. OR tooth decay.mp. OR caries disease.mp.

3 - adolescent/ OR child/ OR infant/ OR teenager.mp. OR exp child, preschool/ OR exp infant, newborn/ OR preschool children.mp

PubMed and Scopus

1 - life event OR life change events (Mesh) OR life course OR lifecourse OR life span OR life-span OR lifespan OR life time OR life-time OR lifetime
2 - dental caries (Mesh) OR dental fissures OR root caries (Mesh) OR dental decay OR dental fissure OR tooth caries OR tooth decay OR caries disease OR tooth disease
3 - adolescent (Mesh) OR teenager OR child (Mesh) OR infant (Mesh) OR preschool child (Mesh) OR newborn (Mesh) OR preschool children OR children

5. Result

Figure 1 illustrates the results of the electronic database search using the PRISMA

flow chart. The electronic search recognised 371 articles after removing duplicates;
347 articles did not meet eligibility criteria after screening title and abstract; 22 articles screened in full; 18 were excluded, and six new articles were found that studied dental caries in the life course. Further details are below:



Figure 1 Prisma flow chart for identifying studies of dental caries and life-course

5.1 Complete bibliographic details for excluded studies

- Arantes, R., Welch, J.R., Tavares, F.G., Ferreira, A.A., Vettore, M.V. and Coimbra Jr, C.E.A. (2018). Human ecological and social determinants of dental caries among the Xavante Indigenous people in Central Brazil. *PloS one.* 13(12), pp. e0208312. No life course model identified
- BaniHani, A., Deery, C., Toumba, J., Munyombwe, T. and Duggal, M. (2018). The impact of dental caries and its treatment by conventional or biological approaches on the oral health - related quality of life of children and carers. *International journal of paediatric dentistry*. **28**(2), pp.266-276. Not life course
- Baxevanos, K., Menexes, G., Lazaridou, A., Coolidge, T., Topitsoglou, V. and Kalfas, S. (2021). Dental caries and psychosocial factors: Testing a conceptual model in adolescents. *Community Dentistry and Oral Epidemiology*. **49**(4), pp. 314-321. No life course model identified
- Bernabe, E., Sabbah, W., Delgado-Angulo, E.K., Murasko, J.E. and Gansky, S.A. (2015). Income gradients in oral health according to child age. *European Journal of Oral Sciences.* **123**(4), pp.260-266. Income Inequality
- Bernabé, E., Ballantyne, H., Longbottom, C. and Pitts, N.B. (2020). Early Introduction of Sugar-Sweetened Beverages and Caries Trajectories from Age 12 to 48 Months. *J Dent Res.* 99(8), pp.898-906. Not available
- Birungi, N. (2016). Early life course factors, early childhood caries and oral health related quality of life among five-year-olds–a prospective and intergenerational study from eastern Uganda. PhD thesis
- Brondani, M., Amin, M., Poon, B. and Nicolau, B. (2016). Life Course Approach to Oral Health Research Workshop: A Summative Report. *J. Can. Dent. Assoc.* 82. Report
- Celeste, R.K., Eyjólfsdóttir, H.S., Lennartsson, C. and Fritzell, J. (2020). Socioeconomic Life Course Models and Oral Health: A Longitudinal Analysis. *J Dent Res.* 99(3), pp.257-263. No life course model identified
- 9. Costa, F.d.S., Agostini, B.A., Schuch, H.S., Correa, M.B., Goettems, M.L. and Demarco, F.F. (2019). Parent-child interaction and stimulation in early life can be related to caries in primary dentition? Hypotheses from a life-course approach. *Medical Hypotheses*. **130**.

Hypothesis

- Emmanuelli, B., Knorst, J.K., Menegazzo, G.R., Mendes, F.M. and Ardenghi, T.M. (2021). The Impact of Early Childhood Factors on Dental Caries Incidence in First Permanent Molars: A 7-Year Follow-Up Study. *Caries Research*. 55(3), pp.167-173. No life course model identified
- 11. Findley, P.A. and Weiner, R.C. (2020). Oral health across the life course: A role for social work. *Journal of Studies in Social Sciences and Humanities*. 6(1), pp.1-10. Review of the literature
- Hall-Scullin, E., Whitehead, H., Milsom, K., Tickle, M., Su, T.L. and Walsh, T. (2017). Longitudinal study of caries development from childhood to adolescence. *Journal of dental research.* **96**(7), pp.762-767. No life course model identified
- 13. Kavaliauskienė, A., Šidlauskas, A., Žemaitienė, M., Slabšinskienė, E. and Zaborskis, A. (2020). Relationships of dental caries and malocclusion with oral health-related quality of life in Lithuanian Adolescents aged 15 to 18 years: A cross-sectional study. *International Journal of Environmental Research and Public Health.* **17**(11), p4072. Not life course
- 14. Paixão-Gonçalves, S., Corrêa-Faria, P., Ferreira, F.M., Ramos-Jorge, M.L., Paiva, S.M. and Pordeus, I.A. (2019). Risk of Dental Caries in Primary Teeth with Developmental Defects of Enamel: A Longitudinal Study with a Multilevel Approach. *Caries research.* **53**(6), pp.667-674. Full article not available
- Peres, M.A., Sheiham, A., Liu, P., Demarco, F.F., Silva, A.E.R., Assunção, M.C., Menezes, A.M., Barros, F.C. and Peres, K.G. (2016). Sugar Consumption and Changes in Dental Caries from Childhood to Adolescence. *J Dent Res.* **95**(4), pp.388-394. Trajectories of feedings
- 16. Reyes, L.T., Knorst, J.K., Ortiz, F.R., Mendes, F.M. and Ardenghi, T.M. (2021). Pathways influencing dental caries increment among children: A cohort study. *Int J Paediatr Dent.* **31**(3), pp.422-432. Not life course
- 17. Teixeira, A.K.M., Roncalli, A.G. and Noro, L.R.A. (2016). Factors related to the dental caries incidence in youth: A cohort study in Brazilian Northeastern. *Ciencia e Saude Coletiva.* 21(12), pp.3871-3878. Young adults, not children
- 18. Zemaitiene, M., Grigalauskiene, R., Andruskeviciene, V., Matulaitiene, Z.K., Zubiene, J., Narbutaite, J. and Slabsinskiene, E. (2017). Dental caries risk

indicators in early childhood and their association with caries polarisation in adolescence: a cross-sectional study. *BMC Oral Health.* **17**(1), pp.1-6. No clear definition for the use of life course model

5.2 Complete bibliographic details for included studies

- Bernabé, E., MacRitchie, H., Longbottom, C., Pitts, N. B. & Sabbah, W. (2017) Birth weight, breastfeeding, maternal smoking and caries trajectories. *Journal* of dental research, 96(2),pp.171-178.
- Birungi, N., Fadnes, L. T., Kasangaki, A., Nankabirwa, V., Okullo, I., Lie, S. A., Tumwine, J. K. & Astrom, A. N. (2017) Assessing causal effects of early lifecourse factors on early childhood caries in 5-year-old Ugandan children using directed acyclic graphs (DAGs): A prospective cohort study. *Community Dent Oral Epidemiol*, **45**(6), pp.512-521.
- Priyanka, G. G., Kote, S., Sravanthi, K. V., Chethan, M., Anand, A., Jain, M. & Singh, S. (2019) Life course approach in the assessment of association between dental caries and health capital with family-related characteristics among 12-year-old school children. *Journal of family medicine and primary care*, 8(7), pp. 2506-2510.
- Silva, M. J., Kilpatrick, N. M., Craig, J. M., Manton, D. J., Leong, P., Burgner, D. P. & Scurrah, K. J. (2019) Genetic and Early-Life Environmental Influences on Dental Caries Risk: A Twin Study. *Pediatrics*, **143**(5). e20183499
- Sun, X., Bernabé, E., Liu, X., Gallagher, J. E. & Zheng, S. (2017) Early life factors and dental caries in 5-year- old children in China. *Journal of Dentistry*, 64,pp. 73-79.
- Velasco, S. R. M., da Silva Bastos, R., Silva, R. M. & de Lima Navarro, M. F. (2018) Association Between Presence of Enamel Defects, Dental Caries and Socioeconomic Conditions on Brazilian Children. *Pesquisa Brasileira em Odontopediatria e Clínica Integrada*, **18**(1), pp.3727.

Appendix 3. Amstar (Shea 2007)

BMC Medical Research Methodology 2007, 7:10

http://www.biomedcentral.com/1471-2288/7/10

Table 2: AMSTAR is a measurement tool created to assess the methodological quality of systematic reviews.

1. Was an 'a priori' design provided? The research question and inclusion criteria should be established before the conduct of the review.	Yes No Can't answer Not applicable
2. Was there duplicate study selection and data extraction? There should be at least two independent data extractors and a consensus procedure for disagreements should be in place.	Yes No Can't answer Not applicable
3. Was a comprehensive literature search performed? At least two electronic sources should be searched. The report must include years and databases used (e.g. Central, EMBASE, and MEDLINE). Key words and/or MESH terms must be stated and where feasible the search strategy should be provided. All searches should be supplemented by consulting current contents, reviews, textbooks, specialized registers, or experts in the particular field of study, and by reviewing the references in the studies found.	Yes No Can't answer Not applicable
4. Was the status of publication (i.e. grey literature) used as an inclusion criterion? The authors should state that they searched for reports regardless of their publication type. The authors should state whether or not they excluded any reports (from the systematic review), based on their publication status, language etc.	Yes No Can't answer Not applicable
5. Was a list of studies (included and excluded) provided? A list of included and excluded studies should <u>be provided</u> .	Yes No Can't answer Not applicable
6. Were the characteristics of the included studies provided? In an aggregated form such as a table, data from the original studies should be provided on the participants, interventions and outcomes. The ranges of characteristics in all the studies analyzed e.g. age, race, sex, relevant socioeconomic data, disease status, duration, severity, or other diseases should be reported.	Yes No Can't answer Not applicable
7. Was the scientific quality of the included studies assessed and documented? 'A priori' methods of assessment should be provided (e.g., for effectiveness studies if the author(s) chose to include only randomized, double-blind, placebo controlled studies, or allocation concealment as inclusion criteria); for other types of studies alternative items will be relevant.	Yes No Can't answer Not applicable
8. Was the scientific quality of the included studies used appropriately in formulating conclusions? The results of the methodological rigor and scientific quality should be considered in the analysis and the conclusions of the review, and explicitly stated in formulating recommendations.	Yes No Can't answer Not applicable
9. Were the methods used to combine the findings of studies appropriate? For the pooled results, a test should be done to ensure the studies were combinable, to assess their homogeneity (i.e. Chi- squared test for homogeneity, I ²). If heterogeneity exists a random effects model should be used and/or the clinical appropriateness of combining should be taken into consideration (i.e. is it sensible to combine?).	Yes No Can't answer Not applicable
10. Was the likelihood of publication bias assessed? An assessment of publication bias should include a combination of graphical aids (e.g., funnel plot, other available tests) and/or statistical tests (e.g., Egger regression test).	Yes No Can't answer Not applicable
11. Was the conflict of interest stated? Potential sources of support should be clearly acknowledged in both the systematic review and the included studies.	Yes No Can't answer Not applicable

Appendix 4. JBI Data Extraction Form for review for the four

systematic reviews

Study details/Author	Ruff, R. R., S. Senthi, S. R. Susser & A. Tsutsui (2018) "Oral health, academic performance, and school absenteeism in children and adolescents: A systematic review and meta- analysis: A systematic review and meta- analysis. The Journal of the American Dental AssociationOral health, academic performance and school absents in children and adolescent"
Objectives	"To conduct a systematic review and meta-analysis of the published literature on the potential association between oral health, school attendance, and academic performance in children aged 5 through 18 years".
Participants	Children 5-18 years, 139,989 children
Setting/Context	Schoolchildren
Phenomena of Interest	School performance and absenteeism (objective and subjective)
Search Details	English, 1945 through December 2017. Exposures included subjectively or objectively measured caries, oral pain, and periodontitis. Outcomes included school absence and school achievement.
Sources searched	PubMed, Embase, and Google Scholar
Range (years) of included studies	1996-2016
Number of studies included	14 studies
Types of studies included	(12 cross-sectional studies, 1 case-control study, and 1 longitudinal study)
Country of origin of included studies	Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia
Appraisal	The modified NOS for nonrandomized studies
Appraisal instruments used	The modified NOS for nonrandomized studies
Appraisal rating	≥4 low risk, < 4 high risk 4 high risk, and the rest low risk
Analysis	Meta-analysis
Method of analysis	Pooling study-Random effect
Outcome assessed	Objective & subjective OH (DMFT, tooth pain, periodontitis)
Results/Findings	Poor oral health was significantly associated with increased odds of poor academic performance (pooled odds ratio, 1.52; 95% confidence interval, 1.20 to 1.83) and absenteeism (pooled odds ratio, 1.43; 95% confidence interval, 1.24 to 1.63).
Significance/direction	
Heterogeneity	X ² and I ²
Comments	

Authors /Study details/	De Paula, J. S. & F. L. Mialhe (2013) "Impact of oral health conditions on school performance and lost school days by children and adolescents: What are the actual pieces of evidence? Brazilian Journal of Oral Sciences, 12, 189-198".
Objectives	"Investigate the pieces of evidence and discuss the methods and results of studies
	that assessed the existence of associations between the oral health status of
	children and adolescents, their school performance and lost school days due to
Participanta	102215 obildron
Satting/Context	schoolchildren
Phenomena of Interest	Academic performance and missing schools' day (Objective & subjective) C-OIDP
Search Details	The initial search -512 studies. One naner was evoluded after dunlicated. The
	titles and abstracts of the remaining 511 papers were reviewed for a more
	detailed evaluation. Only 17 papers included
	English-language papers (Jan/1970– Apr/2013)
Sources searched	Medline, PubMed Central
Range (years) of	1992-2012
included studies	
Number of studies	21
Number of studies included	21
Number of studies included Types of studies included	21 (7ecological+14cross-sectional)
Number of studies included Types of studies included Country of origin of	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda,
Number of studies included Types of studies included Country of origin of included studies	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan
Number of studies included Types of studies included Country of origin of included studies Appraisal	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No NO
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis Outcome assessed	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No No DMFT, CPI, OHI-S- dental trauma
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis Outcome assessed Results/Findings	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No No DMFT, CPI, OHI-S- dental trauma Significant associations were found between clinically detected and self-perceived are basily basily acted and self-perceived
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis Outcome assessed Results/Findings	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No DMFT, CPI, OHI-S- dental trauma Significant associations were found between clinically detected and self-perceived oral health status with school performance and school days lost by children and addressente
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis Outcome assessed Results/Findings	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No DMFT, CPI, OHI-S- dental trauma Significant associations were found between clinically detected and self-perceived oral health status with school performance and school days lost by children and adolescents
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis Outcome assessed Results/Findings	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No DMFT, CPI, OHI-S- dental trauma Significant associations were found between clinically detected and self-perceived oral health status with school performance and school days lost by children and adolescents
Number of studies included Types of studies included Country of origin of included studies Appraisal Appraisal instruments used Appraisal rating Analysis Method of analysis Outcome assessed Results/Findings Significance/direction Heterogeneity Comments	21 (7ecological+14cross-sectional) Greece, China, India, Thailand, Laos, USA, Brazil, Saudi Arabia, Uganda, Canada, Pakistan No No No No No DMFT, CPI, OHI-S- dental trauma Significant associations were found between clinically detected and self-perceived oral health status with school performance and school days lost by children and adolescents

Authors /Study details/	Rebelo, M. A. B., J. M. Rebelo Vieira, J. V. Pereira, L. N. Quadros & M. V. Vettore (2018) "Does oral health influence school performance and school attendance? A systematic review and meta-analysis. International journal of paediatric dentistry".
Obiectives	"To examine the evidence on the possible influence of oral health on school
	performance and school attendance in children and adolescents."
Participants	157325 children, 18 and under
Setting/Context	schoolchildren
Phenomena of Interest	Academic performance and missing schools' day (Objective & subjective)
Search Details	The initial search identified 9308. Of them, 3706 studies remained after excluding
	duplicated documents. All titles and abstracts were then reviewed, and 3638
	documents were excluded. The full text of the remaining 68. 50 were excluded
Courses searched	according to the inclusion criteria. In the end, a total of 18.
Sources searched	MEDLINE, SUOPUS, Web of Science, ScienceDirect, and LILAUS
included studies	1 May 2010
Number of studies	18
included	
Types of studies included	(1cohort+1case control+16cross-sectional)
Country of origin of	Australia, Indonesia, China, India, Thailand, USA, Brazil, Saudi Arabia,
included studies	
Appraisal	Yes
Appraisal instruments	Newcastle-Ottawa Scale (NOS) for cohort studies and case-control
used	- Modified
	- NOS for cross-sectional studies
Appraisal rating	Cohort=6-8 moderate, ≤ 5- low
	Cross-sectional= 4-6 moderate, ≤3=low
Analuaia	I wo low quality, and the rest moderate
Analysis Mathad of analysis	Yes Dendem effect Mete englysis
Method of analysis	Random enect meta-analysis
Outcome assessed	DMFT, dmit, and subjective
Results/Findings	Significant associations were found between clinically detected and self-perceived
	adolescents
Significance/direction	
Heterogeneity	Was not significant X ² and I ²
Comments	

Authors /Study details/	Ribeiro, A. P. D., R. F. Almeida, J. G. A. Medonca & S. C. Leal (2018) "Oral health and its effect on the academic performance of children and adolescents.			
Objectives	"To verify whether the child and adolescent oral health affected academic performance."			
Participants	2527children, 3-29-years old			
Setting/Context	schoolchildren			
Phenomena of Interest	Academic performance (Objective only) school, teacher report and grades			
Search Details	A total of 2,009 studies were retrieved, 14 of which were critically assessed. The reasons for excluding 1,995 studies were related to the use of quality of life questionnaires as the only measure of oral health. Of the 14 studies critically assessed, only six6.7.9.12-14 were included in the			
	gualitative synthesis. March 2017			
Sources searched	PubMed, Lilacs, Web of Science, and Scopus databases			
Range (years) of included studies	2003-2015			
Number of studies included	6			
Types of studies included	(Cross-sectional)			
Country of origin of included studies	USA, Brazil, Sudan, India			
Appraisal	Yes			
Appraisal instruments used	Quality assessment tool for observational cohort and cross-sectional studies developed by the National Heart, Lung and Blood Institute			
Appraisal rating	14=good 7= fair			
	no validity and reliability of exposure (no calibration, no reference regarding the number of examiners, and no kappa value for inter-and intra-examiner) and outcome measures (use of subjective criteria to classify students' performance) and no multivariate statistical analysis of t confounder variables = poor quality. Tw good, three poor, and one fair			
Analysis	No			
Method of analysis	No			
Outcome assessed	DMFT, dmft, and present or absent of caries			
Results/Findings	 4 papers (high risk of bias) children's academic performance and poor oral health were associated 2 papers (good quality) did not show an association between oral health and 			
	academic performance unless mediated by socioeconomic factors.			
Significance/direction	No association unless mediated by SES			
Heterogeneity	Was not significant X ² and I ²			
Comments				

Appendix 5. The summary of the included studies: exposures, outcomes and results

	Exposure	Instrument	Outcome	Instrument	Result
Agaku – 2015 Cross-sectional	Oral Health	Perceived by the parents	School Attendance (absenteeism)	Parents reporting (Subjectively measured questionnaire on school absenteeism)	Unmet treatment dental need & dental condition was associated with increased school absenteeism. (Mean number of days of school absence, 0.25 higher (95%Cl, 0.16 to 0.34) among students with unmet need due to dental condition compared to those with no unmet need).
Astrom & Okull – 2003 Cross-sectional	Caries	WHO (DMFT)	School performance	OIDP (oral impact on daily performance)	44% reported difficulties with eating and 30% didn't like showing their teeth. Then impact was on difficulties with cleaning their teeth 35% was common, followed by difficulties with speaking and completing school activities (34%).
Bernabe - 2007 a Cross-sectional	Malocclusion	Child-OIDP	School performance	OIDP (oral impact on daily performance)	Impact of self-perceived malocclusion primarily affected psychological function such as smiling, emotion and social contact
Bernabe - 2007 b Cross-sectional	Oral health	Child-OIDP	School performance	OIDP (oral impact on daily performance)	Smiling (18.8%) and studying (17.0%) were the most significant impacts on children's everyday activities
Blumenshine – 2008 Ecological	Oral health	Perceived by the parents	School performance	Perceived by the parents	Children with poor OH were 2 times more expected to have poor school performance.
Butani – 2009 Cross-sectional	Oral health	Perceived by the parents	School time lost due to oral health problems	Perceived by the parents	Children in special education classrooms were more likely to miss school for dental care or problems than those in the mainstream classroom
Colares & Feitosa – 2003 Cross-sectional	Caries	Present or not	School performance	Questionnaire (teachers)	Severe ECC was associated with poor academic performance.
David – 2006 Cross-sectional	Caries, OH	WHO; DMFT & OHI-S	School performance	Self-reported (Child report of teacher perceptions of performance)	23% of the children reported poor OH. Factors associated with poor school performance: bad breath, difficulty with eating, visiting the dentist, dissatisfaction

		Questionnaire: SES, self- reports OHK, a single- item teeth self-esteem			with teeth look and caries. Children with poor OH had increased OR (2.5) of poor school performance
De Paula – 2015 Cross-sectional	Caries, gingival health, MO	DMFT, BOP, DAI Questionnaire: family environment (OHRQoL, CPQ11-14, SES, children's health (SUBJECTIVE AND OBJECTIVE)	School performance	Cumulative grades at the end of the academic year (language skills, math, science, social science, and other skills). CPQ11-14,	Socio-environmental factors, self-perception and carious, have a significant impact on school performance. Children with poor OH had increased OR of poor school performance (OR, 5.19; 95% CI, 2.16 to 12.47). Number of people living in household (2.46; 1,15-5.24; P=0.019); Household over- crowding (6.96; 2.73-17.7; P<0.0001); Parental perception (3.18; 1.49-6.81; P=0.003); Carious lesions (5.19; 2.16-12.47; P=0.0002) CPQ† 30 (5.25; 1.87-14.76; P=0.0016).
Detty and Oza- Frank – 2014 Cross-sectional	Caries	Data from the Ohio survey	School performance	Objectively measured school-level	Prevalence of (mean score, 0.065 lower; P ¹ / ₄ .088) untreated caries at the school level was significantly related to lower school performance
Egri & Gunai – 2004 Ecological	Caries	Previously recorded WHO DMFT	School performance: 6 indicators (1-adult literacy rate, 2-mean years of schooling, 3- pupil-teacher ratio for primary level, 4- educational expenditure as a % of GNP, 5-primary enrolment ratio and 6- % completing a primary level	Human development report	High DMFT of 12-year-old children associated with the completion of primary school % in developing countries.
El-Sayed – 2015 Cross-sectional	Caries	DMFT, dmft	School performance and absenteeism	School reports academic performance: poor, acceptable, good, very good, and excellent.	Academic performance association with /dmft (P=0.008)/DMFT (P=0.023), but not school absenteeism (dmft/ (P=0.712/DMFT P=0.975).

Feitosa – 2005 Cross-sectional	Caries	Examination acc. To the Brazilian ministry of health criteria	School time lost due to OH problems	Questionnaire: parents or guardians.	Children with: severe caries, toothache, eating certain foods were absent from school; we are embarrassed to smile and avoid playing with other children because of their teeth. Severe caries was linked to a negative impact on children's OHRQoL
Freire – 2008 Cross-sectional	Caries & Height	DMFT and DMFS index	School performance	Failure at the school examination	Taller adolescents are associated with lower levels of caries. Social class and performance at school were found to be a cause of school failure, and both associated with height
Garg – 2012 Cross-sectional	Caries	Number d & f teeth (df-t index)	School performance	Overall marks from the teacher: excellent, average, below average	Poor OH is associated with low school performance.
Gherunpong – 200486 Cross-sectional	Caries, MO, OH	WHO; IOTN, OHI-S	School performance	Child-OIDP	High prevalence of OIDP. Oral impacts were a result of difficulty with eating and smiling, studying and maintaining social contacts.
Gift – 1992 Ecological	Dental visits/Problems (routine or emergency)	National health survey/ oral health supplement	School time lost due to oral health problems (Over the last 2 weeks)	Perception	Then 51 million hours of school lost annually by school-age children due to visit the dentist or OH problem at the national level. The lost hours increase with the children's age, among girls and children with low SES, and Hispanic.
Gradella – 2011 Cross-sectional	Caries & sequela	DMFT, PUFA	School Attendance	Perceived by the parents	Dental caries was significantly associated with missing school days but not pufa
Guarnizo-Herreno and Wehby – 2012 Cross-sectional	Oral health	Perceived by the mother (By the parents)	School performance and school attendance	Perceived by the parents	Children with OH problems were more likely to have school difficulties (OR 1.52), to miss school (OR 1.42), and were less likely to do homework (OR 0.76)
Jackson – 2011 Cross-sectional	Oral health	Perceived by the parents Toothache or routine dental care	School performance and missed school	Perceived by the parent's School grades	Children with poorer OH were more expected to have dental pain Children with pain 3 times more likely to miss school, and do poorly in school. Absence because pain have no mediation effect between OHS and performance
Jiang – 2005 Cross-sectional	Oral health	Self-reported by adolescents	School performance	Self-reported (WHO structured questionnaire of (1) general health (2) OH and QoL, (3) OH behaviour and lifestyles (4) OH knowledge and	OH, and treatment needs to be associated with gender, age, unhealthy lifestyle, poor school performance and SES. Perceived OH and needs were linked with poor school performance

				attitudes (5) school performance, (6) spare time activities (7) family status and lifestyle.	
Jurgensen & Petersen – 2009 Cross-sectional	Caries, Trauma, Periodontal status, toothache	WHO (DMFT)	School time lost due to OH problems	Questionnaire (self- reported in the previous 12 months)	Missing schools was related to high DMFT (2.6) and 0.2 for no missed days. High decay was seen in children with dental visits and frequent consumption of sweet drinks. Missed school classes, toothache and several impairments of daily life activities were associated with a high dD- component
Kaewkamnerdpong - 2018	Caries, OH	DMFT, OHI-S	School performance	The total score of the national standard examination.	DMFT was associated with lower school performance. The oral hygiene score was not.
Krisdapong - 2013	Caries, toothache	DMFT (untreated caries)	School performance and school absenteeism	Self-reported (difficulties in attending school, learning, homework (Child-OIDP). School absence due to toothache in the last 3 months.	For children aged 12 and 15 years, severe caries was not related to absences in adjusted models (adjusted OR, 1.7; 95%CI, 0.8 to 3.7) and (adjusted OR, 2.5; 95% CI, 0.9 to 6.8, respectively). Toothache associated with absenteeism
Maharani – 2017 Cross-sectional;	Toothache in the previous 12 months, observed oral cleanliness	Toothache by parental the past 12 months. Oral cleanliness by plaque index (Silness & Low-1964).	School performance	Score in mathematics	Children aged 11-12 with toothache had a lower school performance than their peers. Children aged 10–11 years with a toothache in the past 12 months had significantly lower school performance than their peers. For children aged 6-7 years, IT WAS NOT SIGNIFICANT. The presence of substantial plaque was significantly associated with child's school performance at; 6–7 years and 10–11 years
Milgrom – 1998 Cross-sectional	Caries	From school dental screening data	School time lost due to OH problems	Questionnaire by mother	Poor OH is associated with school absence and reduced school performance
Muirhead & Locker – 2006 Ecological	% of children in need of urgent treatment (severe trauma, large cavities, acute	From school dental screening data	School performance	Ontario's educational assessment organisation	School performance was a significant prophet of children in need of dental treatment, especially of urgency

	gingival swelling, infection or pain)				
Muirhead & Marcenes – 2004 Ecological	Caries	BASCD OH survey of 5 years old children	School performance	English, mathematics and literacy tests	Low school attainment, free meals at school, schoolchildren's addresses were good indicators for 5 years children OH.
Naavaal – 2018 Cross-sectional	Acute/unplanned dental care (past 6 months)	Parents reporting	School hours missed	Parents reporting	Emergency/unplanned dental care accounted for 34 million school-hours loss annually.
Nasuuna – 2016 Cohort	Oral health	Perceived by the parents	School performance	Scores (numeracy and reading ≤ national minimum standard)	Dental conditions determined poor reading skills. Developmental delay, epilepsy, dental problems, speech, intellectual disabilities and LBW were linked with poor reading scores than those without these conditions (P<0.05) and had a higher OR of being at (OR: 1.3) or below (OR: 3.7) the national minimum standard for reading).
Neves – 2016 Cross-sectional	Caries, trauma, MO	ICDAS, TDI MO=increased overbite (>2 mm), increased overjet (>2 mm), anterior open bite, anterior crossbite and posterior crossbite.	School days lost because of dental problems	Reported by the parents	Dental caries is associated with absenteeism only. The prevalence rate of preschool absenteeism due to oral problems was 8.4 %.
Pau – 2008 Cross-sectional	Dental pain (the past month only)	Dental pain questionnaire, Child- OIDP.	Attending school, learning, homework	Child-OIDP (oral impact on daily performance)	Students who stated difficulties with playing, relaxing or sleeping, going to school and speaking or pronouncing were more likely to visit a dentist because their 139/500 students had dental pain in the past month. 45 (32.4%) reported impacts of pain on going to school and learning in class compared to 94 (67.6%), in whom it did not impact their school tasks. (45 / 500 x 100 = 9% students with dental pain that reported impacts of dental pain on school attendance and performance
Paula – 2016 Case-Control	Caries, BoP	WHO, DMFT/dmft, BoP	School performance	Cumulative grades at the end of the year	Caries, even if treated, were risk indicators of poor school performance
Petersen – 2008 Cross-sectional	Oral health	Self-reported	School performance	Self-reported	The students with low academic attainments reported: periodic dental visits, sugar

Interdict Interdict Consumption, marking, activation, watching TV all a hand thin, activation of the second cross-sectional Consumption, marking, activation, watching TV all a hand thin, activation of the second cross-sectional Consumption, marking, activation, activation compared to the second cross-sectional Consumption, marking, activation cross-sectional Consumption, marking, activation compared to the second cross-sectional Consumption, marking, activation cross-sectional Consumption, marking, activation, cross- cross-sectional Consumption, marking, activation cross-sectional Consumption, marking, activation cross-sectional Consecton ton the second cross-secton ton cross-sectional <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>						
Provesan - 2012 Oral health, trauma, caries Self-perceived & DMFT (clicitally evaluated a norzero D or d component in DMFT indox and evidence or dental trauma) School performance, and school attendance and school attendance School performance, and school attendance School performance and school attendance Porgichil - 2008 Oral health Corise dental trauma Parents perception Absenteeism School attendance School attendance School attendance Being a pit was associated with ighter traction attendance Secteman - 2012 Oral health Cores-sectional Caries clinically: aschool performance, tractineralise attendance) tractineralise attendance School attendance tractineralise attendance) tractineralise attendance School performance, tractineralise attendance School attendance tractineralise attendance School attendance particineralise attendance School attendance School attendance School attenda	Petridou – 1996 Cross-sectional	Caries	DMFT – DMFS	School performance	Objectively measured grades on a 10-grade scale	consumption, smoking, alcohol, watching Tv for a long time, and playing on the computer. Good school performance is associated with better OH. Teeth filling was not significantly associated with school performance. Children achieved better by 1 grade, had a lower prevalence of DMFT by 0.2 and DMFS by 0.6)
Pongpicht – 2008 Longitudinal Oral pain Caries Parents perception Absenteeism School attendance records 159 missing school hours among children because of dental reasons (117 for dental appointments, 42 for dental pain). Low but considerable Seriewan – 2012 Cross-sectional Oral health Craises, dental cross-sectional Oral health Caries dental cross-sectional Caries clinically: cross-sectional School performance and white spot lesions, nonactivated – demineratication, colour and white spot lesions, nonactivated – demineratication, colour and white spot lesions; reatment (1) 'urgent' for immediate care involving pain, infection, swelling, extensive carious lesions, soft tissue lesions; (2) "early for care needed within 15 days to treat dental cares, mid gingivitis, and minimal calculus; or (3) 'routine' for dental cares, mid gingivitis, and minimal calculus; or (3) 'routine' for dental cares (mid gingivitis, and minimal calculus; or (3) 'routine' for dental care within 6 months for prevention. The school attendance Proficiency levels for attendance) Proficiency levels for absected and sec absected and sec absected and sec absected attendance absected active absected active absected absected active absected active absected active absected ac	Piovesan - 2012	Oral health, trauma, caries	Self-perceived & DMFT (clinically evaluated) (prevalence of untreated caries (corresponding to a nonzero D or d component in DMFT index and evidence of dental trauma)	School performance and school attendance	School reports: score in Portuguese language and number of school days missed in the last 3 months.	Children with DMFT had lower mean school performance, but the difference was not significant. Low household income (P<0.05), higher mean CPQ score (P<0.05) and higher school days missed (P<0.001) were individual determinants for lower school performance. Dental caries was not associated with Brazilian language skills or school days missed. Psychological and SES conditions determined children's academic performance. Being a girl was associated with higher school performance (P < 0.05)
Seirawan - 2012 Ecological Cross-sectional Oral health (creise, dental pain, presence of treatment needs) Caries clinically: cariate dearies - minimum of 0.5mm discontinuity of enamel and white spot lesions, nad white spot lesions, of minimum of 0.5mm School performance (School performance and attendance) Objectively measured school attendance a GPA The LA Unified School District (LAUSD) office Data area is and mathematics, and grade provident," Objectively measured school attendance) Objectively measured school attendance Objectively measured school attendance Objectively measured attendance) Objectively measured school attendance Objectively measured school attendance Objectively measured attendance Objectively measured att	Pongpichit – 2008 Longitudinal	3 Oral pain Caries	Parents perception	Absenteeism	School attendance records	159 missing school hours among children because of dental reasons (117 for dental appointments, 42 for dental pain). Low but considerable
status advanced periodontal conditions, or cuspidors soft tissue lesions; (2) "early" for care needed within 15 days to treat dental caries, mild gingivitis, and minimal calculus; or (3) "routine" for dental care within 6 months for prevention. Attendance was related to acute unplanned dental pain Shaikh - 2016 Toothache (past 6 months) Subjectively measured dental pain School Attendance (school record for 6 months (form was used to specify reasons for absence) Attendance was related to acute unplanned dental care. The prevalence of absenteeism due to toothache =18%)	Seirawan - 2012 Ecological Cross-sectional	Oral health (caries, dental pain, presence of dental caries and treatment needs)	Caries clinically: cavitated caries = minimum of 0.5mm discontinuity of enamel and white spot lesions, nonactivated = demineralisation, colour and translucency of the tooth surface. Pain by parent survey. Treatment: (1) "urgent" for immediate care involving pain, infection, swelling, extensive carious lesions,	school performance (School performance and attendance) Proficiency levels from 1 to 5 where 1 = "far below basic", and 5 = "advanced" 2="below basic 3="basic and 4="provident,"	Objectively measured school attendance a GPA The LA Unified School District (LAUSD) office DATA and accountability. Tests scores: English language, arts and mathematics, and grade point averages, number of absent days from the school register	Childen with toothache 4 times more likely to performe low GPA, and 6 times more likely to miss school days. Objective OH have no significnat reltaions
advanced periodontal conditions, or cuspidors soft tissue lesions; (2) "early" for care needed within 15 days to treat dental caries, mild gingivitis, and minimal calculus; or (3) "routine" for dental care within 6 months for prevention.Image: Care set of the set of th						
Shaikh - 2016 Cross-sectionalToothache (past 6 months)Subjectively measured dental painSchool AttendanceThe school attendance (school record for 6 months (form was used to specify reasons for absence)Attendance was related to acute unplanned dental care. The prevalence of absenteeism due to toothache =18%)			advanced periodontal conditions, or cuspidors soft tissue lesions; (2) "early" for care needed within 15 days to treat dental caries, mild gingivitis, and minimal calculus; or (3) "routine" for dental care within 6 months for prevention.			
	Shaikh – 2016 Cross-sectional	Toothache (past 6 months)	Subjectively measured dental pain	School Attendance	The school attendance (school record for 6 months (form was used to specify reasons for absence)	Attendance was related to acute unplanned dental care. The prevalence of absenteeism due to toothache =18%)

Appendix 6. Summary of the confounders in the included studies

Agaku – 2015 Cross-sectional	8	1- Gender. 2- Age. 3-Ethnicity. 4- Health insurance coverage. 5- Poverty level. 6- Presence of a co-morbid condition. 7- Ever repeated a grade. 8-Presence of smokers in the household.	The mean number of school's missing days related to illness/injury was significantly higher among females than males ($\beta = 0.11$; p b 0.001), those aged ≥ 15 years compared to ≤ 9 -year-olds ($\beta = 0.10$; p = 0.001), those with any health insurance coverage compared to those with none ($\beta=0.24$; p b 0.001), those with a co-morbidity compared to those with none ($\beta=0.59$; p b 0.001), those that had ever repeated a grade compared to those that had not ($\beta=0.09$; p = 0.020), and among those living with a smoker ($\beta = 0.21$; p b 0.001).
Astrom & Okull – 2003 Cross-sectional	5	It is adjusted for 1-Age. 2-Hender. 3- Place of residence (urban / ruler). 4- Religious affiliation. 5-Parental education.	After adjusting for socio-demographics, non-clinical and clinical variables, place of residence did not sustain its statistically significant association. Non-clinical and clinical variables mediated Even its effect. No statistically significant relationship effect between missing teeth and social and behavioural variables.
Bernabe - 2007 a Cross-sectional	0	1- Sex. 2- Age. 3-Education level (Primary school vs high school) of child	Only education level has significantly influenced the prevalence of self-perceived malocclusion and intensity of the impacts associated with self-perceived malocclusion.
Bernabe - 2007 b Cross-sectional	0	0	"Further studies are needed to explore how oral impacts vary by socio-demographic characteristics, such as gender, age or socio-economic status".
Blumenshine – 2008 Ecological	9	Controlling for 1-sex. 2-race. 3-parental education. 4-Medicaid insurance. 5- Public versus private school. 6-Presence of medical. 7-Presence of mental and behavioural conditions. 8-Learning disability. 9-school days missed >2 weeks.	Male (OR) = 1.86, 95% (CI) = 1.34, 2.60], African-American (OR = 1.50, 95\$ CI = 1.03, 2.08), had public or no insurance (OR = 1.72, 95% CI = 1.17, 2.53), parents with high school education or less (OR = 3.66, 95 percent CI = 2.35, 5.69) were more likely to have parentally reported poor school performance, mental disorder (OR = 2.68, 95% CI = 1.58, 4.54), diagnosis of behavioural conditions (OR = 3.98, 95% CI = 2.50, 6.34), and diagnosis of a learning disability (OR = 4.97, 95% CI = 3.10, 7.97) significantly linked to poor school performance. Children with both poor oral and poor general health were 2.3 times more likely to report poor school achievement than those with both good oral and general health (OR of 2.34 (95 % CI = 1.07, 5.67)
Butani – 2009 Cross-sectional	2	1-Age. 2-Gender	Adjusting for age and gender: children from special education classes were 2.4 times as likely to have a parental perception that their oral health was less favourable (good/fair/poor) (p = .001, 95% CI 1.54, 3.67).
Colares & Feitosa - 2003 Cross-sectional	0	0	0
David – 2006 Cross-sectional	4	1- Gender. 2-Place of residence. 3-SES. 4-Bleeding gums. 5-Bad breath 6- Toothache. 6-Food impaction. 7 - Dental visits. 8-Satisfied with the appearance of teeth 9-Oral health knowledge. 10- Caries experience 11- OHI 12-Anterior teeth fracture.	"This analysis showed that the initial results of unadjusted analyses were left essentially unchanged when cluster effects were taken into account".
Detty and Oza- Frank – 2014 Cross-sectional	1	School-based dental sealant program (SBSP)	The prevalence of untreated caries and academic performance at schools without an SBSP ($P = 0.001$) was significant determinants, but not at schools with an SBSP ($P = 0.833$) after adjusting for other school characteristics.
Egri & Gunai – 2004 Ecological			
El-Sayed – 2015 Cross-sectional	0	0	0

Feitosa – 2005 Cross-sectional	0	0	0
Freire – 2008			
Cross-sectional			
Garg – 2012	0	0	0
Cross-sectional			
Gherunpong – 200486	0	0	0
Cross-sectional			
Gift – 1992 Ecological	0	0	0
Gradella – 2011 Cross-sectional	5	Adjusted for 1-Gender. 2-Age. 3-Mother education. 4-House ownership. 5- Household overcrowding.	
Guarnizo-Herreno and Wehby – 2012 Cross-sectional	19	Adjusted: 1-child's age. 2-Gender. 3- Ethnicity. 4-country of birth (the US vs others). 5-Birth order. 6-Numbers of people in the household. 7-Maternal age. 8-Marital status. 9-Maternal educational. 10-household poverty-level. 11-Household employment status. 12-Child's health insurance coverage. 13-learning disability, attention deficit disorder or attention deficit hyperactive disorder. 14-behavioral or conduct problems. 15-autism. 16- developmental delay. 17-speech problems. 18-and hearing problems. 19- State-level	
Jackson – 2011 Cross-sectional	11	 Gender. 2-Race. 3- school grade. 4- Paternal education level. 5- Ethnicity. 6- Health insurance. 7-Special health care need. 8-OHS. 9-General health status. Days for school absence for routine dental care. 11-School absence caused by dental pain 	
Jiang – 2005 Cross-sectional	0	0	Multivariate analysis of perceived dental health: significant variables were gender, age, general hygiene practices, visiting a dentist, alcohol habits, oral hygiene practices, family income, attitudes towards dental health and parents' education. Finally, the most significant factors for experiencing dental symptoms: gender, age, general hygiene practices, visiting a physician, visiting a dentist, alcohol habits, consumption of sugary foods and drinks, and lifestyle of parents.

Jurgensen & Petersen – 2009	0	0	0
Cross-sectional			
Kaewkamnerdpong - 2018	14		
Krisdapong - 2013	8	1-Current level of study. 2-Area of residence. 3-School type. 4-dental care in the current semester. 5- DMFT. 6- Severe untreated caries. 7-Intensity of oral impacts. 8- Toothache as a perceived cause of oral impacts.	Multivariate analyses adjusted both age groups: lower grade of 12-15-year-old children were 2 times (95% CI: 1.2, 3.6) and 3.5 times (95% CI: 1.1, 11.0), respectively, more likely to miss school. Children who received dental care during the semester of the study were significantly more likely to miss school than those who did not. The intensity of overall OHRQoL impacts was also significantly associated with school absence in 12 years. Children with severe/moderate intensity OHRQoL were significantly 3 times more likely to miss school compared with those with no or minimal intensity impacts. The likelihood of absenteeism for children with OHRQoL impact at little intensity increased but was not significantly higher than those with no or minimal intensity impacts. Unadjusted models for both age groups: children with severe DMFT were 3 times more likely to missed school, whereas children with toothache were 2 times more likely to miss school. The relations between absence and severe caries and toothache were not significant in adjusted models.
Maharani – 2017 Cross-sectional:	0	0	0
Milgrom – 1998 Cross-sectional	0	0	0
Muirhead & Locker – 2006 Ecological	2	Educational and deprivation confounders (1-the % of grade3 ESL learners and 2- median family income)	
Muirhead & Marcenes – 2004 Ecological	0	0	
Naavaal – 2018 Cross-sectional	7	1-Gender. 2-Age. 3-Ethnicity. 4-Parental education. 5-Family income. 6-OHS. 7- Dental care.	Multivariate models 1: children with high family incomes (≥\$75,000) were less likely to need critical dental care in contrast to those in low-income families (OR = 0.69; 95% CI (0.51-0.92)) (Table 3). Model 2: Children with good OH were 1.6 times, and those with fair/poor OH were 2.7 times more likely to lose one or more hours compared to those with very good OH.
Nasuuna – 2016 Cohort	6		
Neves – 2016 Cross-sectional	4	1-child's gender, 2-child's age, 3-mother education4-household income	No association was found with the outcome.
Pau – 2008 Cross-sectional	0	0	0
Paula – 2015 Cross-sectional	5	Multivariate logistic model, Remains statistically significant when linked with adolescents' poor school performance in the final model:1- a number of individuals living in the	

		27	
		household, 2- overcrowding, 3-parental	
		perceptions about their children's OH, 4-	
		Carious lesions and a 5-Question from	
		CPQ11-14 (difficulty in paying attention in	
		class because of their teeth, lips, jaws or	
		mouth).	
Paula – 2016	6	1-Caries lesions with no treatment 2011	
Case-Control		2-Monthly household income. 3-Mother	
		education. 4- Overcrowding. 5-Children	
		living with both biological parents.	
Petersen – 2008	8	1-Gender. 2- Age. 3- Parental education	
Cross-sectional		level. 4-Family income. 5-Performance.	
		6-Peer relationship. 7-Attitudes towards	
		OH. 8-Knowledge about prevention of	
		dental disease.	
Petridou – 1996	4	1- Age. 2- Gender. 3- OH. 4. SES	Better DMFT among male students with high SES, brushing at least once per day, & better academic
Cross-sectional			performance.
Piovesan - 2012	7	1-Gender. 2- Ethnicity. 3- Household	The a significant relationship between performance and neighbourhood income ($b = 0.03$).
	~	income. 4- CPQ. 5- School Absenteeism.	
		6-School level average income. 7-	
		Number of professors at school.	
		(2&7 were not significant in the full	
		model)	
Pongpichit – 2008	0	0	0
Longitudinal	12210		
Seirawan - 2012	3	Logistic regression model adjusted the 1-	High school perform less than elementary school students in math P<.001).
Ecological		a type of school, 2- gender, and 3-	High school boys had a lower GPA than did high school girls (P= .002).
Cross-sectional		Ethnicity when needed.	Asian students performed statistically significantly better than did other ethnicities {P< .001), were the
			only ethnicity to achieve an average GPA >3. & advanced in English language arts. The average missing
			days of 6 for elementary school, compared with 2.6 among high school.
			Asian students had fewer absent days per school year than did all others (2.1) {P<,001).
Shaikh – 2016	0	0	The prevalence of absenteeism because of toothache was higher in 16 & 18-year-old female students (P
Cross-sectional			≤ 0.05)

Appendix 7. Search strategies to update evidence related to the impacts of OHCs on children's and adolescent's school performance and attendance

1. Search date

From 2018-July 2021

2.Database

Scopus, MEDLINE, and PubMed

3.Mesh terms

"children" or "adolescence" or "adolescent" and "oral health" or "dental health" or "oral conditions" or "dental conditions" or "dental problems" or "oral problems" or "dental needs" or "dental care" or "dental treatment" or "dental pain" or "dental decay" or "caries" or "oral infections" or "dmft" and "school performance" or "academic performance "or "school attendance" or "school absence" or "school absenteeism"

4.Eligibility

- Studies involving participants aged 19years
- Epidemiologic studies: Clinical trials and observational (cohort, case-control and cross-sectional) studies published in English
- Studies that examined assessed OHCs (clinical-self-reported)
- Available information regard school performance and/or school attendance

5.Results

Figure 1 illustrates the results of the electronic database search using the PRISMA flow chart (Moher et al., 2009). The electronic search recognised 2819 articles after removing duplicates. Two thousand eight hundred three articles did not meet eligibility criteria after screening title and abstract. Eight new articles were found that examined the relationship between school performance and attendance in children and adolescents. Details of included and excluded studies are found below

Figure 1 Prisma flow chart for identifying systematic reviews on the association between OHCs, and school achievement and attendance in children and adolescents (2018-2021)



Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group (2009). Preferred Reporting items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med 6(7): e1000097. doi:10.1371/journal.pmed1000097 For more information, visit www.prisma-statement.org.

5.1 Complete bibliographic details for excluded studies

- Quadri, M.F.A. and Ahmad, B. (2019). Is there evidence for the impact of poor oral health on school performance? *BMC Oral Health*. **19**(1), pp.143-143. Literature Review for Maharani et al., 2017
- Brignardello-Petersen, R. (2019). Children and adolescents with poor oral health are more likely to also have poor school performance and attendance, but there is no evidence that oral health is 1 of the causes. *J Am Dent Assoc.* **150**(5), pp.e51-e51. Literature Review for Rebelo et al., 2019
- Groenewald, C.B., Giles, M. and Palermo, T.M. (2019). School Absence Associated With Childhood Pain in the United States. *Clinical Journal of Pain.* 35(6), pp.525-531. Not accessible
- Karam, S.A., Costa, F.d.S., Schwendicke, F., Correa, M.B. and Demarco, F.F. (2021). Oral health and academic performance or absenteeism: Findings from a University in Southern Brazil. *Community Dent Oral Epidemiol.* 49(3), pp.267-274. The sample was University students, not children
- Martin, M.A. (2019). Oral health problems are associated with worse academic performance. *J Pediatr.* 214, pp.238-241.
 Evidence-based review by Martin of a different article by Guarnizo-Herre~no
- Neves É, T.B., Granville-Garcia, A.F., Dutra, L.D.C., Baccin Bendo, C., Ferreira, F.M., Paiva, S.M. and Horowitz, A.M. (2021). Association of Oral Health Literacy and School Factors with Untreated Dental Caries among 12-Year-Olds: A Multilevel Approach. *Caries Res.* 55(2), pp.144-152. Not accessible
- da Franca Bandeira Ferreira Santos, C., Godoy, F., Menezes, V.A., Colares, V., Zarzar, P.M., Ferreira, R.C. and Kawachi, I. (2020). School academic climate and oral health (tooth loss) in adolescents. *PLoS One.* **15**(5), pp.e0233505-e0233505. School climate as performance evaluated on the national, not individual level

5.2 Complete bibliographic details for included studies

- Almeida, R.F., Leal, S.C., Medonca, J.G.A., Hilgert, L.A. and Ribeiro, A.P.D. (2018). Oral health and school performance in a group of schoolchildren from the Federal District, Brazil: Oral health and its relationship with school performance of schoolchildren from Brazil. *Journal of public health dentistry*. **78**(4), pp.306-312.
- Gopalan, T., Asokan, S., John, J. and Geetha Priya, P. (2018). School absenteeism, academic performance, and self-esteem as proxy measures of oral health status: A cross-sectional study. *Journal of the Indian Society of Pedodontics and Preventive Dentistry.* **36**(4), pp.339-346.
- Guarnizo-Herreno, C.C., Lyu, W. and Wehby, G.L. (2019). Children's Oral Health and Academic Performance: Evidence of a Persisting Relationship Over the Last Decade in the United States. *Journal of Pediatrics*. 209, pp.183-+.
- Karki, S., Päkkilä, J., Laitala, M.L., Humagain, M. and Anttonen, V. (2019). Influence of dental caries on oral health - related quality of life, school absenteeism and school performance among Nepalese schoolchildren. *Community Dent Oral Epidemiol.* 47(6), pp.461-469.
- Cunha, I.P.d., Pereira, A.C., Meneghim, M.d.C., Frias, A.C. and Mialhe, F.L. (2019). Association between social conditions and oral health in school failure. *Rev Saude Publica.* 53, pp.108-108.
- Quadros, L.N., Rebelo, M.A.B., de Queiroz, A.C., Pereira, J.V., Vettore, M.V. and Rebelo Vieira, J.M. (2020). Clinical consequences of untreated dental caries and school performance in low-income adolescents. *Int J Paediatr Dent.* 31(5), pp. 619-626
- Darley, R.M., Karam, S.A., Costa, F.D.S., Correa, M.B. and Demarco, F.F. (2021). Association between dental pain, use of dental services and school absenteeism: 2015 National School Health Survey, Brazil. *Epidemiol Serv Saude.* 30(1), pp.e2020108-e2020108.
- Ortiz, F.R., Ardenghi, T.M., Paiva, S.M., Maroneze, M.C. and Pordeus, I.A. (2021). Impact of Oral Conditions and Subjective Factors on Academic Performance. *Pesquisa Brasileira Em Odontopediatria E Clinica Integrada*. 21, pp. e0233

Appendix 8. Muharraq's schools quality rating (2018-2019)

http://www.bqa.gov.bh/En/Reports/SchoolsReports/Pages/default.aspx

Mubarrag HB1 MALE		Outstanding		Satisfactory	Inadequate	
	Total Grade 2 Students	1	Good 2	3	4	Number of classes
1 Abu Obaid primary boys - not exist	75	1				3
2 Abu Firas alhamadani primary boys' school	97	1				3
3 Al- hidd primary boys' school	147				1	4
4 AI- khawarizmi primary boys' school	76	1				3
5 Al-busaiteen primary boys' school	123			1		4
6 Al-dair primary boys' school	128			1		4
7 Abu alala maari primary boys' school	64		1			2
8 Arad primary boys school	65			1		3
9 Hassan bin Thabit primary boys school	165		1			6
10 Omar bin Abdul Aziz primary boys school	81			1		3
11 Qalali primary boys school	87			1		3
12 Samaheej primary intermediate	87				1	3
13 Sh. Mohamed bin Isa al-Khalifa primary boys school	0		1			0
	1195					41
Muharraq HR1 FEMALE	Total Grade 2 Students	Outstanding 1	Good 2	Satisfactory 3	Inadequate 4	Number of classes
14 Al-busaiteen primary girls' school	160		1			5
15 Al-dair primary, an intermediate girls' school	101		1			3
16 Al-muharraq primary girls' school	150	1				5
17 Al-orouba primary girls' school	96	1				3
18 Aminah bint Wahab primary girls school	198	1				6
19 Arad primary girls school	83	1				3
20 Asma that alnetaqain primary girls' school	160		1			5
21 Ruqaya primary girls school	145			1		5
22 Zubaidah primary girls school	96	1				3
23 Mariam Bent Omran	96	1				3
	1285					41

Appendix 9. Participant Information Sheet and consent form



The Effects of Oral Health Conditions on Children's Academic



Performance and School Attendance

Appendix 2 Participant Information Sheet

You are invited to participate in a research project. Before deciding to or not to participate, kindly read the following information to understand why the study is being done, and why your participation is essential. Please take your time and discuss the information with others if you like. If there is anything not clear, do not hesitate to contact me.

Thank you

1. What is the project's purpose?

This research is requirement of a PhD project which aims to investigate how oral health and other social and environmental variables might restrict children's academic potentials over time. This will be conducted twice in September 2019, and September 2020.

2. Why have I been chosen?

You have been chosen because you live in Muharraq governorate, and your child in grade 2 attending one of Muharraq primary school. I will be asking around 500 parent-child pairs to participate as well.

3. Do I have to take part?

It is entirely up to you to participate. If you decide to take part, then will ask you to sign a consent, and you can withdraw at any time without the need for justification. This will not affect you and children at any time.

4. What will happen to me if I take part? What do I have to do?

If you decide to take part, then we will conduct oral health examination, body weight and height measurement and interview for your child at school, similarly to the one done by a dental hygienist in a regular basis. We will obtain your child birth weight and





The Effects of Oral Health Conditions on Children's Academic

Performance and School Attendance

gestational age from the health records. We will ask you about his academic score and attendance. Then will ask you as a parent to fill out a questionnaire regarding the life course of your child. The questionnaire might take 30 minutes to fill. However, you can have a week to fill it and send it back with your child to school in a sealed envelop.

5. What are the possible disadvantages and risks of taking part?

There are no risks expected from this study. We assured you that your name would never be in any report, we will write about the project. Every answer by you is our priority to keep it confidential, so you do not need to worry about others knowing your answers.

6. What are the possible benefits of taking part?

While there are no immediate benefits for you as a participant in this study, however, the information retrieved might support promotional and interventions directed for children in the future.

7. Will my taking part in this project be kept confidential?

All information collected about you and your family will be kept confidential, and out of reach only for the researcher. The questionnaire, will keep saving in a locked cabinet at the University of Bahrain for three years then will be terminated, and the other information will be saved without names at University of Sheffield google drive. The report of this project will not mention any name for any participants.

8. What will happen when the research project ends?

We will collect the information and analyse the result. We can send you a copy if you wish to know.





Performance and School Attendance

9. Who is organising and funding the research?

The research is organised by Seham Mohamed, who is currently a PhD student at the School of Clinical Dentistry at the University of Sheffield. The project funded by the University of Sheffield at the United Kingdom and Bahrain University. A team of supervisors maintains the study, Professor Sarah Baker, Head Academic Unit of oral health, Dentistry and Society of the School of Clinical Dentistry, University of Sheffield, Dr Mario Vettore, Senior Lecturer in the School of Clinical Dentistry, University of Sheffield, and Professor Chris Deer, Dean School of Clinical Dentistry at University of Sheffield.

10. What if something goes wrong and I wish to complain about the research?

If you feel unhappy or if you have any concerns regard the project, I will be happy to talk to you about it at any time. You can contact me using the below details, or contact the Patient Advice and Liaison Service at the hospital on Tel: (0114)2712450

11. Who has reviewed the study?

The Research Ethics Committee has reviewed the research proposal at the University of Sheffield and the Ministry of Health and Ministry of Education in Bahrain.

12. Contact for further information

There are three two senior members of staff involved besides me, and they are willing to help if you need any further information you need;





The Effects of Oral Health Conditions on Children's Academic

Performance and School Attendance

Name and details
Seham Mohamed
PhD student at the School of Clinical Dentistry University of Sheffield Dental Public Health Department 19 Claremont Crescent on whats up: 0097339070805 alternatively, by email: samohammed1@sheffield.ac.uk
Mario V Vettore
Adjunct Professor Dental School, Federal University of Minas Gerais R. Prof. Moacir Gomes de Freitas, 688 - Pampulha, Belo Horizonte - MG Brazil CEP: 31270-901
Telephone: +55 (31) 32455940
Email: mariovettore@gmail.com
FIDIESSUI Salali n Bakel
Professor of Psychology as applied to Dentistry School of Clinical Dentistry Claremont Crescent Sheffield S10 2TA
Telephone: +44 (0) 114 2159312
Email: s.r.baker@sheffield.ac.uk
riolessor onns Deery
Dean School of Clinical Dentistry School of Clinical Dentistry Claremont Crescent Sheffield
S10 2TA
Telephone: +44 (0) 114 2159415 Email: c.deery@sheffield.ac.uk



Performance and School Attendance

The Effects of Oral Health Conditions on Children's Academic

Please note that there is a consent form needed to be signed if you agree that you and your child participate in this project. If so, please sing it within the sealed enveloped attached in, and send it back with your child to school. Thank you very much for taking time reading the information sheet.





The Effects of Oral Health Conditions on Children's Academic

Performance and School Attendance

Participant Consent Form

Researcher Name: Seham Mohamed

Please tick the appropriate boxes	Yes	No
Taking Part in the Project		
I have read and understood the study information sheet dated 05/09/2019. I had an appropriate time to consider participation, and the study has been explained well to me.		
I have been given the time and the chance to ask questions about the study.		
I agree to take part in the project. I understand that taking part in the project will include oral examination for my child, retrieve his birth weight, gestational age from ministry of health records, school score and attendance from us as the child guardians. I agree to answer the questionnaire.		
I understand that my taking part is voluntary and that I can withdraw from the study at any time without justification.		
I understand my personal details such as name, and phone number will remain confidential, and will be stored, and will not be exposed only to the main researcher. This information will be permanently deleted from the records three years after publication of this study.		
I agree to participate with my child at this study.		
I would like to receive a copy of the study results.		
Name of Researcher Date Signature		
Name of Parent Date Signature		
Name of Student Bahrain Personal Number		
Please sign one copy and return it in the given envelope along with this consent, and return child to school, and keep one copy for your record	it with y	our/
Contact		
Seham Mohamed, PhD candidates in University of Sheffield, Faculty in Bahrain University WhatsApp 0097339070805		
Email: samohammed1@sheffield.ac.uk, sehamdh@hotmail.com		
School of Clinical Dentistry. Claremont Crescent. Sheffield. S10 2TA.		

Appendix 10. Parent's questionnaire (English)



Parent Que	estionnaire
1. Telephone contact:	2. Mobile:
	A Arra of fath and
5. Child Are (In completed verse)	4. Age of father:
Day\ Month\ Year\	
6. Child Gender O Male O Fen	nale
7. Place of Birth:	
8. Ethnicity O Bahraini O Gulf Cooperation Countries O Arab O A	Asian O African O European O Others
9. Area of residence	
O Muharraq governorate	
O The capital governorate	
O The southern governorate	
O The northern governorate	
Section 1 FAMILY STRUCTURE and SOCIOECONOMIC S	TATUS NOW
TO BE ANSWERED BY THE FATHER OR THE MOTHER O	OF THE CHILD
10. What type of a family does the child lives in?	
O Nuclear family (i.e. the parents and the children live togethe	er).
O Extended Family (i.e. the parents and the child/children live	e with one or more of the grandparents, uncles, aunts and
O Single parent family (i.e. Only one parent and the child/chil	dren live together)
11. What is the current marital status of the parents?	
O Married O Divorced O d	one of them widowed
12. How many brothers and sisters do this child have in	total excluding him/her?
13. What is the current mother's level of education?	
O Not educated	
O Primary school education (1 - 6 years)	
O Secondary school education (10 – 12 years)	
O College/ University	
O Post college	
14. What was the total number of educational years comp	pleted by the mother?
15. What is the current employment status of the mother	?
O Unemployed	
O Self-employed	
O Military	
O Retired	
O Government employee on the salary scale for public jobs	
O Government employee on the salary schedule for specialize	ed jobs
O Government employee on the salary schedule for executive	e iobs
O Government employee on the salary scale of academic jobs	5 5
O Government employee on the salary scale of diplomatic job	05
16. If you are an employee please type your job title	
17. What is the current father's level of education?	
O Not educated	
O Primary school education (1 - 6 years)	
U Intermediate school education (7 – 9 years)	
O Secondary school education (10 – 12 years)	
O Post college, how many by years	

	•				
. What is the current employment status of the	father?				
Unemployed					
Employed Employed					
Military					
Retired	licioho				
Government employee on the salary schedule for	specialized jobs	5			
Government employee on the salary scales for juc	lges jobs				
Government employee on the salary schedule for	executive jobs				
Government employee on the salary scale of diplo	matic jobs				
. If you are an employee please type your job t	itle				
. At your best estimate, any of the following w ahraini dinars.	ill appropriate	ly descri	be the monthl	y income of	your family i
BD<200					
BD 200-399, BD 400-599					
BD 600-799					
BD 800-999					
	nindat na su O				
. How can you describe your nome ownership	right now?				
Owned house Rented house					
Owned flat					
Rented flat	,				
Rented flat Others (please specify) , excluding yo	ur kitche	n and bathroo	oms?	
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once	n and bathroc	oms? ≻2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	oms? > 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroc	oms? > 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroc Twice a day	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroc Twice a day	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify) _, excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify A. How many rooms do your home contain now A. How many people lives in this house? A. How often 2 PARENTS and CHILD ORAL HEALTH E Statement Statement 26. How often does the MOTHER brush her teeth 27. How often does the MOTHER floss her teeth 28. How often does the FATHER brush his teeth 29. How often does the FATHER floss his teeth 30. How often does the FATHER use mouth wash 31. How often does the CHILD brush his/her teeth 32. How often does the CHILD floss his/her teeth) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify) , excluding yo BAHAVIOURS	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify A. How many rooms do your home contain now A. How many people lives in this house? A. How often 2 PARENTS and CHILD ORAL HEALTH E B. How often does the MOTHER brush her teeth 26. How often does the MOTHER loss her teeth 27. How often does the MOTHER loss her teeth 28. How often does the FATHER brush his teeth 29. How often does the FATHER use mouth wash 28. How often does the CHILD brush his/her teeth 30. How often does the CHILD brush his/her teeth 31. How often does the CHILD floss his/her teeth 32. How often does the CHILD use mouth wash 33. How often does the CHILD use mouth wash 34. How often does the CHILD use mouth wash 35. How often does the CHILD use mouth wash 36. How often does the CHILD use mouth wash 36. How often does the CHILD use mouth wash 36. How often does the CHILD use mouth wash 36. How often does the CHILD use mouth wash) , excluding yo BAHAVIOURS Never	UT kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify A. How many rooms do your home contain now A. How many people lives in this house? A. How often 2 PARENTS and CHILD ORAL HEALTH E B. How often does the MOTHER brush her teeth 26. How often does the MOTHER loss her teeth 27. How often does the MOTHER loss her teeth 28. How often does the FATHER brush his teeth 29. How often does the FATHER brush his teeth 20. How often does the FATHER loss his teeth 30. How often does the CHILD brush his/her teeth 31. How often does the CHILD brush his/her teeth 32. How often does the CHILD loss his/her teeth 33. How often does the CHILD use mouth wash 34. How often does the MOTHER eat sweet snack) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly
Rented flat Others (please specify A. How many rooms do your home contain now A. How many people lives in this house? A. How often 2 PARENTS and CHILD ORAL HEALTH E B. How often does the MOTHER brush her teeth 25. How often does the MOTHER loss her teeth 26. How often does the MOTHER use mouth wash 28. How often does the FATHER brush his teeth 29. How often does the FATHER loss his teeth 20. How often does the FATHER use mouth wash 31. How often does the CHILD brush his/her teeth 32. How often does the CHILD floss his/her teeth 33. How often does the CHILD use mouth wash 34. How often does the CHILD use mouth wash 35. How often does the CHILD use mouth wash 364. How often does the CHILD use mouth wash 365. How often does the FATHER eat sweet snack) , excluding yo BAHAVIOURS Never	ur kitche NOW Once a day	n and bathroo	> 2 a day	Weekly

		1 Exe	cellent	2 Good	3 Average	4 Fa	ir Poor
37. How would th generally	e <u>MOTHER</u> rate her oral he	alth					
38. How would th generally	e <u>MOTHER</u> rate her health						
39. How would th generally	e <u>FATHER</u> rate his oral hea	llth					
40. How would th generally	e <u>FATHER</u> rate his health						
41. How you wou generally	ıld rate your <u>CHILD</u> oral hea	lth					
42. How you wou	ild rate your <u>CHILD</u> health g	enerally					
43. If parents use	e a toothpaste, please deterr	nine the level	of fluori	de? Or wr	ite the tooth	paste na	me
O 500 O	1000 O 1350-1450	O ≥ 5000	0	Other			
14. If child use a	toothpaste, please determin	e the level of	fluoride	?			
0 500 O	1000 O 1350-1450	0 ≥ 5000	0	Other			
O Rinse with mo O Spit the toothp O I don't know 6. Does the MOT Yes	uth wash aste only without rinsing with HER visit the dentist?	h water	47. Do	es the FA	THER visit	he denti	st?
No			O No				
8. Why does the Emergency Regular checku Don't know	mother visit the dentist in ge	neral?	49. WI O Eme O Reg O Don	hy does th ergency jular chec i't know	ne father visi kup	t the den	itist in general?
0. If it's for check Every 6 months Once a year Once every two Less often I can't remembe	s up, how often does the MC years r	OTHER go?	51. If i O Eve O Onc O Onc O Less O I ca	t's for che ry 6 mont ce a year ce every tw s often n't remem	cks up, how hs vo years ber	often do	bes the FATHER go
2. Does the MOT Yes No 4. If yes, what typ Cigarettes, HOV Pipe Others	HER smoke? ve? V many per day	_	53. Do O Yes O No 55. If y O Ciga O Pipe	es the FA ves, what arettes, H e ers	NTHER smol type? OW many p	ke? er day	
6. Does the child	have any chronic disease fr	om below?					
Asthma	O Iron deficiency	O Sickle o	cell disea	ises	0 Ep	ilepsy	O Cerebral palsy
Thalassemia	O Type I diabetic Mellitus	O Disabili visual, sp	ties; hea eech	ring, men	tal, O Au	tism	O (ADHD)
7. Does the child	currently taking any medica	tion? If yes, p	lease sp	ecify			÷
Yes	0	No					

Section 3 Kindly note that this section of the questions looks alike at the first instant with the previous, however,
it asks about the situation when the child BORN.
FAMILY STRUCTURE and SOCIOECONOMIC STATUS WHEN THE CHILD BORN
58. What was the type of family when the baby was born?
O Nuclear family (i.e. the parents and the children live together).
O Extended Family (i.e. the parents and the child/children live with one or more of the grandparents, uncles, aunts and
cousins)
O Single parent family (i.e. Only one parent and the child/children live together)
59. What was the marital status of the parents?
O Married O Divorced O one of them widowed
60. How many brothers and sisters did this child had in total at birth except for him/her?
61. What was the birth order of this child compared to his/her siblings?
O First O Other, specify
o secolid
62. What was the level of mother's education when the child was born??
O Not educated
O Primary school education (1 - 6 years)
O Intermediate school education (7 – 9 years)
O Secondary school education (10 – 12 years)
62 What was the method's job when the baby was here?
O Self-employed
O Employed in the private sector
O Retired
O Government employee on the salary scale for public jobs
O Government employee on the salary schedule for specialized jobs
O Government employee on the salary scales for judges' jobs
O Government employee on the salary schedule for executive jobs
O Government employee on the salary scale of academic jobs
O Government employee on the salary scale of diplomatic jobs
64. Please type the job title if there was any
65. What was the father's level of education at the time when the child was horn?
0. What was the father is level of education at the time when the clinit was both:
O Not educated
O Primary school education (1 - 6 years)
O intermediate school education $(1 - 9)$ years)
O College Iniversity, how many by years
O Post college on versa, new many by years
or for conego, new many by your contract of the second sec
66. What was the employment status of the father at the time when the child was born?
O Unemployed
O Self-employed
O Employed in the private sector
O Military
O Retired
O Government employee on the salary scale for public jobs
O Government employee on the salary schedule for specialized jobs
O Government employee on the salary scales for judges jobs
O Government employee on the salary schedule for executive jobs
O Government employee on the salary scale of dialogatic jobs
67. Please type the job title if there was any

68. At your best estimate, any of the following would appropriately describe the monthly income of your family in Bahraini dinars when the child was born.
O BD<200
O BD 200-399,
D BD 400-599
D BD 600-799
2 BD 800-999
69. How can you describe the ownership of the house when the child was born?
D Owned house
J Rented noise
/0. How many rooms were in your house at that time, except for the kitchen and the bathrooms?
Mother Status at Pregnancy and at The Child Birth
71. How old was the mother at the time of the birth of this child? (In completed years)
Day\ Month\ Year\
72. Did the mother suffer from diabetes mellitus during pregnancy with this child?
O Yes O No
73. Did the mother suffer from hypertension during pregnancy with this child?
O Yes O No
74. Did the mother suffer from any other condition during pregnancy with this child please specify?
75. When the methor on patience and for a using algorithm or shiple) of the time of this shild programmy?
2. Was the motifier an active shoker (i.e. using cigareties of shisha) at the time of this child pregnancy?
7 Festive did you give bitth for this child?
77 Did you suffer from the postnatal depression after having this baby?
77. Did you suffer from the postnatal depression after having this baby?
77. Did you suffer from the postnatal depression after having this baby? O Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS
77. Did you suffer from the postnatal depression after having this baby? O Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)?
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks)
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth??
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Normal (i.e. less than 2500 g)
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Normal (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food)
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Normal (i.e. equal to or more than 2500 g) 0 Normal (i.e. equal to or more than 2500 g) 0 Normal (i.e. less than 2500 g) 0 Now as the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) 0 Yes O No
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Low-birth-weight (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) O Yes O No 81. If the child relied on breastfeeding only, for how long that was ?
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Low-birth-weight (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) O Yes O No 81. If the child relied on breastfeeding only, for how long that was ? O Less than two years O Equal or more than two years
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Low-birth-weight (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) O Yes O No 81. If the child relied on breastfeeding only, for how long that was ? O Less than two years O Equal or more than two years 82. Did the child rely on drinking a bottle of formula milk before or during his night sleep?
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) O Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Low-birth-weight (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) O Yes O No 81. If the child relied on breastfeeding only, for how long that was ? O Less than two years O Equal or more than two years O Equal or more than two years 0 Equal or more than two years
77. Did you suffer from the postnatal depression after having this baby? 0. Yes 0. No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? 0. Normal (i.e. equal to or more than 37 weeks) 0. Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? 0. Normal (i.e. equal to or more than 2500 g) 0. Low-birth-weight (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) 0. Yes 0 No 81. If the child relied on breastfeeding only, for how long that was ? 0. Less than two years 0. Equal or more than two years 82. Did the child rely on drinking a bottle of formula milk before or during his night sleep? 0. Yes 0 No 83. Did the child use to have sweetened drinks other than milk in his feeding bottle?
77. Did you suffer from the postnatal depression after having this baby? D Yes O No BIOLOGY AND BEHAVIOR OF THE CHILD AT THE AGE OF TWO YEARS 78. How long have you been pregnant with this baby (gestational age)? O Normal (i.e. equal to or more than 37 weeks) D Pre-term birth (i.e. less than 37 weeks) 79. What was the weight of the child at birth?? O Normal (i.e. equal to or more than 2500 g) O Low-birth-weight (i.e. less than 2500 g) O Low-birth-weight (i.e. less than 2500 g) 80. Was the child breastfeed only by mother for the first six months of his life?? (i.e. without other supplemental food) O Yes O No 81. If the child relied on breastfeeding only, for how long that was ? O Less than two years O Equal or more than two years 82. Did the child rely on drinking a bottle of formula milk before or during his night sleep? O Yes O No 83. Did the child use to have sweetened drinks other than milk in his feeding bottle? O Yes (i.e. equal to or more than one time per day), SPECIFY please O Yes (i.e. less than one time per day)
84. Did your child
--
O Yes
85. If the answer
86. How long did l
CHILD BIOLOGY
87. When did the d
O At equal to or less than one year O At more than one year
88. Was the child
O Always O Mostly O Occasionally O Never
89. Did the child vis
O Yes O Never visited the O No, except in em
90. Did the child su
O Yes O No
91. Did the child su
O Yes O No
92. If the answer to
O Yes O No
93. Did the child en
O Yes
94. If the answer to
O One year O Two years O Three years
SECTION 4 THE
95. What was the o
O Excellent
96. How many scho
O No absent
97. Had your child
0 Yes
98. If the previous
<u></u>

SECTION 5 PARENTING STYLE, PLEASE TICK ONE BOX									
Statement	Never	Rarely	Sometimes	Very often	Always				
Mother Parenting Style	R.				- F				
99. I am responsiveness for my chil's feelings and needs									
100. I take my child's wishes into consideration before I ask him/her to do something									
101. I explain to my child how I feel about his/her good/bad behaviour									
102. I encourage my child to talk about his/her feelings and problems									
103. I encourage my child to freely "speak his/her mind", even if he/she disagrees with me									
104. I explain to my child my expectations from him									
105. I provide my child reasons for the expectations I have for him/her									
106. I compliment my child									
107. I consider my child's preferences when I make plans for the family (e.g., weekends away and holidays									
108. I respect my child's opinion and encourage him/her to express them									
109. I allow my child to give input into our family rules									
110.I explain to my child the consequences of his behaviour									
111.I give my child reasons why rules should be obeyed									
112. I provide comfort and understanding when my child is upset									
113. I have warm and intimate times together with my child:									
114. When my child asks me why he/she must do something, I tell him/her is because I said it, and I am your Mother									
115. I punish my child by taking privileges away from him/her (e.g., TV, games, visiting friends)									
116. I yell when I disapprove of my child's behaviour									
117. I explode in anger towards my child									
118. I spank my child when I don't like what he/she does or says									
119. I use criticism to make my child improve his/her behaviour									
120. use threats as a form of punishment with little or no justification									
121. I punish my child by withholding emotional expressions (e.g., kisses and cuddles									
122. I openly criticise my child when his/her behaviour does not meet my expectations									
123. I feel the need to point out my child's past behavioural problems to make sure he/she will not do them again									
124. I remind my child that I am his/her parent									
125. I remind my child of all the things I am doing and I have done for him/her									
126. I find it difficult to discipline my child									
127. I give into my child when he/she causes a commotion about something									

400 1 11 111	T	1	1	1	1
128. I spoli my child					
129. I ignore my child's bad behaviour					
130. I threatened my child with punishment more often than giving it					
Statement	Never	Rarely	Sometimes	Very often	Always
Father Parenting Style					
131. I am responsiveness for my chil's feelings and					
needs					
132. I take my child's wishes into consideration before I ask him/her to do something					
133. I explain to my child how I feel about his/her good/bad behaviour					
134. I encourage my child to talk about his/her feelings and problems					
135. I encourage my child to freely "speak his/her mind", even if he/she disagrees with me					
136. I explain the reasons behind my expectations					
137. I provide comfort and understanding when my child is upset					
138. I compliment my child					
139. I consider my child's preferences when I make plans for the family (e.g., weekends away and bolidays					
140. I respect my child's opinion and encourage him/her to express them					
141. I allow my child to give input into our family rules					
142. I explain to my child the consequences of his behaviour					
143. I give my child reasons why rules should be obeved					
144. I provide my child reasons for the expectations I have for him/her					
145. I have warm and intimate times together with my child:					
146. When my child asks me why he/she must do something, I tell him/her is because I said it, and I am your Father					
147. I punish my child by taking privileges away from him/her (e.g., TV, games, visiting friends)					
148. I yell when I disapprove of my child's behaviour					
149. I explode in anger towards my child					
150. I spank my child when I don't like what he/she does or says					
151. I use criticism to make my child improve his/her behaviour					
152. use threats as a form of punishment with little or no justification					
153. I punish my child by withholding emotional expressions (e.g., kisses and cuddles					
154. I openly criticise my child when his/her behaviour does not meet my expectations					
155. I feel the need to point out my child's past behavioural problems to make sure he/she will not do them again					
156. I remind my child that I am his/her parent					
157. I remind my child of all the things I am doing, and I have done for him/her					
158. I find it difficult to discipline my child					

159. I give into my child when he/she causes a commotion about something					
160. I spoil my child					
161. I ignore my child's bad behaviour					
162. I threatened my child with punishment more often than giving it.					
PARENTS SCHOOL SUPPORT					
Statement	Never	Rarely	Sometimes	Very often	Always
Mother Academic Support					
163. I attend parent teacher meeting at my child school					
164. I discuss my child school progress with his teacher					
165. I attend school events and activities					
166. I make sure my child does his homework					
167. I motivate my child to work harder when he gets poor grade					
168. I volunteer at my child school					
169. I talk with my child regarding my expectations for him/her school work					
Statement	Never	Rarely	Sometimes	Very often	Always
Father Academic Support					
170 I attend parent teacher meeting at my child school					
171. I discuss my child school progress with his teacher					
172. I attend school events and activities					
173. I make sure my child does his homework					
174. I motivate my child to work harder when he gets poor grade					
175. I volunteer at your school					
176. I talk with my child regarding my expectations for him/her school work					

Appendix 11. Children's face-face questionnaire





Children Face-Face Questionnaire									
Student ID: Site	• ID:								
Interviewer Number:									
Statement	\bigcirc	\bigcirc	\bigcirc						
Teacher Perceived Support	Yes	Sometimes	No						
1. My teacher is nice and friendly									
2. My teacher likes me									
3. When I need extra help, my teacher gives it to me									
Classmate Perceived Support									
4. The student in my class enjoy being together									
5. Most of the student in my class are kind and helpful									
6. When a student upset, another student comfort him/her									





Modified Pediatric Oral Quality Of Life (POQL) Child Self-Report												
	Excellent	Good	Average	Fair	Poor							
1. How would rate your oral health?												
2. How would you rate your general health?												

		How	often did	this happe	ən?		How b	othered	were you	u?
		All the time	Some of the times	Once in a while	Did not happe n	Very	someti mes	A little bite	never	Did not happen
		8	8	₿	•	\odot	\odot	•••	\odot	\odot
1.	Did you have pain because of your teeth or mouth?									
2.	Did you have trouble eating any foods (hard/hot/cold) because of your teeth or mouth?									
3.	Did you have trouble being attentive in school because of your teeth and mouth?									
4.	Did you miss school days because of your teeth or mouth?									
5.	Did not want to laugh or smile around others because of your teeth or mouth?									
6.	Did you worry that you were not a good looking to others because of your teeth or mouth?									
7.	Were you unhappy with the way you looked because of your teeth or mouth?									
8.	Were you angry or upset because of your teeth or mouth?									
9.	Did you feel worried because of your teeth or mouth?									
10.	Did you cry because of your teeth or mouth?									
Tot	al									

Appendix 12. Dental examination form

Participant study number	Examiner/Recorder
School Code	Date/Septemper/2019

Examination Form

1- Occlusion For Anetrior teeth

 □
 Lip Competent:
 Yes/ No
 □
 Overjet:

 □
 Ovebite:

 □
 Anterior Openbite:
 Yes / No/ NA
 □
 Anterior Crossbite:
 Yes/ No/ NA

2- Dental trauma index (PERMANENT INCISOR)

12	11	21	22
42	41	31	32

S1	S2	S3
S6	S5	S4

3- Dental biofilm

3- ICDAS , PUFA/pufs and EDD

	17	16	15	14	13	12	11		21	22	23	24	25	26	27
D								м							
0					\geq	$\overline{\times}$	\times	0	\ge	\geq	\geq				
м								D							
В								В							
L								L							
PUFA															
EDD															
			55	54	53	52	51		61	62	63	64	65		
		D						М							
		0			\boxtimes	\times	\times	0	\bowtie	\times	\times				
		м						D							
		в						В							
		L						L							
		pufa													
		EDD													

	47	46	45	44	43	42	41	<u></u>	31	32	33	34	35	36	37
D								М							
0					\geq	\ge	\times	0	\succ	\ge	\ge				
М								D							
В								В							
L								L							
PUFA															
EDD															
			85	84	83	82	81		71	72	73	74	75		
		D						м						l l	
		0			\times	\times	\times	0	\bowtie	\times	\times				
		м						D							
		В						В						l l	
		L						L							
		pufa													
		EDD													

Appendix 13. Weight and Height form

Examiner..... Date..... Date.....

	Student Number	Student Name	Weight KG	Height CM
1.				
2.				
3.				
4.				
5.				
6.				
7.				
8.				
9.				
10.				
11.				
12.				
13.				
14.				
15.				
16.				
17.				
18.				
19.				
20.				
21.				
22.				
23.				
24.				
25.				
26.				
27.				
28.				
29.				
30.				
31.				
32.				
33.				
34.				
	1	1	1	

Grade 2 Weight and Height

Appendix 14. University of Sheffield ethical approval



Downloaded: 28/05/2019 Approved: 28/05/2019

Seham Mohamed Registration number: 180129258 School of Clinical Dentistry Programme: Standard PhD in the School of Clinical Dentistry

Dear Seham

PROJECT TITLE: The Effects of Oral Health Conditions on Childrens Academic Performance and School Attendance Using A Life Course Ap-proach: Longitudinal Study in the Kingdom of Bahrain **APPLICATION:** Reference Number 026444

On behalf of the University ethics reviewers who reviewed your project, I am pleased to inform you that on 28/05/2019 the above-named project was **approved** on ethics grounds, on the basis that you will adhere to the following documentation that you submitted for ethics review:

- University research ethics application form 026444 (dated 08/05/2019).
- Participant information sheet 1061375 version 1 (07/05/2019).
- Participant consent form 1061376 version 1 (07/05/2019).

The following optional amendments were suggested:

1. Place version numbers of the documents and date as footnotes 2. Make sure participants get clean versions minus track changes 3. Amend questions to children to make them simpler for example instead of 'how would you rate your oral health' for children why not do you think your teeth and mouth are healthy? 4. Give the name of someone at the school who would be willing to talk to parents if they are upset

If during the course of the project you need to <u>deviate significantly from the above-approved documentation</u> please inform me since written approval will be required.

Yours sincerely

Janine Owens Ethics Administrator School of Clinical Dentistry

Appendix 15. Bahrain's Ministry of Education ethical approval

KINGDOM OF BAHRAIN Ministry of Education

Number: 03/04/tmb2019

Date: 16 May 2019





[emblem: 2019, 100 years of formal education in the Kingdom of Bahrain]

Dear Respected Teacher Seham Abdulla Saleh

Warm regards,

RE: Approval of the application of research tools by the Primary Education Directorate

With reference to your application on 15/04/2019 regarding the use of research tools entitled, 'The Relationship of Oral Diseases to the Academic Performance of Students: Using the Human Life Cycle Approach' - I am pleased to inform you that the application for research tools to the Primary Education Directorate has been approved.

The use of these tools should be carried out in accordance with the following instructions and conditions:

1. Adherence to the use of the approved tool(s), without addition to, or omission of such tools.

2. During the use of research tools, it must be ensured that the educational process and achievement of students is not harmed.

3. The information gathered must be protected and used only for the purpose of scientific research.

4. Upon its completion, a copy of the research must be provided to the Department of Public Libraries at the Ministry of Education.

5. The researcher must bear responsibility for all procedures and steps necessary to implement the use of the research tools. They are not entitled to rely upon the Ministry staff to fulfil this task.

With our best wishes for your success, please accept our sincerest regards.

[signature]

Dr. Farzana Abdulla Al Maraghi

Principal of the Department of Scientific Research



BAHRAIN

Scientific Research Directorate at the Secretariat-General of the Higher Education Council Tel : (00903) 17873061 – Fax : (00973) 17680168 P.O. Box : 43 Kingdom of Bahrain

Appendix 16. Bahrain's Ministry of Health ethical approval

Kingdom of Bahrain Ministry of Health Office of Asst. Undersecretary For Human Resources & Services



No: AURS/ 402 /2019 Date: 22 July 2019

To: Seham Mohamed School of Clinical Dentistry University of Sheffield

Subject: Letter of Approval for Research Proposal:

"The Effects of Oral Health Conditions on Children's Academic Performance and School Attendance Using A Life Course Approach: Longitudinal Study in the Kingdom of Bahrain"

Dear Mrs. Seham,

Thank you for submitting your research proposal documents, which have been considered by members of the Research Technical Support Team (RTST) on 29th May 2019.

We would like to inform you that the team found no major ethical issues or methodological problems that would hinder the conduct of this survey. We are thus pleased to approve the above application.

This approval is subject to the following conditions:

- 1. We expect that the study will begin within 6 months of the date of this approval.
- Approval from an MOH Research Committee does not automatically imply that the researcher is granted access to data, medical records or biological samples from MOH healthcare facilities. Researchers must seek permission and follow procedures as dictated by the concerned departments after presenting them with a valid MOH approval letter.
- Any significant change, which occurs in connection with this study and/or which may alter its ethical consideration, must be reported immediately to the RTST.
- This approval is valid for up to 1 year from the date of approval. If the study extends beyond this date, a progress report must be sent to the RTST to renew the approval.
- The RTST must be informed when the research has been completed and a copy of the final research report must be submitted for our records.

We wish you all the best in this study.

Yours sincerely,



Fatima A. Wahid Al Ahmed

Assistant Undersecretary for Resources and Services

CC: Team file

Appendix 17. University of Sheffield ethical amendments



School Of Clinical Dentistry.

16th December 2020

Ms Seham Mohammed School of Clinical Dentistry Claremont Crescent Sheffield S10 2TA Professor Chris Deery Dean School of Clinical Dentistry Claremont Crescent Sheffield S10 2TA

Telephone: +44 (0) 114 222 2076 Email: hesdenreshub@sheffield.ac.uk

Dear Seham

Application No: 026444 Project Title: The Effects of Oral Health Conditions on Children's Academic Performance and School Attendance Using A Life Course Approach

I am writing to confirm approval of your request for minor amendments to your ethics submission Number 026444, detailed in the Notice of Amendments Form submitted on 16th December 2020.

Yours sincerely



Professor Paul Hatton Acting Ethics Lead

Appendix 18. MOE ethical approval for schoolchildren attendance

The ethical approval to obtain children attendance from school records was done via emails between the researcher Mrs Mariam Al-Othman; She is an Educational Specialist in Scientific Research Directorate in Secretariat-General of the Higher Education Council. She sent an email as indicated below in the Arabic language, and the translation is below.



The email said:

The respectable Mrs Seham Mohamed

Due and respects

In response to your email below, you can check with the Social Guidance Office in schools where students have been screened to provide you with the presence and absence of students for the period from September 2019 to February 2020, providing the Guidance Office with the names of students who have the consent of their parents to offer approved approvals - and if no school cooperates with providing you with data, you can contact us on 17896948.

Regards

Mariam AI – Othman

Appendix 19. Summary of data screening

Missing values patterns and percentage



Table 1 Summary of variables with missing values

	Mi	ssing	Valid N
	Ν	%	
1. Academic performance G2	16	3.4%	450
2. Father parenting style	16	3.4%	450
3. Mother parenting style	10	2.1%	456
4. Father's level of education at birth	3	0.6%	463
5. Mother use of mouth wash	3	0.6%	463
6. Mother flossing	3	0.6%	463
7. Mother teeth brushing	в	0.6%	463
8. Mother's employment at birth	2	0.4%	464
9. Mother's level education at birth	2	0.4%	464
10. Current household income	2	0.4%	464
11. Current mother's level of education	2	0.4%	464
12. Exclusive breast feeding	1	0.2%	465
13. Birth weight	1	0.2%	465
14. Gestational age	1	0.2%	465
15. Maternal postnatal depression	1	0.2%	465
16. Maternal health conditions	1	0.2%	465
17. Maternal hypertension	1	0.2%	465
18. Maternal diabetes mellitus	1	0.2%	465
19. Family income at birth	1	0.2%	465
20. Current father's level of education	1	0.2%	465

Little's MCAR test

EM Estimated Statistics



Table 2 Results of the univariate analysis

Number of observations with a Z score of below or above 3.29	Variable
7 cases	Absent days
6 cases	School performance
7 cases	Treated primary teeth
9 cases	Enamel caries
4 cases	Dentine caries

	Observation number	Mahalanobis d-squared	P1	P2
1.	77	137.967	.000	.000
2.	380	122.966	.000	.000
3.	20	106.742	.000	.000
4.	440	106.090	.000	.000
5.	49	98.472	.000	.000
6.	117	96.752	.000	.000
7.	419	95.786	.000	.000
8.	121	94.324	.000	.000
9.	324	88.567	.000	.000
10.	275	83.425	.000	.000
11.	7	80.280	.000	.000
12.	34	79.227	.000	.000
13.	432	77.526	.000	.000
14.	332	77.092	.000	.000
15.	142	76.606	.000	.000
16.	217	75.263	.000	.000
17.	104	73.871	.000	.000
18.	364	73.548	.000	.000
19.	233	72.606	.000	.000
20.	89	72.174	.000	.000
21.	16	71.549	.000	.000
22.	452	69.341	.000	.000
23.	71	68.482	.000	.000
24.	157	67.386	.000	.000
25.	351	65.942	.000	.000
26.	463	65.904	.000	.000
27.	189	65.853	.000	.000
28.	387	65.671	.000	.000
29.	192	65.114	.000	.000

Table 3 Results of the multivariate analysis

Table 4 Test of normality

Variable	Min	Мах	Skew	Critical ratio	Kurtosis	Critical ratio
Plaque	.000	100.000	671	-5.911	-1.083	-4.774
EDD	.000	1.000	7.974	70.275	61.587	271.378
Maternal CHCs	1.000	2.000	-1.286	-11.331	347	-1.528
Treated teeth	.000	25.000	2.156	19.004	5.407	23.825
ds_4_6	.000	68.000	.817	7.202	.677	2.983
Supervised TB	1.000	2.000	731	-6.443	-1.465	-6.457
Absent days	.000	22.000	3.970	34.988	23.299	102.667
pufa	1.000	2.000	1.175	10.352	620	-2.733
ds_2_3	.000	11.000	1.655	14.581	2.925	12.888
Exclusive breast feeding	.000	1.000	.986	8.686	-1.028	-4.532
School performance	65.000	100.000	-1.656	-14.596	2.977	13.118
Current dental pain	.000	1.000	.495	4.364	-1.755	-7.732
Birth weight	1.000	2.000	-2.467	-21.743	4.087	18.010
Gestational age	1.000	2.000	-2.502	-22.050	4.260	18.772
Birth order	1.000	2.000	-1.202	-10.590	556	-2.450
income at birth	1.000	3.000	.020	.176	680	-2.998
Mother academic involvement	8.000	28.000	471	-4.151	502	-2.214
Father parenting style	1.000	3.000	-3.418	-30.125	10.256	45.193
Mother parenting style	1.000	3.000	-4.338	-38.229	17.813	78.492
Child's teeth brushing	.000	2.000	.138	1.212	3.553	15.655
Child's flossing	.000	2.000	3.017	26.588	8.930	39.348
Child's use of MW	.000	2.000	1.943	17.125	2.886	12.718
Child's sugar consumption	.000	2.000	038	333	280	-1.235
Mother's teeth brushing	.000	2.000	.547	4.824	.408	1.797
Mother's teeth brushing	.000	2.000	.736	6.485	502	-2.212
Mother's flossing	.000	2.000	1.408	12.411	1.019	4.489
Current income	1.000	3.000	124	-1.095	763	-3.361
Current father's level of education	.000	2.000	-1.821	-16.052	2.447	10.784
Current mother's level of education	.000	2.000	-1.251	-11.027	.388	1.711
Mother job at birth	1.000	3.000	1.443	12.719	.299	1.318
Mother's level of education at birth time	.000	2.000	-1.202	-10.593	.226	.997
father's level of education at birth time	.000	2.000	-1.769	-15.586	2.280	10.047

Appendix 20. The initial CFA model



Appendix 21. Standardised Regression Weights, total, direct and indirect effects, BC 95% CI, and SE for full

model

Parameter			Estimate	Lower	Upper	Р	Total effect	Direct effect	P/direct	Indirect effect	P/indirect	Bootstrap-SE	BC 95% CI
MCHCs	<	BSES	001	121	.116	.970	001	001	.970			.061	121/.116
CSES	<	BSES	.986	.957	1.005	.003	.986**	.986**	.003			.12	.957/1.005
C mother education	<	BSES					.636**			.636**	.002	.063	.514/.758
C father education	<	BSES					.464**			.464**	.003	.068	.321/.585
C family income	<	BSES					.458**			.458**	.001	.071	.323/.599
MOHPs	<	BSES					036			036	.589	.073	178/.112
C mother flossing	<	BSES					011			011	.559	.022	055/.035
C mother mouth wash	<	BSES					028			028	.574	.058	147/.079
C mother brushing	<	BSES					007			007	.497	.016	039/015
Parent's characteristics	<	BSES					.438			.438**	.006	.148	.165/.727
Mother parenting S	<	BSES					.293			.293**	.002	.073	.149/.432
Father parenting S	<	BSES					.231			.231**	.001	.066	.108/.354
Mother academic involvement	<	BSES					.151			.151**	.004	.098	.041/.396
Birth outcomes	<>	BSES	034	211	.184	.752	034	034	.752			.096	022/.021
Birth order	<	BSES					007			007	.611	.020	048/.034.
Gestation age	<	BSES					030			030	.781	.065	194/.137
Birth weight	<	BSES					022			022	.681	.085	137/.124
Supervised TB	<	BSES	.163	.034	.268	.007	.163**	.163**	.007			.059	.034/.268
Child OHPs	<	BSES					190			190**	.004	.068	328/.057
Child sugar	<	BSES					.036			.036**	.002	.019	.008/.080
Child MW	<	BSES					161			161**	.003	.056	274/050
Child flossing	<	BSES					092			092**	.002	.036	169/030
Child TB	<>	BSES					044			044**	.001	.019	089/015
Plaque	<	BSES					006			-0.006	.704	.016	042/.023
Breastfeeding	<	BSES	.078	032	.193	.163	.078	.078	.163			.057	032/.193
Dentine caries	<	BSES	248	356	121	.002	249**	248**	.002	001	.949	.057	356/121
Enamel caries	<	BSES	.089	041	.193	.181	.066	.089	.181	023	.149	.059	041/.193
pufa	<	BSES	.018	131	.135	.844	005	.018	.844	022	.162	.066	131/.135
Dental pain	<	BSES					039			039**	.002	.015	075/014
Treated teeth	<	BSES	.021	101	.133	.830	.003	.021	.830	017	.257	.062	101/.133
B father education	<	BSES	.497	.318	.623	.003	.497**	.497**			1	.075	.318/.623
B mother education	<	BSES	.665	.530	.789	.003	.665**	.665**				.065	.530/.789
B mother job	<	BSES	.481	.377	.658	.000	.481**	.481**	.000			.066	.377/.658
B family income	<	BSES	.413	.267	.575	.001	.413**	.413**	.001			.078	.267/.575
Birth outcomes	<	MCHCs	.159	.040	.304	.012	.159*	.159*	.012			.068	.040/.304

Parameter			Estimate	Lower	Upper	Р	Total effect	Direct effect	P/direct	Indirect effect	P/indirect	Bootstrap-SE	BC 95% CI
Birth weight	<	MCHCs					.102			.102*	.018	.052	.012/.212
Gestational age	<	MCHCs					.137			.137*	.019	.058	.026/.249
Birth order	<	MCHCs					.032			.032*	.010	.020	.005/.092
EDD	<	MCHCs					002			002	.593	.008	024/.009
pufa	<	MCHCs					.002			.002	.768	.009	019/.020
Dentine caries	<	MCHCs					012			012*	.044	.010	042/000
Dental pain	<	MCHCs					002			002	.059	.002	008/.000
Enamel caries	<	MCHCs					.008			.008	.169	.008	004/.031
EDD	<	Birth outcomes	014	140	.054	.735	014	014	.735			.050	140/.054
Plaque	<	Birth outcomes					001			001	.622	.003	010/.004
Treated teeth	<	Birth outcomes					001			001	475	.005	003/.015
Dentine caries	<	Birth outcomes	077	169	.008	.072	076	077	.072	.001	.705	.004	169/.008
Enamel caries	<	Birth outcomes	.048	046	.128	.284	.048	.048	.284			.003	046/.128
Dental pain	<	Birth outcomes					011			011	082	.009	037/.001
pufa	<	Birth outcomes	.010	145	.092	.959	.010	.010	.959			.057	145/.092
Birth order	<	Birth outcomes	.199	.064	.338	.002	.199**	.199**	.002			.070	.064/.338
Gestational age	<	Birth outcomes	.860	.544	1.630	.002	.860**	.860**	.002			.252	.544/1.630
Birth weight	<	Birth outcomes	.642	.316	.932	.004	.642**	.642**	.004			.184	.316/.932
Enamel caries	<	EDD	.000	075	.105	.952	.005	.000	.952	.005	.065	.045	075/.105
Dentine caries	<	EDD	090	146	030	.002	069**	090**	.002	.021	.090	.029	146/030
pufa	<	EDD	032	087	.067	.528	033	032	.528	001	.687	.039	087/.067
Treated teeth	<	EDD	.085	046	.213	.177	.092	.085	.177	.007	.052	.062/.005	046/.213
Dental pain	<	EDD					014			014	.062	.062	033/.017
Plaque	<	EDD	.058	017	.110	.098	.058	.058	.098			.030	017/.110
Dentine caries	<	Breastfeeding	.026	050	.113	.489	.026	.026	.489			.042	050/.113
pufa	<	Breastfeeding	.019	071	.117	.683	.019	.019	.683			.047	071/.117
Treated teeth	<	Brest feeding	011	085	.081	.894	011	011	.894			.043	085/.081
Enamel caries	<	Brest feeding	019	100	.082	.756	019	019	.756			.046	100/.082
Dental pain	<	Breastfeeding					.005			.005	.412	.008	008/.023
Child OHPs	<	Supervised TB	.025	070	.118	.592	.025	.025	.592			.046	070/.118
Child TB	<	Supervised TB					.006			.006	.507	.011	015/.032
Child flossing	<	Supervised TB					.012			.012	.565	.022	035/.056
Child MW	<	Supervised TB					.021			.021	.584	.038	056/.098
Child sugar	<	Supervised TB					005			005	.482	.009	026/.012
Plaque	<	Supervised TB	011	103	.087	.825	011	011	.825			.048	103/.087
Enamel caries	<	Supervised TB	061	157	.039	.229	060	061	.229			.051	157/.039
Dentine caries	<	Supervised TB	061	153	.024	.172	066	061	.172	005	.807	.045	153/.024
pufa	<	Supervised TB	041	146	.045	.331	038	041	.331	.002	.468	.047	146/.045
Treated teeth	<	Supervised TB	063	160	.049	.229	063	063	.229	001	.882	.051	160/.049

Parameter			Estimate	Lower	Upper	Р	Total effect	Direct effect	P/direct	Indirect effect	P/indirect	Bootstrap-SE	BC 95% CI
Dental pain	<	Supervised TB					012			012	.129	.009	033/.001
MOHPs	<	CSES	037	181	.114	.585	037	037	.585			.074	181/.114
Mother TB	<	CSES					008					.016	040/.022
Mother MW	<	CSES				2	028					.058	149/.079
Mother flossing	<	CSES				1	011					.023	056/.036
Parent's characteristics	<	CSES	.444	.167	.733	.006	.444**	.444**	.006			.150	.167/.733
Mother academic involvement	<	CSES					.153					.099	.042/.401
Father parenting S	<	CSES					.234					.067	.109/.358
Mother parenting S	<	CSES				0	.297					.074	.153/.438
Child OHPs	<	CSES	109	284	.129	.238	197	109	.238			.112	284/.129
Child TB	<	CSES					046					.019	090/015
Child flossing	<	CSES					095					.037	175/032
Child MW	<	CSES					166					.058	281/049
Child sugar	<	CSES					.036					.019	026/.085
Plaque	<	CSES					004			-0.004	.690	.013	038/.018
pufa	<	CSES				L	018					.014	035/.002
Dentine caries	<	CSES					.005					.012	016/.038
Enamel caries	<	CSES					010					.013	046/.008
Treated teeth	<	CSES					006					.012	035/.014
C mother education	<	CSES	.645	.521	.770	.002	.645**	.645**	.002			.064	.521/.770
C father education	<	CSES	.470	.323	.596	.003	.470**	.470**	.003			.071	.323/.596
C family income	<	CSES	.464	.333	.607	.001	.464**	.464**	.001			.070	.333/.607
Child OHPs	<	Parent's characteristics	132	381	.082	.258	132	132	.258			.129	381/.082
Child TB	<	Parent's characteristics					031			031	.507	.032	104/.015
Child flossing	<	Parent's characteristics					064			064	.565	.065	201/.038
Child MW	<	Parent's characteristics					112			112	.584	.109	315/.072
Child sugar	<	Parent's characteristics					.025			.025	.482	.028	012/.096
Plaque	<	Parent's characteristics					003			003	.561	.012	041/.014
Treated teeth	<	Parent's characteristics					004			004	.425	.012	035/.010
pufa	<	Parent's characteristics					012			012	.213	.017	063/.005
Dentine caries	<	Parent's characteristics					.003			.003	.449	.011	010/.041

Parameter		Estimate	Lower	Upper	Р	Total effect	Direct effect	P/direct	Indirect effect	P/indirect	Bootstrap-SE	BC 95% CI	
Enamel caries	<	Parent's characteristics				5	007			007	236	.012	049/.006
Mother parenting S	<	Parent's characteristics	.669	.318	.907	.002	.669**	.669**	.002			.154	.318/.907
Father parenting S	<	Parent's characteristics	.527	.225	.699	.001	.527**	.527**	.001			.127	.225/.699
Mother academic involvement	<	Parent's characteristics	.345	.176	.579	.003	.345**	.345**	.003			.103	.176/.579
Plaque	<	MOHPs				17	.015			.015	.719	.046	076/.111
Treated teeth	<	MOHPs					.025			.025	.611	.043	054/.109
pufa	<	MOHPS					.070			.070	.119	.049	016/.178
Dentine caries	<	MOHPs					020			020	.631	.043	109/060
Dental pain	<	MOHPs					.001			.001	.948	.009	017/.018
Enamel caries	<	MOHPs					.041			.041	.257	.043	037/.133
Child OHPs	<	MOHPs	.784	.560	.940	.003	.784**	.784**	.003			.089	.560/.940
Child TB	<	MOHPs					.183			.183**	.001	.057	.083/.304
Child flossing	<	MOHPs					.377			.377**	.003	.082	.186/.534
Child MW	<	MOHPs					.662			.662**	.003	.085	.456/.810
Child sugar	<	MOHPs					148			148**	.002	.046	243/060
Mother flossing	<	MOHPs	.302	.146	.425	.004	.302**	.302**	.004			.069	.146/.425
Mother MW	<	MOHPs	.778	.641	1.028	.002	.778**	.778**	.002			.100	.641/1.028
Mother TB	<	MOHPs	.206	.081	.308	.003	.206**	.206**	.003			.057	.081/.308
Plaque	<	Child OHPs	.020	099	.136	.735	.020	.020	.735			.058	099/.136
Treated teeth	<	Child OHPs	.030	071	.134	.653	.032	.030	.653	.002	.604	.054	011/.021
pufa	<	Child OHPs	.090	024	.206	.138	.089	.090	.138			.060	024/.206
Dentine caries	<	Child OHPs	033	138	.058	.454	025	033	.454	.007	.731	.050	138/.058
Dentine caries	<	Child OHPs	.050	049	.170	.287	.052	.050	.287	.002	.568	.056	049/.170
Dental pain	<	Child OHPs					.001			.001	.952	.011	022/.023
Child sugar	<	Child OHPs	189	292	070	.003	189**	189**	.003			.055	292/070
Child MW	<	Child OHPs	.845	.724	.967	.002	.845**	.845**	.002			.059	.724/.967
Child sugar	<	Child OHPs	.481	.320	.608	.003	.481**	.481**	.003			.072	.320/.608
Child TB	<	Child OHPs	.233	.110	.353	.002	.233**	.233**	.002			.061	.110/.353
pufa	<	Plaque	011	111	.084	.803	011	011	.803			.051	111/.084
Treated teeth	<	Plaque	.116	.019	.192	.013	.116*	.116*	.013			.044	.019 /.192
Dentine caries	<	Plaque	.370	.293	.442	.002	.370**	.370**	.002			.038	.293/.442
Enamel caries	<	Plaque	.094	.006	.181	.033	.094*	.094*	.033			.044	.006/.181
Dental pain	<	Plaque					.055**			.055	.002	.020	.019/.098
Dental pain	<	Dentine caries	.155	.063	.244	.002		.155**	.002			.047	.063/.244
Dental pain	<	pufa	.061	031	.155	.203	.061	.061	.203			.047	.000/.000

Parameter		Estimate	Lower	Upper	Р	Total effect	Direct effect	P/direct	Indirect effect	P/indirect	Bootstrap-SE	BC 95% CI	
Dental pain	<	Treated teeth	011	100	.078	.791	011	011	.791			.045	100/.078
Absent days	<	BSES					025			025	.433	.039	118/.035
Absent days	<	CSES					022					.042	131/.030
Absent days	<	MCHCs					.001			.001	.896	.008	016/.016
Absent days	<	Supervised TB					.001			.001		.006	011/.014
Absent days	<	Birth outcomes					.005	.006	.946			.048	014/.008
Absent days	<	MOHPS					002			002	.482	.005	014.006
Absent days	<	Child OHPs					003			003	.522	.006	018/.009
Absent days	<	Dental pain	045	122	.034	.293		045	.293			.041	122/.034
Absent days	<	Parent's characteristics	051	202	.078	.497	051	051	.497			.075	202/.078
Absent days	<	Enamel caries	017	082	.066	.678		017	.678			.037	082/.066
Absent days	<	Dentine caries	.013	069	.098	.753		.013	.753			.042	069/.098
Absent days	<	pufa	018	096	.066	.620	021	018	.620			.042	096/.066
Absent days	<	Treated teeth	.000	108	.115	.949		.000	.949	.001	.499	.057	108/.115
Absent days	<	Birth outcomes	.006	103	.090	.946		.006	.946	002	.433	.056	103/.090
School performance	<	Absent days	074	162	.013	.082	074	074	.082			.045	162/.013
School performance	<	Dental pain	010	103	.080	.848	007	010	.848	.003	.173	.047	103/.080
School performance	<	Parent's characteristics	.213	.060	.376	.004	.216**	.213**	.004	.003	.497	.081	.060/.376
School performance	<	Enamel caries	.024	052	.103	.496	.025	.024	.496	.001	.422	.040	052/.103
School performance	<	Dentine caries	154	258	055	.006	156**	154**	.006	002	.736	.052	258/055
School performance	<	pufa	018	110	.074	.751	017	018	.751	.001	.743	.047	110/.074
School performance	<	Plaque					043			043*	.026	.021	089/
School performance	<	Treated teeth	.102	.031	.174	.011	.102*	.102*	.011			.038	.031/.174
School performance	<	EDD					.021			.021*	033	.010	.002/.044
School performance	<	Birth outcomes	018	128	.094	.728	006	018	.728	012	.142	.010	128/.094
School performance		CSES					.095					.066	.016/.255
School performance	<	BSES					.137			.137**	.004	.062	.050/.281
School performance	<	MCHCs					001			001	.817	.009	023/.014
School performance	<	Supervised TB					.003			.003	.711	.011	015/.027
School performance	<	MOHPs					.006			.006	.473	.009	011/.026
School performance	<	Child OHPs					.007			.007	.470	.012	014/.032
School performance	<	Breastfeeding				5	006			006	.412	.008	023/.009

The full model from AMOS

